Developing and refining social norms measures for the evaluation of a schoolbased intervention to reduce dating and relationship violence among adolescents in England

REBECCA NICOLE MEIKSIN

Thesis submitted in accordance with the requirements for the degree of Doctor of Philosophy of the University of London October 2024

Department of Public Health, Environments and Society Faculty of Public Health and Policy LONDON SCHOOL OF HYGIENE & TROPICAL MEDICINE

Paper 1 was supported with funding from the Passages Project. Passages is made possible by the generous support of the American people through the United States Agency for International Development (USAID) under the terms of Cooperative Agreement No. AID-OAA-A-15-00042. The contents are the responsibility of the authors and do not necessarily reflect the views of Georgetown University, USAID, or the United States Government. Papers 2, 3 and 4 draw on data from the Project Respect study, which was funded by the NIHR Public Health Research Board (project 15/03/09). The views expressed in these papers are those of the authors and not necessarily those of the NIHR or the Department of Health and Social Care. The funder was not involved in study design; collection, analysis, or interpretation of data; the writing of the articles; or any decisions to submit these articles for publication.

I, Rebecca Nicole Meiksin, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

Volume I

Abstract

Introduction

Dating and relationship violence (DRV) is widespread in England. DRV is associated with increased prevalence of sexual risk behaviour and poor mental health. Interventions often aim to shift harmful social norms underpinning DRV, but lack of valid, reliable measures is a barrier.

Methods

I conducted a systematic review of DRV social-norms measures. I developed three brief measures of social norms concerning DRV and gender. I refined these using cognitive testing and assessed the reliability and validity of resulting measures using student surveys in five secondary schools. I analysed qualitative data from students, staff and parents and carers in ten secondary schools to explore how social norms are implicated in DRV in England and inform further measure refinement.

Results

Most of the 40 social-norms measures identified in the review were associated with DRV outcomes. Other evidence of reliability and validity was mixed and no measure was shared across studies. In cognitive testing of social-norms measures, answerability was improved where items assessed norms salient and publicly manifest among a cohesive, influential reference group. Refined measures were tested among 1,426 students (82.5% response rate). While floor-effects indicate limited sensitivity to low-to-moderate levels of the assessed constructs, all three measures were reliable and valid. Qualitative interviews suggest that DRV is sustained directly by norms tolerating controlling behaviours and inhibiting disclosure of victimisation, and indirectly via sexist norms that subjugate girls to boys and facilitate genderbased harassment and abuse.

Discussion

My findings support the reliability and validity of the three tested measures of social norms, which can be incorporated into evaluations. Research to assess the measures' cross-cultural validity would contribute to improving comparability of norms across contexts. Further research should seek to develop longer-form versions with increased sensitivity for use in

epidemiological research. My findings support recommendations for improving social norms measurement in DRV research.

Dedication

This thesis is dedicated to my Grandpap, Zvi, and to my late Grandma, Jeannine, whose work

inspires me and whose love nurtures me.



"Mother's Purse" Painting by Jeannine Meiksin, circa 1959

Acknowledgements

I'm extremely grateful to the many wonderful people who have made this work possible.

Thank you to the students, staff and parents and carers who generously shared their time and views for this research and without whom, there would be no research.

Thank you to my supervisors, Chris Bonell and Nambusi Kyegombe, for your enduring support and for helping me to clarify my thinking and writing throughout this process. Thank you to my advisors, Anjalee Kohli and GJ Melendez-Torres, for patiently sharing your expertise and guidance. Thank you to Liz Allen and Charles Opondo for taking the time to share your statistical expertise with clear and accessible explanations.

Thank you to Rebecca Glover, Pippa Grenfell, Shari Krishnaratne and Emily Warren – for showing me how this is done, and that it is possible. Thank you to Sujit Rathod for your humour, from start to finish.

Thank you to my teammates Ruth Ponsford and Nerissa Tilouche, for your insights and for your friendship. Thank you to Claire Thompson, Amiya Bhatia and Melissa Palmer for sharing your time and expertise, and thank you to Jocelyn Elmes and Ford Hickson for inspiring new thinking on gender and gender socialisation.

Thank you to Helen Burchett and to Lucy Platt for your support and encouragement, and to Daniel for his invaluable childcare skills.

My deepest love and thanks to my parents, Judy Meiksin and Janice Rensi, for literally everything.

To my brilliant and beautiful partner, Ben Pelhan: this project would not have been possible without your love, your daily encouragement (affirmation love) and your willingness to discuss public health for hours on end. You are my favourite. Also, happy birthday.

And finally, thank you to Poppy, my best and most powerful motivation for finishing this work.

30 March 2024

Volume I Contents

Abstract	7
Dedication	9
Acknowledgements	. 11
List of Appendices	. 17
List of abbreviations	. 18
List of figures	. 19
Introduction and Methods (Chapters 1-3)	. 19
Paper 1 (Chapter 4)	. 19
List of tables	. 20
Introduction and Methods (Chapters 1-3)	. 20
Paper 1 (Chapter 4)	. 20
Paper 2 (Chapter 5)	. 20
Paper 3 (Chapter 6)	. 20
Paper 4 (Chapter 7)	. 21
Introduction	. 22
Structure of thesis	. 23
Chapter 1. Dating and relationship violence	. 25
1.1. Introduction to DRV	. 25
1.2. DRV prevalence	. 26
1.3. Sociodemographic patterning of DRV	. 27
1.4. Health, education and economic impacts of DRV	. 28
1.5. Intervening in adolescence to prevent DRV	. 29
1.5.1. Modifiable risk and protective factors for DRV	. 29
1.6. Approaches to DRV prevention	. 43
Chapter 2. Social norms and DRV	. 71
2.1. Introduction to social norms and DRV	. 71
2.2. Social norms theory	. 72
2.2.1. Social norms scholarship in the social sciences	. 72
2.2.2. Definitions and features of social norms for the present study	. 73
2.2.3. Conceptualising gender norms as a type of social norm	. 75
2.3. Social norms relating to DRV and gender are drivers of DRV	. 76
2.3.1. Evidence of the relationship between DRV norms and DRV outcomes	. 77

2.3.2 Evidence of the relationship between gender norms and DRV outcomes	79
2.4. Incorporation of social norms approaches into DRV prevention	80
2.5. Social norms measurement	81
2.5.1. Survey measures of social norms are limited	82
2.5.2. Measuring social norms: recommendations from GBV and adolescent SRH rese	earch 83
2.6. Research aim, questions and objectives	87
Chapter 3. Methods	90
3.1. Methods for Project Respect pilot cluster RCT	90
3.1.1. Project Respect theory of change and intervention	90
3.1.2. Project Respect pilot cluster RCT	97
3.1.3. Summary of Project Respect study findings	99
3.1.4. My role as research fellow on the Project Respect study	100
3.2. Thesis methods	100
3.2.1. Distinction between the Project Respect study and my thesis	100
3.2.2. Ontological and epistemological assumptions	102
3.2.3. The use, reliability and validity of quantitative measures	106
3.2.4. Systematic review of social norms measures relating to DRV and gender used i research – methods for Chapter 4 (Paper 1)	n DRV 108
3.2.5. Cognitive interviews informing the development of social norms measures – methods for Chapter 5 (Paper 2)	117
3.2.6. Reliability and validity testing of new and adapted social norms measures – me for Chapter 6 (Paper 3)	ethods 124
3.2.7. Relationships between social norms and DRV in student, staff and parent/care accounts – methods for Chapter 7 (Paper 4)	r 132
3.2.8. Ethics	138
3.2.9. Reflexivity and positionality	139
Chapter 4. Paper 1: Social norms measures relating to DRV and gender – systematic revie DRV research	w of 141
4.1. Introduction to Paper 1	141
4.2. Paper 1: Social norms about dating and relationship violence and gender among adolescents: systematic review of measures used in dating and relationship violence re	search
Chapter 5. Paper 2: Cognitive interviews informing the development of social norms measured	sures
	161

5.1	I. Introduction to Paper 2	161
!	5.1.1. Notes on Table 1 of Paper 2	161
	5.1.2. Previous work reporting on this research	162
5.2 vio	2. Paper 2: Assessing survey items on social norms relating to dating and relationship plence and to gender: cognitive interviews with young people in England	163
5.3	3. Follow-up to Paper 2: Development and refinement of social norms measures for pilot	ting 188
	5.3.1. Descriptive DRV norms measure	188
!	5.3.2. Injunctive DRV norms measure	190
	5.3.3. Injunctive gender norms measure	192
	5.3.4. Piloted social norms measures	194
Chap	ter 6. Paper 3: Reliability and validity testing of new social norms measures	196
6.1	L. Introduction to Paper 3	196
6.2 ada	2. Paper 3: Social norms in dating and relationship violence research: testing new and apted measures and examining their relationships to violence outcomes	198
6.3	3. Follow-up to Paper 3: social norms measures refined on the basis of psychometric test	ting
	tor 7. Depart 4. Delationships between social norms and DDV in student, staff and	233
chap parer	nt/carer accounts	236
7.1	L. Introduction to Paper 4	236
7.2 seo	2. Paper 4: Social norms relating to gender and dating and relationship violence in Englis condary schools: exploring student, staff and parent/carer accounts	h 237
Chap	ter 8. Discussion	265
8.1	L. Summary of key findings	265
:	8.1.1. Contextualisation	265
:	8.1.2. Research question 1: Are existing measures of adolescent social norms relating to DRV and gender reliable and valid?	266
	8.1.3. Research question 2: Are new and adapted measures of social norms relating to D and gender understandable and answerable when used in research with adolescents in England?	0RV 268
;	8.1.4. Research question 3: Are new and adapted measures of social norms relating to D and gender reliable and valid when used in research with adolescents in England, and he can they be refined?	0RV 0w 269
	8.1.5. Research question 4: What are student, staff and parent/carer accounts of social norms relating to DRV and gender in schools, and how are these implicated in DRV?	271
8.2	2. Synthesis and interpretation	272

8.2.1. Implications for research	
8.2.2. Implications for practice	
8.3. Limitations	
8.4. Conclusions	
References	

Volume II

List of Appendices

- Appendix 1. Project Respect published protocol, "Protocol for pilot cluster RCT of Project Respect: a school-based intervention to prevent dating and relationship violence and address health inequalities among young people"¹
- Appendix 2. Project Respect process evaluation publication, "Implementing a whole-school relationships and sex education intervention to prevent dating and relationship violence: evidence from a pilot trial in English secondary schools"²
- Appendix 3. Systematic review protocol on Open Science Framework³
- Appendix 4. Database search strategy and yields for systematic review
- Appendix 5. Paper 1 supplemental appendices
- Appendix 6. Cognitive interview information sheet and assent form
- Appendix 7. Cognitive interview demographic form and guide
- Appendix 8. Student baseline survey
- Appendix 9. Student baseline survey information sheets and assent form
- Appendix 10. Optimisation session information sheets and assent/consent forms
- Appendix 11. Expert-led and cascaded trainings information sheets and consent forms
- Appendix 12. Sample of process evaluation information sheet and assent/consent form materials
- Appendix 13. Optimisation session and process evaluation qualitative data collection tools
- Appendix 14. Ethical approval letters
- Appendix 15. Published abstract for Paper 1, "Social norms concerning dating and relationship violence and gender among adolescents: a systematic review of survey measures used in dating and relationship violence research"⁴
- Appendix 16. Paper 2 supplemental appendix
- Appendix 17. Published abstract drawing on data used in Paper 2, "Assessment of survey items on social norms relating to sexual behaviour and dating and relationship violence among young adolescents in England: cognitive interviews within two pilot studies"⁵

Appendix 18. Paper 3 supplemental appendices

List of abbreviations

ALIGN: Advanced Learning and Innovation on Gender Norms

- aOR: adjusted odds ratio
- ATWS: Attitudes Toward Women Scale
- CADRI-s: short Conflict in Adolescent Dating Relationships Inventory
- CFA: confirmatory factor analysis
- CFI: comparative fit index
- CI: confidence interval
- COSMIN: COnsensus-based Standards for the selection of health Measurement Instruments
- CTS: Conflict Tactics Scale
- CTT: classical test theory
- SD: standard deviation
- SE: standard error
- DRV: dating and relationship violence
- EFA: exploratory factor analysis
- GBV: gender-based violence
- GVHC: gender violence and health centre
- HICs: high-income countries
- IDACI: Income Deprivation Affecting Children Index
- IPV: intimate partner violence
- LINEA: Learning Initiative on Norms, Exploitation and Abuse
- LMICs: low- and middle-income countries
- LSHTM: London School of Hygiene & Tropical Medicine
- NIHR: National Institute for Health Research
- NSPCC: National Society for the Prevention of Cruelty to Children
- OR: odds ratio
- PRISMA: Preferred Reporting Items for Systematic Reviews and Meta-Analyses
- PROMS: patient-related outcome measures
- RCT: randomised controlled trial
- RMSEA: root mean square error of approximation

RQ: research question SES: socioeconomic status SGM: sexual- and gender-minority SRH: sexual and reproductive health TLI: Tucker-Lewis Index US: United States UK: United Kingdom

List of figures

Introduction and Methods (Chapters 1-3)

- Figure 1. An example of a social ecological model used for conceptualising violence
- Figure 2. Framework distinguishing between measures of behaviours, attitudes and social norms (adapted from Mackie et al.'s theoretical framework)
- Figure 3. Example of survey item assessing injunctive norms by measuring reference group views, excerpted from Cislaghi & Heise
- Figure 4. Example of survey item assessing injunctive norms by measuring specific sanctions, excerpted from the Social Norms Mentorship Programme training slides
- Figure 5. Example of survey item assessing injunctive norms by measuring sanctions generally, excerpted from the Social Norms Mentorship Programme training slides
- Figure 6. Ajzen's model of the Theory of Planned Behaviour
- Figure 7. Cambron's depiction of the Social Development Model
- Figure 8. Project Respect theory of change, adapted from published study protocol
- Figure 9. Image from Manyukhina & Alderson depicting a metaphor for the relationship between structure and human agency in critical realism
- Figure 10. Bhaskar's transformational model of social activity
- Figure 11. Bhaskar's transformational model applied to theoretical relationship between normative social structure and students' actions in school

Paper 1 (Chapter 4)

Figure 1. Flow diagram of search and study selection

List of tables

Introduction and Methods (Chapters 1-3)

- Table 1. Summaries of selected reviews of DRV interventions
- Table 2. Research questions and objectives mapped to methods, data sources and papers
- Table 3. Survey items subject to cognitive testing
- Table 4. Characteristics of tested items

Paper 1 (Chapter 4)

- Table 1. Eligible constructs assessed by included measures and evidence of measure quality, bydomain and inductive grouping
- Table 2. Implications for practice, policy, and research

Paper 2 (Chapter 5)

- Table 1. Original measures, tested items, and final measures
- Table 2. Background characteristics of cognitive interview participants
- Table 3. Descriptive DRV norms: Original, cognitively tested and piloted measures
- Table 4. Injunctive DRV norms: Original, cognitively tested and piloted measures
- Table 5. Injunctive gender norms: Original, cognitively tested and piloted measures
- Table 6. Piloted social norms measures, refined on the basis of cognitive testing

Paper 3 (Chapter 6)

- Table 1. Completion and distribution of piloted DRV descriptive norms items (N=947)
- Table 2. Completion and distribution of piloted DRV and gender injunctive norms items (N=1,426)
- Table 3. Participant characteristics and DRV prevalence
- Table 4. Rotated factor loadings and ordinal alphas (last two rows of table) of tested DRV and gender items (N=917)
- Table 5. Distribution of EFA-derived social norms measures
- Table 6. Descriptive characteristics of EFA-derived social norms measures by sex

- Table 7. Unadjusted and adjusted regression coefficients of the associations between socialnorms measures and DRV outcomes
- Table 8. Regression coefficients of the associations between social norms measures and DRVoutcomes, adjusted for other norms measures and for covariates
- Table 9. Final social norms measures following psychometric testing

Paper 4 (Chapter 7)

- Table 1. Optimisation session participation
- Table 2. Interview participants

Introduction

Interventions to prevent dating and relationship violence (DRV) among young people have long been informed by social norms approaches. That is, they often seek to change what can be thought of generally as "the often unspoken rules that govern behaviour".^{6(p1)} As DRV researchers Wekerle and Wolfe wrote in an early review of DRV and its prevention (1999), "...peer 'pressure' can be harnessed to serve prosocial ends".^{7(p450)} Reviews of DRV interventions since that time have reported the continued influence of social norms approaches marshalling the "power of the peer group"^{8(p127)} to change pro-DRV and gender-inequitable norms in DRV prevention.^{8–13} An expert consulted for one such review described this approach in practical terms: "...in any classroom of 25 kids...five or even ten of them might be at risk of an abusive relationship. The other 15 are there to keep that from happening..."^{8(p127)}

Today, we know that interventions can reduce DRV, but we know little about what aspects make them effective and how.^{10,11,14} Despite the ongoing incorporation of social norms approaches into DRV interventions,^{8–13} efforts to measure whether norms that underpin DRV are changing as intended have been largely neglected. Evaluations of DRV interventions very rarely measure social norms, ^{11,12} and little guidance is available on how best to do so. Consequently, despite widespread acknowledgement of the importance of social norms to DRV outcomes, we do not know whether changing social norms plays a role in effective DRV interventions, or whether failure to do so helps to explain why many¹¹ DRV prevention interventions fail.

This thesis aims to address these gaps, informed by advancements in social norm theory and measurement over the last decade. Drawing on a conceptualisation of social norms stemming from social psychology, which has been particularly influential in the areas of gender-based violence (GBV) and adolescent sexual and reproductive health (SRH), I distinguish in this thesis between social norms concerning perceptions about what behaviours are (1) typical (descriptive norms), and (2) socially acceptable (injunctive norms) within a social group.^{6,15}

Using this framework, I developed social norms measures for use in Project Respect, a study to optimise and pilot a new DRV intervention for secondary school students in England. Drawing on data from this study, I tested three brief measures of DRV and gender norms among adolescents in England. Based on these findings, I offer the first set of evidence-based recommendations to improve social norms measurement in DRV research.

Structure of thesis

Chapter 1 of this thesis defines DRV and introduces it as a public health problem, providing an overview of its epidemiology and impacts. The chapter then reviews approaches to DRV prevention and the evidence base for interventions, which are often framed in terms of targeting social norms that contribute to DRV. It concludes with a discussion of the limitations to the available evidence base for prevention, which include a gap in evaluating the role of norms change in DRV prevention. Chapter 2 introduces social norms theory, details relationships between norms and DRV and summarises promising approaches to reducing DRV by shifting social norms. It highlights limitations to existing approaches to social norms in mediating intervention impact. My thesis seeks to address these gaps by drawing on data from the pilot trial of Project Respect, a school-based DRV intervention in England. Chapter 2 closes with my thesis aim, objectives and research questions. Chapter 3 summarises the methods and findings of Project Respect and then outlines my role as research fellow on that pilot trial. The chapter then presents an overview of the methods for my thesis research and concludes with a section on reflexivity and consideration of my positionality.

Chapters 4, 5, 6 and 7 each report on the methods for one component of my research and present results within published peer-reviewed papers or (where not yet published) in paper-style. Chapter 4 presents a systematic review of measures of social norms relating to DRV and to gender used in DRV research. Chapter 5 presents qualitative research with young people in England to inform the development of survey measures of gender and DRV social norms. Chapter 6 presents reliability and validity testing of three new and adapted measures of DRV

and gender norms piloted with young people in England. Chapter 7 presents qualitative research with young people in England to explore relationships between social norms and DRV in their accounts.

Chapter 8, the discussion chapter, summarises and contextualises findings from chapters 4, 5, 6 and 7; considers limitations to the thesis research; and addresses implications for research and practice.

Chapter 1. Dating and relationship violence

This chapter begins with an introduction to dating and relationship violence (DRV) and provides an overview of its prevalence, sociodemographic patterning and impacts. It next presents an overview of modifiable risk and protective factors followed by a review of approaches to DRV prevention, which often aim to change social norms relating to DRV and to gender that underpin DRV. The chapter then reviews the evidence supporting existing DRV interventions. It concludes with a discussion of limitations to the evidence base for prevention, highlighting incongruence between the high level of attention that is given to social norms in the framing of DRV interventions and the lack of rigorous research empirically examining their role in DRV prevention.

1.1. Introduction to DRV

DRV¹⁶ refers to intimate partner violence (IPV) involving a young person, defined in this thesis as a person aged 10-19 years.^{a17} DRV comprises abuse by a current or former intimate partner,^{18,19} including physical violence, stalking, psychological aggression, threats, controlling behaviours, economic abuse and coerced, non-consensual or abusive sexual activities.^{16,20} Globally, DRV is widespread^{21–25} and associated with a host of subsequent health problems.²⁶ Among girls ages 15-19, it is the third leading risk factor for death and the fourth for disabilityadjusted life years, increasing in rank from the fifth leading cause in both measures from 1990 to 2013.²⁷ Despite the proliferation of DRV prevention interventions, particularly in North America,⁸ and a rapidly expanding body of reviews synthesising evidence on DRV prevention,^{7–} ^{14,22,28–32} existing interventions demonstrate uneven success in reducing incidents of DRV and relatively little is known about effective approaches to prevention.^{10,11}

^a The age range of 10-19 years comes from the World Health Organization's definition of adolescence.¹⁷ However, studies vary in the age ranges they use to define or research DRV. Where I report DRV data that uses an age range falling outside of 10-19 years in my Introduction, Methods and Discussion chapters I note this. The results of my thesis research, reported in my Results chapters, relate to DRV within the age range of 10-19 years.

1.2. DRV prevalence

Systematic review evidence suggests that DRV is widespread.^{21,22,25} Though rates of each DRV type vary widely by study, prevalence patterns tend to be consistent across North America and Europe with psychological DRV the most frequently reported, followed by physical and then sexual DRV.²² Young people who report experiencing DRV often report multiple types.²² Girls and boys report similar rates of psychological and physical DRV victimisation while girls tend to report higher rates of sexual DRV victimisation.^{b,c,22} This reflects the findings of Wincentak et al.'s 2017 meta-analytic review of 101 DRV studies among participants aged 13 to 18 years.²⁵ While noting high variability across studies, the reviewers report victimisation prevalence estimates of 21% for physical DRV victimisation among girls and boys and, for sexual DRV victimisation, 14% among girls and 8% among boys. Research in the United States (US) with students in grades 8 and 9 (typically aged 13-15 years) suggests that girls tend to report experiencing higher levels than boys of severe physical DRV victimisation.³³ Wincentak et al. estimate prevalence of physical DRV perpetration to be 25% among girls and 13% among boys, and they estimate prevalence of sexual DRV perpetration to be 3% among girls and 10% among boys.²⁵

DRV has historically received little attention in the United Kingdom (UK) but is gaining recognition.^{16,19,34} Surveys have increasingly examined its prevalence in UK samples.^{16,34–38} As they do globally, estimates vary by sampling and by outcome measurement, but available UK evidence suggests that DRV is widespread.^{16,35,37–39} In Project Respect's 2017 baseline surveys, conducted with 1,426 year 8 and year 9 students (aged 12-14 years) from five secondary schools in England, 1,022 (71.7%) reported ever having dated or been in a relationship.³⁹

^b I primarily refer to "girls" and "boys" or "females" and "males" in this thesis and in some cases refer to DRV in heterosexual relationships. This reflects the prevailing use of binary gender and sex categories and a focus on heterosexual relationships in existing DRV research. There is a widely-acknowledged gap in DRV research on sexual- and gender-minority (SGM) young people.^{8,11,14} As of yet, there is little evidence on causes and consequences of DRV among SGM, and little is known about effective approaches to prevention among this population.

^c I primarily use gender terms in this thesis (e.g. "girl" and "boy") but use sex terms (i.e. "female" and "male") where it is clear that sex rather than gender has been measured or where I am following the language of the authors whose work I'm reporting.

Among this dating sample, 72.8% of girls and 64.4% of boys reported psychological DRV victimisation, and 56.8% of girls and 53.4% of boys reported physical DRV victimisation.³⁹ While this survey did not measure sexual DRV, in a 2015 study in England of young people aged 14-17, 41% of girls and 14% of boys who had been in a relationship reported experiencing forced or pressured sexual DRV.³⁵ A 2017 study of 16-19 year-olds in further education in England and Wales also found high DRV prevalence, with more than half of young people who had ever dated reporting psychological, physical or online sexual DRV victimisation.¹⁶ Available evidence suggests that reported perpetration is similarly high. Among daters in Project Respect's baseline surveys, 57.2% of girls and 48.7% of boys reported psychological DRV perpetration and 47.3% of girls and 41.6% of boys reported physical DRV perpetration.³⁹

1.3. Sociodemographic patterning of DRV

Adolescence marks the beginning of a period of heightened vulnerability to relationship violence, which when considered across the life course peaks in late adolescence and young adulthood.^{40,41} DRV rates are higher among sexual- and gender-minority (SGM) young people than their heterosexual and cisgender peers,^{42–44} including among young people reporting a non-heterosexual sexual identity in the UK.¹⁶ Evidence on patterning of DRV by ethnicity is mixed,^{45,46} including in the UK where the significance and direction of this relationship varies by gender, age, DRV type and outcome measure.^{16,34,38,39} Researchers have suggested that evidence on the relationship between ethnicity and DRV might be limited by low representation of ethnic minorities in study samples.⁴⁷ An extensive, systematic review of 128 articles on risk factors for adolescent and adult relationship violence concluded that "the weight of findings" suggests that members of minoritised ethnic groups face higher risk of relationship violence but did not distinguish between adolescent and adult samples.^{40(p8)} Evidence on the relationship between associations vary by sample and by DRV type.^{16,34,38,39,48}

1.4. Health, education and economic impacts of DRV

Romantic relationships in adolescence play an important role in shaping young people's health and development,⁴⁹ and evidence suggests that both experiencing and perpetrating DRV are associated with a range of subsequent negative health and socioeconomic outcomes.

DRV victimisation leads to injuries⁵⁰ and can lead to death. In the US, 6.9% of homicides of adolescents are committed by an intimate partner; of these, 90.0% of the victims are girls.⁵¹ DRV victimisation is also associated with a range of other subsequent health problems including increased depression⁵², increased illicit substance use,^{26,52} antisocial behaviour,^{26,52} sexual risk behaviour,^{53,54} and suicidal ideation²⁶ and attempts⁵⁵ among girls and boys. Girls report additional subsequent harms, including fear,³⁴ more injuries,⁵⁰ increased body mass index,⁵³ and for Black mothers, lower infant birthweight.⁵³ Longitudinal research suggests that partner violence can become chronic, with DRV victimisation predicting IPV victimisation⁴⁹ and perpetration^{d,56} in adulthood. Evidence suggests that DRV perpetration is associated with subsequent substance use^{53,57} and mental ill health⁵³ among males and females, including suicidal ideation which is more strongly associated with DRV perpetration among females than males.⁵³

Few studies have examined the relationship between DRV and educational outcomes longitudinally,⁵⁸ but available evidence suggests that DRV victimisation is associated with worse educational outcomes among girls and boys including dropping out of school.⁵⁹ In a study assessing adolescent DRV victimisation using women's retrospective reporting in adulthood (i.e. at baseline), Adams et al. found that DRV was associated with lower educational attainment at baseline.⁶⁰ Three subsequent waves of data collection (T1-T3) assessed earnings over the next five years, and modelling found that this educational deficit contributed to a loss in both earnings and growth in earnings. Through this mediator, DRV victimisation was associated with a loss of US\$343 in earnings at T1 and with a loss of growth in earnings of US\$442 by T3.

^d Age at DRV victimisation in this analysis is not specified, but DRV was assessed the year following grades 7-12. Participants would typically have been approximately 14-20 years old.

Considering costs of DRV at the national level, these have not been isolated in the UK but a 2019 Home Office report estimates the annual cost of IPV in England and Wales to be £66 billion.⁶¹ These stem primarily from physical and emotional harms.⁶¹ This body of evidence underscores the importance of early intervention to mitigate immediate fear, injuries and mortality caused by DRV as well as its longer-term health, educational and economic harms.

1.5. Intervening in adolescence to prevent DRV

As the developmental phase when young people begin to form close romantic ties⁶² and risk of DRV emerges, adolescence is a critical intervention point to prevent DRV and mitigate its potential harms. Cognitively, adolescents are developing an understanding of risks and consequences associated with their behaviours and at this age tend to be open to learning about personal responsibility and boundaries if "delivered in a blame-free manner".^{32(p139)} As their interest in romance and sex increases, young people are particularly attentive to "issues involving gender, sexuality, and relationship formation" and enter a stage particularly sensitive to the development of attitudes and beliefs relating to "interpersonal relationships and the abuse of power and control."^{62(p362)} Dating norms governing romantic and sexual relationships are most "malleable" as young people first begin to navigate these relationships because the behaviour is new.^{6(p4)} This section summarises modifiable risk and protective factors for DRV and then provides an overview of approaches to DRV prevention and evidence supporting these.

1.5.1. Modifiable risk and protective factors for DRV

An ecological model based on "the notion of embedded levels of causality"^{63(p264)} was popularised by Lori Heise in 1998 as a framework for conceptualising factors driving GBV.⁶³ The model is widely used and adapted to organise risk and protective factors in epidemiological research on violence in romantic and sexual relationships (e.g. Krug et al, Vézina & Hébert, Heise, and Claussen et al.).^{47,64–66} Figure 1 depicts a simple version of this model, presented by Krug et al. in the World Health Organization's 2002 "World report on violence and health".⁶⁶



Figure 1. An example of a social ecological model used for conceptualising violence⁶⁶

Section 1.5.1. will review non-sociodemographic factors associated with DRV as these are, at least theoretically, modifiable to reduce DRV. I organise these in terms of individual, relationship, community (family and peer) and societal factors, reflecting the categories shown in Figure 1.

An extensive body of research, including several meta-analytic reviews published since 2017, have identified a wide array of modifiable factors significantly associated with DRV. Metaanalyses by Garthe et al., 2017⁶⁷ and Hébert et al., 2017⁶⁸ examine factors associated with DRV victimisation among adolescents, focusing respectively on peer factors and on individual, family and peer factors. Two papers by Spencer et al., 2020 and 2021, present findings from metaanalyses exploring risk and protective factors^e across ecological levels for physical DRV victimisation⁶⁹ and perpetration⁷⁰ among US adolescents. A meta-analytic review by Park and Kim, 2018 examines family and community risk and protective factors for DRV in research among adolescents and/or young adults.⁷¹ Each of these reviews draws on data from many individual studies (ranging from 27 studies in reviews by Garthe et al.⁶⁷ and Park and Kim⁷¹ and 87 in the review by Hébert et al.⁶⁸), which can improve the process of estimation and increase

^e This thesis uses the terms 'risk factors' and 'protective factors' to refer to factors associated with increased and decreased likelihood of reporting DRV outcomes, respectively. This is to align with existing literature, which tends to use these terms whether or not existing research has established a causal link between these factors and DRV.

the power to detect significant relationships that individual studies might be too small to detect.^f

While these studies provide valuable syntheses of a broad range of literature, however, they should be read with a note of caution due to high levels of heterogeneity across individual studies.^{67,68,71} Furthermore, authors of reviews of factors associated with DRV note that the evidence base is limited by the cross-sectional nature of most relevant studies, which precludes determining whether identified factors are causally linked to DRV; and if so, whether they contribute to or are consequences of DRV.^{40,67,69,70} Meta-analytic reviews by Garthe et al., Hébert et al. and Spencer et al. include both cross-sectional and longitudinal studies but do not differentiate by study type in their main analyses.^g To explore differences between the estimates derived from cross-sectional and longitudinal studies, Hébert et al. conducted a moderation analyses by study design, the results of which are reported in Section 1.5.1.5., "Key DRV risk factors highlighted in meta-analytic reviews" below.

A number of other systematic reviews have further synthesised quantitative research on factors associated with DRV (e.g., Capaldi et al., 2012; Clausson et al., 2022; Johnson et al., 2015; Leen et al., 2013; Malhi et al, 2020; Vagi et al., 2013; and Vézina and Hébert, 2007).^{22,40,47,64,72–74} Like meta-analytic reviews, these are also based on the results of systematic searches of the DRV literature. Their findings supplement the evidence base from meta-analytic reviews by reporting on factors that are associated with DRV but have not been sufficiently extensively researched for inclusion in meta-analyses. While most of these non-meta-analytic reviews synthesise both cross-sectional and longitudinal research, they often distinguish between these in the presentation of their results. Vagi et al.'s review of factors associated with DRV

^f Garthe et al.,⁶⁷ Hébert et al.⁶⁸ and Spencer et al.^{69,70} each draw on at least two, and for most factors many more, estimates from independent samples to arrive at each of their estimates. Park and Kim do not report on the number of estimates included in the meta-analyses for each of the 17 factors on which they report.⁷¹ However, they report that these analyses draw on 139 correlates of DRV victimisation and 131 correlates of DRV perpetration from 27 included studies in total.

^g Park and Kim do not report on whether the studies included in their meta-analytic review are crosssectional or longitudinal.

perpetration includes only longitudinal studies, focusing on factors that have been shown to precede DRV temporarily.⁷⁴ While this temporal relationship is not sufficient for establishing causality, the authors argue that factors that are both associated with and precede DRV perpetration "represent the best available targets for prevention programs" at the time of their review.^{74(p634)}

Like Park and Kim's meta-analytic review, reviews of factors associated with DRV often aggregate studies of both adolescents and young adults.^{47,71,72,74} Relationships in these developmental stages tend to be more similar to each other than to those of older adults; the latter are more likely to have long relationships, cohabit, marry and have children together.⁷² Meta-analytic evidence on modifiable individual, family and peer factors suggests that the risk profiles are nearly identical for adolescent and young adults, providing empirical evidence in support of this approach.⁶⁸

The following sections (Sections 1.5.1.1. to 1.5.1.6.) summarise available meta-analytic evidence on factors associated with DRV.^{67–71} This is supplemented with further evidence from non-meta-analytic systematic reviews^{22,40,47,72–74} and draws on evidence from individual studies not included in these reviews where these add further insights to the review evidence.^{75–82} They then present a summary of meta-analytic evidence about factors with the strongest relationship to DRV, followed by a summary of available evidence on DRV risk factors from UK studies. Moderation analyses in meta-analytic reviews tend to find little difference in DRV risk factors by sex or gender,^{67,69,71} but where studies report differences in the significance of associations by sex or gender I note this. Finally, in these sections I provide effect estimates where these report on moderation analyses that statistically compare the relative importance of different risk and protective factors and where these report on UK research. As noted above, findings from meta-analytic reviews combine data from cross-sectional and longitudinal studies. Other findings presented in these sections are based on cross-sectional studies unless otherwise specified.

1.5.1.1. Individual factors

Evidence from meta-analyses suggests that DRV is associated with experience of other forms of abuse, with health behaviours and with personal characteristics; and that DRV victimisation and perpetration share several common risk factors. Other systematic reviews and individual longitudinal studies extend this research.

DRV victimisation and perpetration

Studies identify several individual-level factors associated with both DRV victimisation and DRV perpetration. Among girls and boys, meta-analyses find that childhood maltreatment is associated with DRV victimisation among adolescents⁶⁸ and with DRV perpetration in a combined sample of adolescents and unmarried young adults.⁷¹ Both depression and externalising behaviours are also associated with physical DRV victimisation and perpetration in meta-analyses.^{69,70} Further evidence on behavioural factors come from non-meta-analytic systematic reviews, which identify associations between externalising problems and both boys' DRV perpetration²² and adolescent girls' and young women's DRV victimisation.⁴⁷

Considering health-related behaviours, in Spencer et al.'s meta-analyses focusing on physical DRV, adolescents who use substances and report risky sexual behaviours are more likely to report physical DRV victimisation and perpetration.^{69,70} This research also finds that proviolence attitudes are associated with physical DRV victimisation and perpetration.^{69,70} Other studies examine relationships between personal attitudes and DRV outcomes in more depth. Systematic reviews report relationships between specifically pro-DRV attitudes and both DRV victimisation and perpetration.^{22,47} While some research supports a longitudinal relationship between such attitudes and subsequent DRV, findings are inconsistent.²² Studies do show that the pattern of relationships between DRV attitudes and outcomes can differ for attitudes towards girls' versus boys' perpetration of DRV and by respondent gender.^{22,75} Considering views on gender, boys who hold more gender-equitable attitudes or are less supportive of traditional gendered expectations are significantly less likely to perpetrate DRV.^{73,80,81} Longitudinal research by Reyes et al. on adolescent male DRV perpetration suggests that DRV and gender attitudes might interact to influence DRV outcomes.⁸² Their research finds that the relationship between gender-inequitable attitudes and DRV perpetration is attenuated by anti-DRV attitudes: gender-inequitable views are associated with DRV perpetration among boys with high but not low tolerance of DRV.⁸²

DRV victimisation

Exploring the relationship between child maltreatment and adolescent DRV victimisation in more depth, Hébert et al.'s meta-analysis finds significant associations with experiences of child sexual abuse, psychological abuse, physical abuse and neglect.⁶⁸ Considering health and behavioural factors, Spencer et al.'s meta-analysis focusing on physical DRV victimisation identifies associations with disordered eating and suicide attempts as well as carrying a weapon, while adolescents with good physical health are less likely to report physical DRV victimisation in their analysis.⁶⁹ Their work also suggests that modifiable personal characteristics might play a role in DRV risk: communication skills and self-esteem are associated with less physical DRV victimisation.⁶⁹

DRV perpetration

In Spencer et al.'s meta-analysis, anger, delinquency and controlling behaviours all show a significant relationship to physical DRV perpetration while conflict resolution skills and responsibility are associated with less of this type of perpetration.⁷⁰ Disaggregated findings from non-meta-analytic systematic reviews provide some further insights. This research suggests that several individual-level DRV risk factors predict DRV perpetration longitudinally: mental health difficulties (depression, anxiety and emotional distress), substance use, risky sexual behaviour, and a history of aggression or anti-social behaviour.^{22,40,74,77} Leen et al.'s review further identifies internalisation (withdrawal, somatic complaint, anxiety, depression, obsession and compulsion) as a risk factor for girls' DRV perpetration.²²

1.5.1.2. Relationship factors

Meta-analytic and other systematic reviews identify several characteristics of young people's intimate partners and relationships that are associated with increased DRV victimisation and perpetration.

DRV victimisation and perpetration

Meta-analysis by Spencer et al. finds associations between DRV victimisation and DRV perpetration, as well as associations between different types of DRV.^{69,70} Specifically, adolescents are more likely to report physical DRV victimisation if they also report DRV perpetration, or victimisation from other types of DRV.⁶⁹ In parallel, adolescents are more likely to report physical DRV victimisation or other types of DRV.⁶⁹ In parallel, adolescents are more likely to report physical DRV perpetration if they also report DRV victimisation or other types of DRV.

DRV victimisation

In their systematic review of risk factors among adolescent girls and young women Vézina and Hébert report that having an older partner is associated with DRV victimisation.⁴⁷ Their findings also suggest that dyadic power dynamics might help to explain this relationship, considering evidence that adolescent girls and young women who perceive that their partner has more control in the relationship are more likely to report DRV victimisation.⁴⁷

DRV perpetration

Evidence from non-meta-analytic systematic reviews highlights characteristics of intimate relationships that are associated with DRV. Longitudinal research suggests that conflict and hostility within adolescent relationships (assessed at 14-19 years) are associated with subsequent physical DRV perpetration (assessed at age 15-20 years).⁷⁴ Considering partner characteristics, evidence from DRV research among gang members suggests that boys in age-disparate relationships with younger female partners are more likely to perpetrate DRV than are boys who are younger than or closer in age to their partners.⁷³

1.5.1.3. Community factors

Several meta-analytic reviews have examined relationships between family- and peer-related factors and DRV outcomes, and this work is extended by other systematic reviews and by individual studies exploring these factors. This research suggests that family relationships, parenting practices, peers' behaviours and views and young people's perceptions of what their peers think and do are associated with DRV, as are young people's experience with their peers outside of intimate relationships.

Family factors

DRV victimisation and perpetration

Adolescents who witness IPV between their parents are more likely to report both victimisation⁶⁹ and perpetration⁷⁰ in Spencer et al.'s meta-analyses. In their meta-analysis of adolescent and young adult research, Park and Kim find that family relationship problems and reporting fear of family violence are also associated with both DRV victimisation and perpetration.⁷¹ Their review also presents aggregate indicators of 'positive' and 'negative' parenting.⁷¹ Positive approaches include parental warmth, monitoring, support and communication, while negative approaches comprise harsh or inconsistent discipline, harsh or hostile parenting, negative interactions, parent-child boundary violations and low trust or support.⁷¹ Young people reporting negative parenting have higher levels of DRV victimisation and perpetration while positive parenting is associated with lower levels of both.⁷¹

DRV victimisation

Considering specific parenting practices, meta-analytic reviews suggest that parental support^{68,69} and monitoring⁶⁸ are associated with less adolescent DRV victimisation.

DRV perpetration

Meta-analytic reviews suggest that parental separation (not living together, or divorcing)⁷¹ and child abuse victimisation within the family are both associated with DRV perpetration.⁷⁰
Multiple longitudinal studies suggests that problems within parental relationships^h (parental marital conflict and exposure to parental IPV) and poor parenting practicesⁱ (low parental monitoring and harsh or unskilled parenting) are associated with subsequent DRV perpetration.⁷⁴

Peer factors

DRV victimisation and perpetration

Meta-analytic reviews find that adolescents reporting peers' aggressive and antisocial behaviour or DRV among their peers are more likely to report their own DRV victimisation or perpetration,⁶⁷ as are adolescents and young adults reporting deviant peers.⁷¹ Experience with other types of peer abuse are also associated with increased DRV. Meta-analytic evidence suggests that adolescents and young adults who report bullying (a combined indicator of victimisation and perpetration)⁷¹ and adolescents who are violent towards their peers^{69,70} report higher levels of both DRV victimisation and perpetration. Conversely, peer relationships might also play a protective role. Park and Kim's meta-analytic review found higher friendship quality to be associated with lower levels of adolescent and young adult DRV victimisation and perpetration.⁷¹ As individual studies increasingly examine relationships between friends' attitudes towards DRV and DRV outcomes, some are finding cross-sectional associations between friends' approval of DRV, and DRV victimisation^{76,78} and perpetration.⁸³

DRV victimisation

Considering relationships between DRV victimisation and other forms of abuse among young people, peer victimisation^{67,68} and peer sexual harassment⁶⁸ are associated with increased DRV victimisation while peer support is associated with less DRV victimisation⁶⁸ among adolescents in meta-analyses. Vézina and Hébert highlight a potential relationship between friends' attitudes towards violence, and adolescent girls' and young women's DRV victimisation in their non-meta-analytic systematic review.⁴⁷ These findings are supported by longitudinal research

^h Based on three studies reported by Vagi et al.⁷⁴

ⁱ Based on three studies reported by Vagi et al.⁷⁴

suggesting that friends' involvement in DRV predicts subsequent DRV victimisation among girls.³³

DRV perpetration

Non-meta-analytic systematic reviews report on longitudinal research identifying bullying and friends' DRV involvement as risk factors for subsequent DRV perpetration. In their systematic review of literature on boy's DRV perpetration, Malhi et al. report research suggesting that bullying is associated with subsequent DRV perpetration among boys.⁷³ In their review of longitudinal risk factors for DRV perpetration, Vagi et al. report that having friends who have experienced DRV is associated with subsequent DRV perpetration among girls while having friends who have perpetrated DRV is associated with DRV perpetration both among boys and in a combined sample of girls and boys.⁷⁴

1.5.1.4. Societal factors

Evidence from systematic reviews suggests that young people's DRV risk is associated with characteristics of their social system, referred to here as societal factors. These findings are extended by findings from other systematic reviews and from individual cross-sectional and longitudinal studies. Evidence at this level of the social-ecological model tends to focus on associations between neighbourhood and school factors and DRV outcomes.

DRV victimisation and perpetration

In their meta-analytic review, Park and Kim report that a measure combining neighbourhood hazards and ethnic heterogeneity is associated with adolescent and young adult DRV victimisation and perpetration, but they do not disaggregate findings for these two disparate neighbourhood characteristics.⁷¹ Other research suggests that young people's relationship to and perceptions of their school also appear to play a role in DRV risk. Other systematic reviews report that a sense of attachment to school is associated with less subsequent DRV perpetration⁷⁴ and with less DRV victimisation among girls and young women.⁴⁷ Conversely, school-level bullying victimisation,⁷⁹ and rating the school low on safety, connectedness and

maintenance,⁷³ are positively associated with DRV victimisation among US adolescents and with DRV perpetration among boys in South Africa, respectively.

DRV victimisation

In the meta-analysis conducted by Spencer at al. neighbourhood disorganisation is associated with higher levels of physical DRV victimisation among adolescents.⁶⁹ Other studies provides further insights into relationships between neighbourhood and school environments and DRV victimisation. Some research suggests an association between neighbourhood violence and more DRV victimisation among girls and young women.⁴⁷ Longitudinal research finds that the school environment interacts with family-level factors to influence subsequent DRV risk: among girls, family disadvantage is more strongly associated with subsequent DRV victimisation in "more economically advantaged" schools.^{40(p10)} Evidence from a large cross-sectional survey of more than 100,000 adolescents in California schools also suggests that associations between school-level factors and DRV victimisation can be attenuated by other school-level factors.⁷⁹ Among students who were in a relationship in the previous year, school-level school connectedness, caring relationships with school staff, opportunities for participation and a sense of safety in school were associated with lower DRV victimisation.⁷⁹ These relationships (with the exception of opportunities for participation), however, were moderated by schoollevel bullying victimisation; for example, school connectedness was more strongly associated with lower DRV victimisation in schools with lower levels of bullying victimisation.⁷⁹

DRV perpetration

Park and Kim's meta-analysis suggests that young people who report positive neighbourhood characteristics like support, monitoring and collective efficacy are less likely to perpetrate DRV.⁷¹ Evidence from the same review links the school environment to DRV outcomes, identifying an association between a composite factor of school attachment, support and attainment with less DRV perpetration.⁷¹

Considering further evidence from non-meta-analytic studies, longitudinal research finds that school-related factors interact with individual- and family-level factors to influence subsequent DRV risk.⁴⁰ School bonding (assessed in terms of feeling like school is like a family) decreases girls' risk but increases boys' risk of subsequently perpetrating a combination of peer violence and DRV versus peer violence alone.^{40,84} School-related factors have also been found to moderate the relationship between family violence and subsequent DRV perpetration differently by a combination of ethnic group and sex.⁴⁰ In a US longitudinal study assessing DRV between ages 16 and 20 years among African American, Hispanic and White/other ethnic groups, lack of school safety exacerbated the relationship between parental IPV and subsequent perpetration among African-American males only.^{40,85} Early school involvement, on the other hand, had this effect among Hispanic females only.^{40,85}

1.5.1.5. Key DRV risk factors highlighted in meta-analytic reviews

Several meta-analytic reviews report moderation analyses, enabling comparisons of effect sizes across study designs, outcome measures, participant characteristics and risk and protective factors. Concerning study methods, Hébert et al. (whose moderation analyses include both adolescent and young adult samples) report that effect sizes for DRV victimisation are attenuated for some risk factors in longitudinal versus cross-sectional studies.⁶⁸ They also report that effects can be moderated by DRV type and by the instrument and approach (binary versus continuous) used to measure DRV outcomes.⁶⁸ Several reviews examined effect modification by gender, and the vast majority of these analyses identify no significant differences.^{67,69,71} Where they do find differences, reviewers report that physical abuse is more strongly associated with DRV victimisation among girls (r=0.12) than boys (r=0.07), $Q_B(1)$ =4.25, p=0.39, and that sexual harassment is also more strongly associated with DRV victimisation among girls (r=0.26) than boys (r=0.14), $Q_B(1)=6.72$, p=0.010).⁶⁸ Other gender differences are reported for depression, which is more strongly associated with physical DRV perpetration among girls (r=0.11, 95%CI=0.07, 0.14) than boys (r=0.03, 95%CI=-0.01, 0.06) and controlling behaviours, which are more strongly associated with physical DRV perpetration among boys (r=0.28, 95%Cl=0.14, 0.40 versus 0.09, 95%Cl=-0.03, 0.22).⁷⁰

Evidence is mixed on risk factors with the strongest relationship to DRV. Hébert et al. report no significant difference between the 12 child maltreatment, peer and parenting risk factors identified in their study,⁶⁸ but other meta-analytic reviewers highlight the comparative strength of a number of factors.^{67,69,71} Factors that Spencer et al. highlight as most strongly associated with both physical DRV victimisation⁶⁹ and perpetration⁷⁰ at the individual level include substance use (r=0.55, 95%CI=0.47-0.63 substance use and victimisation; r=0.09, 95%CI=0.02=0.17 alcohol use and perpetration), risky sexual behaviours (r=0.34, 95%CI=0.24-0.44 victimisation; r=0.16, 95%CI=0.08-0.23 perpetration) and some mental health challenges (r=0.30, 95%Cl=0.28-0.32 suicide attempts and victimisation; r=0.30, 95%Cl=0.25, 0.36 disordered eating and victimisation; *r*=0.08, 95%CI=0.04-0.12 depression and perpetration). They further identify weapon-carrying as among the strongest individual risk factors for physical DRV victimisation (r=0.31, 95%CI=0.22-0.39)⁶⁹ and externalising behaviours (r=0.33, 95%CI=0.18-0.46), pro-violence attitudes (r=0.19, 95%CI=0.14, 0.24) and delinquency (r=0.06, 95%CI=0.00-0.11) as the strongest at this level for perpetration.⁷⁰ At the relationship level, Spencer et al.'s research suggests that DRV victimisation and perpetration and different types of DRV are closely associated with each other. They highlight emotional DRV victimisation as an important risk factor for physical DRV, both in terms of victimisation (r=0.51, 95%CI=0.42-(r=0.49, 95%Cl=0.37, 0.59).⁷⁰ They further highlight physical DRV perpetration (*r*=0.66, 95%CI=0.61, 0.70) and sexual DRV victimisation (*r*=0.53, 95%CI=0.45-0.59) as important risk factors for physical DRV victimisation.⁶⁹ At this level they identify physical DRV victimisation (r=0.66, 95%CI=0.56, 0.74), emotional DRV perpetration (r=0.37, 95%CI=0.30, 0.63) and past physical DRV perpetration (r=0.41, 95%CI=0.27, 0.53) as the factors most strongly associated with physical DRV perpetration.⁷⁰ Park and Kim highlight witnessing parental IPV (ESr=0.48, 95%CI=0.36, 0.60), a family-level factor, for its strength of association with DRV victimisation among adolescents and young adults.⁷¹ Considering peer factors, metaanalytic reviewers highlight antisocial behaviour (r=0.29, 95%CI=0.20, 0.37) for the strength of its associations with DRV victimisation and perpetration⁶⁷ and, among adolescents and young

adults, peer deviance (ESr=0.46, 95%CI=0.39, 0.52) for the strength of its association with DRV perpetration.⁷¹

1.5.1.6. Modifiable risk factors identified in UK surveys

Most evidence on modifiable DRV risk factors in the UK comes from a few cross-sectional studies.^{16,34,35} At the individual level, victimisation is associated with ever having sent a sexually explicit image (aORs=2.91-7.97^{j16,35}) and (for girls) with living independently (aOR=4.03, 95%CI=2.19, 7.41 threatening behaviours; aOR=1.74, 95%CI=1.33, 2.28 controlling behaviours).¹⁶ Multivariable analyses by Barter et al. have further identified relationship, family and peer factors significantly associated with DRV risk among UK young people at the level of p < 0.05.³⁴ In this research partner age, family violence and peer group violence emerge as the modifiable risk factors most strongly associated with both DRV victimisation and perpetration. Their findings suggest that at the relationship level, having a younger (compared to same-age) partner is associated with lower risk of both physical DRV victimisation (aOR=0.41) and emotional DRV perpetration (aOR=0.45) for boys while having an older (as opposed to younger) partner is associated with physical DRV victimisation for girls (aOR=4.91, p<0.05).³⁴ At the family-level, family violence is associated with both DRV victimisation among girls (aOR=2.77 physical; aOR=1.80 emotional; aOR=2.36 sexual) and boys (aOR=2.77 sexual) and perpetration among girls (aOR=2.18 physical; aOR=3.97 sexual).³⁴ Considering peer factors, peer violence is associated with DRV victimisation among both girls (aOR=2.22 physical; aOR=2.46 emotional) and boys (aOR=2.30 physical, aOR=2.06 emotional) as well as perpetration among both girls (aOR=2.69 physical; aOR=3.83 sexual) and boys (aOR=3.12 physical; aOR=2.17 emotional; aOR=3.06 sexual).³⁴

^j For girls/boys respectively: aOR=2.31, 95%CI=2.04-2.62/aOR=2.91, 95%CI=2.01-4.23 threatening behaviours; aOR=4.25, 95%CI=3.43-5.26/aOR=2.49, 95%CI=2.05-3.02 controlling behaviours; aOR=7.97, 95%CI=3.63-17.52/aOR=4.25, 95%CI=3.43-5.26 online sexual violence¹⁶

1.6. Approaches to DRV prevention

A number of narrative and systematic reviews synthesise evidence on approaches to DRV prevention. Many focus primarily on adolescents.^{7,8,11,13,14,22,28–30} Others include interventions targeting adolescents and young adults into their early 20s,^{10,12,31} a reasonable approach because adolescent and young adult relationships are more similar to each other than to relationships in later adulthood.⁷² Some discuss DRV prevention alongside broader IPV prevention.^{9,32} Given their shared risk factors, DRV and non-partner GBV are considered together in some reviews,^{10–12,14,30} while others focus exclusively on abuse within the context of intimate relationships.^{7,8,8,9,9,13,22,28,29,31,31,32} Table 1 summarises the methods and key findings from 14 reviews, published between 1999 and 2024, of interventions that aim to reduce DRV. In this section I synthesise the findings of these reviews, noting where evidence comes solely from reviews that combine DRV and GBV interventions and/or interventions for adolescents and young adults.

Table 1. Summaries of selecte	d reviews of DRV interventions
-------------------------------	--------------------------------

Year	Review	Methods	Dates of eligible	# of included interventions/	Eligible design(s)	Eligible interventions	Key outcome findings	Other key findings and recommendations
1999	Wekerle ⁷	Review (does not specify whether systematic)	1990 and later, though manual searches included journals from 1980 and later	6 interventions	Quasi- experimental and intervention- only designs	Adolescent relationship violence prevention programmes	 Studies find evidence of impact on knowledge, attitudes about dating aggression, and behavioural intentions, maintained over short follow-up periods (e.g., 3 months) Two studies found intervention participants report less perpetration than control at post-test; however, this is based on self-report and may be subject to social desirability bias Unclear whether there is a short-term impact on victimisation 	 Identified 4 school-based programmes, 1 community- based and 1 combined Most interventions integrated feminist and social learning approaches School-based programmes provided practical benefits (e.g. access to participants; space), and the benefit of staff support (e.g. programme facilitators, and guidance counsellors to address incidents and to follow up post-programme) Suggests harnessing peer pressure "to serve prosocial ends" (p. 451) Importance of teaching pro- social skills (e.g. assertive communication; help-seeking)
1999	Wolfe ³²	Review (non- systematic)	N/A	N/A	N/A	Describes 2 public health models for intimate partner violence (IPV) prevention, gives examples of prevention programmes	 Cites a 1999 review finding that school- based dating and relationship violence (DRV) prevention programmes targeting adolescents have had positive impacts on knowledge, attitudes and DRV perpetration 	 Schools provide a good opportunity for IPV prevention to a wide range of young people because Most children attend school Much of their social learning takes place in school, and social learning "can play a role in the development of behaviors and attitudes that

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible	interventions/	design(s)	interventions		recommendations
			reports	studies/reports				
								 By mid-adolescence, romantic relationships become more important "Clear messages about personal responsibility and boundaries, delivered in a blame-free manner, are generally acceptable to this age group" (p. 139) Need to move beyond small, scattered local programming to comprehensive research and evaluations to support broader prevention
2006	Whitaker ¹³	Systematic review	1990 – April 2003	11 interventions (15 reports)	Comparison group (e.g., quasi- experimental, randomisation to intervention versus control, or randomisation between 2+ interventions), and pre/post designs	Interventions for the primary prevention of partner violence perpetration (initially intended to be broad, but all eligible studies targeted adolescents)	 Of the 9 studies reporting positive outcomes, most report effects on knowledge or attitudes; unclear whether this will lead to behaviour change Two programmes found positive impacts on behaviour (Safe Dates and the Youth Relationships Project) 	 All interventions "had some emphasis of a feminist orientation to partner violence" (p. 159), discussing how concepts like gender norms, gender-based coercion or power and control contribute to DRV. Most were underpinned by a combination of feminist theory, either social-cognitive or cognitive-behavioural theory (NB, the narrative was unclear as to whether authors were referring to both of these, or to one or the other) and educational methods All targeted middle- or high- school age students; all but one were universal and school- based interventions

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible	interventions/	design(s)	interventions		recommendations
			reports	studies/reports				
								All interventions were delivered
								to mixed-sex groups
								 Interventions tended to be
								brief, with 6 of the 11 shorter
								than 5 hours, and only 2 longer
								than 5 hours (excluding
								activities outside of the
								structured curriculum)
								 Overall quality of evaluation
								designs was low, with short
								follow-up periods; high attrition
								rates; little fidelity monitoring;
								lack of measurement of
								perpetration; and it was
								uncommon to conduct
								mediator analyses
								While DRV prevention
								programmes are promising for
								preventing IPV perpetration,
								"strong conclusions about
								[their] effectivenessare
								premature." (p. 160)
								More work is needed to
								understand the mechanisms by
								which DRV programmes change
2007	Corpolius ²⁸	Doviour	Not	11	Not coolified.	Drimony and		Denaviour
2007	Comenus	REVIEW	specified	interventions	discusses a	secondary	 N/A; Discusses 	 Wost DKV prevention programmes target secondary
			specifieu,	Interventions	range of design	interventions to	programmes,	school aged or university aged
			referenced		and reporting	nrevent	avidence base and	young people "usually within
			range from		from a reporting	adolescent	recommendations for	the school or the curriculum"
			1987 -		on an	dating violence	future research but	(n 366)
			2004		intervention		does not synthesise	 Though several programmes
					without		accontrol synthesise	have been implemented since

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible	interventions/	design(s)	interventions		recommendations
			reports	studies/reports	outcome data, to randomised controlled trials (RCTs)		existing evidence on outcomes	 the 1980s, many have not been evaluated to assess impact on attitudes and behaviours Limited comparability between studies which use bespoke scales, without necessarily examining reliability and validity, to measure outcomes Studies using self-report data to measure outcomes rarely assess social desirability; self-report data may underestimate incidence of dating violence, especially among adolescents "who may be accustomed to responding in ways in order to please a perceived authority figure" (p. 372) Only some programmes include/describe a skill-building component, which the review authors argue "need to be incorporated consistently and methodically" (p. 373) to change behaviour
								examine longitudinal behaviour
2013	Leen ²²	Systematic review (not described as systematic, but methods	2000 – 2011	9 interventions	Not specified; must be published in a peer-reviewed journal	"Primary" adolescent dating violence interventions for young people aged 12-18 years	 Several programmes demonstrated positive behavioural change Several demonstrate effects on interpersonal and 	 All included studies evaluated interventions in North America; none identified in Europe All included a focus on healthy relationship skills

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible	interventions/	design(s)	interventions		recommendations
			reports	studies/reports				
		appear to be systematic. Forms part of a broader review of DRV prevalence, risk factors and intervention efficacy)				(this is how the authors describe the interventions, although the review includes both primary prevention interventions and those for young people considered to be at high risk of DRV, including those with previous DRV experience)	relationship attitudes but "marginal or no behavioral change" (p. 169). Changes in attitudes alone does not necessarily lead to changes in behaviour; "The link between effecting attitudinal change and effecting behavioural change appears far from straightforward." (p. 171) • Though there is limited evidence to make a determination, authors note that interventions that focus on awareness- raising and knowledge tend to have less success in effecting long-term behaviour change than those that focus on relationship behaviour, skills and attitudes • Some interventions report negative effects on DRV behaviour, though it's unclear whether these	 Most interventions (6 of 9) were school-based and took place during scheduled lessons; another was located at a school but took place during after- school time Findings on the impact of intervention delivery in groups are conflicting; it is "unclear what effects group dynamics have as a driver of attitudinal and behavioral change" (p. 171) Given that post-test intervention effects are not necessarily sustained at later follow-up, evaluations should use longer follow-up periods

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible	interventions/	design(s)	interventions		recommendations
			reports	studies/reports				
2013	Fellmeth ³¹	Cochrane systematic review and meta- analysis	Until 7 May 2012	interventions/ studies/reports 38 studies (41 reports); 33 studies included in meta-analysis	design(s) RCTs, cluster RCTs and quasi- randomised studies with a control	interventions Educational and skills-based interventions targeting adolescents and young adults (aged 12- 25 years) to prevent relationship and dating violence.	 are actual DRV behaviour changes or changes in reporting Significant increase in knowledge about DRV (but suggest interpreting this with caution due to high heterogeneity between studies). Moderated by setting, which is correlated with age: university- based interventions more effective at 	 All studies took place in the USA (N=37) or the Republic of Korea (N=1). Most studies were in educational settings (25 in universities, 10 in high schools) Outcomes did not vary by total contact hours, number of sessions or timing of outcome measurement Limitation: Safe Dates not included in this review
						Studies with a wider age- range were included if at least 80% of the participants were aged 12- 25 years.	 increasing knowledge; effect of school- and community-based interventions on knowledge was not significant. However, because no other outcomes were moderated by setting, the authors conclude this moderation arose by chance. No significant effect on episodes of DRV (whether measured using categorical or continuous data) No significant effect on attitudes towards DRV or behaviour in 	 Most research on impact of DRV prevention among adolescents and young adults focuses on changes in attitudes and knowledge Future research should measure DRV incidence itself, and involve larger RCTs with longer follow-up periods

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible reports	interventions/ studies/reports	design(s)	interventions		recommendations
2014		Comphall	1060 July	22 studies (21	Quantitative	School based	DRV (e.g., not DRV itself but related behaviours), or skills related to DRV	
2014	De la Rue ³³	campbell systematic review and meta- analysis	2013	23 studies (21 reports)	experimental and quasi- experimental designs with a control group	interventions to reduce dating and sexual violence (mental, physical, and sexual violence and coercion), implemented with students in 4 th – 12 th grade that focused on middle and high schools	 Authors conclude that prevention programmes "show promise in increasing knowledge and awareness", but "impacts on behaviors are less clear and indeed are often not reported" (p. 50) and that the review "did not show substantial changes in perpetration or victimization experiences" (p. 54). Moderate, significant and sustained increases in DRV knowledge Small but significant reductions in attitudes supportive of DRV (but decreased slightly at follow-up) Large reduction in support for rape myths at post-test (but only measured in 4 studies, and unknown if sustained 	 All studies took place in the USA (N=22) or Canada (N=1) Only 1 measured effect on bystander behaviour Relatively few studies measured DRV perpetration and victimisation (4 and 5, respectively) "Prevention programs can have a positive impact, however, the plethora of programs presented and the limited evidence to support behaviour change creates challenges in recommended specific approaches for schools" (p. 55) Important for interventions to consider social context Future research should use RCTs and should go beyond knowledge and attitude measures to explore changes in DRV victimisation and perpetration

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible	interventions/	design(s)	interventions		recommendations
			reports	studies/reports				
							at follow-up – only	
							measured this in 1	
							study)	
							 Moderate, significant 	
							increase in positive	
							conflict management	
							skills at post-test as	
							measured by Conflict	
							Tactics Scale (CTS)	
							 Small but significant 	
							reduction in DRV	
							victimisation	
							incidence at post-test	
							(-0.21), but not	
							sustained at follow-up	
							 No effect on DRV 	
							perpetration at post-	
							test but small but	
							significant decrease at	
	20						follow-up	
2014	De Koker ²⁹	Systematic	(unclear;	6 studies (8	RCTs, cluster	Interventions to	 Interventions that 	Future research should assess
		review	beginning	reports)	RCTs and quasi-	prevent	have been effective	differences in effects based on
			of 2004?)		randomised	primary and	have been based in	gender and prior experience of
			Until end		trials	secondary	multiple settings (both	DRV
			Of			victimisation	school and	Eligible trials took place in the
			February			and	community),	US, Canada and South Africa
			2013			perpetration of	addressed	
						adolescent	relationship skills and	
						numale	focused on key	
						violonco	adults in the	
						targeting young	audiescents	
						neonle aged	toochors paronts and	
						10-19 years	community	
						evoluding	mombors)" (n. 11)	
						excluding	members) (p. 11)	

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible reports	interventions/ studies/reports	design(s)	interventions		recommendations
						interventions focused on specific populations (e.g. young drug users)	 Findings suggest that "comprehensive IPV prevention interventions based in both school and community are effective in preventing IPV perpetration and victimization among adolescents" (p. 11) 	
2015	Stanley ⁸	Systematic review, review of UK grey literature, and consultation with young people and experts	1990 – Feb. 2014 (March 2014 for grey lit)	Database search: 22 interventions UK grey lit: 18 interventions	Meta-analyses, research reviews, controlled studies, before- and-after studies, independent case evaluations, qualitative and ethnographic studies	Interventions with children and young people under 18 years old to prevent them from experiencing and/or perpetrating domestic abuse (paper focuses on schools but search does not seem to have been limited to school-based programmes)	 Information-based programmes can increase short-term knowledge; less evidence that knowledge is retained in the longer term 	 Most programmes underpinned "by an explanation of domestic abuse that drew on social norms and feminist or gender theories and those interventions utilising the 'bystander approach'" (p. 127) Most programmes targeted knowledge and awareness rather than behaviour. Experts discussed an aim of prevention interventions as shifting the climate; discussed "opportunities to use the power of the peer group to construct social norms that challenge domestic abuse" (p. 127) Identified difficulties of transferring programmes across cultures and populations Expert consultation identified importance of organisational readiness in schools, and importance of supporting

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible	interventions/	design(s)	interventions		recommendations
			reports	studies/reports				
			reports	studies/reports				 "across all aspects of a school's work and curriculum" (p. 122) Evidence from UK grey literature and expert consultation that some teachers are not prepared (in terms of confidence or values) to deliver DRV programmes There is an increasing focus on targeting boys to reduce male perpetration rather than targeting girls to reduce their victimisation (found from expert consultations) In consultations, generally agreed that "messages for boys should be positively framed and should avoid a blaming approach that could provide resistance" (p. 127) Some evidence of boys finding programmes to be "anti-men" or "sexist" and resisting their messages (p. 127) Consultation groups emphasised lack of materials
								 emphasised lack of materials for lesbian, gay, bisexual and transgender young people Consultations with young people and experts found they support the involvement of young people in designing and delivering programmes; young people emphasised the

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible	interventions/	design(s)	interventions		recommendations
			reports	studies/reports				
								importance of authenticity,
								which this could aid
2015	Lundgren ¹⁰	Review (describes search methods but review not described as systematic)	1990 and later	61 interventions	Evaluations; excluded editorials, conference abstracts and opinion pieces	Qualitative and quantitative evaluations of interventions to prevent IPV and sexual violence among young people aged 10-19 years (included studies targeting up to 26 years old); excluded programmes from higher- income countries without strong evidence	 Evidence indicates that longer-term interventions with "repeated exposure to ideas delivered in different settings over time" (p. S49) are more effective than single awareness- raising or discussion sessions School-based interventions are promising but have only been implemented in high- income countries (HICs). These should be adapted for other settings and evaluated Evidence for school- based programmes to promote gender- equitable norms is considered to be emerging, because impact on experience and perpetration of violence "remains to be seen" (p. S49) Gender- transformative community based 	 Most included programmes aimed to affect factors like inequitable gender norms, tolerance of sexual violence, and relationship conflict Limited rigorous evidence available; there is a need for "more robust evaluation of promising interventions" (p. S44). Only 6 studies were RCTs and 8 used quasi-experimental designs. Studies tend to be underpowered and not to evaluate outcomes over a period long enough to assess their effects on future perpetration and victimisation There is a lack of "robust standardized measures for behavioral outcomes" (p. S44) Of 61 interventions, identified only 17 in low- and middle- income countries (LMICs) Studies with strong research designs are disproportionately conducted in high-income settings; need to expand the base of rigorous evidence in LMICs Sexual and reproductive health outcomes not often measured

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible	interventions/	design(s)	interventions		recommendations
			reports	<u>studies/reports</u>			 interventions have been effective in preventing IPV and sexual violence; however, feasibility in terms of human resources and cost is unclear Evidence suggests that parenting interventions can be effective in reducing child maltreatment, a risk factor for later IPV and sexual violence; however, no longitudinal research has been done yet to see if such programmes do affect these outcomes Limited evidence suggests it is important that microfinance initiatives include educational, skills and 	• There is little evidence about the "essential elements of effective programs," e.g. ideal dosage and whether single- or mixed-sex programming is more effective (S49)
							components	
2021	McNaughton- Reyes ¹⁴	Systematic review	Before 1 Jan 2020	45 studies of 52 intervention evaluations (61 reports)	Experimental and controlled quasi- experimental evaluations	DRV primary prevention programmes for young people aged 10-19 years;	 Half of included studies reported effective victimisation and/or perpetration prevention 	 The number of evaluations published from LMICs increased steeply from 2010 LMIC and HIC evaluations had similar follow-up periods; 56%

Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
		eligible	interventions/	design(s)	interventions		recommendations
		reports	studies/reports				
		eligible reports	interventions/ studies/reports	design(s)	interventions also included if assessed sexual violence outcomes without specifying the context of dating/ relationship	 There was no significant difference in effectiveness by HIC/LMIC, implementation setting, study design or follow-up period There was a trend towards higher effectiveness with more exposure time, but no significant difference Programmes tended to be more effective in preventing perpetration than victimisation Half of the effective HIC interventions reported prevention of both victimisation and perpetration among both girls and boys Some evidence supports (1) use of self-defence and 	 of studies followed participants for at least 1 year post-baseline Studies in LMICs were more likely than those in HICs to assess girls (45%) and victimisation (60%) only Most HIC studies assessed both victimisation and perpetration outcomes (78%) and measured outcomes among both girls and boys (91%), reflecting "a more gender-neutral focus" than LMIC interventions (p. 7) Most interventions were school-based and universal, not targeted LMIC interventions were more likely to be gender- transformative while HIC interventions were more likely to include healthy relationship education/training More research is needed to understand how DRV prevention programmes work and which programme components trigger important mechanisms
						self-defence and assertiveness training to reduce sexual victimisation among girls, and (2) programme activities	 mechanisms More research is needed on transferability of programme effects across settings and subgroups More than half of the 29
							 Some evidence supports (1) use of self-defence and assertiveness training to reduce sexual victimisation among girls, and (2) programme activities triggering the

Year	Review	Methods	Dates of	# of included	Eligible design(s)	Eligible	Key outcome findings	Other key findings and recommendations
			reports	studies/reports	acsign(s)			
							 mechanisms of delayed sexual debut, fewer sexual partners, reduced acceptance of DRV, gender- equitable norms, increased awareness of DRV community services, conflict management skills, and/or increased family cohesion Sex moderated intervention effects in 25% of the studies that explored this but there was no clear pattern of this moderation 	 programmes examined moderation by subgroup but few (N=4) reported mediation analyses Most LMIC studies took place in a few sub-Saharan African or South Asian countries; most HIC studies took place in the US No interventions focused on sexual- or gender- minority young people Few programmes targeted changes at the levels of community, family or peer networks
2022	Lowe ⁹	Realist review	No date restrictions reported	11 interventions (15 reports)	Intervention studies assessing impact on DRV victimisation or perpetration, supplemented by: protocols, cross-sectional studies on risk factors, qualitative studies of experiences, adult IPV prevention	Primary DRV prevention interventions in LMICs for young people aged 10-19 years, supplemented by literature on adult IPV prevention in LMICs and on implementation of adolescent health and	 Gender- transformative content led to critical reflection on gender and violence attitudes, and on participants' own relationship behaviours, ultimately "reconceptualising what constitutes violence, and what is acceptable behaviour" (p. 15). These processes were facilitated by the 	 Most interventions (64%) were school-based and most (64%) targeted both girls and boys Most interventions (82%) were gender-transformative, focusing on changing gender and violence attitudes and norms. They "used participatory group-based educationfor social norm change" (p. 9) with content on gender, violence, relationships and reproductive health

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible	interventions/	design(s)	interventions		recommendations
			reports	studies/reports				
					studies, and reports on implementation of adolescent health and social interventions	social interventions	 peer-group (often single-sex) format, creating both safe spaces for discussion and opportunities for communication and interpersonal skill- building, increasing self-confidence and expanding peer networks Most interventions (73%) showed reductions in victimisation and/or perpetration 	 Though most interventions aimed to shift gender norms, they tended to measure attitudes rather than social norms Improving measurement of social norms is needed for understanding mechanisms of change and long-term intervention impact Further research is needed to understand the potential of gender- transformative interventions at different stages of adolescent development Interventions tended to neglect drivers of DRV other than gender norms and attitudes, though mixed findings suggested that targeting other factors (e.g., girls' school attendance, agency, assertiveness) can be effective
2023	Verbeek ¹²	Systematic	Until	15 studies of 13	Quantitative	Group-based,	Most significant	Programmes targeted
		review	March	interventions	studies such as	facilitated,	effects were on short-	experiences of sexual and
			2022	(17 reports)	KCIS, quasi-	sexual and	term attitudes and	dating violence or related
					experimental studies and		long-term benaviours	attitudes or norms
					nre-/nost tost	prevention		
					pre-/post-test	programmes		

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible	interventions/	design(s)	interventions		recommendations
			reports	studies/reports				
					evaluations	for males <=25	 Effectiveness tended 	Little is known about impact on
					without a	years old,	to be demonstrated	"theoretical proxies" including
					control group	excluding	more at follow-up	norms and perceived
						programmes	than at direct post-	behavioural control, which
						that were:	test	were "sparsely investigated" (p.
						single-session;		2899)
						treatment for		 Of the Theory of Planned
						perpetrators;		Behaviour constructs
						and/or mixed-		(behaviours, attitudes, norms,
						gender		perceived behavioural control,
								intentions), behaviours and
								attitudes were most assessed;
								social norms were assessed in
								only 1 study
								 Studies assessed impact on
								perpetration and on bystander
								behaviour; not on victimisation
								 LMIC interventions tended to
								be community-based,
								facilitated by peers or
								community leaders; HIC
								interventions tended to be
								school-based, facilitated by
								professionals or teachers
								 Programmes tended to address
								gender and violence but not
								attitudes relating to masculinity
								• Evaluations could be improved
								by matching intended and
								assessed outcomes (e.g. by
								assessing norms if they are a
								target of the intervention).
								Evaluations tended to assess
								outcomes at a single time-point

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible	interventions/	design(s)	interventions		recommendations
			reports	studies/reports				and might be improved by
								multiple and longer-term
								follow-ups
2024	Melendez- Torres ¹¹	Systematic review	Database inception to June 2021	107 interventions assessed in process/ implementation studies; 57 interventions assessed in outcome evaluations	RCTs, process evaluations	School-based interventions addressing DRV victimisation/ perpetration among children aged 5-18 years (review also included gender-based violence [GBV] interventions but results extracted here were disaggregated by DRV versus GBV)	 Meta-analyses found long-term impacts on DRV victimisation and perpetration Heterogeneity within and across studies; differences in effectiveness not explained by intervention type Effectiveness sometimes seen in reducing DRV frequency but not prevalence Found short-term improvements in DRV knowledge and in DRV and personal help- seeking attitudes Effects on long-term victimisation might be improved when intervention excludes a parental component (long-term victimisation), includes "a range of opportunities for guided practice of skills and attitudes 	 Teaching about gendered aspects of DRV could alienate male staff and students Few studies (N=3) reported social norms outcomes; norms outcomes were reported much less commonly than were knowledge or attitude outcomes It might be that interventions impacted social norms but this was not, or not effectively, measured. Further research on measure reliability and validity "would be useful" (p. 237) Violence attitudes (two studies) and gender attitudes (one study) mediated impact on DRV victimisation Violence attitudes (three studies), gender stereotyping (one study) and belief in the need for help (one study) mediated impact on DRV perpetration Interventions that improved DRV attitudes did not necessarily improve DRV outcomes, suggesting that attitude shifts "may not be

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible	interventions/	design(s)	interventions		recommendations
			reports	studies/reports				
							 and interpersonal components focusing on student relationships" (long- term perpetration, p25), is single- component (short- and long-term victimisation and perpetration) and/or is multilevel; and when sample includes a higher proportion of girls (victimisation) Interventions more effective in reducing perpetration (particularly emotional and physical) among boys than girls, and (according to 1 study) among sexual majority compared to sexual minority students Mixed findings on whether interventions are more effective in reducing perpetration among those with or without a prior history of perpetration Interventions might work by a "basic 	 sufficient for affecting violent behaviour" (p. 154) Mixed findings on whether knowledge mediates impact on DRV victimisation or perpetration Stakeholders noted gap in evidence on DRV among sexual- and gender- minority young people Stakeholders emphasised common factors, including gender stereotypes, underlying DRV, GBV, homophobia and bullying

Year	Review	Methods	Dates of	# of included	Eligible	Eligible	Key outcome findings	Other key findings and
			eligible	interventions/	design(s)	interventions		recommendations
			reports	studies/reports				
							safety" mechanism of	
							communicating	
							unacceptability of	
							violence to increase	
							"student capabilities	
							and motivations	
							concerning the	
							unacceptability of	
							violence" (p. 237)	

CTS=Conflict Tactics Scale

DRV=dating and relationship violence

HICs=high-income countries

IPV=intimate partner violence

LMICs=Low- and middle-income countries

RCT=randomised controlled trial

Approaches to DRV prevention vary by setting, content, duration, delivery model and target population. School- and community-based interventions are especially prominent in the review literature,^{7,8,11,13,22,28,30} and some interventions are designed to be delivered across multiple settings.^{7,10,29} Interventions commonly aim to foster protective social norms at the community level of the social ecological model as well as knowledge, attitudes and skills at the individual level to prevent DRV.^{11,22,29,30} Though some DRV interventions target young people considered to be at high DRV risk, most are universal.^{10,13} Targeted interventions can be "selective", focusing on young people identified as having risk factors associated with DRV, or they can be "indicated", offered to young people who have already experienced or perpetrated DRV.^{13(p162)} Universal interventions, in contrast, are offered to everyone within a defined population (e.g., a participating school) regardless of their individual DRV risk.¹³

Supporting a universal approach, early DRV literature emphasises the principle that young people have an important role to play in helping to protect their peers from abuse.^{7,32} As Werkerle and Wolfe put it, "peer 'pressure' can be harnessed to serve prosocial ends".^{7(p450)} Subsequent reviews report that DRV interventions have continued to target protective social norms as a mechanism to reduce DRV,^{8,11} including via explicit 'bystander' interventions that promote prosocial actions by others at the community level of the social ecological model.^{8,11,86} In their work on social norms and bystander behaviour among university students, Deitch-Stackhouse et al. outline five stages through which an individual must progress to intervene in violence: noticing an event, interpreting it as a problem, feeling responsible to address it, having the skills to do so, and intervening.⁸⁶ Alongside other factors like skills and self-efficacy,¹¹ bystander theories suggest that both individual-level attitudes and community-level social norms can influence progression through these stages.⁸⁶ Bystander interventions use a variety of approaches to address barriers to taking action to reduce violence.^{11,86} Though less common than approaches targeting DRV victimisation or perpetration behaviours directly, more than 25% of interventions included in a 2024 systematic review of school-based DRV and GBV interventions used a bystander approach.¹¹

Theoretical work on the influence of gender inequality on DRV has also long underpinned DRV interventions. In an early review of DRV prevention interventions published in 1999, Wekerle et al. report that most of the six included interventions integrated feminist and social learning approaches.⁷ Subsequent reviews have continued to trace the influence of feminist theory and of 'gender-transformative' approaches – those that "seek to transform gender roles and create more gender-equitable relationships"^{87(p10)} – promoting gender-equitable norms for DRV prevention.^{8,9,13,14} This approach is supported by the broader GBV literature, which documents mechanisms through which gender-inequitable norms contribute to male perpetration of violence against women and girls globally.^{88,89} In their scholarship advocating for a shift in GBV prevention towards changing inequitable gender norms, Jewkes et al. trace how social expectations of dominant forms of masculinity support IPV and non-partner sexual violence both directly and indirectly.⁸⁹ Most directly, these expectations support male dominance and control over women as social ideals, alongside male attributes of physical strength and toughness. Considering individual-level risk factors for GBV perpetration, Jewkes et al. outline the influence of patriarchal norms in behaviours displaying male sexual prowess (having multiple partners, engaging in transactional sex) and in male involvement in other forms of violence. Finally, they highlight that male perpetrators are more likely than other men to report depression, substance use and social or economic marginalisation, suggesting that males who "struggle to live up to a masculine ideal in other respects" are more likely to perpetrate violence against women and girls.^{89(p1584)} Situating inequitable gender norms as an overarching influence, both direct and indirect, over GBV perpetration, this work suggests that gendertransformative interventions might usefully target perpetration itself and a number of its risk factors.89

1.5.2.1. Evidence on approaches to DRV prevention

While much work has been done to synthesise existing evidence and recent reviews report evidence of effectiveness,^{9,11,14} prevention science in the DRV field is still at an early stage. Meta-analyses in 2013 and 2014 systematic reviews of education- and skills-based DRV interventions among adolescents and young adults³¹ and school-based DRV and sexual violence interventions³⁰ found improvements in knowledge. However, evidence of attitude changes was

mixed and these meta-analyses found no³¹ or little³⁰ change to DRV behaviours. A decade later, in 2024, meta-analyses in Melendez-Torres et al.'s systematic review found weak long-term impacts of school-based DRV interventions on both DRV victimisation and perpetration.¹¹ Their findings support previous narrative reviews of DRV and sexual violence interventions,¹⁴ and of DRV interventions for adolescents and young adults in low- and middle-income countries (LMICs),⁹ which found that such interventions are often effective in reducing victimisation and/or perpetration. Melendez-Torres et al.'s review suggests that interventions are more effective in reducing perpetration among boys than girls and that current interventions might be more effective for sexual-majority than sexual-minority young people.¹¹ Findings on whether interventions are more effective for participants with or without a prior history of DRV perpetration are mixed.¹¹

Little is known, however, about how DRV interventions work, which components trigger important mechanisms of change or which intervention models are most effective.^{10,11,14} Reviews report conflicting findings on the role of intervention dose. While a systematic review of DRV interventions for adolescents and young adults suggests that interventions with "repeated exposure to ideas...over time" are more effective,^{10(p549)} the number of contact hours and sessions in education and skills-based interventions among this population made no significant difference to DRV outcomes in meta-analyses³¹ and school-based interventions were more effective when they were single- rather than multi-component.¹¹ In terms of delivery model, review evidence suggests that there might be a benefit to implementing interventions in multiple settings, such as both schools and communities,^{10,29} and across multiple levels of the social ecological model.¹¹ Some evidence supports incorporating a skills component,^{11,22,29} including guided practice of new skills and attitudes, and including a higher proportion of girls in DRV programming.¹¹ Findings are mixed on whether components focusing on the parents of adolescents help or hinder effectiveness.^{11,29}

Evaluations rarely assess what factors mediate DRV reduction,^{11,13,14} but those that do offer some insights into mechanisms of change. DRV studies included in Melendez-Torres et al.'s

2024 review assessed mediation by changes in knowledge, individual attitudes, conflict management skills, belief in the need for help, school belonging and bystander actions.¹¹ Results suggest that increases in DRV knowledge and changes in attitudes towards violence and gender stereotyping can mediate reductions in DRV victimisation and perpetration. However, findings vary across studies and DRV type, and despite the influence of gender-transformative approaches to prevention,^{8,9,13,14} only one study assessed the mediating role of attitudes towards gender stereotypes.¹¹ Furthermore, reviewers report that changes in attitudes do not always lead to reductions in DRV and interventions can be effective without detecting significant attitudinal changes,^{11,22} concluding that the relationship between attitude and behaviour change "appears far from straightforward".^{22(p171)} These findings suggest that social norms might be at work, influencing behaviour independently of attitudinal influences.¹⁵

Reviewer syntheses also point to other mechanisms that might be important, including delayed sexual debut, fewer sexual partners and increased family cohesion for DRV and sexual violence interventions;¹⁴ and communication and interpersonal skills, self-confidence and the expansion of peer networks for DRV prevention among adolescents and young adults in LMICs.⁹ However, conclusive findings on key components and mechanisms to reduce DRV remain elusive.

1.5.2.2. School-based DRV prevention

Two decades of research has consistently highlighted school-based programming as a promising approach to reducing DRV.^{7,10,11,13,22,32} Schools offer an infrastructure for intervention delivery and enable DRV interventions to reach students at-scale.^{7,90} As key sites of social learning³² and gender socialisation,^{91,92} schools can play an important role in the formation of DRV-related attitudes and behaviours and they employ teachers who are positioned to "motivate students to consider new ways of thinking and behaving".^{32(p138)}

As noted in Section 1.6.2., earlier meta-analyses found little impact of school-based interventions on DRV and sexual violence outcomes³⁰ or of education and skills-based interventions on DRV among adolescents and young adults,³¹ but more recent meta-analyses found weak but significant long-term reductions in DRV victimisation and perpetration.¹¹ While

this is encouraging in terms of the potential of school-based interventions, heterogeneity amongst included studies was high and was not explained by differences in intervention models.¹¹ Available evidence suggests that school-based interventions can be effective in reducing DRV but concludes little about what intervention designs are likely to be most effective and the mechanisms underlying intervention effects.¹¹

1.5.2.3. Limitations to the DRV prevention evidence base

Reviews identify limitations to the evidence base for DRV prevention in terms of its geographic representation, applicability for SGM young people, impact and mechanisms of change. First, the overwhelming majority of evaluations of DRV and sexual violence interventions and a disproportionate number of those with strong research designs have taken place in high-income settings,¹⁰ primarily in North America.^{30,31} Melendez-Torres et al. identified only three DRV outcome evaluations from the UK, all cluster randomised controlled trials (RCTs), published in 2012, 2014 and 2020. None reported DRV reductions.¹¹ Second, reviewers have noted a dearth of DRV programming targeting SGM young people^{8,11,14} despite ample evidence of their elevated risk.^{42–44} These gaps present a challenge to reducing DRV globally and equitably, as evidence suggests DRV interventions cannot be simply transferred from one culture or population to another,⁸ even between high-income Western settings.⁹³

Third, DRV intervention evaluations often focus on attitude and knowledge changes,¹³ and many do not measure effects on DRV victimisation and perpetration.^{13,28,30} A lack of robust, standardised outcome measures limits comparability across studies^{10,28} and the evidence base is further limited by short follow-up periods.^{12,13,22,31} Fourth, as noted in Section 1.6.2., evaluations rarely assess which factors mediate intervention success. While much is known about risk factors for perpetrating and experiencing DRV, and behavioural theories suggest theoretical antecedents,¹² existing research tells us little about how interventions impact these factors and whether these impacts lead to reductions in DRV.

Several reviewers draw particular attention to gaps in research on the role and measurement of social norms. Experts see shifting harmful social norms or fostering protective ones as an important aspect of DRV prevention⁸ and reviewers report that this approach remains a common underpinning of interventions.^{8–10,12,13} Interventions that engage with social norms might reduce DRV via different pathways, depending on whether attitudes and behaviours in the reference group align or conflict with members' perceptions of them.⁹⁴ Where young people overestimate their peers' involvement in or support for DRV, interventions might focus on correcting these misperceptions,⁸⁶ an approach that has been widely used in the area of alcohol prevention among university students but has demonstrated little effectiveness.⁹⁵ Other interventions aim to foster collective changes in attitudes that support violence and to do so in a way that is visible and public within a reference group so that it is clear that social expectations are shifting. For example, group discussions using critical reflection can change both individual attitudes and norms^{96,97} and have been effective in reducing violence against women.⁹⁸ Behaviour change theory and empirical evidence suggest that relationships between attitudes, norms and behaviours are complex^{89,99,100} and variable (e.g., Jewkes et al. 2015,⁸⁹ Bicchieri & Mercier 2014,⁹⁹ Ajzen 1991,¹⁰⁰ Mackie et al. 2015,¹⁰¹ Enosh 2007,⁷⁶ Hunt et al. 2022⁷⁸ and Chung & Rimal 2016¹⁰²). Drawing on the Theory of Planned Behaviour, attitudes and norms can be thought of as influencing each other reciprocally and as varying in their relative influence over behaviour across outcomes and contexts. Evaluations of DRV interventions that aim to address norms have to-date shed light on these pathways: they rarely measure impact on norms,^{11,12} and none have assessed whether norms changes have mediated impacts on DRV.¹¹ Reviewers report that gaps in methods for social norms measurement present a barrier to reliably and validly conducting these analyses.^{9,11}

Overall, available evidence suggests DRV interventions can increase knowledge and protective attitudes and, to some extent, improve DRV outcomes.^{7,8,11,13,22,30–32} However, little is known about which types of interventions are most effective and the evidence to support DRV behaviour change is limited,¹¹ making it difficult to recommend particular approaches. Reviews suggest that more research is needed to elaborate the mechanisms that lead to behaviour

change, and that methods for measuring social norms must to be improved in order to examine the role of what is thought to be a key component of DRV prevention.^{9,9,11–14}

Chapter 2. Social norms and DRV

This chapter begins with a brief introduction to the role of social norms in DRV research and prevention and then provides an overview of social norms theory. Next it reviews evidence on relationships between social norms and DRV outcomes and efforts to incorporate social norms into DRV prevention, highlighting limitations to the measurement of social norms in DRV research. The chapter then provides an overview of social norms measurement, drawing on work in the fields of GBV and adolescent SRH. The chapter concludes with my thesis aims, objectives and research questions, which outline my approach to building on existing knowledge about social norms measurement to improve methods for their measurement in DRV research.

2.1. Introduction to social norms and DRV

Social norms are the informal rules that determine acceptable behaviour in a group.¹⁰³ They can act as a "brake on social change"^{104(p7)} or serve to hasten it: they can impede behaviour change even when individual-level attitudes are shifting or, alternatively, foster it even in the absence of changes in individual attitudes.^{15,104} Theoretical and empirical literature suggests that social norms might play an important role in DRV behaviours. DRV researchers have long recognised the importance of harnessing peer influence to protect against DRV,^{7,32} and feminist approaches to addressing gender norms and gendered power commonly underpin DRV interventions.¹³ As social norms theory has gained influence in public health,¹⁰⁵ its influence on DRV interventions has become more explicit.^{8,13} In a 2015 systematic review, Stanley et al. report that most included DRV interventions were informed by social norms, feminist or gender theories or used a bystander approach, which aims to foster protective intervention by peers.^{k,8} Despite social norms' theories or line mediating intervention impact, few evaluations of DRV interventions have measured changes in social norms and none have assessed social norms as potential mediator.¹¹

^k Bystander interventions can also aim to foster protective intervention by adults.¹¹

2.2. Social norms theory

2.2.1. Social norms scholarship in the social sciences

At its core a study of human interaction, the study of social norms and their influence has been a topic of wide interest across the social sciences.¹⁰² While terminology, definitions of key concepts and operationalisation vary across disciplines, in their 2016 review of social science scholarship on social norms Chung and Rimal report that conceptualisations typically coalesce around social norms as "customary rules that constrain behavior by eliciting conformity".^{102(p3)} Theorists make a key distinction between the framing of social norms as properties of groups (e.g., by sociologists) versus perceptions of individuals (e.g., by social psychologists).^{94,102} Informal rules that determine "acceptable, appropriate, and obligatory" behaviour in a group are considered collective norms,^{106(p2)} while perceived norms are individuals' perceptions (whether correct or incorrect) about what others do and approve of.¹⁰²

Situated at the level of the society, collective norms can be thought of both as shaping experiences and behaviours and as shaped through individuals' actions. According to sociologist Anthony Giddens' structuration theory, for example, social norms comprise an aspect of social structure that both enables and constrains social practices.¹⁰⁷ This structure is, in turn, maintained or modified by those practices, and individuals can choose to take action to maintain or modify prevailing norms.¹⁰⁷ Scholarship on gender norms, emerging from feminist work to advance gender equality, tends to conceptualise social norms as collective norms situated at the level of society and embedded in institutions.⁹⁴

Perceived norms feature as a key construct in prominent behavioural theories underpinning public health interventions,^{6,100,108} including interventions to reduce DRV.^{1,11,13} The Theory of Planned Behaviour, for example, posits that subjective norms (perceptions about what is socially expected)¹⁰² have reciprocal influences on personal attitudes and perceived behavioural control, and that these norms influence behaviour via behavioural intentions.¹⁰⁰ According to Social Cognitive Theory, norms (framed as social outcome expectancies) are thought to work alongside non-social outcome expectancies to influence behaviour both directly and via goal-
setting.¹⁰⁹ This theory presents social modelling as an important source of behavioural learning,¹⁰⁸ aligning with the theorised influence of descriptive norms – perceptions of typical behaviour – in social psychology.¹⁰²

Though theorists use different terminology to describe these constructs, the influence of social expectations on health-related behaviours is widely recognised and DRV interventions often seek to modify the social environment to foster protective norms.

2.2.2. Definitions and features of social norms for the present study

A conceptualisation of social norms emerging from social psychology has been particularly influential among health researchers and practitioners who focus on GBV and on adolescent SRH.^{6,15,106,110} This approach situates social norms within the mind, framing them as beliefs about people in a valued reference group of important others.¹¹¹ Within this framework, theorists distinguish between two types of social norms:^{15,106}

- Descriptive norms: Beliefs about what others in the group do (i.e. what behaviour is typical)
- Injunctive norms: Beliefs about what others in the group think should be done (i.e. what behaviour is considered to be appropriate)

Where descriptive norms are supportive of DRV, young people might believe for example that their peers commonly experience or perpetrate abusive behaviours within their intimate relationships. Where descriptive norms are protective against DRV, they might perceive DRV to be rare among their peers. Where injunctive norms are supportive of DRV, young people might believe that their family or friends would disapprove of the use of violence in intimate relationships or would support their decision to break up with an abusive partner. While these beliefs are subjective, they form based on observations of behaviours in a reference group and of how reference group members react to others' behaviours.¹⁰⁴ For example, descriptive norms about DRV will be based on observations of the typicality of experiencing or perpetrating DRV. Injunctive norms will be based on observations of the reference group's reaction to others who experience or perpetrate DRV, including bystander action to intervene in this type of

abuse. Norms theorists make a critical distinction between personal attitudes, which are *internally* motivated preferences or judgements, and social norms, which represent *social* expectations.^{101,112} The reference group or groups important for influencing behaviour can change over time and depending on the behaviour of interest.^{15,105}

There is no consensus on causal models of normative influence (i.e. on why individuals comply with prevailing norms); theorists suggest that it might be for a number of reasons.^{103,105} Some include, for example: the internalisation of operative norms, the material benefits of coordinated behaviour, and the fulfilment of a sense of social identity.¹⁰⁵ A focus of theorists and interventionists, and perhaps the strongest mechanism of normative influence, however, is the anticipation of social sanctions enacted by a reference group.¹⁰³ These sanctions can take the form of social rewards for complying with a norm and of social punishments for deviating from it.^{15,103} Theorists disagree about whether descriptive norms, injunctive norms and social consequences must all be in place to confirm existence of a social norm or whether these components work collectively to strengthen its influence.⁶ Theorists also disagree about the relationship between descriptive and injunctive norms,⁶ with some positing that injunctive norms moderate the behavioural impact of descriptive norms.¹⁰²

Norms are thought to exert a stronger behavioural influence where the behaviour is interdependent and visible, norms relate directly to the behaviour of interest and social sanctions are anticipated to be likely and strong.¹⁰³ Theorists suggest that the strength of a social norm's influence determines whether it makes a behaviour obligatory (e.g. female genital cutting), appropriate (e.g. adolescent drinking), tolerated (e.g. sexual harassment) or merely conceivable, which determines the approach to intervention.¹⁰³ Practitioners might also usefully consider the constellation of relevant social norms, as some are more resistant to change than others.^{113(p31)} Despite norms being likely to be more influential where they relate directly to a specific behaviour,¹⁰³ more distal norms can influence behaviours indirectly.¹⁰⁶ Considering violence outcomes, female genital cutting is sustained by direct social norms about

this specific practice.¹⁰⁶ For partner violence, on the other hand, indirect but influential social norms might include norms against divorce or interfering in others' private lives.^{15,106}

2.2.3. Conceptualising gender norms as a type of social norm

Historically gender theory and social norm theory have developed independently, but efforts over the last decade have sought to join up these two areas of scholarship to inform public health practice.^{6,104,114} Gender norms, "informal rules and shared social expectations that distinguish expected behaviour on the basis of gender"^{114(p4)} can be situated as one aspect of a broader gender system that privileges maleness and masculinity over femaleness and femininity in the allocation of "resources, roles, power and entitlements".^{94(p410)} In their work to align scholarship on gender norms emerging from feminist scholarship and shaped by sociological theory, and broader social norms as framed largely by social psychologists, Cislaghi and Heise, 2020 highlight key differences in how these two bodies of work traditionally conceptualise how norms are situated and reproduced. While social norms in public health tend to be framed as beliefs, situated in the mind, gender norms have traditionally been framed (like collective norms; see Section 2.2.1.) as a feature of society, embedded in institutions whose characteristics and practices sustain male dominance. While both gender norms and social norms are characterised as being reproduced via social interactions, the gender norms literature has traditionally taken into more consideration the role of power as a motivation for enforcing maintaining inequitable gender norms.⁹⁴

As psychologist Sandra Bem wrote in 1981, the differentiation of roles based on gender "serves as a basic organizing principle for every human culture", driving the socialisation of children into gendered self-concepts, traits, personality attributes and skills.^{115(p354)} Cislaghi and Heise endorse the prominence of gender as a "primary frame for social relations",⁹⁴ suggesting that gender norms can be thought of as a particularly powerful type of social norm.^{94,116} This thesis draws on the social norms framework of descriptive and injunctive norms to consider the role and measurement of gendered social expectations, one aspect of the broader gender system, in DRV research.

Though particular manifestations of gender norms vary, core aspects of social expectations of girls and boys – and their inequitability – are remarkably consistent worldwide.^{92,117,118} Across settings, femininity is associated with beauty, attractiveness, propriety and compliance.⁹² Girls, who are viewed as weak and vulnerable, are expected to submit to male authority.⁹² Though their social value may depend on having a stable male partner,^{92,117} girls' and women's sexuality is "universally restricted"^{117(p1582)} as gender norms prescribe their innocence and romantic and sexual passivity.^{117,118} Girls and women who are seen as too sexual or promiscuous face social stigma and isolation, and in some settings physical harm.^{92,117} Boys, on the other hand, are expected to be strong and tough, and to eschew ostensibly feminine behaviours such as physical weakness or displays of emotion.⁹² In contrast to the sexual role prescribed to girls, gender norms dictate male sexual and romantic agency and dominance: men and boys are expected to pursue women and girls, take a dominant role in relationships and "demonstrate manhood" ^{92(p8)} by having sex with many female partners,^{92,118} for which they are socially rewarded.¹¹⁷

Parents and peers are central in shaping young people's attitudes towards gendered expectations⁹² and schools can promote gender-equitable attitudes or reinforce inequitable gender norms via their rules, traditions and structure.^{91,92} Pressure to conform to gendered expectations intensifies in early adolescence and peer sanctions are a powerful mechanism of gender norms compliance.⁹² However, regional variation in dominant gender norms, cultural shifts in prevailing gender expectations over time and existence of young people who challenge inequitable norms indicate that these norms are not inevitable and in fact can be transformed.^{82,92,118,119}

2.3. Social norms relating to DRV and gender are drivers of DRV

Section 1.5.1.3. discussed peer factors influencing DRV victimisation and perpetration, including the influence of peers' experience of and attitudes towards DRV. Drawing on the framework offered by social norms theorists, this section reviews in more depth existing evidence on

relationships between descriptive and injunctive DRV and gender norms, and DRV outcomes. The significance and strength of the relationships between DRV norms and DRV outcomes vary within and across studies, and differences in measurement and analysis methods limit comparability. To provide an indication of the magnitude of reported relationships between social norms and DRV outcomes based on the strongest evidence currently available I report effect size estimates where these are available from longitudinal studies.¹

2.3.1. Evidence of the relationship between DRV norms and DRV outcomes

Observational studies with adolescents find that descriptive and injunctive norms relating to DRV are associated with DRV victimisation and perpetration, and predict DRV longitudinally. While findings vary to some extent by study, norms and outcome measurement, and adjustment for other factors, they provide evidence of significant and independent relationships between pro-DRV norms and increased DRV.

Considering the role of descriptive norms, young people who believe that their friends or peers have experienced or perpetrated DRV are more likely to report perpetrating^{77,82,120–124} or experiencing^{121,122,125,126} DRV themselves. Researchers have examined the relationship between descriptive norms and DRV perpetration longitudinally, finding that it remains significant (aOR=1.34, p<0.05;^{77,82} *r*=0.12-0.27, p<0.001-p<0.05^{82,124}). Similarly, young people who report pro-DRV injunctive norms are more likely to experience^{76,78} and perpetrate^{83,127} DRV. While little evidence exists on whether injunctive norms predict subsequent DRV, a study by Nardi-Rodriguez et al. provides some evidence of this.¹²⁸ They used two combined measures, each comprising three descriptive and three injunctive DRV norms items: one measure assessed norms for DRV perpetration and the other for DRV victimisation.¹²⁸ Unadjusted correlations were significant for and both boys' DRV perpetration and girls' DRV victimisation (*r*=0.22-0.47, p<0.01).¹²⁸ Using structural equation modelling, they found that these norms measures were

¹ Appendix F to Paper 1 (thesis Appendix 5) details the methods and results of studies assessing relationships between social norms and DRV based on my systematic review.

each associated with DRV intentions, which were in turn associated with subsequent DRV perpetration and victimisation, respectively.¹²⁸

While results are inconsistent, several studies have found that descriptive^{77,122,123} and injunctive^{76,129} DRV norms remain cross-sectionally associated with DRV behavioural outcomes when controlling for individual attitudes towards DRV. Some evidence suggests that for descriptive DRV norms, this relationship might persist longitudinally (aOR=1.35-1.44, p<0.05).^{77,82} In research with older secondary school students in Haiti,^m Gage found that injunctive DRV norms were more strongly associated with girls' and boys' physical and psychological DRV perpetration than were personal attitudes towards DRV or gender stereotypes.¹²⁹ Similarly, others have found stronger relationships between descriptive¹³⁰ and injunctive^{78,83} DRV norms and DRV victimisation⁷⁸ or perpetration^{83,130} than between DRV¹³⁰ or gender^{78,83} attitudes and DRV outcomes. Heterogeneity in the relative importance of attitudes and DRV norms within^{76,77,82} and across^{76,78,83,122,127,130–133} studies suggests that these relationships might vary by gender, measure, type of norms, DRV involvement (victimisation or perpetration), DRV type and other factors. This is in-line with the Theory of Planned Behaviour's suggestion that the relative strength of attitudinal and normative influence will vary across outcomes and contexts.¹⁰⁰ The weight of the evidence suggests that pro-DRV norms can influence the behaviours even of young people who personally disapprove of this type of violence, as social norms theory would suggest.^{101,104} Considering that individual attitudes could theoretically be on the causal pathway between norms and DRV outcomes, controlling for attitudes in these analyses might actually underestimate the impact of social norms.^{100,134}

Qualitative research in the UK provides some insight into the mechanisms of normative influence, finding that fear of being blamed can be a barrier to seeking help for DRV victimisation and that when girls do seek support from peers, controlling and abusive behaviour can be normalised.³⁴

^m Participants were in grades 10-12, aged 14 years and older. Of these, 61.4% were aged 19 years or older.

2.3.2 Evidence of the relationship between gender norms and DRV outcomes

While theoretical and qualitative DRV literature engages extensively with the relationship between gender norms and DRV, quantitative research in this area remains in the early stages. Studies exploring relationships between gender-inequitable injunctive norms are sparse but provide some evidence of cross-sectional associations with DRV outcomes,^{133,135,136} including in models controlling for personal attitudes towards DRV¹³⁵ and gender.¹³³ Emerging, crosssectional evidence suggests that DRV is more common among young people who report gender-inequitable injunctive norms relating to violence against girls and women generally (i.e. not DRV-specific violence),¹³⁶ household gender roles¹³⁵ and female sexual availability.¹³³ Quantitative research on associations between gender norms and DRV is otherwise limited because the role of gender norms tends to be assessed by measuring personal gender-related attitudes^{80,137} rather than perceptions of others' views.

Drawing on social and psychological theoretical perspectives, Reyes et al. suggest that traditional gender norms advance scripts of male relationship dominance that promote DRV, and they posit that DRV injunctive norms play a role in determining whether boys enact these scripts.⁸² Considering UK evidence, qualitative research offers further insights into how inequitable gender norms might manifest to drive and sustain male DRV in heterosexual relationships. Interviews with UK adolescents suggest that boys can lose social status if their girlfriend is unfaithful¹³⁸ and that jealousy can feed into controlling behaviours.³⁴ Abusive boys use DRV as a tactic to assert control and dominance within the relationship³⁴ and, particularly among boys in disadvantaged groups, to advance a violent and powerful public image.¹³⁸ In line with social expectations of girls' chastity, boys report that it is considered acceptable to sexually pressure girls who are seen as sexually experienced.³⁴ In a context where boys face social pressure to have sex³⁴ and are celebrated for doing so,¹³⁸ for girls resisting sex can precipitate severe physical DRV and coercive threats of abandonment.³⁴ However, norms prescribing durable heterosexual relationships for girls can make it difficult for them to leave an abusive partner.^{34,139}

2.4. Incorporation of social norms approaches into DRV prevention

Adolescence offers a critical window of opportunity to promote protective anti-DRV and gender-equitable norms. As noted in Section 1.6, norms governing sexual and romantic relationships are particularly sensitive to influence as young people first begin to navigate these relationships in adolescence.⁶ This period of norm formation overlaps with adolescents' growing awareness of and self-consciousness about how others view them,¹⁴⁰ a shifting affiliation from family towards peers¹⁴¹ and the increasing importance of peer influence^{32,92} at this age. While pressure to conform to gendered expectations intensifies in early adolescence,⁹² studies suggest adolescence is also a stage when young people's own views on gender are in flux: attitudes become more gender-equitable and less stereotypical during early adolescence, before boys' views tend to become less egalitarian in middle- and late-adolescence.^{82,92} Peer sanctions are a powerful mechanism of gender norms compliance, especially for boys,¹¹⁹ but variations in manifestations of dominant gender norms and young people who challenge inequitable norms – girls, more commonly than boys^{92,118} – suggest that inequitable gender norms can be successfully challenged.

Many DRV interventions capitalise on this window of opportunity, incorporating efforts to promote more gender-equitable norms and to reduce the social acceptability of DRV itself. Bystander interventions, for example, can train participants to intervene to reduce violence,¹⁴² often targeting both sexual violence and DRV.^{142–144} While existing evidence is limited¹¹ and mixed (e.g. Edwards et al., 2019; Miller et al., 2012; Coker et al., 2017),^{142–144} cluster RCTs demonstrate that bystander interventions can increase DRV bystander behaviours¹⁴³ and reduce DRV perpetration.^{142,144}

Effective DRV interventions incorporating a social norms approach have taken a range of forms; evidence does not point to a single model for effectiveness. Green Dot, a "gender-neutral" bystander intervention in US secondary schools,^{142(p8)} demonstrated significant reductions in DRV victimisation and perpetration in a cluster RCT but these were sustained only for girls' perpetration by the end of the four-year follow-up. Coaching Boys into Men, a gender-

transformative bystander intervention for male secondary school athletes in the US, was found to be effective in a cluster RCT in reducing DRV perpetration and negative bystander behaviours.¹⁴⁵ Two of the earliest interventions effective in reducing DRV in RCTs – Safe Dates and Shifting Boundaries – targeted social norms alongside other factors as potential mediators of change.^{137,146} Safe Dates, implemented in US secondary schools, aimed to shift norms relating to gender and DRV and it significantly reduced perpetration of psychological, moderate physical, and sexual DRV perpetration as well as moderate physical DRV victimisation.¹³⁷ Implemented in US middle schools, Shifting Boundaries targeted bystander behaviours and DRV norms and significantly reduced sexual DRV.¹⁴⁷

Despite the prominence of social norms in intervention theories of change and the effectiveness of interventions that incorporate norms-based approaches, existing literature provides little information about whether these interventions do impact norms as hypothesised, and if so, whether this mediates reductions in DRV. Emerging evidence suggests that DRV interventions can successfully shift DRV-related social norms, including in interventions that reduce DRV.^{11,148,149} Evaluations rarely measure norms directly, however, and none appear to have assessed them as mediators of impact on DRV outcomes.¹¹ Reviewers suggest that this might be due to limitations to methods for reliably and validly measuring social norms.¹¹ Refinements to these methods would be a valuable step towards improving our understanding of whether, to what extent and to what end DRV interventions are activating this potential mechanism of change.

2.5. Social norms measurement

As attention to social norms and its measurement in the areas of both GBV and adolescent SRH has intensified over the last decade, empirical and theoretical literature on these topics has proliferated.^{104,150–154} This body of literature, informed by the dynamics of gendered violence and by social and cognitive factors in adolescent development, provides useful insights for social norms measurement in DRV research.

2.5.1. Survey measures of social norms are limited

Definitions of beliefs, attitudes and social norms, and the relationships between these constructs, have historically been unclear, as have implications for intervention development and evaluation.¹⁵ Research informed by social norms concepts, including evaluations of DRV interventions, typically measures participants' individual attitudes and not their perceptions of the views of others in their reference group.^{6,155} Political theorist Gerry Mackie and colleagues offer a useful framework to illustrate distinctions between "standard" measures of behaviour and attitudes and measures of social norms.^{n,101(p49)} Adapted based on this framework, Figure 2 illustrates the distinction between measures of behaviours, attitudes and social norms by categorising these into "beliefs about the self" (second column) and "beliefs about others" (third column). Each type of belief can be though of as descriptive (what happens) or injunctive (what *should* happen). "Standard" measures of behaviours) and about what one thinks should be done (attitudes). Social norms measures assess beliefs about others, or social expectations: beliefs about what others do (descriptive norms) and about what others think should be done (injunctive norms).

	Beliefs about self	Beliefs about others	
Descriptive	What I do	What others do	
	(behaviour)	(descriptive norm)	
Injunctive What I think should be done		What others think should be done	
	(attitude)	(injunctive norm)	

Figure 2. Framework distinguishing between measures of behaviours, attitudes and social norms (adapted from Mackie et al.'s theoretical framework)^{101(p49)}

Assessing social norms and social norms change would mean adding measures of descriptive and injunctive norms – beliefs about what others do and what others think should be done in relation to a specified behaviour, respectively – to evaluation surveys. Measuring these

ⁿ Mackie uses the term "empirical expectations" to refer to a construct similar to what we call "descriptive norms"; and "normative expectations" to refer to a construct similar to what we call "injunctive norms".^{101(pp24-25)}

constructs alongside individual behaviours and attitudes is important because research suggests that attitudes and social expectations each correlate with DRV outcomes.^{76,77,122,123,129,133,135} Bystander research further suggests that, along with these factors correlating with intentions⁸⁶ and actions to intervene in violence as a bystander,^{156,157} levels of alignment between attitudes and injunctive norms also correlate with intentions to intervene as a bystander in IPV.⁸⁶

A challenge, however, is that there is little consensus on how to measure social norms and a limited evidence base of norms measures validated in multiple settings.⁶ Furthermore, some research suggests that very young adolescents (aged 10-14 years) might struggle to distinguish between their personal attitudes and the views of others^{158,159} when responding to social norms items. Without valid and reliable quantitative measures of salient social norms relating to DRV and to gender, appropriate for adolescents' stage of cognitive development, researchers are limited in our ability to assess normative change in intervention surveys and test its theorised role as a mediator of DRV behaviour change.

2.5.2. Measuring social norms: recommendations from GBV and adolescent SRH research

Methods for measuring social norms are still being developed. Little evidence is available to support specific approaches in DRV research, but work on social norms measurement has been rapidly expanding in GBV and adolescent SRH research. Recommendations from experts in these fields, drawing on their own experience and that of their colleagues, offer some insights and raise areas for further research.

Qualitative and quantitative methods can be used to measure social norms, and vignettes exploring norms and social sanctions in realistic but fictional scenarios can be incorporated into either approach.^{104,160} The length and complexity of vignettes, however, can make these difficult to fit feasibly within evaluation surveys.⁶ A few DRV evaluations have used survey measures of descriptive and injunctive norms effectively, and their findings demonstrate that these measures can be sensitive to change over the course of an evaluation.^{11,148,149} Wider use

of social norms measures is hindered, however, by evidence gaps and a lack of consensus on best practice.

Recognising that the relevance and rigidity of norms and the nature of social sanctions vary across settings and populations, experts emphasise the importance of formative research to identify and test relevant, influential norms that are amenable to change.^{6,101,104} Experts also highlight the importance of specifying bounded reference groups and disaggregating findings where data include norms among more than one.^{6,104} Evaluators can draw on formative research to identify salient reference groups and incorporate these into survey items or can alternatively use survey items to collect this information.^{6,160}

Researchers report difficulty balancing the need to include enough items to validly measure norms with the need to keep surveys from becoming unwieldly and labour-intensive.^{101,104} The multifaceted nature of social norms means that a wide array of survey items can provide information about a single norm. However, surveys that ask too many similar questions can confuse and fatigue participants.⁶ For example, researchers for the Voices for Change project in Nigeria adapted Mackie's framework¹⁰¹ (see Figure 2) to ask six questions about each norm of interest.¹⁰⁴ They found that this approach was too time- and resource-intensive and that the distinctions between the nuanced items were unclear to participants.¹⁰⁴ Confusion about the meaning and nature of norms items is of particular concern for young respondents.

Recommendations coalesce around focusing on measurement of two aspects of social norms, beliefs about what others do and what others think should be done, and specifying a reference group.^{6,104} Experts advise considering carefully whether items should be phrased positively (i.e. if someone does X) or negatively (i.e. if someone doesn't do X) because social sanctions – the causal mechanism of injunctive norms influence – are not necessarily levied in parallel.^{101,104,161} Cislaghi and Heise offer the example of bringing a cake to the office for one's birthday: while a respondent might say that colleagues would think this was good, it doesn't necessarily follow that *not* bringing the cake would elicit negative social sanctions.¹⁰⁴ If we wanted to know

whether social norms compel colleagues to bring in the cake, we would need to ask about the social consequences of not doing so.¹⁰⁴

Another challenge is to ensure that item meaning is clear to participants, including that they are being asked what they think and not what is objectively the case.⁶ It is important to bear in mind that, for norms items, researchers are interested in participants' perceptions – not true numbers – of what others in their reference group do and think. Norms can persist when many oppose a specific behaviour but incorrectly believe that others in their reference group favour it, a scenario theorists term "pluralistic ignorance".^{101(p23)} It is perceptions themselves, whether accurate or inaccurate, that wield normative influence.¹⁰¹

Considering the structure of social norms measures, recommendations on measure length vary. Multi-item scales can be useful to capture the social norms' multidimensional nature and reduce the impact of response errors.⁶ On the other hand, evidence suggests that single-item measures might be adequate¹⁰⁴ and these place less burden on respondents. Deciding on the number of response options also involves trade-offs. Binary yes/no items don't allow for nuanced responses indicating the degree of agreement, and field experience suggests that responses to this type of item can be biased towards a "yes" response.¹⁰⁴ Likert scales can be good for collecting nuanced data, but the number of response options needs careful consideration: a greater number provides more granularity, but items with fewer response options are simpler to answer.¹⁶⁰

Finally, a variety of different item formats can elicit information on norms. Items assessing descriptive norms might ask about a behaviour's prevalence (i.e. "how many")^{104(p18)} or about its frequency (i.e, "how often").^{104(p18)} To assess injunctive norms, researchers might ask separately about the reference group's views and potential social sanctions. An example of the former, tested for an adolescent SRH study in Honduras, was formulated as shown in Figure 3.¹⁰⁴ A different set of response options could be offered instead, for example ranging from "Strongly agree" to "Strongly disagree".¹⁰⁴

The people in your community believe that fathers should attend pregnancy check-ups with their pregnant wives/ companions 1 = No 2 = Yes sometimes 3 = Yes mostly 4 = Yes always

Figure 3. Example of survey item assessing injunctive norms by measuring reference group views, excerpted from Cislaghi & Heise^{104(p17)}

Alternatively, experts suggest that measurement of injunctive norms can sometimes be simplified by asking only about social sanctions.^{104,160} Field experience suggests that participants might find questions about social sanctions easier to answer because the observable actions of others are easier to call to mind than others' thoughts.¹⁶² A limitation to this approach is that specifying sanctions within the survey item requires detailed knowledge about social sanctions governing the norm in the study population and setting. With this information, researchers can ask about the perceived likelihood of specific sanctions, as they do in an example from the Social Norms Mentorship Project in Figure 4.¹⁶⁰



Figure 4. Example of survey item assessing injunctive norms by measuring specific sanctions, excerpted from the Social Norms Mentorship Programme training slides^{160(Day 4, slide 25)}

Otherwise, more general response options can identify the existence of social sanctions with less specificity, as in two other examples offered by the Social Norms Mentorship Programme, shown in Figure 5.¹⁶⁰ Injunctive norms might also ask about perceptions of others' present views, as shown in Figure 3, or frame these items in terms of a hypothetical scenario, as shown in Figure 5.



Figure 5. Example of survey item assessing injunctive norms by measuring sanctions generally, excerpted from the Social Norms Mentorship Programme training slides^{160(Day 4, slide} 26)

Studies testing different types of descriptive and injunctive norms items are needed to assess which formulations work best,¹⁰⁴ and how this might vary by setting and population. For DRV research, particular attention should be paid to what features support valid and reliable norms measurement among adolescents, whose reference groups and ability to distinguish between their own and others' views might differ from adults'.^{158,159}

2.6. Research aim, questions and objectives

DRV is widespread and poses a risk to young people's health. Evidence suggests that social norms play an important role in underpinning this type of abuse, and social norms concerning DRV and gender have long been recognised as important to DRV prevention. Although DRV interventions often seek to foster protective DRV and gender norms, measurement of social norms in this field has been largely neglected. A lack of reliable, valid measures for what is thought to be a key mediator of DRV prevention contributes to crucial gaps in our

understanding of why some DRV interventions work and others do not. Informed by social norms theory, my thesis research seeks to address this by building on recent advancements in methods for measuring social norms. Drawing on data from Project Respect, I aim to refine measures of social norms as hypothesised mediators of a school-based intervention to reduce DRV in England and to inform methods of social norms measurement in DRV research.

To achieve these aims, I will address four research questions:

- 1) Are existing measures of adolescent social norms relating to DRV and gender reliable and valid?
- 2) Are new and adapted measures of social norms relating to DRV and gender understandable and answerable when used in research with adolescents in England?
- 3) Are new and adapted measures of social norms relating to DRV and gender reliable and valid when used in research with adolescents in England, and how can they be refined?
- 4) What are student, staff and parent/carer accounts of social norms relating to DRV and gender in schools, and how are these implicated in DRV?

The specific objectives of this thesis research are:

- a) To conduct a systematic review of social norms measures related to gender and DRV used in research with adolescents
- b) To develop, cognitively test and refine social norms measures related to gender and DRV
- c) To pilot new and adapted social norms measures and assess their reliability and validity
- d) To recommend refinements to piloted social norms measures
- e) To analyse qualitative data from Project Respect to identify social norms contributing to DRV in England

Table 2 presents the objectives associated with each research question, the data sources and methods on which I draw to address each research question, and the corresponding papers in which I present my findings.

	Research questions	Obj	ectives	Methods/Data sources	Paper
1.	Are existing measures of adolescent social norms relating to DRV and gender reliable and valid?	a)	To conduct a systematic review of social norms measures related to gender and DRV used in research with adolescents	Systematic literature review	Paper 1
2.	Are new and adapted measures of social norms relating to DRV and gender understandable and answerable for adolescents in England?	b)	To develop, cognitively test and refine social norms measures related to gender and DRV	Qualitative analysis of Project Respect cognitive interviews	Paper 2
3.	Are new and adapted measures of social norms relating to DRV and gender reliable and valid when used in research with adolescents in England, and how can they be refined?	c) d)	To pilot new and adapted social norms measures and assess their reliability and validity To recommend refinements to piloted social norms measures	Quantitative analysis of Project Respect baseline student surveys	Paper 3
4.	What are student, staff and parent/carer accounts of social norms relating to DRV and gender in schools, and how are these implicated in DRV?	d) e)	To recommend refinements to piloted social norms measures ^o To analyse qualitative data from Project Respect to identify social norms contributing to DRV England	Qualitative analysis of Project Respect data: • Optimisation sessions • NSPCC-delivered trainings • All-staff trainings • Student and staff interviews (intervention and control schools) • Parent/carer interviews (intervention schools)	Paper 4

Table 2. Research questions and objectives mapped to methods, data sources and papers

^o Objective (d) contributes to addressing both research question 3 and research question 4.

Chapter 3. Methods

This chapter opens with an overview of the methods for the Project Respect pilot RCT, summarises its findings and then describes my role as a research fellow on the study. It then outlines the distinction between the Project Respect pilot RCT and my thesis and presents the ontological and epistemological underpinnings of my thesis. Next it provides an overview of reliability and validity as indicators of the quality of quantitative measures, before outlining the methods for each of the four components of my thesis. The first component is a systematic literature review and the remaining three draw on data from Project Respect. I provide further detail on the methods for Chapters 4-6 in the papers presented in those chapters.

Finally, this chapter provides information on the ethical approvals for this work and concludes with a section on reflexivity and my positionality in undertaking this work.

3.1. Methods for Project Respect pilot cluster RCT

The Project Respect study was a pilot cluster RCT of a DRV prevention intervention of the same name, conducted with adolescents in England. In this section I will summarise the intervention, the design of its pilot cluster RCT and the findings of the overall study. Further detail on the Project Respect intervention, study and findings are available in publications of the protocol¹ (Appendix 1), process evaluation findings (Appendix 2)² and full research report.³⁷

3.1.1. Project Respect theory of change and intervention

Project Respect was a school-based, complex intervention¹⁶³ informed by the Safe Dates¹⁶⁴ and Shifting Boundaries¹⁴⁶ school-based interventions, both of which were effective in reducing DRV among US school students.^{147,165} Its core components and theory of change were developed prior to the study, informed by existing evidence. The research team and our implementing partner, the National Society for the Prevention of Cruelty to Children (NSPCC), led further elaboration of the intervention and the development of programme materials from March to September 2017. The programme was optimised for use in the UK via sessions with UK secondary school staff and students, input from the Advice Leading to Public Health Advancement young researchers group¹⁶⁶ and consultation with stakeholders.

The Safe Dates DRV prevention intervention was implemented among 8th and 9th grade students (typically ages 13-15 years old) in 14 public schools in rural North Carolina in November 1994-March 1995.¹⁶⁷ Baseline questionnaires were completed by 1,886 students.¹⁶⁷ Foshee et al. describe programme activities.¹⁶⁷ These included 20 hours of teacher training on DRV and the Safe Dates curriculum, a ten-session curriculum (45 minutes per session) and a DRV-themed poster contest. These in-school components were supplemented by workshops for community service providers. Results from a two-arm cluster RCT suggest that the programme reduced psychological, moderate physical and sexual DRV perpetration and moderate physical DRV victimisation, with effects persisting at least three years postintervention.¹⁶⁵ This reduction in DRV was mediated by attitudes demonstrating lower DRV acceptance and less support for gender stereotypes and by an increase in awareness of community services.¹⁶⁵

The Shifting Boundaries intervention aimed to reduce DRV, peer sexual violence and sexual harassment and was implemented with 2,655 6th and 7th grade students (typically ages 11-13 years old) in New York City public middle schools.¹⁶⁸ The programme had two components: a classroom-based six-session curriculum and a schoolwide building-based component.¹⁶⁸ The latter comprised school-based restraining orders, DRV/sexual harassment-themed posters around the school and hotspot mapping by students to identify physical areas of violence risk in the school for increased teacher surveillance. Participating schools were randomly assigned to one of four arms of a cluster RCT conducted in 2009-2010: (1) building-only, (2) classroom-only, (3) combined building and classroom or (4) control.¹⁶⁸ Taylor et al. report the results of the RCT six months post-intervention.¹⁶⁸ Effects on sexual harassment victimisation were mixed, showing an increase in prevalence but a decrease in frequency in the building-only arm and a decrease in frequency in the combined arm. Findings for other outcomes were more consistent, showing reductions in sexual harassment perpetration (building-only arm), peer sexual violence

victimisation (building-only and combined arms), peer sexual violence perpetration (buildingonly and combined arms) and sexual DRV victimisation (building-only arm). The trial found no impact on sexual DRV perpetration. The study authors concluded that their findings support the use of a building-only intervention as well as the addition of this type of school-wide component to curriculum interventions.¹⁶⁸

Drawing on the designs of these two effective interventions, Project Respect was developed by the London School of Hygiene & Tropical Medicine (LSHTM) and optimised for use in secondary schools in England by LSHTM and the NSPCC via sessions with secondary school staff and students. The programme is underpinned by two behavioural theories, the Theory of Planned Behaviour¹⁰⁰ and the Social Development Model.¹⁶⁹ The Theory of Planned Behaviour posits that behavioural intentions are the immediate antecedent to behaviours themselves.¹⁰⁰ As depicted in Ajzen's model of this theory in Figure 6, personal attitudes, perceived behaviour via intentions, while perceived behavioural control is also thought to exert a direct behavioural influence. As discussed in Section 2.4, subjective norms are a construct analogous to what social norm theorists call injunctive norms.



Figure 6. Ajzen's model of the Theory of Planned Behaviour^{100(p182)}

The core tenet of the Social Development Model is that a person's values, shaped by the social groups to which they are strongly bonded, drive behaviour. The model proposes that people develop prosocial and antisocial behaviours through two parallel pathways of socialisation, as depicted in Figure 7.¹⁷⁰ Considering the prosocial pathway, the model suggests that when a person has the opportunity to take part in pro-social activities, their involvement leads to rewards for involvement with these pro-social groups *if* they have the social, emotional and cognitive skills necessary to access these rewards. Receiving these rewards promotes bonding to others with prosocial orientations and whose influence shapes belief in prosocial values, leading to the adoption of prosocial behaviours. The antisocial pathway is nearly identical but allows for the possibility that rewards for antisocial behaviour and bonding to antisocial groups can foster antisocial behaviour even where they conflict with a person's own values. The model recognises the influence of three external factors on the opportunities, skills and rewards critical to these pathways: positioning in the social order, external constraints and personal characteristics.



Figure 7. Cambron's depiction of the Social Development Model^{170(p43)}

Turning to Project Respect, I will summarise the programme's design and then review its underlying theory of change. The programme was comprised of eight core components targeting staff, students, parents and carers and the school environment:

Staff

- 1. NSPCC-delivered 2-3 hour training for senior leadership and key school staff involved in programme delivery; and programme manual
- 2. All-staff 1-1.5-hour training, cascaded by staff who attended the NSPCC-delivered training

Students

- 3. Teacher-delivered student curriculum (six lessons for year 9, two lessons for year 10)
- Opportunity to download the Circle of 6 app (version 2.0.5, Tech for Good, New York, NY, USA)^p facilitating requesting support from friends and local services in unsafe situations

Parents and carers

5. Written information for parents and carers

School environment

- 6. School policy review to assure appropriate response to DRV
- 7. Hotspot mapping to identify and prompt increased surveillance in areas of risk on the school grounds
- 8. Student-led campaigns against gender-based harassment and DRV

Drawing on the Theory of Planned Behaviour,¹⁰⁰ Project Respect aimed to reduce DRV in part by shifting attitudes to be less supportive of DRV and gender stereotypes; and by fostering supportive social norms relating to DRV and to gender. Informed by the successes of Safe Dates and Shifting Boundaries, the programme included a substantial package of curriculum lessons alongside whole-school elements which were theorised to work synergistically to reduce gender-based harassment observable on the school site and to promote protective attitudes and norms. In the process of operationalising programme mediators for measurement in student surveys, we elaborated the construct of "perceived norms"^{1(p9)} to delineate three types: injunctive DRV and gender norms and descriptive DRV norms. Figure 8 depicts the revised theory of change.

^p This app has since been discontinued and is no longer available for download.

Targeting another important behavioural influence in the Theory of Planned Behaviour,¹⁰⁰ Project Respect aimed to promote students' sense of control over their behaviours in relationships via lesson content focused on building communication and anger management skills.¹⁶⁴ Drawing on evidence from Safe Dates and Shifting Boundaries, lessons also addressed gender roles,^{165,168} healthy relationships^{165,168} and help-seeking.¹⁶⁴ Informed by findings from the Safe Dates study, signposting in lessons and promotion of the Circle of 6 app targeted increases in knowledge of and access to support for students experiencing DRV.¹⁶⁵ Underpinned by the Social Development Model,¹⁷⁰ Project Respect was designed to offer opportunities for student participation in the curriculum and in leadership of whole-school campaigns, promoting school bonding, adherence to gender-equitable and anti-DRV peer norms and adoption of attitudes aligned to these values.





3.1.2. Project Respect pilot cluster RCT

The Project Respect pilot cluster RCT aimed to elaborate, optimise and pilot the Project Respect intervention and assess the feasibility and acceptability of the intervention and trial methods. An embedded process evaluation explored implementation, mechanisms of change and context.¹⁷¹ An embedded economic evaluation explored feasibility of economic evaluation methods. The study sought to address nine specific research questions:

- 1) Is progression to a Phase III RCT justified in terms of prespecified criteria? These criteria are as follows: randomisation occurs, and four or more schools (out of six) accept randomisation and continue in the study; the intervention is implemented with fidelity in at least three of the four intervention schools; the process evaluation indicates that the intervention is acceptable to ≥ 70% of the year 9 and 10 students, and staff involved in implementation; computer-assisted self-interviewing surveys of students are acceptable and achieve response rates of at least 80% in four or more schools; and methods for economic evaluation in a Phase III randomised controlled trial are feasible.
- 2) Which of two existing scales the Safe Dates and the short CADRI (CADRI-s) is optimal for assessing DRV violence victimisation and perpetration as primary outcomes in a Phase III RCT, judged in terms of completion, interitem reliability and fit?
- 3) What are likely response rates in a Phase III RCT?
- 4) Do the estimates of prevalence and intracluster correlation coefficient of DRV derived from the literature look similar to those found in the UK, so that they may inform a sample-size calculation for a Phase III RCT?
- 5) Are secondary outcome and covariate measures reliable, and what refinements are suggested?
- 6) What refinements to the intervention are suggested by the process evaluation?
- 7) What do qualitative data suggest about how contextual factors might influence implementation, receipt or mechanisms of action?
- 8) Do the qualitative data suggest any potential harms and how might these be reduced?
- 9) What sexual health- and violence-related activities occur in and around control schools?

School recruitment for the pilot trial proceeded via letters and telephone calls to schools in southern England and to school networks. Private schools, pupil referral units and schools exclusively for students with learning disabilities were not eligible. Of eligible schools expressing interest, we selected three from south-east England and three from south-west England purposively by deprivation (assessed using Income Deprivation Affecting Children Index [IDACI] score)¹⁷² and school-level value-added attainment (assessed using Progress 8 score).¹⁷³ We conducted two waves of optimisation sessions with students and staff in four secondary schools not taking part in the pilot trial, two in the south-east region of England and two in south-west, in April and July 2017. We pre-piloted the baseline student survey on electronic tablets with one year 8 class in a south-east England optimisation school in April 2017. In the same month I conducted cognitive interviews to test and refine student survey measures in one south-east England secondary school that had expressed interest in the study but not consented in time to take part in the pilot RCT.^q The head teacher of each participating school consented for the school to take part.

Baseline surveys were conducted with staff using paper and with year 8 and 9 students using electronic tablets from June to July 2017. LSHTM's clinical trials unit then stratified schools by region (south-east and south-west England) and randomly assigned schools 2:1 to the intervention or control condition. Intervention schools were to deliver Project Respect to year 9 and year 10 students in the 2017-2018 school year while control schools were to continue with usual provision. Follow-up surveys with staff and students were conducted using electronic tablets approximately 16 months post-baseline, in September to November 2018.

My thesis draws on optimisation sessions, cognitive interviews, student baseline surveys and process evaluation data to address research questions 2-4 (see Section 2.6.). Details on aspects of Project Respect data collection not used for this thesis are available in the published study protocol¹ (see Appendix 1) and full study report.³⁷ The Project Respect pilot trial was registered with the ISRCTN registry on 8th June 2017 (ISRCTN 65324176).¹⁷⁴

^q This school later joined the pilot RCT to replace a school that had withdrawn from the study.

3.1.3. Summary of Project Respect study findings

We completed elaboration and optimisation of the Project Respect intervention. Results from pre-piloting of the student baseline survey and cognitive interviews suggested that students generally understood survey items but also informed some refinements to item wording. One school dropped out prior to baseline surveys. It was replaced but without sufficient time to arrange baseline surveys in the replacement school, resulting in five schools taking part in baseline surveys and six taking part in follow-up surveys. The use of electronic tablets for student surveys was acceptable to students but posed logistical challenges to the study team, requiring intense planning and higher than anticipated levels of staffing. High response rates of 82.5% and 78.2% were achieved in participating schools at baseline and follow-up, respectively. Both the Safe Dates and CADRI-s DRV measures had high completion rates (around 99%) and reliability (Cronbach's and ordinal alphas around 0.9); the CADRI-s was therefore recommended for future use over the Safe Dates measure due to its brevity. School staff surveys achieved very low response rates at baseline (7.5%) and follow-up (6.5%), suggesting that staff surveys were unfeasible.

All six schools accepted randomisation to the intervention or control condition. However, the Project Respect intervention did not meet the criteria for progression to a phase III RCT due to limited fidelity and acceptability in the pilot trial. Implementation in the four intervention schools of school-wide elements was particularly low, including cascaded training for all school staff (delivered by two schools), policy review (delivered by two schools) and reorientation of staff surveillance according to identified hotspots (delivered in no schools). Process evaluation interviews suggested that staff and students viewed DRV as an important topic to address in schools. However, their views on the curriculum were mixed and delivery was undermined where schools were focused on addressing other, emerging challenge (such as poor Ofsted^r results or budgetary issues), where too few staff shared a commitment to delivery and where staff struggled with timetabling lessons or with insufficient time for planning.

^r Office for Standards in Education, Children's Services and Skills

3.1.4. My role as research fellow on the Project Respect study

Project Respect data collection took place from April 2017 to November 2018. I joined the study team as a research fellow at the start of the project and led its management under the direction of the principal investigator. In this role, I helped to refine study methods; led amendments to the bodies responsible for ethical oversight; developed cognitive interview guides and conducted all cognitive interviews; contributed to optimisation, survey and process evaluation data collection tools; led fieldwork in south-east England including optimisation sessions and most process evaluation interviews in the region; led analyses of qualitative and fidelity data; and led reporting on study findings.

3.2. Thesis methods

In this section I outline the distinction between the Project Respect study and my role as a research fellow, and my thesis and the tasks I undertook in my capacity as a PhD student. I then discuss the ontological and epistemological positions underpinning my thesis. I report on the methodologies and methods used for Chapters 4-7 as follows:

- In this section I present the methodologies and an overview of the methods used for each of Chapters 4-7
- In this section I also present methods details that are not critical to interpreting the research presented in Chapters 4-7 but that elaborate on considerations underlying my approach
- The papers I present in Chapters 4-7 provide all of the key information on methods that is required for interpreting the research I present in each paper

3.2.1. Distinction between the Project Respect study and my thesis

The Project Respect study set out to optimise and pilot the Project Respect intervention and trial methods and was guided by a published protocol¹ (see Appendix 1). When I joined the study, student survey measures had been identified and one of my tasks was to compile these and prepare the survey for pre-piloting. I noticed at this stage that, while social norms were theorised mediators of the intervention, measures of social norms had not been identified for

inclusion in student surveys. Having worked on a previous adolescent health study that drew on social norms theory to distinguish between descriptive and injunctive norms, ¹⁷⁵ I proposed the inclusion of specific measures to assess these two constructs in Project Respect. Finding no established, appropriate measures in the DRV literature, I identified a need for further work in the area of social norms measurement in DRV research. With the approval of the principal investigator, I initiated and led streams of work to (1) develop, refine and test social norms measures for Project Respect, and (2) identify social norms protecting against and contributing to DRV in England by drawing on staff, student and parent and carer accounts. This original work is the topic of my thesis.

My role as a research fellow centred on managing the study, developing data collection tools to answer the study's research questions (see Section 2.6.), data collection in south-east England, analysis and reporting. To address thesis research questions 2-4 I undertook additional work, beyond the scope of my research fellow role:

- I developed three new and adapted measures of social norms relating to DRV and to gender. To test and refine these measures, I conceptualised and led the incorporation of gender and DRV social norms and (for comparison) attitude measures into cognitive testing, analysed the resulting data, reported on findings and refined measures for inclusion in student surveys. I present this work in Chapter 5.
- I conceptualised, designed and conducted the analysis of data from student baseline surveys to assess the performance of the three new and adapted social norms measures, presenting this work in Chapter 6.
- 3) I incorporated questions into optimisation and process evaluation data collection tools to identify social norms contributing to or protecting against DRV. These questions explored the gender regime,⁹¹ gender dynamics and gender and DRV norms within schools; and (for intervention schools) potential intervention impacts on social norms. I analysed data on relevant social norms and report findings in Chapter 7.

I conducted the literature review presented in Chapter 4 independent of the Project Respect study.

3.2.2. Ontological and epistemological assumptions

The work in this thesis is informed by the philosophy of critical realism, developed by Roy Bhaskar, which proposes three levels of ontology:^{176,177}

- 1) The "empirical" level refers to what is observed
- The "actual" level refers to objects and events that exist, whether or not they are observed
- 3) The "real" level refers to the unseen causes of what takes place at the level of the "actual"

Critical realism accepts a realist ontology – the view that there exists a real world, independent of what we know, observe or measure, which contains both natural and social "causal forces".^{178(p46)} These causal mechanisms represent "tendencies",^{178(p46)} which have the potential to activate to produce outcomes. When they do, they manifest in the realm of the actual where they can, at the level of the empirical, be observed and measured.¹⁷⁷ Critical realist approaches acknowledge that the ways in which researchers come to know about the world are shaped by our perspectives, rendering knowledge inherently incomplete and leaving open the possibility of alternative, valid explanations for what we observe.¹⁷⁹ However, critical realists assert that we can use rational thought as well as empirical observation and experimentation to test, evaluate and refine our ideas to "get closer to truth".¹⁸⁰

Critical realism frames the relationship between structure and human agency as "interdependent and partly independent".^{176(p88)} That is, it posits that humans act with agency but their behaviour is also shaped by the structures of which they are a part; and those structures both shape and are shaped by human action.^{176,181,182} Maniykhina & Alderson use an illustrative metaphor (Figure 9), asking: "Do rivers shape landscapes or landscapes shape rivers?"^{182,slide 14}



Figure 9. Image from Manyukhina & Alderson depicting a metaphor for the relationship between structure and human agency in critical realism^{182,slide 16}

Bhaskar's transformational model of social activity^{181(p12)} (Figure 10) suggests that human action – enabled or constrained as it is by existing structures – is also what reproduces those structures or transforms them over time.¹⁸¹



Individuals



Though they act with intention,¹⁸¹ humans are taken to be only partially aware of the motivations for and consequences of their behaviours.¹⁷⁶ Within this model, norms can be thought of as an aspect of the societal structures influencing behaviour (whether or not humans are conscious of this influence); and it is people's behaviours that sustain or challenge these norms. Figure 11 applies Bhaskar's transformational model to an imaginary secondary

school, using a three-part model to illustrate one theoretical pathway through which social norms might change. In this school, existing norms that constitute an aspect of the school's social structure influence the spectrum of behaviours that students feel are permitted (part 1). When students comply with existing norms and enact social sanctions that encourage others to do the same, their actions reproduce those norms within the school. But there is also potential for change: when students act in conflict with existing norms or weaken them by limiting the social sanctioning that encourages compliance (part 2), these acts can begin to shift, or transform, the normative societal structures that influence behaviour in the school. These structural changes might make it easier for students to take action to weaken or challenge prevailing norms going forward (part 3).



Figure 11. Bhaskar's transformational model applied to theoretical relationship between normative social structure and students' actions in school

DRV interventions attending to social norms seek to foster this kind of transformation. In their elaboration of the concept of realist mechanisms, Dalkin et al. distinguish between resource and reasoning mechanisms.¹⁸³ Resource mechanisms represent intervention inputs, which aim

to produce changes in participants' reasoning and subsequent changes in behaviour. Project Respect, for example, provides school staff with lesson plans (resources) to deliver to their students. These lessons aim, in part, to change injunctive norms about the social acceptability of DRV (reasoning), which it is theorised will reduce incidents of DRV. Mapping this to Bhaskar's transformational model of social activity, we can conceptualise this process as intervention resources (lesson plans) producing changes in perceptions of what behaviour is acceptable in the group (reasoning), which changes social structures (social norms) that enable or constrain behaviour in an intervention group. Those changes in reasoning lead to changes in behaviours (DRV victimisation and perpetration, for example) and their social consequences. Those behavioural shifts consequently transform the social structures themselves, making it a bit easier to act in concordance with health-promoting norms (or to reject harmful ones).

Within this framework, social norms are real reasoning mechanisms with the potential to sustain or prevent actual experiences of DRV. While DRV interventions commonly offer resources to activate this mechanism,⁸ methodological challenges impede efforts to measure norms at the empirical level and, consequently, to gather evidence to suggest whether (1) the norms are changing, and (2) these changes lead to reductions in DRV. I take a mixed methods approach to addressing this gap. While many critical realists reject the use of quantitative methods in social research,¹⁸⁴ mixed methods researchers outline how studies underpinned by critical realism can in practice draw on the strengths of both qualitative and quantitative approaches to develop a deeper understanding of social phenomena.¹⁸⁵ I subscribe to this view, integrating in my thesis research quantitative and qualitative methods to explore aspects of social norms measurement in more breadth and depth than would be possible using either approach on its own.^{178,186} I draw on the strengths of qualitative interviews to explore phenomena in depth and to generate theory¹⁸⁶ to address research questions about young peoples' experiences of social norms themselves (research question 4) and of responding to survey items about these (research question 2). This is complemented by use of quantitative, deductive¹⁸⁶ methods to test the performance of social norms measures in a large sample of young people in England (research question 3).

3.2.3. The use, reliability and validity of quantitative measures

In this section I provide an overview of the use of multi-item scales for measuring social phenomena. I then introduce the concepts of reliability and validity and discuss the role of each as an indicator of measure quality. This introduction presents the basis for the research presented in Paper 1 and Paper 3 (see Chapters 4 and 6, respectively), which involve assessing the quality of social norms measures by assessing, in part, their reliability and validity.

3.2.3.1. The use of multi-item scales in social science

Social scientists use multi-item scales to measure latent variables: constructs that theory suggests exist but that can't be measured directly.¹⁸⁷ Each scale item should share the construct that the scale is intended to measure as their "single common cause".^{187(p205)} If they do, the scale can be described as unidimensional: it measures a single construct.¹⁸⁷ A scale's composite score (made up of the scores of each individual item) is therefore taken to represent the level of the latent variable.¹⁸⁷ We can consider as an example the six-item scale designed to measure injunctive DRV norms in Project Respect student baseline surveys, refined on the basis of the cognitive testing presented in Paper 2 (see Chapter 5). This measure asks respondents about their friends' views on six statements about the acceptability of physical DRV. These six items represent different aspects of the phenomenon, two of which are support for girls perpetrating DRV (e.g. "Boys sometimes deserve to be hit by their girlfriends") and support for DRV under specific conditions (e.g. "It is okay for a boy to hit a girl if she hit him first") (see Chapter 5). Items are intended to be indicators of the latent variable's effects: e.g. because a respondent believes that their friends support DRV, they would report that their friend agrees with the statement "Girls sometimes deserve to be hit by their boyfriends" (see Chapter 5). When scores of all six item responses are added together,^s resulting composite scale scores are expected to be higher than among those of respondents who perceive friends' support for DRV to be low.

^s Statements framed in the negative are reverse-scored before computing the scale's composite score.

3.2.3.2. Reliability and validity as indicators of measure quality

Classical test theory (CTT) is a traditional social science approach to psychometric testing, assessing the performance of a measure by quantitatively testing its reliability and validity.¹⁸⁸ CTT assumes that a scale's score is made up of two components: the respondent's true score (i.e. the real value of the latent variable) and random, normally distributed error.¹⁸⁸ Scale scores are therefore expected to remain stable if the true value of the latent variable hasn't changed.¹⁸⁷ Tests of reliability assess to what extent this is the case. As Devellis defines the concept, "scale reliability is the proportion of variance attributable to the true score of the latent variable".^{187(p39)} Reliable scales measure a construct consistently. One way of examining reliability is by calculating a scale's internal consistency, or the correlations between the items in the scale.¹⁸⁷ Internal consistency is an indicator of the extent to which the scale's items assess the same underlying construct.¹⁸⁷ Other indicators of reliability include split-half reliability, which involves splitting a multi-item scale in half and testing the correlation between scores of each half of the scale, and test-retest reliability, which involves administering a measure to the same participant sample at two timepoints and testing the correlation between these two scores.¹⁸⁷ While tests of internal consistency and split-half reliability require multiitem scales, the test-retest approach can also be used to assess the reliability of single-item measures.

The extent to which a scale measures the construct it is intended to is a question of **validity**. Whether and how validity should be sub-categorised is a matter of debate,¹⁸⁹ but textbooks tend to distinguish between three types:^{187,189}

- **Construct validity** is the extent to which a scale correlates with measures of other constructs that are theoretically associated with the latent variable.¹⁸⁷
- **Content validity** is the extent to which a scale's items reflect the construct of interest and cover all relevant aspects of its domain. ^{t,187,190}

^t A scale's content validity is often framed in terms of whether its items assess all aspects of the construct of interest.¹⁸⁷ However, this is particularly difficult to assess for constructs such as beliefs and attitudes, for which it can be challenging to determine the range of potentially relevant items.¹⁸⁷ This thesis draws on definitions of content validity that take into account both the extent to which a scale

 Criterion validity refers to the correlation between a scale and an established "gold standard" measure of the construct of interest.¹⁸⁹

Within CTT, tests of reliability and validity assume that the scale being assessed is unidimensional.¹⁸⁷ Exploratory and confirmatory factor analysis (EFA and CFA, respectively) are two statistical approaches to testing this assumption.¹⁸⁸ By exploring correlations between scale items, EFA can be used to determine how many latent constructs (or factors) a scale is measuring.^{187,188} Results of EFA can also help to identify scale items that are not performing well in terms of their fit with the construct(s) being measured.¹⁸⁷ In CFA, on the other hand, researchers draw on theory or on the results of previous analysis to predict a scale's structure and then statistically assess the extent to which scale data fit that structure.^{187,188} EFA and CFA can both be considered tests of a scale's **convergent validity**,¹⁹¹ which refers to correlations between measures of the same construct.¹⁹² EFA and CFA can alternatively be described as assessing "item convergence within scales",^{188(p652)} or a scale's internal or factor structure.¹⁹¹

Methods for Paper 1 and Paper 2 (see Chapters 4 and 6, respectively) draw on these conceptualisations of reliability and validity to assess the quality, and test the performance, of social norms measures.

3.2.4. Systematic review of social norms measures relating to DRV and gender used in DRV research – methods for Chapter 4 (Paper 1)

I conducted a systematic review to address my first research question, "Are existing measures of adolescent social norms relating to DRV and gender reliable and valid?" This component of my thesis is presented as a peer-reviewed publication,¹⁹³ Paper 1, in Chapter 4. The following provides an overview of the methodology and methods used for this component of my thesis and reports on additional details that fall outside the scope of Paper 1. All methods critical to understanding this systematic review and interpreting its results are detailed in Paper 1. The

assesses all domains of the construct of interest¹⁸⁷ and the extent to which all of a scale's items are relevant to that construct.¹⁹⁰
review protocol was registered in advance on the Open Science Framework³ and is provided in Appendix 3.

3.2.4.1. Systematic review methodology

Methods for scientific approaches to evidence synthesis began to develop in earnest in the 1900s, though they took hold in health research only in the latter half of the century.¹⁹⁴ Systematic reviews address the need in health research to collate the universe of existing evidence on a particular topic^{194,195} and to present a "comprehensive" synthesis with minimal bias and transparently reported and reproducible methods.¹⁹⁵ While early systematic reviews focused on synthesising RCTs, newer methodologies have proliferated to address a range of different types of research questions.¹⁹⁶ These systematic approaches share nine key features that distinguish them from unsystematic literature reviews,¹⁹⁶ the latter of which report selectively on existing literature and provide more subjective, potentially biased overviews.¹⁹⁵ These nine distinguishing features, reported in Munn et al.'s typology of systematic reviews, are as follows:¹⁹⁶

- 1) Clear objectives and research questions
- 2) Explicit, a priori inclusion and exclusion criteria to determine study eligibility
- 3) Comprehensive search strategy to identify all relevant studies
- 4) Study screening and selection process
- 5) Appraisal of study quality and of the validity of study findings
- 6) Extraction and analysis of data from included studies
- 7) Presentation and synthesis of the extracted findings
- 8) Interpretation of results
- 9) Transparent reporting of the review methodology and methods used

Psychometrics is the branch of social science "concerned with measuring psychological and social phenomena".^{187(p3)} Munn et al. define psychometric systematic reviews as "systematic reviews of measurement properties".¹⁹⁶ This type of review involves systematically searching for and assessing the quality of health-related measurement tools in terms of available

evidence on aspects of their performance such as their reliability and validity.¹⁹⁶ Psychometric systematic reviews can be used to assess a particular measurement tool, a set of common tools for measuring a particular construct, all measurement tools for a particular population, or, as it does in this thesis, all measures of a particular construct within a specific population.¹⁹⁶

Guidelines for conducting psychometric reviews (though they don't use this term) have been developed by the COnsensus-based Standards for the selection of health Measurement Instruments (COSMIN) steering committee.^{190,196} COSMIN seeks to improve the selection of what the initiative refers to as "patient-related outcome measures" (PROMS) – questionnaires completed by patients to report directly on their own health.¹⁹⁰ While reports on one's health reflect a construct different from reports on one's perceptions (as social norms measures require), the COSMIN guidance lays out useful steps and practical quality assessment criteria for psychometric systematic reviews such as the one conducted here.

In-line with standard systematic review approaches,¹⁹⁶ the COSMIN guidelines begin with developing the aim of the review; establishing criteria for eligible measures and the studies in which they are reported; and systematically searching and screening the literature.¹⁹⁰ They next offer detailed steps and criteria for appraising the quality of included measures and grading the quality of the underlying evidence of their measurement properties. Measures are evaluated based on their content validity, internal structure, reliability, measurement error, criterion validity,^{190(p1152)} construct validity and responsiveness (sensitivity to changes).¹⁹⁰ COSMIN recommendations for the transparent reporting of psychometric systematic reviews align with the PRISMA Statement's 2016 guidelines for the reporting of systematic reviews.¹⁹⁰

Embedded in the COSMIN guidelines is an assumption that the literature under review has reached a level of maturity such that studies designed to develop or evaluate measures are available and that included measures have been assessed in multiple studies. However, evidence on measurement properties of social norms measures is still in the early stages. In

110

Section 3.2.4.3. I describe modifications made to the COSMIN inclusion and quality appraisal criteria to reflect the emerging state of the available literature.

3.2.4.2. Overview of systematic review methods

Eligibility criteria

This review aimed to systematically identify and assess existing measures of descriptive and injunctive social norms relating to DRV and to gender that have been used in DRV research. Eligible studies were empirical research published in English in 1997 and later. This timeframe was chosen because measures might become less meaningful and therefore appropriate for young people over time due to cultural changes⁸² and because 1997 can be considered the advent of social media,¹⁹⁷ which now plays an important role in the formation of adolescent relationships.¹⁹⁸ Eligible studies presented at least one quantitative measure of descriptive DRV norms, injunctive DRV norms, descriptive gender norms and/or injunctive gender norms and assessed construct validity by conducting at least one test of association between an eligible social norms measure and a DRV behavioural outcome among young people aged 10-18 years. An assessment of this aspect of construct validity was required because norms measures that are not associated with DRV outcomes would not be suitable for use in evaluations of DRV evaluations regardless of other aspects of measure quality. DRV behavioural outcomes were defined as DRV victimisation, DRV perpetration and DRV bystander behaviours. Studies could be reported in peer-reviewed published papers or grey literature.

Search strategies

I employed seven complementary search approaches to ensure comprehensive coverage of available literature:

- 1) Searched databases
- 2) Searched Google Scholar
- 3) Searched websites of relevant organisations
- 4) Contacted subject experts
- 5) Reviewed known literature contained in my existing database of DRV research

- 6) Reviewed the references of included reports
- 7) Screened programme evaluations included in known reviews of DRV interventions

Search terms were based on three concepts, linked by the Boolean search term "OR": (1) social norms relating to DRV and/or gender; (2) DRV; and (3) adolescents. Specific search terms were informed by known studies that included relevant measures. I piloted and refined the search strategy using known studies on social norms and DRV to assess its sensitivity. The search strategy was then reviewed by an LSHTM librarian according to Peer Review for Electronic Search Strategies guidance and refined based on their feedback.^{199,200} The full and final search strategy is available in Appendix 4. Paper 1 (see Chapter 4) provides further detail on the search methods employed.

Screening and data extraction

After deduplication,^u database search results were screened on title and abstract and then on full text as described in Paper 1. From included reports I extracted the following data for eligible social norms measures: development; content, comprising: title, number of items, item(s) and response options, type of social norm (injunctive or descriptive, DRV or gender) and reference group(s); mode of data collection; evidence of reliability and of construct, content and convergent validity (including setting, sample size and characteristics, DRV behavioural outcome measure, analysis method and results); and statistical properties (measures of the item or scale data's central tendency and distribution such as its mean, median, mode, skewness and standard distribution or standard error; responsiveness to change; and evidence of floor or ceiling effects). For eligible studies I extracted data on title, author, publication year, type of literature, study region, study design, eligibility and recruitment.

^u Deduplication followed the University of Leeds method as outlined by the LSHTM Library & Archives Service,²⁰¹ modified slightly to include Medline in-process records with the initial set of Medline record imports.

Quality appraisal and data synthesis

I assessed the quality of each included measure against seven criteria for which I assigned scores as described in Paper 1 (see Chapter 4): (1) the use of participatory development; (2) defined reference group; (3) reliability; (4) content validity; (5) construct validity (assessed as association with DRV behavioural outcome); (6) other evidence of validity (association with theoretically associated constructs, or structure affirmed by factor analysis); and other statistically desirable properties (lack of floor or ceiling effects, responsiveness to change or the availability of evidence on central tendency and distribution^v). Section 3.2.4.3. provides further detail on the sources and rationale supporting this approach. Further detail on these quality criteria are available in Supplemental Appendix B of Paper 1 (provided in Appendix 5 of this thesis). Reporting was guided by the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) checklist.²⁰³

3.2.4.3. Methodological considerations for systematic review

Decisions on the eligibility criteria and quality appraisal of social norms measures needed careful consideration. In this section I discuss the rationales underpinning key decisions which could not be addressed in detail in Paper 1 (see Chapter 4) due to journal limitations on article length.

Eligibility criteria for studies

COSMIN guidelines recommend including only studies designed to develop or evaluate the measurement properties of included measures.¹⁹⁰ Initial scoping of available literature suggested that this approach would be unfeasible given the early stage of social norms measurement in DRV research. To gather all available evidence in this nascent area of study I did not exclude studies on the basis of their aims.

^v The availability of data on the central tendency and distribution of a measure's score is useful because this information aids in interpreting the measure's scores for future use.²⁰²

Eligible social norms measures had to have been used with respondents ages 10-18 years. Age 10 was selected because this marks the start of adolescence;¹⁷ pressure to conform to gendered expectations begins to intensify at the start of puberty;⁹² and students aged 11-13 years report DRV, suggesting primary prevention would ideally begin earlier;¹⁴⁶ and this marks a turning point in the feasibility and developmental appropriateness of self-administered surveys.²⁰⁴ While social norms prior to age 10 could be salient for early primary prevention, differences in cognitive and reading skills before and after around age 10²⁰⁴ suggest that measures appropriate for younger respondents are unlikely to also be suitable for older adolescents. Age 18 was selected as the upper age limit because sexual violence research suggests prevention should begin prior to university²⁰⁵ and to maintain a focus on school-aged adolescents, excluding studies primarily comprised of university samples. To avoid excluding relevant studies that included some participants outside of this age range, I operationalised the age criteria by including studies for which >50% of participants were age 18 years or younger. If this information was unavailable, studies were included if the mean age was younger than 19 years (e.g. a study with a mean age of 18.9 years would be eligible for inclusion).

I aimed to explore the pool of existing measures that have been assessed for construct validity as norms associated with DRV behavioural outcomes among young people. Reports with relevant social norms measures but that did not report on their use among young people aged 10-18 years, and those that did not assess their association with a DRV behavioural outcome, were excluded. This is because, as Ashburn et al. observe, "there is considerable literature on theoretical ways to measure norms...[but] far fewer examples of social norm measures that have been utilized and shown valid in multiple contexts".⁶ In addition to DRV victimisation and perpetration, bystander behaviours (i.e. intervening to prevent or address DRV) were included as eligible outcome measures because they represent a concrete behavioural outcome that we would expect to correlate with social norms measures.

114

Eligibility criteria for social norms measures

Eligible descriptive norms measures assessed perceptions of how (1) typical or (2) frequent DRV or gendered behaviours were. Eligible injunctive norms measures assessed perceptions of (1) the social acceptability of DRV or (2) social expectations based on gender. Measures that asked whether anyone in the participant's reference group had been involved in DRV or if the participant had been told about or witnessed DRV were excluded, as were studies which assessed DRV rates among reference groups by asking the reference groups directly. This is because these approaches do not directly assess social perceptions of how widespread the behaviour is – the key feature of descriptive norms' influence. Witnessing or being told about an instance of DRV would not necessarily lead to a perception that DRV is typical within the reference group, while DRV behaviours that were common could be discreet. Similarly, measures that assess perceived consequences of DRV but do not specify these consequences as social (e.g. "bad things would happen" to the participant if they hit a partner)^{206(p71)} were excluded because they do not clearly assess the construct of social expectations key to the influence of injunctive norms. "Bad things" could, for example, refer to regulatory outcomes such as arrest, or personal harm such as feeling guilty. Finally, scales that assess social norms relating to sexual or other types of interpersonal violence without specifying the context of an intimate partnership were excluded. This is because non-partner violence might be driven by a different constellation of social norms than those underpinning partner-specific violence.

Rationale for conducting quality appraisal of measures but not studies

In line with the approaches of previous systematic reviews on measures in menstrual health research²⁰⁷ and of social norms relating to contraception,²⁰⁸ I focused on the characteristics and quality of included measures and did not grade the underlying evidence by assessing the quality of the studies themselves. This is because, reflecting the early stage of social norms measurement in DRV research, their measurement properties tend to be reported incidentally within studies with non-psychometric aims (e.g. epidemiologic and other observational studies) and the same measures did not tend to be referred to in different studies. COSMIN methods for grading the quality of evidence for specific measurement properties of each included

115

measure¹⁹⁰ will be more appropriate when findings from multiple studies can be synthesised to assess measurement properties for more established measures.

Quality appraisal of included measures

In the absence of established criteria for appraising the quality of social norms measures, I developed a tailored assessment tool. This approach was informed by the work of Doherty et al., who developed a novel quality appraisal tool for their systematic review of measures to assess the mental health of people who have experienced human trafficking.²⁰⁹ Informed by COSMIN guidelines,¹⁹⁰ methods of existing systematic reviews of measures,^{209–211} and literature on social norms measurement,^{6,208} I selected quality criteria inductively based on initial scoping of available evidence. In line with COSMIN guidelines, I assessed quality on the basis of available evidence on reliability (including internal consistency, split-half reliability and testretest reliability), content validity, construct validity, assessment of factor structure, floor or ceiling effects and responsiveness.¹⁹⁰ Cross-cultural validity/measurement invariance, which refers to an assessment of how consistently a measure performs across different populations,¹⁹⁰ was not assessed given the lack of studies focused on measure development and evaluation across populations. Feasibility¹⁹⁰ was not assessed because the review did not aim to make recommendations about specific measures for use. Criteria were added to assess availability of benchmark data on the central tendency and distribution of the measure's total score;²¹⁰ use of a defined reference group;^{6,208} and use of formative work involving the target population to inform measure development.⁶

Sample size thresholds for determining measurement properties with sufficient precision (e.g. the minimum sample required to establish properties like reliability and construct validity) are assessed through evidence grading in the COSMIN guidelines.¹⁹⁰ Drawing on detailed sample size thresholds offered by Terwee et al. and Lewis et al.,^w I embedded these criteria within the quality appraisal of included measures. Supplemental Appendix B of Paper 1 (see thesis

^w Terwee et al.'s criteria²⁰² are a predecessor to the COSMIN guidelines,¹⁹⁰ while Lewis et al. offer criteria for assessing the quality of measures of implementation.²¹⁰

Appendix 5) details sample size criteria for assessments of reliability, construct and convergent validity, floor or ceiling effects, responsiveness and measurements of central tendency and distribution.

A score was assigned for each criterion but not for the overall measure. This was to avoid obscuring specific strengths and weaknesses for each measure, which were presented by type of norm: descriptive DRV, injunctive DRV, descriptive gender and injunctive gender. Supplemental Appendix B of Paper 1 (see thesis Appendix 5) specifies the criteria and scoring for each quality criterion.

3.2.5. Cognitive interviews informing the development of social norms measures – methods for Chapter 5 (Paper 2)

Data from cognitive interviews testing new and adapted social norms measures were used to answer my second research question, "Are new and adapted measures of social norms relating to DRV and gender understandable and answerable when used in research with adolescents in England?" This component of my research is presented as a peer-reviewed publication,²¹² in Chapter 5. In this section, I provide an overview of the methodology and methods for this component of my thesis and report on details that could not be included in Paper 2 due to space. All methods critical to understanding this research and interpreting its results are reported Paper 2.

Cognitive testing of norms items was conducted as part of a broader cognitive interviewing component of the Project Respect study which also tested survey elements outside the scope of this thesis. In this section, I report on methods relating to the testing of social norms items.

3.2.5.1. Cognitive interviewing methodology

Cognitive interviewing, also referred to as cognitive testing,²¹³ is a qualitative approach to exploring whether participants consistently understand survey items as intended and how they approach responding to these items.²¹⁴ The "question-and-answer model", originating in

cognitive psychology, is a widely accepted theory of participants' processes for engaging with survey items.^{214(p231)} The model specifies four steps: comprehension (understanding the item's intended meaning), retrieval (accessing the necessary information from memory), judgement (assessing which information is needed to respond to the question) and response (selecting a response in line with the survey format and that the participant is willing to give).²¹⁴ Directly studying the question-and-answer process in cognitive interviews²¹⁴ enables researchers to identify problems with proposed survey items¹⁸⁹ and refine items before administering surveys. This can reduce measurement error stemming from participants' misunderstanding of items, from their not retrieving the necessary information or from problems with communicating their responses.²¹⁴

Cognitive interviewers typically combine the method's two techniques, "think-aloud interviewing" and "verbal probing".^{213(p354)} The think-aloud method asks participants to narrate their thoughts as they answer survey items.^{214,215} This approach benefits from being openended and minimising interviewer bias, but it can be difficult for participants and takes time for them to learn.²¹⁵ A warm-up exercise at the start of the interview can help to train participants in the technique.²¹⁵ Willis offers this example:

"Try to visualize the place where you live, and think about how many windows there are in that place. As you count up the windows, tell me what you are seeing and thinking about."^{215(p4)}

A warm-up gives the participant the opportunity to become comfortable with verbalising their thought process in front of the interviewer and for the interviewer to provide feedback on the participant's technique.

Verbal probing involves asking specific questions about the participant's experience of responding to items.^{214,215} Proactive probes are determined before the interview and reactive probes are developed *ad hoc* in response to what happens in the interview.²¹³ Interviewers can

use verbal probes concurrently, immediately after the participant has responded to a tested item and while the experience is still "fresh in the subject's mind";^{215(p7)} or retrospectively, after they've finished responding to all tested items. The latter approach might be more useful when items are self- rather than interviewer-administered and in later stages of development when the aim is to test items as they would ultimately be administered.²¹⁵ Complementing the think-aloud method, verbal probing allows the interviewer to ask about aspects of survey items they suspect might be a source of response error,²¹⁵ and places less burden on participants.²¹⁴ On the other hand, probes risk introducing bias from leading questions and so using them requires careful consideration of wording.²¹⁵ Cognitive interviews can be conducted with children as well as adults.²⁰⁴ For young people aged 13-15 years, the age range of the cognitive interviewing sample for this thesis, combining the think-aloud and verbal probing approaches and beginning with a warm-up to practice thinking aloud is thought to work well.²⁰⁴

Cognitive interview samples tend to be small (i.e. between ten and 30 participants).²¹³ Researchers should aim to recruit samples that are similar to the survey's target population and reflect the diversity present in that population.²¹³ Researchers can analyse audio-recordings of interviews or written item-by-item notes.²¹³ Willis recommends aggregating interview notes by item, looking for common themes both across interviews for a single item and that might be shared across items.^{213,215} Analyses should be sensitive to both common problems and problems that were uncommon but pose a serious risk to data quality.²¹⁵ Resource permitting, researchers can subject refined items to further rounds of cognitive testing.²¹⁵

3.2.5.2. Overview of cognitive interview methods

Measure development

I conducted cognitive interviews to test items designed to assess the following social norms constructs: descriptive DRV norms (two items), injunctive DRV norms (one item) and injunctive gender norms (two items). Interviews did not test items assessing descriptive gender norms because no suitable existing measures were identified to serve as a basis for this type of item. In light of emerging evidence suggesting that young people might struggle to distinguish between their own views and the views of others,^{158,159} I also tested one item assessing personal attitudes towards the same gender-stereotyped behaviour assessed in one injunctive norms item. Comparing results from parallel attitudinal and norms items enabled exploration of whether and how participants distinguished between these two constructs in their processes of making sense of and responding to these items. Tested items are shown in Table 3.

I adapted the descriptive DRV norms items from an existing measure with respectable reliability^x (Cronbach's *a*=0.70) in the evaluation of the Green Dot DRV and sexual violence intervention in US high schools.²¹⁶ To develop injunctive DRV and gender norms items, I adopted the approach used by Gage¹²⁹ (among others), which involves adapting existing attitudinal scales to ask participants about others' views rather than their own. The injunctive DRV norms item was adapted from a DRV attitudes scale with minimally acceptable reliability (Cronbach's *a*=0.69) used with 8th and 9th grade US students (typically aged 13-15 years) in the evaluation of the Safe Dates DRV intervention.⁷⁷ Injunctive gender norms items were adapted from the Attitudes Towards Women Scale, an attitudinal scale used with very good reliability (Cronbach's *a*=0.82) in a 2011 study of young people's attitudes towards gender roles and stereotypes and DRV in Greece (mean participant age=16.6 years).²¹⁷

^x In this thesis I use Devellis' subjective assessments of reliability, which ascribe the following labels: *a*<0.60, unacceptable; *a* between 0.60 and 0.65, undesirable; *a* between 0.65 and 0.70, minimally acceptable; *a* between 0.70 and 0.80, respectable; between 0.80 and 0.90, very good; "much above" 0.90, consider shortening scale.^{187(p145)}

Construct	Item	Item(s) tested	Response options
Descriptive DRV norms	<i>π</i>	Please tick one box on each line ^a to show how many students in your school you think has done each of the following:	
	DD1	a) How many boys in your school insult their girlfriend, swear at her, or try to control everything she does?	 None Some Many Most
	DD2	b) How many girls in your school insult their boyfriend, swear at him, or try to control everything he does?	 None Some Many Most
Injunctive DRV norms	ID1	 Please tick one box on each line^a to show how most other students in your school would feel if a student in your school did each of the following: a) A boy hit his girlfriend to get her back under control 	 Approve Disapprove Neither
Injunctive gender norms	IG1	 Please tick one box on each line^a to show how most other students in your school would feel about each of the following scenarios: a) A girl and a boy go on a date, and the boy pays all the expenses 	 Approve Disapprove Neither
	IG2	 Please tick one box on each line^a to show how most other students in your school would feel about a girl or a boy in your school who does each of the following: a) A girl in your school who has a lot of sex partners. 	 Approve Disapprove Neither
Attitudes towards gender roles/ stereotypes	AG1	 Please tick one box on each line^a to show how much you personally agree or disagree with each statement. a) On a date, the boy should pay all the expenses. 	 Strongly agree Agree Disagree Strongly disagree

Table 3.	Survey	, items	subject	to co	gnitive	testing
						0

^a Instructions for items refer to ticking "one box on each line" because the tested items form part of multi-item measures in which each item appears on a separate line. The wording was retained for cognitive testing including where only one item was tested.

Sampling and recruitment

I conducted cognitive testing to refine survey items in one London secondary school, which had expressed interest in the Project Respect pilot cluster RCT but was not yet involved (this school later replaced a school that withdrew from the pilot trial). I asked the school to purposively sample eight girls and eight boys of varying academic abilities from year groups 8 to 10, including at least two girls and two boys per year group. We recommended that students with personal experience of DRV not be selected due to the sensitive nature of the survey items to be tested. Participants' parents or carers received an information sheet prior to the interview and could opt out their child if they wished. On the day of their interview, I reviewed the assent form with participants and they had the opportunity to ask questions before completing the form and beginning the interview. The information sheet and assent form for cognitive interviews are provided in Appendix 6.

Data collection

I conducted cognitive interviews in a private room in the participating school in April 2017. Participants self-completed a brief demographic form (Appendix 7) before participating in a warm-up exercise to gain familiarity with the think-aloud method. Interviews followed a written guide (see Appendix 7) that combined the think-aloud method with proactive verbal probes; reactive probes were used as the need arose. Using show-cards to demonstrate how survey items and response options would appear in the survey, I read each tested item aloud and took detailed notes on participants' responses to each think-aloud task and verbal probe.

Analysis

I produced written summaries of responses to each think-aloud task and verbal probe by participant year group and then overall. Summaries were subjected to thematic analysis.¹⁷⁸ Drawing on Young et al.'s survey development work, initial codes related to understandability (assessing comprehension) and answerability (assessing recall, judgement and response).^{214,218} I developed sub-codes inductively to reflect item characteristics enhancing or impeding item clarity and participants' ability to respond.^{178,213} Finally, I applied this coding framework to the coding of notes on individual interviews to identify the participant-specific evidence underlying the themes emerging from analysis of written summaries.

3.2.5.3. Methodological considerations for cognitive interviews

In this section, I discuss my rationale for the selection of specific items for cognitive testing and the challenges preventing the inclusion of a descriptive gender norms item.

Measure development

I selected items for cognitive interviews that would enable me to test a range of item characteristics, as shown in Table 4. In addition to covering four distinct constructs (injunctive DRV norms, injunctive gender norms, descriptive DRV norms and attitudes towards gender roles and stereotypes), tested items varied by DRV type (physical, psychological), perpetrator (girls, boys) and gender role construct. Considering gender role constructs, gender attitude and norms items assessed indicators of benevolent sexism²¹⁹ and the sexual double-standard. The former refers to an ideology conferring protection and other benefits to girls and women who adopt traditional roles,²²⁰ while the latter refers to a common societal double-standard conferring more sexual permissiveness to boys and men than to girls and women.^{117,221}

Construct	Item #	DRV type		Perpetrator		Gender role construct	
		Physical	Psychological	Girls	Boys	Benevolent sexism	Sexual double- standard
Descriptive DRV norms	DD1		Х		Х		
	DD2		Х	Х			
Injunctive DRV norms	ID1	Х			Х		
Injunctive gender	IG1					Х	
norms	IG2						Х
Attitudes towards gender roles/ stereotypes	AG1					X	

Table 4.	Characteristics	of tested	items
----------	-----------------	-----------	-------

I did not test an item assessing gender descriptive norms because I found no existing measures to serve as a basis for such an item and existing literature linking gender norms to DRV tends to focus on the role of social expectations and social sanctions (injunctive norms) rather than on beliefs about typical gendered behaviour. Developing an evidence-based measure of descriptive gender norms associated with DRV would be a standalone research project outside of the scope of this thesis.

3.2.6. Reliability and validity testing of new and adapted social norms measures – methods for Chapter 6 (Paper 3)

The Project Respect pilot cluster RCT offers an analytic opportunity to test the performance of two new and one adapted measure of social norms. Following their refinement based on cognitive testing, these measures were piloted in student baseline surveys. I draw on these data to answer my third research question, "Are new and adapted measures of social norms relating to DRV and gender reliable and valid when used in research with adolescents in England, and how can they be refined?" I present this component of my thesis as Paper 3 (prepared for publication) in Chapter 6.

As reflected in the student baseline survey instrument (Appendix 8) and detailed in the full Project Respect study report,³⁷ student baseline surveys collected data on a number of sociodemographic, mediator and outcome variables. In this section I report on methods for the collection and analysis of survey data used to assess the performance of social norms measures, including details that could not be included in Paper 3. All methods essential to interpreting this research are reported in Paper 3.

3.2.6.1. Method for testing social norms measures

In line with the CTT approach described in Section 3.2.3.2., I conducted psychometric testing to assess the reliability and validity of three social norms measures and to identify potential refinements. Paper 3 (see Chapter 6) tests the construct validity of these measures by testing hypotheses about how they should correlate with other measures. I address content validity in the Discussion section of my thesis (see Chapter 8), where I draw on findings from Paper 4 (see Chapter 7) to reflect on the extent to which social norms underpinning DRV in England are incorporated into the social norms measures presented in Paper 3. Criterion validity cannot be assessed for these measures because, as reported in Paper 1 (see Chapter 4), there exists no gold standard, no established measures of DRV norms or of gender norms in DRV research.¹⁹³

3.2.6.2. Overview of methods used to test social norms measures

Sampling and recruitment

One school in the Project Respect trial did not take part in baseline surveys because it joined the study shortly before baseline surveys were administered, replacing a school that had withdrawn. Year 8 and 9 students from the other five study schools were invited to take part in student baseline surveys, excluding students whom school staff judged as not competent to provide assent. Prior to data collection, students and their parents/carers received a study information sheet and could opt out/opt out their child ahead of time if they wished. On the day of data collection, students received a copy of the student information sheet. LSHTM fieldworkers, trained in safeguarding, research procedures and administration of the Project Respect survey, described the study. Students had the opportunity to ask questions before completing an assent form to take part. Information sheets and the assent form for baseline surveys are available in Appendix 9.

Data collection

Trained fieldworkers administered student baseline surveys in schools in June-July 2017. Students self-completed surveys anonymously using electronic tablets, which were replaced by paper surveys on the rare occasion when technical issues inhibited survey completion. School staff remained in the classroom but were instructed to refer questions about the survey to fieldworkers. Eligible students with learning or language difficulties who required support to complete baseline surveys were supported to do so by fieldworkers. Copies of the survey and assent forms were left for eligible students who were not in school on the day of data collection to self-complete and return via post. Further details on fieldwork are available on the full study report.³⁷ See Appendix 8 for the full student baseline survey.

Survey measures

After refinement based on the results of cognitive testing (see Chapter 5), three measures of social norms were piloted and tested in student baseline surveys: one adapted measure of descriptive DRV norms (3 items), one new measure of injunctive DRV norms (6 items) and one

new measure of injunctive gender norms (5 items). Participants reporting having friends with girlfriends/boyfriends, based on a routing question, were eligible to respond to descriptive DRV norms items. All participants were eligible to respond to injunctive norms items.

Informed by cognitive testing (see Chapter 5), measures of DRV and gender attitudes were adapted based on measures used in the Safe Dates trial⁷⁷ and in a study of gender attitudes and DRV in Greece,²¹⁷ respectively. As reported in Section 3.2.5.2., these original measures had also served as the bases for the new measures of injunctive DRV and gender norms, respectively. All participants were eligible to respond to DRV and gender attitudes items.

DRV victimisation and perpetration were measured using the 58-item Safe Dates scale¹⁶⁷ and the 18-item CADRI-s scale,²²² both with slight adaptations informed by cognitive testing.⁹ Participants reporting ever having dated were routed to Safe Dates items and those reporting having a girlfriend or boyfriend in the past year were routed to both Safe Dates and CADRI-s items. Scales assessed eight DRV outcomes:

- The Safe Dates measure assessed overall victimisation and perpetration, and psychological and physical subscales assessed type-specific victimisation and perpetration.
- The CADRI-s assessed overall victimisation and perpetration.

Sociodemographic variables for all participants included sex assigned at birth,²²³ gender identity, age, year group, ethnicity,²²⁴ sexual identity, religion²²⁵ and socioeconomic status.²²⁶ Participants were categorised as SGM if they reported a minoritised sexual or gender identity, including (in addition to direct responses to sex and gender items) those reporting their gender as female if assigned male sex at birth or their gender as male if assigned female sex at birth.^z

^y Cognitive testing and refinements to DRV victimisation and perpetration measures are reported in detail in the full study report.³⁷

^z "Female" and "male" are sex rather than gender terms; however, wording here follows the wording of survey items.

Statistical analysis

Sample characteristics

I first explored characteristics of the overall sample and of the sample eligible to respond to descriptive norms items ("descriptive norms sample"). Subsequent analyses using descriptive norms items were conducted using the descriptive norms sample and other analyses were conducted using the full sample. To gain familiarity with the data, I examined relationships between DRV outcomes and sociodemographic and attitudinal variables. I examined DRV prevalence by group for categorical sociodemographic variables and assessed associations between DRV outcomes and continuous sociodemographic and attitudinal variables using univariable linear regression.

Characteristics of social norms items

I used descriptive analyses to assess completion rates and distributions of each of the 14 social norms items.

Assessing factor structure

Tests of reliability and validity assume that multi-item measures are unidimensional – that is, that they assess a single construct.¹⁸⁷ In line with Taylor et al.'s approach to attitudinal measures,¹⁴⁶ I conducted EFA on the full set of social norms items to determine whether items intended to measure the same construct loaded together. Items that loaded together were considered a single measure. I then conducted CFA to statistically test the fit of the resulting factor structure, i.e. the set of measures emerging from EFA.

Assessing reliability

I assessed each measure's reliability (how consistently it measures the assessed construct) using ordinal alpha, which is considered to be a more suitable index of reliability for Likert data than the more commonly reported Cronbach's alpha.²²⁷ I then tested whether removing any item would improve reliability. I carried out subsequent testing on the measures derived from factor loadings that demonstrated minimally acceptable or better reliability according to the criteria proposed by Devellis.¹⁸⁷

Exploring statistical properties

I used descriptive analyses to explore the distribution of each measure's total and mean response scores, determining the measures' sensitivity to different levels of the assessed norms. I compared mean response scores by sex using t-tests, accounting for clustering within schools.

Assessing validity

Tests of construct validity were guided by three *a priori* hypotheses:

- Pro-DRV and gender-inequitable norms will be associated with Safe Dates measures of DRV victimisation and perpetration in univariable analyses
- Pro-DRV and gender-inequitable norms will be associated with CADRI-s measures of DRV victimisation and perpetration in univariable analyses
- Sex will moderate one or more relationships between pro-DRV and genderinequitable norms and DRV

I tested these hypotheses using unadjusted linear regressions to assess associations between piloted social norms measures and DRV outcomes. I then added a sex*social norms interaction term to each regression model to assess whether relationships between piloted social norms measures and DRV outcomes were moderated by sex.

Linear regressions assume independence, normality and homoskedasticity of the residuals. Violations of these assumptions can result in heteroskedasticity, which means that the residuals (the differences between predicted and actual values of the outcome variable) are not randomly distributed across all values of the independent variable. Heteroskedasticity, which can reduce the accuracy of standard error estimates for regression coefficients (and the resulting 95% confidence intervals and tests of statistical significance), can be assessed using scatterplots which show the distribution of the residuals across values of the independent variable. To inform my analyses, I therefore tested whether the outcome variables used in my regressions (DRV victimisation and perpetration scores) were normally distributed. Finding skewed distributions, I examined residual plots for a sample of my univariable regressions, which showed heteroskedasticity. Data transformations were precluded by a high proportion of participants with DRV outcome scores of 0 because under these conditions transformations cannot approximate a normal distribution.²²⁸ I therefore used bootstrapping, an approach robust to heteroskedasticity. to improve estimates for these and subsequent regressions. With bootstrapping, an analysis is repeated many times^{aa} using smaller, randomly selected subsamples of the overall sample to simulate many smaller studies conducted from among an overall population defined by the full sample. The resulting estimates are normally distributed. These results are used to calculate the mean and standard error, which in bootstrapped regressions are taken as the bootstrapped regression coefficient and bootstrapped standard error.

Linear regressions also by default assume independence between observations, but clustered data like the data from Project Respect violates this assumption. I accounted for clustering within schools in my univariable and subsequent regressions by using robust cluster standard errors. This approach improves the accuracy of standard errors under these conditions by relaxing the assumption of independence between observations and imposing an assumption of independence between the sample's clusters (schools). Resampling for bootstrapped analyses took into account the clustered sampling design.

Linear regression with addition of covariates

To identify where refinements to the piloted measures might be needed, I conducted exploratory analyses using bootstrapped multivariable linear regressions that accounted for

^{aa} My bootstrapped analyses used 1000 replications, which is typically considered sufficient for calculating acceptable bootstrapped estimates.²²⁹

clustering using robust cluster standard errors. These regressions explored associations between piloted social norms measures and DRV outcomes, adjusting for sociodemographic and attitudinal covariates. Models were sex-stratified where sex*social norms interactions had been significant. The weight of existing evidence, including longitudinal research,^{77,82} suggests that some social norms and DRV outcomes have a significant association that is independent of the effects of personal attitudes.^{76,77,82,122,123} Gaps in existing literature and heterogeneity in measurement and methods preclude conclusions about which types of social norms are associated with which DRV outcomes in this way, and behavioural theory suggests that the relative importance of attitudes and norms is likely to vary by outcome and context.¹⁰⁰ Consequently, I did not develop *a priori* hypotheses about which tested measures would show an independent relationship with which DRV outcomes. Instead, I drew on the results of these analyses to assess whether the tested measures were sensitive to an independent relationship with DRV outcomes, and, if not, to consider refinements to improve their sensitivity.

3.2.6.3. Methodological considerations for testing social norms measures

Statistical analysis – testing validity

Evidence is mixed on how relationships between social norms and DRV might vary by sex. For example, Foshee et al. found that descriptive DRV norms were associated with DRV perpetration among boys in some analyses but not among girls⁷⁷ while Nardi-Rodriguez et al. found consistent correlations between descriptive DRV norms and boys' and girls' victimisation and perpetration.¹²⁸ Mixed evidence is also found for injunctive DRV norms (e.g. see Enosh 2007 and Hunt et al. 2022),^{76,78} and my systematic review (see Paper 1) identified no studies that compare associations between injunctive gender norms and DRV (either victimisation or perpetration) among both girls and boys.¹⁹³ So while within- and across-study evidence suggests that relationships between social norms and DRV can vary by sex, it does not support hypotheses about which norm-DRV relationships will vary in this way.

The tests of construct validity undertaken for my thesis share two underlying assumptions: (1) social norms correlate with DRV risk, and (2) this relationship varies by sex for some norm-DRV

outcomes pairings. However, there is no definitive way to assess construct validity.¹⁸⁹ If the piloted social norms measures don't correlate with DRV or vary by sex in their correlations as hypothesised, this could be for a number reasons. It could be due to a problem with theory (i.e. social norms do not underpin DRV or this relationship doesn't vary by sex) and/or with the piloted measure (i.e. social norms underpin DRV and do so differently for girls and boys, but the piloted measure does not really measure these norms).¹⁸⁹ It could also be due to a problem with the comparator measure (i.e. our outcome measure does not really measure DRV) or with a sample size too small to detect real correlations and effect-modification. Given this uncertainty, Streiner and Norman advise that "the weight of the evidence should be in favor of a positive relationship...the burden of evidence in testing construct validity arises not from a single powerful experiment, but from a series of converging experiments".^{189(p11)} I therefore assessed construct validity against different types of DRV (overall, psychological and physical) and both victimisation and perpetration. While a reliable and valid social norms measure might not correlate with every outcome, hypothesis-driven tests exploring outcome measures provide the opportunity to build up a picture of how the tested measures perform. The Safe Dates and CADRI-s scales were both selected because they are the two most commonly used relationship violence scales developed for adolescents²³⁰ and they provide different types of evidence for validity-testing. The Safe Dates scale is more sensitive, while the CADRI-s is more established and shorter, and so potentially more suitable for use in evaluation studies.³⁷

Statistical analysis – exploratory analysis to inform refinements

If social norms do play a role in shaping DRV outcomes, they should theoretically show an independent relationship to DRV outcomes after adjusting for sociodemographic and attitudinal covariates. Several studies identified in my systematic review (see Paper 1) have examined this relationship empirically, adjusting for DRV and/or gender attitudes. Given the range of different possible DRV outcomes (e.g. experience and perpetration of DRV overall or by type; and in samples of girls, boys or all genders), however, the picture they paint is quite patchy. There is currently no evidence on whether relationships between descriptive gender norms persist when accounting for DRV or gender attitudes. Considering descriptive DRV norms, attitude-

131

adjusted analyses variously report associations with DRV perpetration among boys^{77,82,123} or report no such association among girls,⁷⁷ boys^{77,81,82} or overall.¹³⁰ Similarly, findings on injunctive DRV norms are conflicting. Some studies report that these are not associated with either victimisation or perpetration among girls⁷⁶ or overall^{131,132} in attitude-adjusted analyses. However, others find that they are associated with victimisation⁷⁸ and perpetration^{83,127} among girls^{78,127} and overall^{78,83} in these analyses. Among boys, evidence suggests that injunctive DRV norms are associated with DRV victimisation^{76,78} but not perpetration^{76,127} in attitude-adjusted analyses. Similar analyses exploring gender-inequitable injunctive norms report an association with girls' victimisation¹³⁵ and with perpetration but not victimisation in a gender-aggregated sample.¹³³

The weight of available evidence suggests that some norms-DRV associations can be expected to persist when attitude variables are added to these models, but it does not pinpoint norms and DRV outcomes for which this is likely to be the case. I therefore used multivariable regressions to test whether the piloted measures were sensitive enough to demonstrate this relationship, and I reflect on findings to inform recommendations in Paper 3 (see Chapter 6) and in the Discussion section of my thesis (see Chapter 8) on potential refinements of the tested measures.

3.2.7. Relationships between social norms and DRV in student, staff and parent/carer accounts – methods for Chapter 7 (Paper 4)

Content validation refers to ensuring that a scale sufficiently covers the construct it is intended to measure and includes only items that reflect that specific construct.¹⁸⁹ This is easiest to ascertain for well-defined constructs. Devellis gives the example of a measure assessing all vocabulary words taught to sixth grade students.¹⁸⁷ Content validity is generally more difficult to ensure for constructs like attitudes or beliefs because it is difficult to define the universe of potentially relevant items.¹⁸⁷ Assessing and improving the content validity of the social norms measures tested in Chapter 6 requires evidence on which social norms are important to DRV behaviour in the UK, which could also inform preliminary development of items on descriptive

gender norms. Qualitative research with young people in the UK offers some insights^{34,138,139} but has not sought explicitly to explore these questions or to analyse data through a social norms lens. The Project Respect pilot trial provided an opportunity to explore prevalent social norms relating to DRV and gender with young people, parents/carers and school staff and to identify relevant norms implicated in their discussions about or justifications for DRV.

I drew on data from Project Respect optimisation sessions, trainings and process evaluation interviews and focus groups to address my fourth thesis research question, "What are student, staff and parent/carer accounts of social norms relating to DRV and gender in schools, and how are these implicated in DRV?". I present this component of my thesis as Paper 4 (prepared for publication) in Chapter 7. Details of optimisation and process evaluation methods have been previously reported in full.³⁷ In this section, I report on methods relating to the collection and analysis of the data I draw on for this component of my thesis, including details that could not be included in Paper 4 due to length. All methods essential to interpreting the results of this research are reported in Paper 4.

3.2.7.1. Method for exploring social norms and DRV in student, staff and parent/carer accounts

In contrast to quantitative research, which primarily produces numerical data, qualitative research primarily uses interviews and observations to produce written or oral language data.¹⁷⁸ While quantitative research focuses on quantifying phenomena,¹⁷⁸ qualitative research focuses on the contexts, meanings and explanations of participants' experiences and can be used to generate theory and hypotheses.¹⁸⁶ By exploring questions like "how" and "why",^{178(p8)} this approach can provide insight into processes over time, detailed accounts of setting and context, and a deep understanding of the concepts explored.¹⁸⁶

Conducted individually or in a group, qualitative interviews provide participant accounts of the phenomenon of study.¹⁷⁸ They are particularly well-suited to exploring participants' perceptions of why they and others do what they do.¹⁷⁸ This approach is suited well to my fourth research question because the conceptualisation of social norms in this thesis focuses on perceptions of

133

what others think and do and social mechanisms through which these factors influence DRV behaviour. While observational methods typically function to provide direct data on phenomena,¹⁷⁸ the observational data on which I draw for the present analysis (recorded trainings) are more akin to group interview data because they feature comments from staff about their perceptions of DRV and of behaviours in their schools that they find concerning. These data complement individual staff interviews for two reasons: (1) participants might feel more comfortable discussing sensitive topics, such as negative or critical views, in a group environment,¹⁷⁸ and (2) interview data are subject to social desirability bias,¹⁷⁸ but hesitance to share information that staff feel could cast their school in a negative light might be mitigated in a learning setting where it is a trainer rather than an evaluator posing questions. In a similar vein, utilising a mixture of group, paired and individual interviews with students provided opportunities to access both "shared social meanings"¹⁷⁸(p¹⁵⁵) that become evident in interactions between participants and more "in-depth accounts"¹⁷⁸(p⁶¹) as well as socially proscribed views that participants might hesitate to raise amongst their peers.¹⁷⁸

3.2.7.2. Overview of methods used to explore social norms and DRV in student, staff and parent/carer accounts

Sampling and recruitment

Optimisation sessions

From the list of secondary schools that had expressed an interest in but were not selected for the Project Respect pilot RCT (see Section 3.1.2.), we selected four to take part in focus groups to optimise the Project Respect intervention. We sampled schools purposively by region (southeast and south-west England) and deprivation, assessing the latter using the IDACI score assigned to the school's postcode. Each school's head teacher signed a consent form. For each optimisation session we aimed to include three girls and three boys from each of years 9 and 10 alongside three or more staff. We aimed for the latter to include: a school safeguarding lead; staff involved in Personal, Social, Health and Economic education; and senior leadership. Participants and the parents/carers of students invited to take part received an information sheet ahead of time and could opt out if they wished. Participants completed an assent/consent form at the start of each session. When feasible, participants from the first wave of optimisation sessions also took part in the second (final) wave. Information sheets and assent/consent forms for optimisation sessions are available in Appendix 10.

Process evaluation – trainings and interviews

For this component of my thesis, I drew on training and interview data collected as part of the Project Respect process evaluation. We aimed to audio-record NSPCC-led trainings and the cascaded all-staff trainings (delivered by school staff participating in the NSPCC-led trainings) in each of the four intervention schools. For NSPCC-led trainings, the trainer and each participant received a study information sheet ahead of time and completed a consent form at the start of the session. Trainers for cascaded all-staff trainings also received an information sheet ahead of time and completed a consent form before the session. Information sheets and consent forms for NSPCC-led and cascaded trainings are available in Appendix 11.

For interviews in each intervention school we aimed to recruit four staff, purposively sampled by seniority and programme involvement, and two parents/carers, purposively sampled by their child's year group and sex. We aimed to recruit eight students per intervention school and four per control school, purposively sampled by year group and gender. We also aimed to recruit two staff members per control school, purposively sampled by seniority. Interview participants, and the parents/carers of students invited to take part, received an information sheet ahead of time with information on how to opt out. At the start of the session, they received a copy of the information sheet and completed an assent/consent form. A sample of information sheets and assent/consent forms for process evaluation interviews are available in Appendix 12.

Data collection

Optimisation sessions

We sought to conduct one session per school for each of two waves of optimisation sessions. The first informed the intervention's content and format. The second sought feedback on the planned intervention and draft materials, and explored terminology used among young people, the role of social media in romantic and sexual relationships, and factors that could affect programme implementation. We began each session with a slide presentation outlining key information. The first session also included prompts for whole-group discussions in the course of this presentation. Staff and students were then separated for focus groups, each of which was led by a member of the NSPCC or research team using semi-structured discussion guides (optimisation session and process evaluation guides are available in Appendix 13). We noted in the first session that students were more forthcoming once separated from staff and that younger students were reluctant to speak, which informed the decision for subsequent optimisation sessions to shorten the whole-group portion and to further separate year 9 and year 10 students for focus group discussions. NSPCC and study team members took notes on optimisation sessions. Sessions in the second wave were also audio-recorded and transcribed.

Process evaluation – trainings and interviews

NSPCC-led trainings aimed to enable school and intervention leaders to plan and deliver Project Respect, while cascaded all-staff training delivered by school staff leading the intervention aimed to prepare all school staff to recognise, prevent and respond to DRV. Trainings included discussions among staff about signs of DRV and about behaviours in the school that concern them, providing data on school context. We conducted interviews using semi-structured guides (see Appendix 13). Where schools requested, student interviews were conducted in pairs or in groups. All interviews explored school context, including how gender was negotiated in schools, for example in terms of school gender balance, power relations and gender norms, and the extent to which the school environment reinforced gender role differences.⁹¹ Intervention school interviews further explored programme implementation, acceptability, impact and mechanisms of change. Control school interviews further explored provision of relationship and sex education, social and emotional learning and violence prevention.

Data analysis

I had initially gained familiarity with Project Respect optimisation, training and interview data through my work on the per-protocol process evaluation. To address my fourth thesis research

question, I conducted a secondary analysis of these data focusing on dynamics and expectations concerning gender, dating and relationships and sexual harassment.

Taking a common approach to analysing qualitative data in public health, I conducted thematic analysis complemented by techniques drawn from grounded theory.¹⁷⁸ Blending deductive and inductive approaches,¹⁷⁸ my research question and a review of background theoretical and empirical literature guided starting codes and subsequent analysis. As new themes emerged, new codes were added inductively to capture accounts of social norms and how they present and operate within schools. These could come underneath starting codes or be added as new independent codes in the coding framework. Axial coding was then used to explore relationships between codes. As analysis progressed, codes were combined where there was significant overlap and separated where distinct constructs emerged, to arrive at a final coding scheme.

3.2.7.3. Methodological considerations for exploring social norms and DRV in student, staff and parent/carer accounts

Data collection and analysis

Direct and indirect evidence on social norms and DRV

Collecting qualitative data for the optimisation and process evaluation of Project Respect provided an opportunity to explore young people's accounts of social norms relating directly and indirectly to DRV. Personal experiences with DRV and adolescent sexual abuse in the UK have been explored elsewhere^{34,138,139} and were not the focus of this research. We therefore did not purposively select interview participants with experience of DRV, and we oriented interview questions more broadly to explore perceptions of behaviours and norms within participating schools. Where participants drew direct links between social norms and DRV, I drew this out in my analysis. Acknowledging that people are not fully and consciously aware of the factors driving their and others' behaviours,^{176,231} I drew on empirical and theoretical literature to interpret participant accounts of social norms and gender relations in their contexts and how these might contribute to or protect against DRV.¹⁷⁸

Sequencing of analysis

Bearing in mind that qualitative research can provide a deep but relatively narrow perspective from a limited number of participants compared to quantitative surveys, I staged my analyses to move from the data sources providing the broadest perspective and most general application (i.e., from optimisation and control schools) to those exploring experience with the Project Respect intervention. I anticipated that interviews from intervention schools would provide richer data on DRV and social norms because the intervention itself addressed these. In addition to participants having concrete lessons and potential programme impacts on which to draw, I anticipated that they were also likely to have considered these issues in more depth over the preceding year and to be more practised in discussing them, yielding more in-depth observations and reflections. I therefore conducted initial coding of optimisation and process evaluation data in the following order:

- 1. Optimisation sessions
- 2. Control school interviews
- 3. Intervention school data

3.2.8. Ethics

Project Respect received ethical approval from the LSHTM Ethics Committee (reference: 11986) and the NSPCC Research Ethics Committee (R/17/106). The latter provided separate approvals for each component (baseline surveys, intervention implementation, process evaluation and follow-up surveys). Ethical approval letters are provided in Appendix 14. These approvals cover all data collection and per-protocol analyses of optimisation sessions, process evaluation data, cognitive interviews and baseline surveys. I conceptualised and obtained ethical approval for an amendment to the study's original ethics application to examine the attitude and social norms measures in the cognitive interviews. The literature review I conducted for Paper 1 of this thesis, presented in Chapter 4, is exempt from ethical review. I conducted secondary analyses of data collected for Project Respect for Papers 2-4, presented in Chapters 5-7, respectively. I obtained ethical approval for this secondary analysis from the LSHTM Ethics Committee (ref: 28163), and this ethical approval letter is included in Appendix 14.

3.2.9. Reflexivity and positionality

I share the critical realist view that social science should seek to improve societal well-being.¹⁷⁷ Shaped by my communities and the feminist values with which I was raised, before coming to public health I studied and worked in politics and policy advocacy. I became interested in public health because of its practical, integrative approach to assessing risks to health and wellbeing, and how they might be mitigated most effectively. What drew me to DRV research, in particular, was its potential for contributing to transformation towards a more genderequitable society. Aware of how these core values have shaped the paths I have pursued, I have been alert to the importance of approaching my research and this thesis with curiosity and not dogma. Studying behavioural theories in previous research on theory synthesis²³² and then in the early work on my thesis was formative in expanding my view of the wide-ranging factors that interact to produce health outcomes. I hope with my thesis to make a useful contribution to social norms measurement in DRV research, a narrow but important area that is still in its early days of development. The data from Project Respect offered an analytic opportunity for this work. Having seen measurement as a black box in the past, I shaped a project that would provide the opportunity to expand my qualitative skillset while taking my first steps into the vast and somewhat intimidating field of psychometrics.

As a university student, I trained and volunteered in peer patient-centred counselling, and I drew on key skills from that work in my qualitative interviews. These included asking openended questions and reflecting back what is said, techniques to avoid biasing participant responses and to check that I have understood participants' meanings as they intended. Familiarity with these skills also helped to offset an imbalanced dynamic in my interactions with study participants. I sensed that in meeting with me as a university researcher coming to evaluate the programme they had been tasked with delivering, school staff were sometimes inclined to portray their schools and their work in a positive light. Many faced challenges with implementing Project Respect due to structural and resource constraints in the school, as we have previously reported,² which might have made participants feel defensive of their work or their school. Similarly, students were sometimes hesitant to speak openly in interviews with adult professionals they had never previously met. At the start of my interviews, I reiterated my

139

role – to evaluate survey questions (for cognitive interviews) and the programme (for process evaluation interviews) – and that I was interested in knowing both what they thought worked and what they thought didn't. Though levels of rapport and openness varied by participant, we were able to have insightful and engaged discussions and the data collected represent a range of participant views.

Chapter 4. Paper 1: Social norms measures relating to DRV and gender – systematic review of DRV research

4.1. Introduction to Paper 1

When developing student surveys for the Project Respect pilot RCT, we encountered a practical challenge: we could identify no established measures of social norms relating to DRV and gender as theorised mediators in the intervention's theory of change. Given the time constraints of an ongoing trial, there was no scope to conduct a systematic review, so I instead conducted an *ad hoc* review of available literature to try to locate appropriate measures. Aware that social norms experts had drawn attention to the common practice of measuring personal attitudes but not social norms,^{6,155} and to the dearth of established, validated norms measures in the field of adolescent SRH,⁶ I was unsurprised to identify no established measures in DRV research.

Through this process I identified a gap in existing DRV research: though social norms were widely viewed as important,⁸ measures to assess them had not been systematically collated and assessed. I therefore set out to conduct a systematic review and quality assessment of measures of the types of social norms most extensively linked to DRV (DRV norms and gender norms) and to answer the first research question of my thesis: "Are existing measures of adolescent social norms relating to DRV and gender reliable and valid?" The practicalities of field research meant that the Project Respect study had ended before findings from this review were available and the review could not be used to inform the tested measures. Instead, findings from the review fill in the broader picture of social norms measurement in DRV research. They reveal a disjointed field of study with some promising approaches but little coherence, cumulation and synthesis of knowledge, strategic direction or shared principles. This work sets the stage for the subsequent results chapters of my thesis, which describe the systematic testing and refinement of three specific measures and draw on evidence from this work to, informed by the review presented in this chapter, offer recommendations to further social norms measurement in DRV research.

141

The published supplemental appendices to which Paper 1 refers are provided in Appendix 5 of this thesis.

I have previously presented preliminary findings from this systematic review at the Lancet's UK Public Health Science conference (2022);²³³ in the associated abstract published in a special issue of *The Lancet* (2022)⁴ and provided in Appendix 15; and in a webinar for the Learning Initiative on Norms, Exploitation and Abuse (LINEA) Project (2020).²³⁴ 4.2. Paper 1: Social norms about dating and relationship violence and gender among adolescents: systematic review of measures used in dating and relationship violence research



London School of Hygiene & Tropical Medicine Keppel Street, London WC1E 7HT

T: +44 (0)20 7299 4646 F: +44 (0)20 7299 4656 www.lshtm.ac.uk

RESEARCH PAPER COVER SHEET

Please note that a cover sheet must be completed for each research paper included within a thesis.

SECTION A - Student Details

Student ID Number	452277	Title	Ms.		
First Name(s)	Rebecca				
Surname/Family Name	Meiksin				
Thesis Title	Developing and refining social norms measures for the evaluation of a school-based intervention to reduce dating and relationship violence among adolescents in England				
Primary Supervisor	Professor Chris Bonell				

If the Research Paper has previously been published please complete Section B, if not please move to Section C.

SECTION B - Paper already published

Where was the work published?	Trauma, Violence, and Abuse				
When was the work published?	February 2023				
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion	N/A				
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes		

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C - Prepared for publication, but not yet published

Where is the work intended to be published?	
Please list the paper's authors in the intended authorship order:	

Improving health worldwide

www.lshtm.ac.uk
Stage of publication	Choose an item.
----------------------	-----------------

SECTION D - Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	I led protocol development, conducted the literature searches, led screening, conducted data extraction and led analysis, writing and submission. I trained Manika Garg and Amiya Bhatia in the review protocol for their roles in supporting title/abstract screening and checking data extraction, respectively. Anjalee Kohli was the lead advisor on the protocol. Chris Bonell, GJ Melendez- Torres and Nambusi Kyegombe helped to interpret the data. All authors provided feedback on the manuscript and aproved the manuscript prior to publication.
---	---

SECTION E

Student Signature	Rebecca Meiksin
Date	15/03/2024

Supervisor Signature	Chris Bonell
Date	15/03/2024

www.lshtm.ac.uk

Social Norms About Dating and Relationship Violence and Gender Among Adolescents: Systematic Review of Measures Used in Dating and Relationship Violence Research

TRAUMA, VIOLENCE, & ABUSE I-15 The Author(s) 2023 C 0 0 Article reuse guidelines: sagepub.com/journals-permissions DOI: 10.1177/15248380231155526 journals.sagepub.com/home/tva SAGE

Rebecca Meiksin¹^(b), Chris Bonell¹, Amiya Bhatia¹, G.J. Melendez-Torres²^(b), Nambusi Kyegombe¹, and Anjalee Kohli³

Abstract

Adolescent dating and relationship violence (DRV) is widespread and associated with increased risk of subsequent poor mental health outcomes and partner violence. Shifting social norms (i.e., descriptive norms of perceived behavior and injunctive norms of acceptable behavior among a reference group of important others) may be important for reducing DRV. However, few DRV studies assess norms, measurement varies, and evidence on measure quality is diffuse. We aimed to map and assess how studies examining DRV measured social norms concerning DRV and gender. We conducted a systematic review of DRV literature reporting on the use and validity of such measures among participants aged 10–18 years. Searches included English peer-reviewed and grey literature identified via nine database; Google Scholar; organization websites; reference checking; known studies; and expert requests. We identified 24 eligible studies from the Americas (N=15), Africa (N=4), and Europe (N=5) using 40 eligible measures of DRV norms (descriptive: N=19; injunctive: N=14) and gender norms (descriptive: N=17; injunctive: N=6). No measure was shared across studies. Most measures were significantly associated with DRV outcomes and most had a defined reference group. Other evidence of quality was mixed. DRV norms measures sometimes specified heterosexual relationships but rarely separated norms governing DRV perpetrated by girls and boys. None specified sexual-minority relationships. Gender norms measures tended to focus on violence, but missed broader gendered expectations underpinning DRV. Future research should develop valid, reliable DRV norms and gender norms measures, and assess whether interventions' impact on norms mediates impact on DRV.

Keywords

social norms, dating violence, dating and relationship violence, domestic violence, intimate partner violence, measurement

Introduction

Dating and relationship violence (DRV) refers to intimate partner violence (IPV) involving a young person (Young et al., 2017), defined here as aged 12–18 years. It comprises physical, psychological, and/or sexual abuse perpetrated or experienced by a current or former intimate partner (Barter & Stanley, 2016; Breiding et al., 2015; Young et al., 2017). DRV is widespread among girls and boys (Leen et al., 2013; World Health Organization, 2021): in systematic reviews, psychological DRV victimization rates range from 47% to 88% (Exner-Cortens et al., 2016a) and meta-analyses suggest prevalence of 21% for physical and 14% (among girls) and 8% (among boys) for sexual DRV (Wincentak et al., 2017). While specific prevalence rates vary widely by measurement and sampling (Exner-Cortens et al., 2017), patterns tend to be consistent: psychological DRV is the most common, followed by physical and then sexual DRV, often with multiple types co-occurring (Leen et al., 2013). Experiencing DRV can lead to injuries (Foshee, 1996) and is associated with increased risk of subsequent depression (Roberts et al., 2003), substance use, antisocial behavior (Exner-Cortens et al., 2013; Roberts et al., 2003), suicidal ideation (Exner-Cortens et al., 2013), and suicide attempts (Castellvi et al., 2017) among girls and boys. In addition, it is a leading risk

¹London School of Hygiene & Tropical Medicine, UK ²University of Exeter, UK ³Georgetown University, Washington, DC, USA

Corresponding Author:

Rebecca Meiksin, London School of Hygiene & Tropical Medicine, 15-17 Tavistock Place, London WCIH 9SH, UK. Email: rebecca.meiksin@lshtm.ac.uk factor for morbidity and mortality among girls aged 15– 19 years (Mokdad et al., 2016), with girls who experience DRV reporting harms additional to those reported by boys including fear (Barter et al., 2009), increased substance use (Exner-Cortens et al., 2013), and more injuries (Foshee, 1996). DRV victimization is a longitudinal risk factor for IPV victimization (Exner-Cortens et al., 2017; Herrenkohl & Jung, 2016) and perpetration (Manchikanti Gómez, 2011) in adulthood, highlighting the influence of adolescent relationships on future development (Exner-Cortens et al., 2017) and the importance of early intervention.

Systematic reviews report that interventions have been successful in increasing DRV knowledge (De La Rue et al., 2014; Fellmeth et al., 2013) and changing personal attitudes (De La Rue et al., 2014), but demonstrate little impact on DRV perpetration or victimization (De La Rue et al., 2014; Fellmeth et al., 2013). While little is known about effective DRV prevention, social norms theory posits that harmful social norms can hinder behavior change despite changes in knowledge or attitudes (Alexander-Scott et al., 2016), while protective norms can support behavior change (Cislaghi & Heise, 2018). Social norms comprise perceptions of typical behaviors (descriptive norms) and acceptable behaviors (injunctive norms) among a reference group of important others, with social sanctions playing an important role in holding norms in place (Alexander-Scott et al., 2016; Cislaghi & Heise, 2018).

Empirical research finds that DRV norms are associated with DRV victimization and perpetration. Considering descriptive norms, young people who believe that their friends experience or perpetrate DRV are more likely to report perpetrating DRV themselves (Kinsfogel & Grych, 2004; Reed et al., 2011), including in longitudinal studies (Foshee et al., 2001; Vagi et al., 2013), even when controlling for their own attitudes toward DRV (Foshee et al., 2001). Girls who report having friends involved in violent relationships are at increased risk for subsequent victimization (Arriaga & Foshee, 2004). Although injunctive norms are less explored in the literature, data also suggest that injunctive pro-DRV norms in secondary school are associated with sexual violence perpetration prior to university (Salazar et al., 2018).

Gender norms, "collective beliefs about what behaviors are appropriate for women and men and the relations between them" (The Social Norms Learning Collaborative, 2021, p. 8), can be thought of as a particularly powerful type of social norms (Lokot et al., 2020) and play an important role in DRV risk. Qualitative research finds myriad ways in which inequitable gender norms operate to underpin male DRV in heterosexual relationships, including by forming a basis for the social acceptability of sexual coercion (Barter et al., 2009) and by grounding girls' status in having a boyfriend (Marston & King, 2006), which could present a barrier to ending abusive relationships (Barter, 2006; Barter et al., 2009). In interviews with young people, norms supporting the legitimacy of male dominance in relationships emerge as drivers of both physical violence and controlling behaviors (Barter et al., 2009; Wood et al., 2011). Although less explored in quantitative research, evidence also suggests that inequitable injunctive norms relating to household gender roles (Shakya et al., 2022) and female sexual availability (Wesche & Dickson-Gomez, 2019) are associated with an increased risk of DRV.

This evidence suggests that interventions may need to shift social norms concerning DRV and gender that support DRV. Social norms theory, and the approach of fostering protective DRV and gender norms among peers, has long informed DRV interventions (Offenhauer & Buchalter, 2011; Stanley et al., 2015; Wolfe & Jaffe, 1999). This is evident in the popularity of bystander interventions, which encourage young people to intervene in DRV (Stanley et al., 2015), and of gender-transformative approaches (Stanley et al., 2015; Whitaker et al., 2006), which aim to reshape gender roles and promote "more gender-equitable relationships" (Gupta, 2000, p. 10). Evaluations suggest that norms-based interventions can be effective in reducing intra-marital and domestic violence (Fulu et al., 2014), and there is emerging evidence that interventions with young people (Plourde et al., 2016) or their parents (Ehrensaft et al., 2018) can shift DRV-specific social norms among adolescents. However, evaluations of DRV interventions rarely assess impact on social norms (Coker et al., 2017; Foshee et al., 2005; Miller et al., 2014; Taylor et al., 2011), and to our knowledge none have assessed social norms as a potential mediator of intervention effects, limiting what is known about intervention mechanisms.

This may be due in part to the lack of consensus on how to measure social norms and to limited evidence as to the reliability and validity of existing measures (Ashburn et al., 2016). Valid measures assess the construct in question (DeVellis, 2017): in this case, social norms that are important to DRV outcomes. Reliable measures do so consistently (DeVellis, 2017). Measures of social norms used with adults are unlikely to be suitable for adolescents due to likely differences between these populations in reference groups, behaviors, and cognitive ability to distinguish between personal attitudes and the views of others (Moreau, 2018; Moreau et al., 2021). We therefore reviewed existing DRV literature to explore (1) what measures exist of adolescent descriptive and injunctive social norms concerning DRV and gender and (2) the validity and reliability of these measures.

Methods

This review was guided by a study protocol registered on the Open Science Framework (Meiksin, 2020) and is exempt from ethical review.

Eligibility, Search Strategy, and Screening

Eligible reports were studies published in English since 1997. We selected this timeframe because cultural shifts

might render older measures meaningless or inappropriate for young people today (Reyes et al., 2016), and because 1997 marks the advent of social media (History Cooperative, n.d.), which plays an important role in the initiation and formation of relationships among adolescents (McGeeney & Hanson, 2017). Reports were required to the assess the construct validity of one or more quantitative measure of norms relating to DRV and/or gender (including bystander norms in these domains) by testing these against DRV behavioral outcomes (i.e., by exploring their association with DRV victimization, perpetration, and/or bystander behavior). Measures were assessed among participants aged 10-18 years and comprised one or more survey items, with at least 50% of items assessing one of four domains: descriptive DRV norms, injunctive DRV norms, descriptive gender norms, or injunctive gender norms. Where eligible measures comprised subscales, subscales were also included as unique measures if they independently met eligibility criteria.

Measures of DRV and gender norms overlap where those relating to perceptions of the typicality or social acceptability of DRV are "gender specific," by which we mean they assess norms governing girls and boys separately (e.g., a measure assessing the social acceptability of a boy hitting his girlfriend). We categorized all measures of DRV norms as DRV norms whether or not they were gender specific. Broader gender norms measures, that is, those that did not focus on violent behaviors in the context of adolescent relationships or dating, were categorized as gender norms. Descriptive norms were operationalized as perceptions of the typicality or frequency of (1) DRV and (2) gendered behaviors, excluding DRV behaviors. Injunctive norms were operationalized as perceptions of (1) DRV's social acceptability and (2) social expectations based on gender, excluding social acceptability of DRV.

Our search strategy used free-text and controlled vocabulary terms linked by the Boolean connector "OR" for three concepts: (1) social norms concerning DRV and/or gender; (2) DRV; and (3) adolescents. The search terms used within each concept were linked by the Boolean connector "AND" (see Supplemental Appendix A for Medline search strategy). The search strategy was peer reviewed based on the Peer Review for Electronic Search Strategies guidance (McGowan et al., 2016; Shamseer et al., 2015). After piloting the strategy in Medline, in June 2019 we searched nine databases containing reports relevant to our topic: IBSS; Popline; Medline; PsychINFO; PsychEXTRA; EMBASE; Web of Science; Global Health; and Scopus. We conducted additional searches via Google Scholar (July 2019; limited to the first 100 results), websites of relevant organizations (June 2020) (Care Evaluations, n.d.; Explore Our Resources, n.d.; Find a Report, n.d.; Girl Effect, n.d.; Global Early Adolescent Study, n.d.; Publications, n.d.; Resources, n.d.), two online databases of relevant measures (June 2020) (EMERGE, n.d.a; EMERGE, n.d.b; Gender and Power Metrics, n.d.), contacting subject experts (February-March

2020) (Advancing Learning and Innovation on Gender Norms (ALIGN), n.d.; Gender Violence and Health Centre (GVHC), n.d.; Learning Collaborative to Advance Normative Change—IRH, n.d.; Sexual Violence Research Initiative, n.d.), our study team's internal database of DRV literature, and reference checking. We also screened for eligibility all evaluations included in eight reviews of DRV intervention studies (Cornelius & Resseguie, 2007; De Koker et al., 2014; De La Rue et al., 2014; Fellmeth et al., 2013; Leen et al., 2013; Lundgren & Amin, 2015; Stanley et al., 2015; Whitaker et al., 2006).

Search results were imported into EndNote X9 (The EndNote Team, 2013), de-duplicated ("Removing Duplicates from an EndNote Library," 2018), and dual-screened on title and abstract by the first author (RM) and another reviewer in batches of 50 until reaching 85% agreement. These reviewers discussed records of uncertain eligibility to reach a consensus. RM then single-screened remaining records on title and abstract and screened all retained records on full text, discussing records of uncertain eligibility with AK and CB.

The database search was updated in March 2022, excluding IBSS (due to lack of institutional access) and Popline (retired in September 2019) (USAID, n.d.).

Data Extraction

From all included reports, RM extracted study information and the following data for each eligible measure: method of development; content; mode of data collection; evidence of reliability, construct validity, content validity, and convergent validity; and statistical properties. A second reviewer (AB) checked all extracted data, flagging areas of disagreement which were then resolved through discussion. We requested missing information on social norms measures and analysis results from study authors.

Analysis and Synthesis

Informed by previous reviews of measures (Costenbader et al., 2017; Hennegan et al., 2020), we report on the quality of included norms measures rather on than the overall quality of included studies. This quality assessment is the key focus of this review. Drawing on existing methods for assessing survey measure quality (Doherty et al., 2016; Lewis et al., 2015, 2018; Pocock et al., 2021; Prinsen et al., 2018; Terwee et al., 2007), we assessed each measure against seven criteria: participatory development; defined reference group; reliability (internal consistency, test-retest reliability, or split-half reliability); content validity (assessed as 75% or more items aligning with a relevant domain); construct validity (significant association with DRV behavior); other evidence of validity (factor analysis; or significant association with theoretically related constructs: DRV/gender attitudes, DRV intentions or perceived behavioral control over DRV); and statistically desirable properties (responsiveness, lack of



Figure 1. Flow diagram of search and study selection.

floor/ceiling effects, or data available on measures of central tendency and distribution of total score for the full measure-or, where absent, for all measure subscales) (Lewis et al., 2015; Terwee et al., 2007). Significance of associations for construct validity criteria was assessed using the threshold of $p \le .05$, or a lower p value where lower values were used by the authors of included reports. Reliability was scored to reflect poor reliability (Cronbach's alpha or correlation of <0.70) (-1), no evidence (0) or good reliability (Cronbach's alpha or correlation of ≥ 0.70) (+1). Construct validity was scored to reflect an inverse relationship between pro-DRV/inequitable gender norms and DRV (-1), no evidence of a significant relationship (0), or pro-DRV/inequitable gender norms associated with increased risk of DRV (+1). All other criteria were scored as evidence absent (0) or present (1). Supplemental Appendix B further details our methods assessing for assessing measure quality.

Within each of the four social norms domains considered in this review, we inductively created categories of constructs assessed by included measures (Hennegan et al., 2020). We then created tables summarizing features of included measures and evidence on their quality; and summarizing characteristics of the measures and of the samples in which their reliability and construct validity were assessed. Drawing on these tables and other extracted data, we summarized the following: features of included studies; features of included measures; and evidence for measures' validity and reliability, identifying strengths and limitations of existing measures.

Results

Literature Search

Database and Google Scholar searches identified 7,347 unique records (Figure 1), of which 477 were retained to screen on full text and 21 were eligible for inclusion (Aizpitarte et al., 2017; Antônio et al., 2012; Enosh, 2007; Flisher et al., 2007; Foshee et al., 2001; Gagné et al., 2005; Gonzalez-Mendez et al., 2019; Hébert et al., 2019; Helland, 1998; Hopper, 2011; Kernsmith & Tolman, 2011; Kinsfogel & Grych, 2004; Peskin et al., 2017; Pöllänen et al., 2021; Price, 2002; Reed et al., 2011; Reyes et al., 2016; Shamu et al., 2016; Shorey et al., 2018; Van Ouytsel et al., 2020; Wesche & Dickson-Gomez, 2019), reporting on 21 unique studies. Two reports presented analyses of different social norms measures from the same randomized controlled trial (Foshee et al., 2001; Reyes et al., 2016) and were therefore treated as two unique studies. Our updated search identified one new eligible report (Hunt et al., 2022), and two additional reports were identified by screening known studies (Nardi-Rodríguez et al., 2022; Shakya et al., 2022), resulting in the inclusion of 24 eligible reports of 24 unique studies.

Included Studies

Of the included studies, 11 were conducted in the United States (Foshee et al., 2001; Helland, 1998; Hopper, 2011; Hunt et al., 2022; Kernsmith & Tolman, 2011; Kinsfogel & Grych, 2004; Peskin et al., 2017; Reed et al., 2011; Reyes et al., 2016; Shorey et al., 2018; Wesche & Dickson-Gomez, 2019), three in Canada (Gagné et al., 2005; Hébert et al., 2019; Price, 2002), three in South Africa (Flisher et al., 2007; Pöllänen et al., 2021; Shamu et al., 2016), three in Spain (Aizpitarte et al., 2017; Gonzalez-Mendez et al., 2019; Nardi-Rodríguez et al., 2022), one in Belgium (Van Ouytsel et al., 2020), one in Brazil (Antônio et al., 2012), one in Israel (Enosh, 2007), and one in Niger (Shakya et al., 2022). All studies were observational, and seven (Enosh, 2007; Foshee et al., 2001; Peskin et al., 2017; Pöllänen et al., 2021; Reyes et al., 2016; Shakya et al., 2022; Shamu et al., 2016) analyzed data collected as part of an evaluation. One report presented only longitudinal associations between social norms measures and DRV (Nardi-Rodríguez et al., 2022), all other reports presented cross-sectional analyses, and three presented both (Foshee et al., 2001; Reyes et al., 2016; Shorey et al., 2018). Considering participants, 17 studies sampled girls and boys, four included only girls and three included only boys (Supplemental Appendices C and D provide further details of study and sample characteristics, respectively). All studies assessed relationships between social norms measures and DRV victimization and/or perpetration; none assessed relationships with bystander behaviors.

Included Measures

Most studies included a single eligible social norms measure assessing a single domain of interest (N=15) (Aizpitarte et al., 2017; Antônio et al., 2012; Enosh, 2007; Foshee et al., 2001; Hébert et al., 2019; Helland, 1998; Hunt et al., 2022; Kernsmith & Tolman, 2011; Peskin et al., 2017; Price, 2002; Reed et al., 2011; Reyes et al., 2016; Shorey et al., 2018; Van Ouytsel et al., 2020; Wesche & Dickson-Gomez, 2019). Six studies included two eligible measures (Flisher et al., 2007; Gonzalez-Mendez et al., 2019; Hopper, 2011; Kinsfogel & Grych, 2004; Pöllänen et al., 2021; Shakya et al., 2022) and one study included three (Shamu et al., 2016). In addition, one study included a single measure for which half the items assessed one social norms domain and half assessed another (Gagné et al., 2005), and a second study included four such measures (Nardi-Rodríguez et al., 2022). Since half of a measure's items must assess a domain of interest for inclusion as a measure of that domain, each of these five measures was eligible for inclusion in two separate domains and therefore counted twice for this review. The review therefore includes 35 unique measures, assessed as 40 measures across four domains: 19 measuring descriptive DRV norms, 14 measuring injunctive DRV norms, one measuring descriptive gender norms, and six measuring injunctive gender norms.

Characteristics of Measures

Measures were generally quite short, comprising a median of six items (range=1-28, mean=7). For most measures (58%), all items assessed the domain of interest. Where information was provided on measure development, reports suggested that six measures were adapted from measures of DRV outcomes (Aizpitarte et al., 2017; Antônio et al., 2012; Hopper, 2011; Kinsfogel & Grych, 2004; Van Ouytsel et al., 2020); two were adapted from a measure of personal attitudes (Shakya et al., 2022); two were tools used in previous studies (Hébert et al., 2019; Kernsmith & Tolman, 2011); one was adapted from a previous study to ask about physically rather than sexually aggressive behaviors (Helland, 1998); and six were newly developed (Flisher et al., 2007; Peskin et al., 2017; Pöllänen et al., 2021; Wesche & Dickson-Gomez, 2019). For detailed information on each included measure please, see Supplemental Appendices D (measure wording, variable calculation) and E (initial development, reference group, content validity, reliability, and construct validity).

Descriptive DRV norms. The review identified 19 eligible measures of descriptive DRV norms from 14 included reports (Table 1) (Aizpitarte et al., 2017; Antônio et al., 2012; Foshee et al., 2001; Gagné et al., 2005; Gonzalez-Mendez et al., 2019; Hébert et al., 2019; Helland, 1998; Kinsfogel & Grych, 2004; Nardi-Rodríguez et al., 2022; Peskin et al., 2017; Price, 2002; Reed et al., 2011; Reyes et al., 2016; Shorey et al., 2018). Measures ranged from 1 to 26 items (mean=6, median=4). Most specified reference groups of friends or peers. Only two referenced social rewards or consequences for adhering to/violating a norm (Flisher et al., 2007; Pöllänen et al., 2021). Questions were typically framed to ask for perceptions of the number or proportion of referencegroup members who had experienced or perpetrated DRV, or for perceptions of whether "most" reference-group members had done so (Foshee et al., 2001, p. 133; Nardi-Rodríguez et al., 2022, pp. 12-13; Reyes et al., 2016, p. 353). Most measures referred to specific DRV behaviors among the reference group (e.g., hitting, yelling, threatening, forcing sex).

Most measures were gender neutral, that is, they did not specify gender or they included items about girls and boys within the same measure. Most asked about perceptions of DRV perpetration alone. We identified three inductive categories of constructs measured (Table 1). In all, 10 "gender/sexuality-neutral DRV" measures did not specify

Table 1. Eligible Constructs Assessed by Included Measures and Evidence of Measure Quality, by Domain and Inductive Grouping.

	Construct Assessed						Measure Quality (Possible Score)										
		DRV	Туре		Involv	ement		Gender		Specifies Heterosexual	Participatory	Defined Reference	Reliability	Content Validity ^a	Construct Validity ^b	Other Validity	Statistically Desirable
Report	Psych.	Phys.	Sex.	Gen.	Vict.	Perp.	F	м	GN	Relationship	Dev. (0 to 1)	Group (0 to 1)	(-1 to 1)	(0 to 1)	(-1 to 1)	(0 to 1)	Properties (0 to 1)
Descriptive DRV norms																	
Gender/sexuality-neutral DRV																	
Aizpitarte et al. (2017)	1	1	1			1			1		0	1.00	1	1.1	1	0	1
Antonio et al. (2012)	1	1	1			1			1		0	1	1	1.1	1.1	0	0
Gonzalez-Mendez et al. (2019) (#1)				1		1			1		0	1.1	0	1.1	1.1	0	1
Gonzalez-Mendez et al. (2019) (#2)				1	1				1		0	1	0	1	1	0	1
Hebert et al. (2019)		1	1		1				1		0	1	-1	1	1	0	0
Helland (1998)		1				1			1		0		0		- i -	1	0
Kinsforel and Grych (2004) (#1)	1	5							1		0		ĩ	ò		0	ő
Kinsford and Grush (2004) (#2)	1								1					ĩ		ĩ	
Parkin at al. (2017)	1	1				1			1		ő						
Shorey et al. (2017)	*	1							1		ő					ő	
Silorey et al. (2018)						~						1 A A A A A A A A A A A A A A A A A A A					1 C C
Mixed DRV		,			,	,			,	,							
Gagne et al. (2005) (#1)		×			~	~	×		×	×	0	1.1.1	0	0		0	0
Heterosexual DRV																	
Foshee et al. (2001)		~				1	~	×		~			1	0		0	0
Nardi-Rodriguez (2022) (#1)	1					1		×.		~	0	1	1	0	1	1	1
Nardi-Rodriguez (2022) (#2)	~					~		~		~	0	1 I I I I I I I I I I I I I I I I I I I	1	0	1	1	1
Nardi-Rodriguez (2022) (#3)	~				~		1			1	0	1	1	0	1	1	1
Nardi-Rodriguez (2022) (#4)	~				1		1			~	0	1	1	0	1	1	1
Price (2002)	~					~		~		1	1	1	1	1.1	1	1	1
Reed et al. (2011)		1	1			1		~		1	0	1	0	1.1	1	0	0
Reyes et al. (2016)		1				1	1	~		1	1	1	-1	1.1	1	1	1
Injunctive DRV norms																	
Respondent DRV																	
Kernsmith and Tolman (2011)				1		1	1				0	1	1	1.1	0	1	1
Nardi-Rodriguez (2022) (#5)	1					1		1			0	1	1	0	1	1	1
Nardi-Rodriguez (2022) (#6)	1					1		1			0	1	i i	0	1	i i	
Nardi-Rodriguez (2022) (#7)	1				1		1				0		i	0	- i -	i	
Nardi-Rodriguez (2022) (#8)	1				1		1				ő		- i -	õ		- i	
Politinen et al. (2019) (#1)			1			1	1	1			ĩ			ő			
Pollinen et al. (2019) (#1)			1			1	1	1				ĩ		ĩ		ő	
Gender peutral beterorexual DRV						*						1.1					1.1.1
Easth (2007)			1			1	1	1		1			0			0	0
Elisher et al. (2007) (#2)		1	~			×,	×.	×,		×.							0
Hisner et al. (2007) (#2)					,	×,	×.	×,		×,			-				
Hopper (2011) (#1)	×	,			×.	×,	×.	×,		×,							
Hopper (2011) (#2)		×,			×.	×.	×.	×,		×.	0						
Hunt et al. (2022)		×			~	×	×	~		×	0	1.1.1				U	1.1
Mixed or unspecified DRV																	
Flisher et al. (2007) (#1)		1				1	1	~		~	0		-1		0		0
van Ouytsel et al. (2020)	~					~			1		0						1
Descriptive gender norms																	
Gagné et al. (2005) (#2)	# of frie	nds involv	ed in male	-perpetrat	ed sexual	violence ag	ainst fem	iale peer			0	1	0	0	1	0	0
Injunctive gender norms																	
Gendered violence																	
Shakya et al. (2022) (#2)	Beliefs a	bout peop	de in villag	e's attitude	es toward	phys. viole	nce again	st women			0	1	0	1	0	0	0
Shamu et al. (2016) (#1)	Beliefs about family/friends' attitudes toward gender-based violence					1	1	-1	0	1	1	1					
Shamu et al. (2016) (#2)	Beliefs a	bout famil	y's attitud	es toward	gender-ba	used violence	e				1	1.1	-1	1	1	0	1
Shamu et al. (2016) (#3)	Beliefs a	bout frien	ds' attitud	es toward	gender-ba	used violence	e				1	1	-1	1.1	1.1	0	1
Gendered expectations																	
Shakya et al. (2022) (#1)	Beliefs a	bout peop	de in villag	e's attitude	es toward	household	gender n	oles			0	1.00	1	1.1	1.1	0	1
Wesche and Dickson-Gomez (2019)	Beliefs a	bout sexu	al expecta	tions of fe	male gang	members					0	0	1	1.1			1

Note. DRV = dating and relationship violence; F = female; gen. = general; GN = gender neutral; M = male; participatory dev. = participatory development; perp. = perpetration; psych. = psychological; phys. = physical; sex. = sexual; vict. = victimization.

*≥75% items assessing social norms domain.

^bRelationship to DRV behavioral outcome.

heterosexual or sexual-minority relationships (Aizpitarte et al., 2017; Antônio et al., 2012; Gonzalez-Mendez et al., 2019; Hébert et al., 2019; Helland, 1998; Kinsfogel & Grych, 2004; Peskin et al., 2017; Shorey et al., 2018). One "mixed DRV" measure assessed perceptions of gender-neutral perpetration and female victimization within heterosexual partnerships (Gagné et al., 2005), and eight "heterosexual DRV" measures assessed perceptions of DRV within heterosexual relationships (perpetration by girls and boys within one measure (Foshee et al., 2001; Reyes et al., 2016), boys' perpetration (Nardi-Rodríguez et al., 2022; Price, 2002; Reed et al., 2011) or girls' victimization (Nardi-Rodríguez et al., 2022).

Injunctive DRV norms. We identified 14 eligible measures of injunctive DRV norms from eight included reports (Table 1) (Enosh, 2007; Flisher et al., 2007; Hopper, 2011; Hunt et al., 2022; Kernsmith & Tolman, 2011; Nardi-Rodríguez et al., 2022; Pöllänen et al., 2021; Van Ouytsel et al., 2020). These ranged from two to 28 items (mean=8, median=6). Six specified a single reference group of respondents' friends and six referred to multiple reference groups, one of which also assessed the importance of each (Kernsmith & Tolman, 2011). One measure did not specify a reference group (Pöllänen et al., 2021).

Measures asked respondents to report their perceptions of the views of reference group members, or the extent to which the respondent thought that DRV perpetration would "make me seem successful" (Pöllänen et al., 2021, p. 9). Nine (64%) asked about norms governing DRV perpetration alone (Enosh, 2007; Flisher et al., 2007; Kernsmith & Tolman, 2011; Nardi-Rodríguez et al., 2022; Pöllänen et al., 2021; Van Ouytsel et al., 2020). Most measures used Likert scale response options.

Half of the measures were gender specific (Kernsmith & Tolman, 2011; Nardi-Rodríguez et al., 2022; Pöllänen et al., 2021), and all but one measure (Kernsmith & Tolman, 2011) specified a single type of DRV. We identified three inductive categories of included measures (Table 1). Seven "respondent DRV" measures assessed injunctive norms governing DRV among survey respondents (Kernsmith & Tolman, 2011; Nardi-Rodríguez et al., 2022; Pöllänen et al., 2021) (e.g., asked to select a response for what will happen "[i]f I put pressure on my boyfriend or girlfriend to have sex. . .") (Pöllänen et al., 2021, p. 9). Five "gender-neutral heterosexual DRV" measures combined DRV among girls and boys and focused on heterosexual partnerships (Enosh, 2007; Flisher et al., 2007; Hopper, 2011; Hunt et al., 2022). Two "mixed-or-unspecified DRV" measures assessed a combination of DRV perpetrated by girls and boys in heterosexual relationships and by young people responding to the survey (without specifying partner gender) (Flisher et al., 2007), or gender-neutral DRV perpetration (Van Ouytsel et al., 2020).

Descriptive gender norms. The review identified one eligible measure of descriptive gender norms from one included report (Table 1). This measure assessed perceptions of the prevalence of male-perpetrated sexual coercion of females (without specifying a dating/relationship context) among friends from the past year (Gagné et al., 2005).

Injunctive gender norms. The review identified six eligible measures of injunctive gender norms from three included reports (Table 1) (Shakya et al., 2022; Shamu et al., 2016; Wesche & Dickson-Gomez, 2019). Measures ranged from one to 15 items (mean=6, median=5) and where response options were described, measures used Likert scales. Four "gendered-violence" measures assessed injunctive norms governing male-perpetrated violence and violence against girls/women (e.g., "My family thinks that there are times when a woman deserves to be beaten") (S. Shamu, personal communication, May 2, 2019), without specifying the context of adolescent dating/relationships. Two "gendered expectations" measures assessed social norms concerning broader gender roles (Shakya et al., 2022; Wesche & Dickson-Gomez, 2019), including sexual expectations of female gang members (Wesche & Dickson-Gomez, 2019) and gender roles within the family or household (Shakya et al., 2022).

Quality of Measures

Table 1 shows the quality of included measures by domain and inductive category. Further details on the evidence underpinning our quality assessment are available in Supplemental Appendices D (study samples, DRV outcome measures), E (summaries of initial development, reference group, content validity, reliability, and construct validity) and F (construct validity: analysis methods, results, and summary findings showing alignment between norm and outcome measures).

Descriptive DRV norms. Among the 19 included measures of descriptive DRV norms, three (16%) were informed by participatory development and all had defined reference groups. In all, 11 (58%) had good reliability and two (11%) had poor reliability. In total, 12 measures (63%) had good content validity. All showed a significant association between higher levels of perceived DRV prevalence and higher DRV risk. Eight measures (42%) also had other evidence of validity and 12 (63%) had statistically desirable properties.

Most measures were tested separately against DRV perpetration and/or victimization outcomes. Although most descriptive DRV norms measures were gender neutral, almost all were tested against gender-specific DRV outcomes, primarily standalone measures of girls' victimization and/or boys' perpetration.

The six gender-specific descriptive DRV norms measures were tested against DRV outcomes that matched the gender of the norms measure (Nardi-Rodríguez et al., 2022; Price, 2002; Reed et al., 2011) (i.e., norms concerning DRV among boys tested against DRV outcomes among boys). In all, 15 measures were tested against DRV outcomes that matched on type of DRV involvement (victimization or perpetration), and 11 were tested against outcomes that matched on type(s) of DRV (psychological, physical, and/or sexual). Five measures matched the DRV outcome against which they were assessed in all three dimensions, which focused on boys' perpetration (Nardi-Rodríguez et al., 2022; Price, 2002) and girls' experience (Nardi-Rodríguez et al., 2022) of psychological DRV.

Injunctive DRV norms. Of the 14 included measures of injunctive DRV norms, three (21%) were informed by participatory development. In all, 13 (93%) included a defined reference group, 11 (79%) had good reliability, and two (14%) had poor reliability. Nine measures (64%) had good content validity. In total, 11 (79%) showed a significant association between pro-DRV norms and higher DRV risk. Ten (71%) had other evidence of validity and 11 (79%) had statistically desirable properties.

Most injunctive DRV measures were tested against standalone DRV perpetration outcomes and against gender-specific outcomes. All seven gender-specific measures were tested against DRV outcomes specifying the same gender (Kernsmith & Tolman, 2011; Nardi-Rodríguez et al., 2022; Pöllänen et al., 2021). Ten measures were tested against outcomes that matched on victimization (Nardi-Rodríguez et al., 2022) or perpetration (Enosh, 2007; Flisher et al., 2007; Nardi-Rodríguez et al., 2022; Pöllänen et al., 2021; Van Ouytsel et al., 2020), and 13 were tested against DRV outcomes that matched on type of DRV (Enosh, 2007; Flisher et al., 2007; Hopper, 2011; Hunt et al., 2022; Nardi-Rodríguez et al., 2022; Pöllänen et al., 2021; Van Ouytsel et al., 2020). Six gender-specific measures aligned with assessed DRV outcomes in all three dimensions, focusing on girls' and boys' perpetration of sexual DRV (Pöllänen et al., 2021) and on boys' perpetration and girls' experience of psychological DRV (Nardi-Rodríguez et al., 2022).

Descriptive gender norms. The single measure of descriptive gender norms had a defined reference group and showed a significant association with DRV outcomes: Girls who reported more inequitable descriptive gender norms (i.e., more friends involved in sexual coercion) were significantly more likely to report DRV victimization (Gagné et al., 2005). The measure met no other quality criteria.

Injunctive gender norms. Of the six included measures of injunctive gender norms, three (50%) were informed by participatory development and five (83%) had a defined reference group. Two (33%) had good reliability and three had poor reliability. Five (83%) had good content validity. For five measures, inequitable gender norms were significantly associated with higher DRV risk. Two measures had other evidence of validity and five had statistically desirable properties. Five were tested against gender-specific DRV outcomes (Shakya et al., 2022; Shamu et al., 2016).

Discussion

Summary of Key Findings

Our findings suggest that social norms measures relating to DRV that are valid and reliable among young people can be developed, but that measurement is inconsistent and evidence supporting the quality of existing measures is limited. We found no eligible measure used more than once, limiting comparability across studies. Geographic diversity was also limited, with more than half of included studies taking place in the Region of the Americas. We found no eligible measures used in the South-East Asian, Eastern Mediterranean, or Western Pacific Regions (World Health Organization, 2021).

Most measures reviewed had evidence of construct validity, assessed as a significant association between pro-DRV/ inequitable gender norms and increased DRV risk. Measures were typically tested against gender-specific DRV outcomes, most commonly girls' victimization and boys' perpetration. Psychological, physical, and sexual DRV all featured frequently among the behavioral outcomes explored. Evidence on reliability and on other types of validity was mixed.

Though under a third of included measures had evidence of being informed by participatory development with young people, nearly all specified a defined reference group. However, all reference groups were pre-defined; no measure asked respondents to identify who held the most influence over them in relation to the assessed norms (Costenbader et al., 2017) and only one assessed the importance of each reference group to the respondent (Kernsmith & Tolman, 2011).

Two-thirds of gender norms measures asked about the respondent's friends and/or family, two groups that are particularly influential in gender socialization (Kågesten et al., 2016). However, several measures combined items asking about multiple reference groups, including unbounded groups of "others" and "people important to you" (Flisher et al., 2007, p. 622): features that limit their usefulness for gathering valid data about norms among a clear, coherent group and the relationship between these norms and DRV. Only two measures of injunctive norms referenced social sanctions, both without specifying the reference group applying these (Flisher et al., 2007; Pöllänen et al., 2021).

Several measures specified norms within heterosexual partnerships. Though sexual-minority youth face significantly higher risk of DRV than their heterosexual peers (Dank et al., 2014; Luo et al., 2014; Young et al., 2017), no measures specified norms governing sexual-minority relationships and no studies explicitly explored associations between included measures and DRV within sexual-minority relationships. Little is known about social norms contributing to DRV among same-sex partners and the key reference groups among which these norms are held. Some experts have suggested minority-stress theory (Dietz, 2019; Martin-Storey & Fromme, 2021; Reuter & Whitton, 2018) as a framework for understanding the elevated DRV risk among sexual-minority youth, which would suggest that homophobia, underpinned by gender norms (Solomon, 2015; Whitley, 2001), could play an important role. Formative research is needed to explore the social norms influencing same-sex DRV, and its findings should form the basis of social norms measures used with sexual-minority youth.

Considering measures of DRV norms, several studies explored the relationship between descriptive DRV norms and DRV outcomes, while fewer explored the relationship between injunctive DRV norms and DRV outcomes. DRV norms measures most commonly focused on DRV perpetration, and most were gender neutral. While studies usually explored DRV norms as predictors of gender-specific DRV outcomes, this was less common for measures of injunctive than descriptive DRV norms, despite evidence suggesting that predictors of DRV differ for girls and boys (Ali et al., 2011; Arriaga & Foshee, 2004; Capaldi et al., 2012; Foshee et al., 2001, 2011; Leen et al., 2013). A minority of DRV norms measures were tested against DRV outcomes focusing on the same gender, involvement (victimization/ perpetration), and DRV type. This presents an important limitation to existing measures of DRV norms: social norms theorists hypothesize that norms relating directly to a behavior of interest (as the most salient at the time of the behavior) generally exert a stronger influence than do more distal norms (Cislaghi & Heise, 2018). Empirical literature suggests that this may be the case for DRV. DRV norms may affect DRV outcomes via gender-specific pathways (Foshee et al., 2001; Pöllänen et al., 2021; Shorey et al., 2018), and in Gagné et al.'s (2005) research physical DRV norms predicted physical and psychological but not sexual DRV. The relationship between attitudes and DRV outcomes has been more widely explored, finding that young people tend to view male-perpetrated DRV more negatively than female-perpetrated DRV (Exner-Cortens et al., 2016b; Reeves & Orpinas, 2012; Rogers et al., 2019), and that attitudes toward DRV vary by DRV type (Exner-Cortens et al., 2016b; Reeves & Orpinas, 2012), with attitudes most strongly predicting DRV outcomes of the same type (Exner-Cortens et al., 2016b). Omitting or combining genders, victimization/perpetration, and/or types of DRV in measures of DRV norms (and the outcomes these might predict) therefore risks missing important differences in norms and their influence.

Far fewer studies explored the relationship between gender norms and DRV. Compared to measures of DRV norms, gender norms measures tended to have less evidence of reliability and of validity assessed as an association with theoretically related constructs aside from DRV behaviors. As a strength, most gender norms measures were assessed

for their relationship with gender-specific DRV outcomes. We identified only one measure of descriptive gender norms, which did not appear to be conceptualized as such given that only half of its items assessed this domain. Injunctive gender norms measures were more conceptually consistent, with the vast majority showing good content validity. However, both tended to focus on the social acceptability of violence by males and/or against females. This is a limitation to existing measures, as evidence points to the importance of separating gender norms from violence norms to avoid conflating the relationships between these distinct constructs and DRV behavior (Reyes et al., 2016). Only two measures asked about broader gendered expectations, assessing norms governing female sexual roles and gender roles within the family/household. No measures explored other gendered expectations that qualitative research suggests contribute to DRV, such as the social importance of sustained heterosexual relationships for girls (Barter et al., 2009; Marston & King, 2006) and of being sexually active for boys (Wood et al., 2011).

No eligible measures assessed bystander norms, reflecting limited evidence on the relationship between norms supporting protective, DRV-specific bystander behaviors and DRV outcomes. However, it is important to note that research with adolescents that reports on measures of DRV, gender, and bystander norms not assessed for their relationship to DRV outcomes, or on measures of related norms (e.g., sexual violence norms), can offer insights into norms measurement among this population.

Limitations

Like all reviews, this review might have missed eligible reports published after our search was completed. However, our database search was extensive and updated near the end of the study period, and no additional reports were identified through our expert requests. Eligible reports might also have been missed where abstracts did not indicate that relevant norms measures were used. However, we mitigated this risk by full-text screening evaluations of DRV interventions identified via reviews, and reports for which abstracts referenced "attitudes" or any terminology suggestive of norms.

We did not undertake dual data extraction, but worked with a second reviewer to check data extraction and identify and reconcile disagreements. We used a novel, tailored tool for quality assessment rather than an existing tool.

Implications

Findings from this review support a number of recommendations for practice, policy, and research (Table 2). We recommend that future research build on existing measures where evidence supports their reliability and validity among similar populations, and where measures distinguish between victimization/perpetration among girls and boys

Table 2. In	plications for Practice, Policy, and Research.
Practice	 Public health practitioners should draw on studies that use valid and reliable measures of social norms, where these are available in their setting, to identify the social norms contributing to DRV and to pinpoint areas for intervention
Policy	 When funding evaluations of interventions that aim to reduce DRV by modifying social norms, policymakers should ensure that these use reliable and valid measures to assess norms and that such measures are gender specific where norms exert gendered influence
Research	 Evaluators should use existing valid and reliable measures of DRV and gender norms, if these are available in their setting, to assess social norms as a mediator in DRV interventions aiming to modify social norms Researchers should adapt existing measures to be gender specific where norms exert gendered influence, and to focus on the DRV type(s) of interest, where no such measures exist for their setting. Where no such measures exist for similar settings, researchers should develop new measures informed by existing literature and participatory research with young people Researchers should develop and validate measures of gender norms governing the broad range of gendered behaviors underpinning DRV and measures of social norms influencing same-sex DRV.

DRV=dating and relationship violence.

and focus on the DRV type(s) of interest; or where they can be adapted to do so. New measures should be informed by existing literature and participatory research with young people to develop and refine measures and to select reference groups (Costenbader et al., 2019). Researchers should report on the development, piloting, refinement, reliability, and validity of such measures, which in addition to enhancing social norms measurement in DRV research would also contribute to learning on best practices for social norms measurement among adolescents. Future research should synthesize this learning with findings from other areas of norms research among adolescents to inform methodological approaches with this population.

Future research should inform the development of gender norms measures that predict DRV but are distinct from norms about gendered violence itself, including descriptive gender norms. New research is also needed to inform the development of measures of social norms influencing samesex DRV, considering the higher risk of DRV among sexualminority youth.

New measures should specify a bounded reference group (Ashbum et al., 2016), and where more than one reference group is pertinent, norms among each should be measured separately. Finally, future research should use valid and reliable measures to explore relationships between descriptive and injunctive DRV and gender norms and subsequent DRV outcomes, assess the impact of interventions on these norms and explore their role in reducing DRV.

Conclusions

Developing valid, reliable measures of social norms associated with DRV is possible, but measurement methods are currently inconsistent. Researchers should report on the development, reliability, and validity of such measures, which should be gender-specific where norms exert gendered influence, consider sexual-minority relationships, and assess gender norms beyond gendered violence.

Acknowledgments

The authors gratefully acknowledge Manika Garg, Ben Pelhan and Nicola Pocock for their contributions to and support for this research.

Declaration of Conflicting Interests

The author(s) declared no potential conflicts of interest with respect to the research, authorship, and/or publication of this article.

Funding

The author(s) disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: This article was supported with funding from the Passages Project. Passages is made possible by the generous support of the American people through the United States Agency for International Development (USAID) under the terms of Cooperative Agreement No. AID-OAA-A-15-00042. The contents are the responsibility of the authors and do not necessarily reflect the views of Georgetown University, USAID, or the United States Government.

ORCID iDs

Rebecca Meiksin D https://orcid.org/0000-0002-5096-8576 G.J. Melendez-Torres D https://orcid.org/0000-0002-9823-4790

Supplemental Material

Supplemental material for this article is available online.

References

- Advancing Learning and Innovation on Gender Norms (ALIGN). (n.d.). Align Platform. Retrieved February 22, 2022, from https://www.alignplatform.org/
- Aizpitarte, A., Alonso-Arbiol, I., & Van de Vijver, F. J. R. (2017). An explanatory model of dating violence risk factors in Spanish adolescents. *Journal of Research on Adolescence*, 27(4), 797–809. https://doi.org/10.1111/jora.12315
- Alexander-Scott, M., Bell, E., & Holden, J. (2016). Shifting social norms to tackle violence against women and girls (VAWG). VAWG Helpdesk.

- Ali, B., Swahn, M., & Hamburger, M. (2011). Attitudes affecting physical dating violence perpetration and victimization: Findings from adolescents in a high-risk urban community. *Violence* and Victims, 26(5), 669–683. https://doi.org/10.1891/0886-6708.26.5.669
- Antônio, T., Koller, S. H., & Hokoda, A. (2012). Peer influences on the dating aggression process among brazilian street youth: A brief report. *Journal of Interpersonal Violence*, 27(8), 1579–1592. https://doi.org/10.1177/0886260511425794
- Arriaga, X. B., & Foshee, V. A. (2004). Adolescent dating violence: Do adolescents follow in their friends', or their parents', footsteps? *Journal of Interpersonal Violence*, 19(2), 162–184. https://doi.org/10.1177/0886260503260247
- Ashburn, K., Costenbader, B., Igras, S., Pirzadeh, M., & Homan, R. (2016). Learning collaborative background reader: Advancing research and practice on normative change for adolescent sexual and reproductive health and well-being. Developed for the convening meeting, December 5-6, 2016. Institute for Reproductive Health at Georgetown University and FHI 360.
- Barter, C. (2006). Discourses of blame: Deconstructing (hetero) sexuality, peer sexual violence and residential children's homes. *Child & Family Social Work*, 11(4), 347–356. https:// doi.org/10.1111/j.1365-2206.2006.00425.x
- Barter, C., McCarry, M., Berridge, D., & Evans, K. (2009). Partner exploitation and violence in teenage intimate relationships. NSPCC.
- Barter, C., & Stanley, N. (2016). Inter-personal violence and abuse in adolescent intimate relationships: Mental health impact and implications for practice. *International Review of Psychiatry*, 28(5), 485–503. https://doi.org/10.1080/0954026 1.2016.1215295
- Breiding, M., Basile, K., Smith, S., Black, M., & Mahendra, P. (2015). Intimate partner violence surveillance: Uniform definitions and recommended data elements, version 2.0. National Center for Injury Prevention and Control, Center for Disease Control and Prevention.
- Capaldi, D. M., Knoble, N. B., Shortt, J. W., & Kim, H. K. (2012). A systematic review of risk factors for intimate partner violence. *Partner Abuse*, 3(2), 231–280. https://doi.org/10.1891/1946-6560.3.2.231
- Care Evaluations. (n.d.). Search evaluations. Retrieved July 25, 2020, from http://www.careevaluations.org/
- Castellví, P., Miranda-Mendizábal, A., Parès-Badell, O., Almenara, J., Alonso, I., Blasco, M. J., Cebrià, A., Gabilondo, A., Gili, M., Lagares, C., Piqueras, J. A., Roca, M., Rodríguez-Marín, J., Rodríguez-Jimenez, T., Soto-Sanz, V., & Alonso, J. (2017). Exposure to violence, a risk for suicide in youths and young adults. A meta-analysis of longitudinal studies. *Acta Psychiatrica Scandinavica*, 135(3), 195–211. https://doi. org/10.1111/acps.12679
- Cislaghi, B., & Heise, L. (2018). Four avenues of normative influence: A research agenda for health promotion in low and midincome countries. *Health Psychology*, 37(6), 562–573. https:// doi.org/10.1037/hea0000618
- Coker, A. L., Bush, H. M., Cook-Craig, P. G., DeGue, S. A., Clear, E. R., Brancato, C. J., Fisher, B. S., & Recktenwald, E. A. (2017). RCT testing bystander effectiveness to reduce violence. *American Journal of Preventive Medicine*, 52(5), 566–578. https://doi.org/10.1016/j.amepre.2017.01.020

- Cornelius, T. L., & Resseguie, N. (2007). Primary and secondary prevention programs for dating violence: A review of the literature. Aggression and Violent Behavior, 12(3), 364–375. https://doi.org/10.1016/j.avb.2006.09.006
- Costenbader, E., Cislaghi, B., Clark, C. J., Hinson, L., Lenzi, R., McCarraher, D. R., McLarnon-Silk, C., Pulerwitz, J., Shaw, B., & Stefanik, L. (2019). Social norms measurement: Catching up with programs and moving the field forward. *The Journal* of Adolescent Health: Official Publication of the Society for Adolescent Medicine, 64(4S), S4–S6. https://doi.org/10.1016/j. jadohealth.2019.01.001
- Costenbader, E., Lenzi, R., Hershow, R. B., Ashburn, K., & McCarraher, D. R. (2017). Measurement of social norms affecting modern contraceptive use: A literature review. *Studies in Family Planning*, 48(4), 377–389. https://doi.org/10.1111/ sifp.12040
- Dank, M., Lachman, P., Zweig, J. M., & Yahner, J. (2014). Dating violence experiences of lesbian, gay, bisexual, and transgender youth. *Journal of Youth and Adolescence*, 43(5), 846–857. https://doi.org/10.1007/s10964-013-9975-8
- De Koker, P., Mathews, C., Zuch, M., Bastien, S., & Mason-Jones, A. J. (2014). A systematic review of interventions for preventing adolescent intimate partner violence. *Journal of Adolescent Health*, 54(1), 3–13. https://doi.org/10.1016/j.jadohealth.2013.08.008
- De La Rue, L., Polanin, J. R., Espelage, D. L., & Pigott, T. D. (2014). School-based interventions to reduce dating and sexual violence: A systematic review. *Campbell Systematic Reviews*, 10(1), 1–110. https://doi.org/10.4073/csr.2014.7
- DeVellis, R. F. (2017). Scale development: Theory and applications (4th ed.). SAGE.
- Dietz, J. (2019). Dating and relationship violence in sexual and gender minority youth: A systematic review of the literature and cost-effectiveness analysis of a targeted school-based preventative intervention. London School of Hygiene & Tropical Medicine.
- Doherty, S., Oram, S., Siriwardhana, C., & Abas, M. (2016). Suitability of measurements used to assess mental health outcomes in men and women trafficked for sexual and labour exploitation: A systematic review. *The Lancet Psychiatry*, 3(5), 464–471. https://doi.org/10.1016/S2215-0366(16)30047-5
- Ehrensaft, M. K., Westfall, H. K., Niolon, P. H., Lopez, T., Kamboukos, D., Huang, K.-Y., & Brotman, L. M. (2018). Can a parenting intervention to prevent early conduct problems interrupt girls' risk for intimate partner violence 10 years later? *Prevention Science: The Official Journal of the Society for Prevention Research*, 19(4), 449–458. https://doi.org/10.1007/ s11121-017-0831-z
- EMERGE. (n.d.a). EMERGE Home: Identify, Extract, and Evaluate. Retrieved July 25, 2020, from http://emerge.ucsd.edu/
- EMERGE. (n.d.b). Quantitative Measurement of Gender Equality and Empowerment (EMERGE). Retrieved July 25, 2020, from http://geh.ucsd.edu/quantitative-measurement-of-genderequality-and-empowerment-emerge/
- Enosh, G. (2007). Cognition or involvement? Explaining sexualcoercion in high-school dating. *Sexual Abuse: A Journal* of Research and Treatment, 19(3), 311–329. https://doi. org/10.1177/107906320701900308

- Exner-Cortens, D., Eckenrode, J., Bunge, J., & Rothman, E. (2017). Revictimization after adolescent dating violence in a matched, national sample of youth. *Journal of Adolescent Health*, 60(2), 176–183. https://doi.org/10.1016/j.jadohealth.2016.09.015
- Exner-Cortens, D., Eckenrode, J., & Rothman, E. (2013). Longitudinal associations between teen dating violence victimization and adverse health outcomes. *Pediatrics*, 131(1), 71–78. https://doi.org/10.1542/peds.2012-1029
- Exner-Cortens, D., Gill, L., & Eckenrode, J. (2016a). Measurement of adolescent dating violence: A comprehensive review (Part 1, behaviors). Aggression and Violent Behavior, 27, 64–78. https://doi.org/10.1016/j.avb.2016.02.007
- Exner-Cortens, D., Gill, L., & Eckenrode, J. (2016b). Measurement of adolescent dating violence: A comprehensive review (Part 2, attitudes). Aggression and Violent Behavior, 27, 93–106. https://doi.org/10.1016/j.avb.2016.02.011
- Explore Our Resources. (n.d.). Institute for Reproductive Health. Retrieved July 25, 2020, from http://irh.org/resource-library/
- Fellmeth, G. L., Heffernan, C., Nurse, J., Habibula, S., & Sethi, D. (2013). Educational and skills-based interventions for preventing relationship and dating violence in adolescents and young adults. *The Cochrane Database of Systematic Reviews*, 6, CD004534. https://doi.org/10.1002/14651858.CD004534.pub3
- Find a Report. (n.d.). Save the Children International. Retrieved July 25, 2020, from https://www.savethechildren.net/researchreports/search
- Flisher, A. J., Myer, L., Mèrais, A., Lombard, C., & Reddy, P. (2007). Prevalence and correlates of partner violence among South African adolescents. *Journal of Child Psychology and Psychiatry*, 48(6), 619–627. https://doi.org/10.1111/j.1469-7610.2007.01711.x
- Foshee, V. A. (1996). Gender differences in adolescent dating abuse prevalence, types and injuries. *Health Education Research*, 11(3), 275–286. https://doi.org/10.1093/het/11.3.275-a
- Foshee, V. A., Bauman, K. E., Ennett, S. T., Suchindran, C., Benefield, T., & Linder, G. F. (2005). Assessing the effects of the dating violence prevention program "Safe Dates" using random coefficient regression modeling. *Prevention Science*, 6(3), 245–258. https://doi.org/10.1007/s11121-005-0007-0
- Foshee, V. A., Linder, F., MacDougall, J. E., & Bangdiwala, S. (2001). Gender differences in the longitudinal predictors of adolescent dating violence. *Preventive Medicine*, 32(2), 128–141. https://doi.org/10.1006/pmed.2000.0793
- Foshee, V. A., McNaughton Reyes, H. L., Ennett, S. T., Suchindran, C., Mathias, J. P., Karriker-Jaffe, K. J., Bauman, K. E., & Benefield, T. S. (2011). Risk and protective factors distinguishing profiles of adolescent peer and dating violence perpetration. *Journal of Adolescent Health*, 48(4), 344–350. https://doi. org/10.1016/j.jadohealth.2010.07.030
- Fulu, E., Kerr-Wilson, L., & Lang, J. (2014). What works to prevent violence against women and girls? Evidence review of interventions to prevent violence against women and girls. Medical Research Council.
- Gagné, M.-H., Lavoie, F., & Hébert, M. (2005). Victimization during childhood and revictimization in dating relationships in adolescent girls. *Child Abuse & Neglect*, 29(10), 1155–1172.
- Gender and Power Metrics. (n.d.). Population Council. Retrieved July 25, 2020, from https://gendermetrics.popcouncil.org/
- Gender Violence and Health Centre (GVHC). (n.d.). London School of Hygiene & Tropical Medicine. Retrieved July 25,

2020, from https://www.lshtm.ac.uk/research/centres-projectsgroups/gender-violence-health-centre

- Global Early Adolescent Study. (n.d.). Global Early Adolescent Study. Retrieved July 25, 2020, from https://www.geastudy.org
- Gonzalez-Mendez, R., Aguilera, L., & Ramírez-Santana, G. (2019). Weighing risk factors for adolescent victimization in the context of romantic relationship initiation. *Journal of Interpersonal Violence*, 36(15–16), NP8395–NP8413. https:// doi.org/10.1177/0886260519843284
- Gupta, G. R. (2000, July 12). Gender, sexuality, and HIV/AIDS: The what, the why, and the how. XIIIth International AIDS Conference, Durban, South Africa.
- Hébert, M., Moreau, C., Blais, M., Oussaïd, E., & Lavoie, F. (2019). A three-step gendered latent class analysis on dating victimization profiles. *Psychology of Violence*, 9(5), 504–516. https:// doi.org/10.1037/vio0000225
- Helland, T. A. (1998). The role of the peer group on individual use and acceptance of physical aggression in adolescent dating relationships. *Dissertation Abstracts International: Section B: The Sciences and Engineering*, 58(8-B), 4450.
- Hennegan, J., Brooks, D. J., Schwab, K. J., & Melendez-Torres, G. J. (2020). Measurement in the study of menstrual health and hygiene: A systematic review and audit. *PLoS One*, 15(6), e0232935. https://doi.org/10.1371/journal. pone.0232935
- Herrenkohl, T. I., & Jung, H. (2016). Effects of child abuse, adolescent violence, peer approval and pro-violence attitudes on intimate partner violence in adulthood. *Criminal Behaviour* and Mental Health: CBMH, 26(4), 304–314. https://doi. org/10.1002/cbm.2014
- History Cooperative. (n.d.). The history of social media: Social networking evolution!. Retrieved June 3, 2019, from https://historycooperative.org/the-history-of-social-media/
- Hopper, L. (2011). Contributions of individual and friend attitudes to dating violence experience in adolescents. *Dissertation Abstracts International: Section B: The Sciences and Engineering*, 73(4-B), 2563.
- Hunt, K. E., Robinson, L. E., Valido, A., Espelage, D. L., & Hong, J. S. (2022). Teen dating violence victimization: Associations among peer justification, attitudes toward gender inequality, sexual activity, and peer victimization. *Journal* of Interpersonal Violence, 37(9–10), 5914–5936. https://doi. org/10.1177/08862605221085015
- Our Work. (n.d.). Girl Effect. Retrieved July 25, 2020, from https:// www.girleffect.org/elementor_libraryour-work-4/
- Kågesten, A., Gibbs, S., Blum, R. W., Moreau, C., Chandra-Mouli, V., Herbert, A., & Amin, A. (2016). Understanding factors that shape gender attitudes in early adolescence globally: A mixedmethods systematic review. *PLoS One*, *11*(6), e0157805. https://doi.org/10.1371/journal.pone.0157805
- Kernsmith, P. D., & Tolman, R. M. (2011). Attitudinal correlates of girls' use of violence in teen dating relationships. *Violence Against Women*, 17(4), 500–516. https://doi. org/10.1177/1077801211404312
- Kinsfogel, K. M., & Grych, J. H. (2004). Interparental conflict and adolescent dating relationships: Integrating cognitive, emotional, and peer influences. *Journal of Family Psychology*, 18(3), 505–515. https://doi.org/10.1037/0893-3200.18.3.505
- Learning Collaborative to Advance Normative Change—IRH. (n.d.). Institute for Reproductive Health. Retrieved July 25,

2020, from http://irh.org/projects/learning-collaborative-toadvance-normative-change/

- Leen, E., Sorbring, E., Mawer, M., Holdsworth, E., Helsing, B., & Bowen, E. (2013). Prevalence, dynamic risk factors and the efficacy of primary interventions for adolescent dating violence: An international review. Aggression and Violent Behavior, 18(1), 159–174. https://doi.org/10.1016/j.avb.2012.11.015
- Lewis, C. C., Mettert, K. D., Dorsey, C. N., Martinez, R. G., Weiner, B. J., Nolen, E., Stanick, C., Halko, H., & Powell, B. J. (2018). An updated protocol for a systematic review of implementation-related measures. *Systematic Reviews*, 7(1), 66. https://doi.org/10.1186/s13643-018-0728-3
- Lewis, C. C., Stanick, C. F., Martinez, R. G., Weiner, B. J., Kim, M., Barwick, M., & Comtois, K. A. (2015). The society for implementation research collaboration instrument review project: A methodology to promote rigorous evaluation. *Implementation Science*, 10(1), 2. https://doi.org/10.1186/s13012-014-0193-x
- Lokot, M., Bhatia, A., Kenny, L., & Cislaghi, B. (2020). Corporal punishment, discipline and social norms: A systematic review in low- and middle-income countries. *Aggression* and Violent Behavior, 55, 101507. https://doi.org/10.1016/j. avb.2020.101507
- Lundgren, R., & Amin, A. (2015). Addressing intimate partner violence and sexual violence among adolescents: Emerging evidence of effectiveness. *Journal of Adolescent Health*, 56(1), S42–S50. https://doi.org/10.1016/j.jadohealth.2014.08.012
- Luo, F., Stone, D. M., & Tharp, A. T. (2014). Physical dating violence victimization among sexual minority youth. *American Journal of Public Health*, 104(10), e66–e73. https://doi. org/10.2105/AJPH.2014.302051
- Manchikanti Gómez, A. (2011). Testing the cycle of violence hypothesis: Child abuse and adolescent dating violence as predictors of intimate partner violence in young adulthood. *Youth & Society*, 43(1), 171–192. https://doi.org/10.1177/0044118X09358313
- Marston, C., & King, E. (2006). Factors that shape young people's sexual behaviour: A systematic review. *Lancet*, 368(9547), 1581–1586. https://doi.org/10.1016/S0140-6736(06)69662-1
- Martin-Storey, A., & Fromme, K. (2021). Mediating factors explaining the association between sexual minority status and dating violence. *Journal of Interpersonal Violence*, 36(1–2), 132–159. https://doi.org/10.1177/0886260517726971
- McGeeney, E., & Hanson, E. (2017). Digital romance: A research project exploring young people's use of technology in their romantic relationships and love lives. National Crime Agency and Brook. https://www.thinkuknow.co.uk/professionals/guidance/digital-romance/
- McGowan, J., Sampson, M., Salzwedel, D. M., Cogo, E., Foerster, V., & Lefebvre, C. (2016). PRESS peer review of electronic search strategies: 2015 guideline statement. *Journal of Clinical Epidemiology*, 75, 40–46. https://doi.org/10.1016/j. jclinepi.2016.01.021
- Meiksin, R. (2020). Systematic review of social norms measures relating to dating and relationship violence. Open Science Framework. https://osf.io/5472t/
- Miller, E., Das, M., Tancredi, D. J., McCauley, H. L., Virata, M. C. D., Nettiksimmons, J., O'Connor, B., Ghosh, S., & Verma, R. (2014). Evaluation of a gender-based violence prevention program for student athletes in Mumbai, India. *Journal of Interpersonal Violence*, 29(4), 758–778. https://doi.org/10.1177/0886260513505205

- Mokdad, A. H., Forouzanfar, M. H., Daoud, F., Mokdad, A. A., El Bcheraoui, C., Moradi-Lakeh, M., Kyu, H. H., Barber, R. M., Wagner, J., Cercy, K., Kravitz, H., Coggeshall, M., Chew, A., O'Rourke, K. F., Steiner, C., Tuffaha, M., Charara, R., Al-Ghamdi, E. A., Adi, Y., . . . Murray, C. J. L. (2016). Global burden of diseases, injuries, and risk factors for young people's health during 1990–2013: A systematic analysis for the Global Burden of Disease Study 2013. *The Lancet*, 387(10036), 2383– 2401. https://doi.org/10.1016/S0140-6736(16)00648-6
- Moreau, C. (2018, February 1). Development and validation of cross-cultural gender norms scales for early adolescents [Webinar]. http://www.geastudy.org/webinars/
- Moreau, C., Li, M., Ahmed, S., Zuo, X., & Cislaghi, B. (2021). Assessing the spectrum of gender norms perceptions in early adolescence: A cross-cultural analysis of the global early adolescent study. *Journal of Adolescent Health*, 69(1), S16–S22. https://doi.org/10.1016/j.jadohealth.2021.03.010
- Nardi-Rodríguez, A., Pastor-Mira, M. Á., López-Roig, S., Pamies-Aubalat, L., Martínez-Zaragoza, F., & Ferrer-Pérez, V. A. (2022). Predicting abusive behaviours in Spanish adolescents' relationships: Insights from the reasoned action approach. *International Journal of Environmental Research and Public Health*, 19(3), 1441. https://doi.org/10.3390/ijerph19031441
- Offenhauer, P., & Buchalter, A. (2011). Teen dating violence: A literature review and annotated bibliography. https://www.ncjrs. gov/pdffiles1/nij/grants/235368.pdf
- Peskin, M. F., Markham, C. M., Shegog, R., Temple, J. R., Baumler, E. R., Addy, R. C., Hernandez, B., Cuccaro, P., Gabay, E. K., Thiel, M., & Emery, S. T. (2017). Prevalence and correlates of the perpetration of cyber dating abuse among early adolescents. *Journal of Youth and Adolescence*, 46(2), 358–375. https://doi. org/10.1007/s10964-016-0568-1
- Plourde, C., Shore, N., Herrick, P., Morrill, A., Cattabriga, G., Bottino, L., Orme, E., & Stromgren, C. (2016). You the man: Theater as bystander education in dating violence. *Arts & Health*, 8(3), 229–247. https://doi.org/10.1080/17533015.201 5.1091017
- Pocock, N. S., Chan, C. W., & Zimmerman, C. (2021). Suitability of measurement tools for assessing the prevalence of child domestic work: A rapid systematic review. *International Journal of Environmental Research and Public Health*, 18(5), 2357. https://doi.org/10.3390/ijerph18052357
- Pöllänen, K., de Vries, H., Mathews, C., Schneider, F., & de Vries, P. J. (2021). Beliefs about sexual intimate partner violence perpetration among adolescents in South Africa. *Journal of Interpersonal Violence*, 36(3–4), NP2056–NP2078. https://doi. org/10.1177/0886260518756114
- Price, E. L. (2002). Risk factors for boys' psychologically abusive behaviour in dating relationships. *Dissertation Abstracts International: Section B: The Sciences and Engineering*, 62(10-B), 4800.
- Prinsen, C. A. C., Mokkink, L. B., Bouter, L. M., Alonso, J., Patrick, D. L., de Vet, H. C. W., & Terwee, C. B. (2018). COSMIN guideline for systematic reviews of patient-reported outcome measures. *Quality of Life Research*, 27(5), 1147–1157. https:// doi.org/10.1007/s11136-018-1798-3
- Publications. (n.d.). ODI. Retrieved July 25, 2020, from https:// www.odi.org/publications
- Reed, E., Silverman, J. G., Raj, A., Decker, M. R., & Miller, E. (2011). Male perpetration of teen dating violence: Associations

with neighborhood violence involvement, gender attitudes, and perceived peer and neighborhood norms. *Journal of Urban Health*, 88(2), 226–239. https://doi.org/10.1007/s11524-011-9545-x

- Reeves, P. M., & Orpinas, P. (2012). Dating norms and dating violence among ninth graders in Northeast Georgia: Reports from student surveys and focus groups. *Journal* of Interpersonal Violence, 27(9), 1677–1698. https://doi. org/10.1177/0886260511430386
- Removing duplicates from an EndNote library. (December 7, 2018). Library & Archives Service Blog. http://blogs.lshtm. ac.uk/library/2018/12/07/removing-duplicates-from-an-endnote-library/
- Resources. (n.d.). Promundo. Retrieved July 25, 2020, from https:// promundoglobal.org/resources/
- Reuter, T. R., & Whitton, S. W. (2018). Adolescent dating violence among lesbian, gay, bisexual, transgender, and questioning youth. In D. A. Wolfe, & J. R. Temple (Eds.), Adolescent dating violence (pp. 215–231). Elsevier.
- Reyes, H. L. M., Foshee, V. A., Niolon, P. H., Reidy, D. E., & Hall, J. E. (2016). Gender role attitudes and male adolescent dating violence perpetration: Normative beliefs as moderators. *Journal of Youth and Adolescence*, 45(2), 350–360. https://doi. org/10.1007/s10964-015-0278-0
- Roberts, T. A., Klein, J. D., & Fisher, S. (2003). Longitudinal effect of intimate partner abuse on high-risk behavior among adolescents. Archives of Pediatrics & Adolescent Medicine, 157(9), 875. https://doi.org/10.1001/archpedi.157.9.875
- Rogers, M., Rumley, T., & Lovatt, G. (2019). The change up project: Using social norming theory with young people to address domestic abuse and promote healthy relationships. *Journal* of Family Violence, 34(6), 507–519. https://doi.org/10.1007/ s10896-018-0026-9
- Salazar, L. F., Swartout, K. M., Swahn, M. H., Bellis, A. L., Carney, J., Vagi, K. J., & Lokey, C. (2018). Precollege sexual violence perpetration and associated risk and protective factors among male college freshmen in Georgia. *Journal of Adolescent Health*, 62(3), S51–S57. https://doi.org/10.1016/j. jadohealth.2017.09.028
- Sexual Violence Research Initiative. (n.d.). Sexual Violence Research Initiative. Retrieved July 25, 2020, from https:// www.svri.org/
- Shakya, H. B., Cislaghi, B., Fleming, P., Levtov, R. G., Boyce, S. C., Raj, A., & Silverman, J. G. (2022). Associations of attitudes and social norms with experiences of intimate partner violence among married adolescents and their husbands in rural Niger: A dyadic cross-sectional study. *BMC Women's Health*, 22(1), 180. https://doi.org/10.1186/s12905-022-01724-y
- Shamseer, L., Moher, D., Clarke, M., Ghersi, D., Liberati, A., Petticrew, M., Shekelle, P., & Stewart, L. A. & the PRISMA-P Group. (2015). Preferred reporting items for systematic review and meta-analysis protocols (PRISMA-P) 2015: Elaboration and explanation. *BMJ*, 350(jan02 1), g7647. https://doi. org/10.1136/bmj.g7647
- Shamu, S., Gevers, A., Mahlangu, B. P., Jama Shai, P. N., Chirwa, E. D., & Jewkes, R. K. (2016). Prevalence and risk factors for intimate partner violence among Grade 8 learners in urban South Africa: Baseline analysis from the Skhokho Supporting

Success cluster randomised controlled trial. International Health, 8(1), 18–26. https://doi.org/10.1093/inthealth/ihv068

- Shorey, R. C., Wymbs, B., Torres, L., Cohen, J. R., Fite, P. J., & Temple, J. R. (2018). Does change in perceptions of peer teen dating violence predict change in teen dating violence perpetration over time? Aggressive Behavior, 44(2), 156–164. https:// doi.org/10.1002/ab.21739
- Solomon, S. D. (2015). "Run Like a Girl? That's So Gay!" Exploring Homophobic and Sexist Language among Grade 7 and 8 Students in the Toronto District School Board [Factor Inwentash Faculty of Social Work, University of Toronto]. https://tspace.library.utoronto.ca/handle/1807/71345
- Stanley, N., Ellis, J., Farrelly, N., Hollinghurst, S., & Downe, S. (2015). Preventing domestic abuse for children and young people: A review of school-based interventions. *Children and Youth Services Review*, 59, 120–131. https://doi.org/10.1016/j. childyouth.2015.10.018
- Taylor, B., Stein, N. D., Woods, D., & Mumford, E. (2011). Shifting boundaries: Final report on an experimental evaluation of a youth dating violence prevention program in New York City Middle Schools (No. 236175). U.S. Department of Justice.
- Terwee, C. B., Bot, S. D. M., de Boer, M. R., van der Windt, D. A. W. M., Knol, D. L., Dekker, J., Bouter, L. M., & de Vet, H. C. W. (2007). Quality criteria were proposed for measurement properties of health status questionnaires. *Journal of Clinical Epidemiology*, 60(1), 34–42. https://doi.org/10.1016/j. jclinepi.2006.03.012
- The EndNote Team. (2013). EndNote (EndNote X9). Clarivate.
- The Social Norms Learning Collaborative. (2021). Social norms atlas: Understanding global social norms and related concepts. Institute for Reproductive Health, Georgetown University.
- USAID. (n.d.). Are you looking for an article or resource from POPLINE? Knowledge Success.
- Vagi, K. J., Rothman, E. F., Latzman, N. E., Tharp, A. T., Hall, D. M., & Breiding, M. J. (2013). Beyond correlates: A review of risk and protective factors for adolescent dating violence perpetration. *Journal of Youth and Adolescence*, 42(4), 633–649. https://doi.org/10.1007/s10964-013-9907-7
- Van Ouytsel, J., Ponnet, K., & Walrave, M. (2020). Cyber dating abuse: Investigating digital monitoring behaviors among adolescents from a social learning perspective. *Journal of Interpersonal Violence*, 35(23–24), 5157–5178. https://doi. org/10.1177/0886260517719538
- Wesche, R., & Dickson-Gomez, J. (2019). Gender attitudes, sexual risk, intimate partner violence, and coercive sex among adolescent gang members. *Journal of Adolescent Health*, 64(5), 648–656. https://doi.org/10.1016/j.jadohealth.2018.10.292
- Whitaker, D. J., Morrison, S., Lindquist, C., Hawkins, S. R., O'Neil, J. A., Nesius, A. M., Mathew, A., & Reese, L. (2006). A critical review of interventions for the primary prevention of perpetration of partner violence. Aggression and Violent Behavior, 11(2), 151–166. https://doi.org/10.1016/j. avb.2005.07.007
- Whitley, B. E., Jr. (2001). Gender-role variables and attitudes toward homosexuality. Sex Roles, 45(11/12), 691–721. https:// doi.org/10.1023/A:1015640318045
- Wincentak, K., Connolly, J., & Card, N. (2017). Teen dating violence: A meta-analytic review of prevalence rates. *Psychology* of Violence, 7(2), 224–241. https://doi.org/10.1037/a0040194

- Wolfe, D. A., & Jaffe, P. G. (1999). Emerging strategies in the prevention of domestic violence. *The Future of Children*, 9(3), 133–144.
- Wood, M., Barter, C., & Berridge, D. (2011). 'Standing on my own two feet': Disadvantaged Teenagers, Intimate Partner Violence and Coercive Control. NSPCC.
- World Health Organization. (2021). Violence against women prevalence estimates, 2018: Global, regional and national prevalence estimates for intimate partner violence against women and global and regional prevalence estimates for non-partner sexual violence against women. World Health Organization. https://apps.who.int/iris/handle/10665/341337
- Young, H., Turney, C., White, J., Bonell, C., Lewis, R., & Fletcher, A. (2017). Dating and relationship violence among 16–19 year olds in England and Wales: A cross-sectional study of victimization. *Journal of Public Health (Oxford, England)*, 40(4), 738–746. https://doi.org/10.1093/pubmed/fdx139

Author Biographies

Rebecca Meiksin, MPH, is a Research Fellow in Social Science at the London School of Hygiene & Tropical Medicine. Her research focuses on sexual and reproductive health and on the role of social and gender norms in the prevention of violence in young people's dating and relationships.

Professor Chris Bonell, PhD, is a Professor of Public Health Sociology at the London School of Hygiene & Tropical Medicine. His research on adolescent health focuses on how schools and schoolbased interventions can benefit or harm young people's health. His research on sexual health examines the broader social determinants of adolescent sexual health, and interventions to address these.

Amiya Bhatia, PhD, is an Assistant Professor in Social Epidemiology and Child Protection at the London School of Hygiene & Tropical Medicine. Her research examines how child health and child protection outcomes are unevenly and unfairly distributed in low- and middle-income countries and the data used to measure these outcomes.

Professor G.J. Melendez-Torres, DPhil, MPH, RN, FAAN, is a Professor of Clinical and Social Epidemiology in the Faculty of Health and Life Sciences at the University of Exeter, where he leads the Peninsula Technology Assessment Group. His research focuses on health technology assessment, intimate partner violence, and child and adolescent health.

Nambusi Kyegombe, PhD, is the head of social sciences at the MRC/UVRI & LSHTM Uganda Research Unit. Her work on adolescent health focuses on low- and middle-income country settings. This includes formative research on the context in which health is located, methodological research to improve measurement, and research on the conduct of high quality, ethical research, particularly among marginalized populations.

Anjalee Kohli, PhD, is an Assistant Professor-Adjunct at Georgetown University and an independent consultant. Through participatory and mixed methods research and learning partnerships, Dr. Kohli advances the evidence and practice to prevent violence against women and children, to improve gender equity and sexual and reproductive health.

Chapter 5. Paper 2: Cognitive interviews informing the development of social norms measures 5.1. Introduction to Paper 2

As described in the introduction to Paper 1, we identified in the course of the Project Respect pilot RCT a need for measures that could be used to assess social norms as a theorised mediator of the intervention's effects. Finding no established measures of social norms concerning DRV and gender, we developed descriptive norms items based on an existing measure of descriptive DRV norms that had been used with university students and proposed for use with adolescents.²¹⁶ We developed injunctive DRV and gender norms items based on measures of personal attitudes that had been used with adolescents.^{77,217} A planned phase of cognitive testing to refine the trial's measures of DRV victimisation and perpetration provided the opportunity to also incorporate testing of these three types of social norms items. Paper 2 presents the development and cognitive testing of these items and how this work informed the three social norms measures ultimately piloted in Project Respect student surveys. Further, I draw out features of tested items that helped and hindered participants' understanding of items' intended meanings and their ability to respond to these items. Based on these findings, I offer broader recommendations for social norms measurement among adolescents and suggest areas for future research.

The published supplemental appendix to which Paper 2 refers is provided in Appendix 16 of this thesis.

5.1.1. Notes on Table 1 of Paper 2

Table 1 (pp. 6-7) of Paper 2 shows the progression of measures used in cognitive testing from the measures under consideration as the basis for Project Respect survey measures immediately preceding cognitive testing (first column) to the measures tested in cognitive testing (second column) and the final measures which were refined on the basis of cognitive testing and ultimately piloted in Project Respect (third column). As noted in footnotes (a) and (b) to Table 1, the "attitudes towards DRV" and "descriptive DRV norms" measures under consideration as the basis for Project Respect survey measures were adapted from their original sources. As Paper 2 does not specify these adaptations, I describe them here. From the "attitudes towards DRV" measure, the ninth item ("If someone hits their boyfriend or girlfriend, their boyfriend or girlfriend should break up with them") was adapted from the original used by Foshee et al.⁷⁷ ("If I hit a dating partner, he/she would break up with me") to more closely align with the construct of a personal attitude. From the "descriptive DRV norms" measure, the second item ("How any of your friends have used physical force, such as hitting to solve fights with their girlfriends or boyfriends?") was adapted from the original used by Cook-Craig et al.²¹⁶ ("How many of your friends have used physical force, such as hitting to solve fights with their boyfriends or girlfriends have used physical force, such as hitting to solve fights with their boyfriends or girlfriends have used physical force, such as hitting to solve fights with their boyfriends or girlfriends have used physical force, such as hitting to solve fights with their boyfriends or girlfriends have used physical force, such as hitting to solve fights with their boyfriends or girlfriends have used physical force, such as hitting to solve fights with their boyfriends or girlfriends have used physical force, such as hitting to solve fights with their boyfriends or girlfriends.

Please also note two corrections to the published version of this paper:

- In the last column of Table 1 (p. 7), the last line should read "First tested item removed" rather than "Second tested item removed".
- 2) Footnote (c) to Table 1 should read "Items from Sotiriou et al." rather than "Items adapted from Sotiriou et al." because the "attitudes towards gender roles and stereotypes" measure appears here as it does in the referenced source, Sotiriou et al.²¹⁷

5.1.2. Previous work reporting on this researchWith my colleague, Dr. Ruth Ponsford, the second author of Paper 2, I had previously conducted an analysis of data from the cognitive testing of the Project Respect social norms items alongside data from the cognitive testing of social norms items relating to sexual behaviour that Dr. Ponsford had undertaken as part of the Positive Choices pilot trial.²³⁵ For Paper 3 and this thesis, I draw only on Project Respect data. However, I presented preliminary findings from the previous analysis drawing on both studies at the LINEA Biennial Meeting (2020);²³⁶ in a webinar for the LINEA Project (2020);²³⁴ at the Lancet's UK Public Health Science conference (2018);²³⁷ and in the associated abstract published in a special issue of *The Lancet* (2018),⁵ presented in Appendix 17.

5.2. Paper 2: Assessing survey items on social norms relating to dating and relationship violence and to gender: cognitive interviews with young people in England



London School of Hygiene & Tropical Medicine Keppel Street, London WC1E 7HT

T: +44 (0)20 7299 4646 F: +44 (0)20 7299 4656 www.lshtm.ac.uk

RESEARCH PAPER COVER SHEET

Please note that a cover sheet must be completed for each research paper included within a thesis.

SECTION A - Student Details

Student ID Number	452277	Title	Ms.
First Name(s)	Rebecca		
Surname/Family Name	Meiksin		
Thesis Title	Developing and refining social norr evaluation of a school-based interve relationship violence among adoles	ns measures ention to rec cents in Eng	s for the luce dating and gland
Primary Supervisor	Professor Chris Bonell		

If the Research Paper has previously been published please complete Section B, if not please move to Section C.

SECTION B - Paper already published

Where was the work published?	Journal of Interp	ersonal violence	
When was the work published?	October 2023		
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion	N/A		
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C - Prepared for publication, but not yet published

Where is the work intended to be published?	
Please list the paper's authors in the intended authorship order:	

Improving health worldwide

www.lshtm.ac.uk

Choose an item.

SECTION D - Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	I conceptualised the cognitive interviewing of social norms items, developed cognitive interview guides, conducted interviews and led analysis, writing and submission. Nambusi Kyegombe and Chris Bonell provided feedback on interview guides. Ruth Ponsford helped to develop the analytical framework. All authors contributed to interpretation of findings, provided feedback on the manuscript and approved the final mansucript prior to publication. Chris Bonell designed the overall study.
---	---

SECTION E

Student Signature	Rebecca Meiksin
Date	15/03/2024

Supervisor Signature	Chris Bonell	
Date	15/03/2024	

Assessing Survey Items on Social Norms Relating to Dating and Relationship Violence and to Gender: Cognitive Interviews with Young People in England Journal of Interpersonal Violence I-22 © The Author(s) 2023 O O O The Author (s) 2023 Article reuse guidelines:

sagepub.com/journals-permissions DOI: 10.1177/08862605231204561 journals.sagepub.com/home/jiv



Rebecca Meiksin¹, Ruth Ponsford¹, Nambusi Kyegombe¹, Anjalee Kohli², and Chris Bonell¹

Abstract

Widespread among adolescents in England, dating and relationship violence (DRV) is associated with subsequent injuries and serious mental health problems. While DRV prevention interventions often aim to shift harmful social norms, no established measures exist to assess relevant norms and their role in mediating DRV outcomes. We conducted cognitive interviews exploring the understandability and answerability of candidate measures of social norms relating to DRV and gender roles, informing measure refinement. In all, 11 participants aged 13 to 15 from one school in England participated. Cognitive interviews tested two items assessing descriptive norms (beliefs about what behaviors are typical), three assessing injunctive norms (beliefs about what is socially acceptable), and (for comparison) one assessing personal attitudes. Findings were summarized by drawing on

Corresponding Author:

Rebecca Meiksin, Department of Public Health, Environments and Society, London School of Hygiene & Tropical Medicine, 15-17 Tavistock Place, London WC1H 9SH, UK. Email: rebecca.meiksin@lshtm.ac.uk

¹London School of Hygiene & Tropical Medicine, London, UK ²Georgetown University, Washington, DC, USA

interview notes. Summaries and interview notes were subjected to thematic analysis. For some participants, injunctive norms items required further explanation to clarify that items asked about others' views, not their own. Lack of certainty about, and perceived heterogeneity of, behaviors and views among a broad reference group detracted from answerability. Participants were better able to answer items for which they could draw on concrete experiences of observing or discussing relevant behaviors or social sanctions. Data suggest that a narrowed reference group could improve answerability for items assessing salient norms. Findings informed refinements to social norms measures. It is possible to develop social norms measures that are understandable and answerable for adolescents in England. Measures should assess norms that are salient and publicly manifest among a cohesive and influential reference group.

Keywords

dating and relationship violence, adolescents, social norms, measurement, qualitative

Introduction

Background

"Dating and relationship violence" (DRV) among young people refers to physical, sexual, or psychological abuse by a current or former intimate partner (Barter & Stanley, 2016; Young et al., 2017). DRV is widespread in England, with 49.1% of young people with a mean age of 13.4 years (SD=0.6) reporting psychological victimization and 39.5% reporting physical victimization (Meiksin et al., 2020). Among those aged 14 to 17 years in England who have ever been in a relationship, 41% of girls and 14% of boys report experiencing sexual DRV, assessed as having been pressured or physically forced into intimate touching or sexual intercourse (Barter et al., 2014). In addition to causing injuries (Foshee et al., 1996), DRV victimization is associated with subsequent antisocial behavior, substance misuse (Exner-Cortens et al., 2013; Roberts et al., 2003), and mental health problems (Castellví et al., 2017; Exner-Cortens et al., 2013; Roberts et al., 2003), including suicidal ideation (Exner-Cortens et al., 2013) and suicide attempts (Castellví et al., 2017).

As social affiliation shifts from adults to peers in adolescence (Spear, 2000), young people are particularly sensitive to peer influence (Bonell et al., 2019). Social norms theory suggests that a person's behaviors are influenced

by beliefs about what behaviors are typical ("descriptive norms") and appropriate ("injunctive norms") among a reference group of others whose views are important to them (Alexander-Scott et al., 2016; Cislaghi & Heise, 2018). According to this conceptualization of social norms, which has been particularly influential in gender-based violence (Alexander-Scott et al., 2016) and adolescent sexual and reproductive health (Costenbader et al., 2019) research, these norms are sustained by anticipation of social rewards (for complying with them) and social punishment (for violating them) enacted by the reference group (Alexander-Scott et al., 2016; Cislaghi & Heise, 2018).

Empirical studies demonstrate the role of peer influence in DRV victimization and perpetration, finding that inequitable gender norms (Barter et al., 2009; Shakya et al., 2022; Wesche & Dickson-Gomez, 2019) and social norms supportive of DRV (Foshee et al., 2001; Gage, 2016; Salazar et al., 2018; Vagi et al., 2013) contribute to DRV risk, even when controlling for personal attitudes toward DRV (Foshee et al., 2001; Gage, 2016; Shakya et al., 2022). Interventions to reduce DRV often incorporate strategies to influence the peer social norms that contribute to sustaining this type of abuse (Stanley et al., 2015) but have not assessed social norms as a mediator of intervention effects (Meiksin et al., 2023).

Social Norms Measurement in DRV Research

A recent global systematic review reported on the use and quality of measures assessing social norms about DRV and gender, where measures had been tested for their association with DRV outcomes (Meiksin et al., 2023). None of the 40 descriptive and injunctive measures identified by the review were used in more than one study, and the review identified no evaluations of DRV interventions that explored whether changes in social norms mediated intervention impact (Meiksin et al., 2023). Fewer than one-quarter of included measures had been developed using input from young people (Meiksin et al., 2023) despite evidence suggesting that young people might struggle to distinguish between their own and their friends' views when responding to survey items (Moreau, 2018; Moreau et al., 2021).

In preparation for the evaluation of a new intervention in England that aimed, in part, to change social norms to reduce DRV (Meiksin 2020), we sought to test candidate social norms measures via cognitive interviews. Cognitive interviewing is a qualitative method for pretesting survey measures by exploring whether survey items function as intended and the cognitive processes participants use to answer these items (Willis & Artino, 2013). The approach allows researchers to identify any problems (Streiner & Norman, 2008) and refine items before administering surveys. The recommended approach for adolescents uses a combination of the think-aloud method and verbal probing (de Leeuw et al., 2002). The former is more open ended, asking participants to narrate their thoughts as they answer survey items (Collins, 2003; Willis, 1999). The latter involves asking specific questions about participants' experience responding to tested items, allowing the interviewer to explore aspects they suspect might be a source of response error (Collins, 2003; Willis, 1999).

In the present study, we conducted cognitive interviews with adolescents in England to assess the understandability and answerability of candidate measures of social norms relating to DRV and gender and to refine these survey measures based on our findings.

Methods

Study Overview

We conducted cognitive interviews to refine measures used in student surveys administered for Project Respect, a pilot cluster randomized controlled trial of a school-based intervention to reduce DRV in England (Meiksin 2020). Cognitive interviews tested selected survey items from measures of descriptive and injunctive DRV norms, and injunctive gender norms. They also tested survey instructions explaining safeguarding procedures, items on attitudes toward gender roles and stereotypes, items on relationship history, and two DRV measures. Findings from the testing of social norms and attitudes items are the focus of this paper. Ethical approval for this research was granted by the London School of Hygiene and Tropical Medicine Ethics Committee (11986).

Recruitment and Informed Consent

Drawing on our existing networks, we recruited one London state secondary school to take part. We asked school staff to select students of diverse academic ability across years 8, 9, and 10 (aged 13–15 years), based on their overall knowledge of the students, including at least two girls and two boys from each year-group. Students deemed by school staff to be unable to give informed consent due to severe cognitive limitations were not eligible to take part. Due to the sensitive nature of tested items, we recommended that students with known experience of DRV not be selected.

Participants' parents/carers received information describing the study and could opt their child out of taking part. Before beginning the interview, the researcher reviewed the written informed consent form with participants and explained that responses would be kept confidential except in the case of safeguarding concerns, which would include the following: reports of sexual activity before age 13, ongoing risk of serious harm, or disclosures for which the participant asked the researcher to breach confidentiality. A safeguarding concern arose for one participant, which was reported to the school's safe-guarding officer per our policy. Participants had the opportunity to ask questions before providing consent.

Interviews

Cognitive interviews took place at the participating school during the school day. Lasting around 40 min, interviews tested instructions, survey items, and response options for measures of descriptive and injunctive DRV norms, injunctive gender norms, and (for comparison) attitudes toward gender roles and stereotypes (Table 1). We did not test descriptive gender norms items because we found no appropriate measure in the DRV literature and developing a new measure would require formative research outside the scope of this study.

After each participant self-completed a brief demographic questionnaire, the interviewer explained that the participant would be asked to "think aloud," describing their thought process as they responded to each tested item. To practice carrying out this process, participants completed an exercise adapted from Willis (1999) which instructed them to "try to imagine your home, and think about how many windows there are in it. As you count up the windows, tell me what you are seeing and thinking about" (Willis, 1999, p. 4). Interviews then proceeded using a combination of the think-aloud and verbal probing approaches (de Leeuw et al., 2002). Participants were asked to think aloud as they answered tested items, which were displayed on show cards as they would appear on a survey. Verbal probes explored: alternative reference groups for norms items (i.e., the participant's friends; or their friends in the school); how easy/difficult items were to answer; understanding of terminology; alternative terminology (i.e., how the participant would phrase the question to their friends); and experiences of answering attitudes versus social norms items. The interviewer (RM) used a laptop to type detailed notes on participants' responses during interviews (Willis, 1999).

Measures

The demographic questionnaire asked for participants' age, year-group, ethnic group (White British; any other White background; Asian or Asian British; Black, African, Caribbean, or Black British; Mixed/multiple ethnic

Table I. Original Measures, Tested Items, and Final Measures.

Original measure (tested items in **bold**, retained items <u>underlined</u>) Attitudes toward DRV^a

- It is OK for a boy to hit his girlfriend if she did something to make him mad.
- It is OK for a boy to hit his girlfriend if she insulted him in front of friends.
- 3. Girls sometimes deserve to be hit by the boys they date.
- A girl who makes her boyfriend jealous on purpose, deserves to be hit
- 5. Boys sometimes deserve to be hit by the girls they date.
- 6. Sometimes boys have to hit their girlfriends to get them back under control.
- 7. It is OK for a boy to hit a girl if she hits him first.
- 8. It is OK for a girl to hit a boy if he hits her first.
- If someone hits their boyfriend or girlfriend, their boyfriend or girlfriend should break up with them
- ROs: Strongly agree, Agree, Disagree, Strongly Disagree

Descriptive DRV norms^b

- How many of your friends have forced someone to have sexual activity with them that caused their partner to cry, scream, plead, hit, or fight back?
- 2. How many of your friends have used physical force, such as hitting to solve fights with their girlfriends or boyfriends?
- 3. How many of your friends insult their girlfriend or boyfriend, swear at them, or try to control everything their boyfriend or girlfriend does?

ROs: 0 friends, 1-2, 3-5, 6+

Items tested in cognitive interviews Injunctive DRV norms

- Please tick one box to show how most other students in your school would feel if a student in your school did each of the following:
- a. A boy hit his girlfriend to get her back under control

Please tick one box on each line to show how

control everything she does?

control everything he does?

many students in your school you think have

How many **boys** in your school insult

their girlfriend, swear at her, or try to

How many girls in your school insult

their boyfriend, swear at him, or try to

ROs: Approve, Disapprove, Neither

Descriptive DRV norms

a.

b.

done each of the following:

ROs: None, Some, Many, Most

Final measure (tested items in **bold**) Injunctive DRV norms

Please tick a box to show whether **your friends** would agree or disagree with each statement:

- It is NOT okay for a boy to hit his girlfriend if she did something to make him mad.
- b. Girls sometimes deserve to be hit by their boyfriends.
- c. Boys sometimes deserve to be hit by their girlfriends.
- d. It is okay for a boy to hit a girl if she hit him first.
- e. It is NOT okay for a girl to hit a boy if he hit her first.
- f. If someone hits their boyfriend or girlfriend, the boyfriend or girlfriend should break up with them

ROs: My friends would agree, My friends would disagree, My friends would neither agree nor disagree

Tested item removed.

Descriptive DRV norms

Do you have friends who have girlfriends or boyfriends? **ROs:** Yes, No

(if Yes)

- Please tick a box to show your best guess of how many of your friends have done the following:
- a. How many of your friends have used physical force, such as hitting, to solve fights with their girlfriend or boyfriend?
- b. How many of <u>your friends</u> insult or swear at their girlfriend or boyfriend?
- c. How many of <u>your friends</u> try to control everything their girlfriend or boyfriend does? ROs: None, Some, Many, Most

(continued)

Table I. (continued)

Attitudes	toward	gender	roles &	stereotypesc
100100000	co mana	School	10103 0	. stereotypes

- Swearing is worse for a girl than for a boy.
- 2. On a date, the boy should be expected to pay all expenses.
- 3. On the average, girls are as smart as boys.
- More encouragement in a family should be given to sons than daughters to go to college.
- 5. It is all right for a girl to want to play rough sports like football.
- 6. In general, the father should have greater authority than the mother
- in making family decisions.7. It is all right for a girl to ask a boy out on a date.
- 8. It is more important for boys than girls to do well in school.
- 9. If both husband and wife have jobs, the husband should do a share of the housework such as washing dishes and doing the laundry.
- 10. Boys are better leaders than girls.
- Girls should be more concerned with becoming good wives and mothers rather than desiring a professional or business career.
- 12. Girls should have the same freedom as boys.
- 13. Most girls like to show off their bodies.
- 14. Most boys like to go out with girls just for sex.
- 15. Most girls cannot be trusted.
- 16. It is more accepted for a boy to have many sexual partners than for a girl.

ROs: Strongly agree, Agree, Disagree, Strongly disagree

Attitudes toward gender roles & stereotypes

- Please tick one box on each line to show how much **you personally** agree or disagree with each statement.
- On a date, the boy should pay all the expenses.
- ROs: Strongly agree, Agree, Disagree, Strongly disagree

Please tick one box on each line to show how

most other students in your school

Please tick one box on each line to show how

most other students in your school

would feel about a girl or boy in your school

A girl in your school who has a lot of

A girl and a boy go on a date, and the

would feel about each of the following

boy pays all the expenses

ROs: Approve, Disapprove, Neither

who does each of the following:

sex partners. ROs: Approve, Disapprove, Neither

Injunctive gender norms

scenarios:

a.

a.

Attitudes toward gender roles & stereotypes

Please tick a box to show how much **you personally** agree or disagree with each statement.

- a. Swearing is worse for a girl than for a boy.
- b. It is more acceptable for a boy to have a lot of sexual partners than for a girl.
- c. Most girls can't be trusted.
- d. On average, girls are as smart as boys.
- e. Girls should have the same freedom as boys.

ROs: I strongly agree, I agree, I disagree, I strongly disagree

Tested item removed.

Injunctive gender norms

- Please tick a box to show whether **your friends** would agree or disagree with each statement.
- a. Swearing is worse for a girl than for a boy.
- b. It is more acceptable for a boy to have a lot of sexual partners than for a girl.
- c. Most girls can't be trusted.
- d. On average, girls are as smart as boys.
- e. Girls should have the same freedom as boys.
- **ROs:** My friends would agree, My friends would disagree, My friends would neither agree nor disagree

Second tested item removed.

Note: DRV=dating and relationship violence; ROs=response options. ^aItems adapted from Foshee et al. (2001). ^bItems adapted from Cook-Craig et al. (2014). ^cItems adapted from Sotiriou et al. (2011). background; or any other ethnic group), sex assigned at birth, gender (male; female; transgender male; transgender female; or do not identify as male, female, or transgender), and religious group (none; Christian; Jewish; Muslim/Islam; Hindu; Buddhist; Sikh; I don't know/not sure; other religious group).

Injunctive DRV Norms. We developed an item measuring injunctive DRV norms (see Table 1) based on a scale assessing attitudes toward DRV which was used with adequate reliability (Lewis et al., 2015) of α =.69 in a trial of the Safe Dates DRV intervention (Foshee et al., 2001). The new measure instructed participants to indicate the views of "most other students in your school" on a series of behaviors attributed to students at the school, assessing norms at the site of intervention. We adapted the item "Sometimes boys have to hit their girlfriends to get them back under control" to ask about injunctive norms governing this behavior. We simplified response options from four levels of agreement ("strongly agree," "agree," "disagree," and "strongly disagree") to "approve," "disapprove," and "neither" (Cislaghi, 2016).

Descriptive DRV Norms. We adapted the descriptive DRV norms item "How many of your friends insult their girlfriend or boyfriend, swear at them, or try to control everything their boyfriend or girlfriend does?" to create two items concerning psychological DRV, complementing the injunctive norms item concerning physical DRV (see Table 1). The original item was drawn from a descriptive norms measure used with good reliability (Lewis et al., 2015) of $\alpha = .70$ in a trial of Green Dot, a DRV and sexual violence intervention (Cook-Craig et al., 2014). We simplified response options from asking for the number of people to four options: "none," "some," "many," or "most" (Cislaghi, 2016). We changed the reference group from "your friends" to "girls in your school" and (in a separate item) "boys in your school," assessing norms at the site of intervention and separately by gender given that reported rates of DRV can differ between girls and boys (Barter et al., 2014; Leen et al., 2013). While evidence suggests that DRV rates might be higher among gender minorities as compared to cisgender young people (Dietz, 2019), we restricted this item to the two gender reference groups used in existing valid and reliable DRV descriptive norms measures (Meiksin et al., 2023).

Attitudes Toward Gender Roles and Stereotypes. We adapted an item from the 16-item Attitudes Toward Women Scale (ATWS), a measure of attitudes toward gender roles and stereotypes that combines items from previous measures (Sotiriou et al., 2011) and was used with excellent reliability (Lewis et al., 2015) of α = .82 in a 2011 study in Greece (see Table 1) (Sotiriou et al.,

2011). We identified this measure via an ad hoc search for relevant measures used with good reliability in gender-based violence research among adolescents within the previous decade. We selected an item to test that concerned gender roles in dating, and instructed participants to indicate "how much you personally agree or disagree." We simplified language from "On a date, the boy should be expected to pay all expenses" to "On a date, the boy should pay all the expenses."

Injunctive Gender Norms. We adapted two items from the ATWS (Sotiriou et al., 2011) to develop injunctive gender norms items asking participants to indicate the views of "most other students in your school" on a series of behaviors and scenarios (see Table 1). The first item assessed norms governing sexual behavior. To simplify language and focus on norms at the site of the intervention, we adapted the original item ("It is more accepted for a boy to have many sexual partners than for a girl") to ask about "a girl in your school who has a lot of sex partners." The second item was paired with the tested item on attitudes toward gender roles and stereotypes, allowing for comparison between responses about participants' own and others' views on the same behavior. We simplified response options for both items from four levels of agreement to "approve," "disapprove," and "neither" (Cislaghi, 2016).

Analysis

The interviewer took detailed notes on each participant's response to each interview question and probe during the interview (Willis, 1999) and, after reading and re-reading these notes after data collection, produced written summaries of the results for each question and probe by participant year-group and then overall sample (Willis & Artino, 2013). Summaries detailed both "dominant trends" and "discoveries" (i.e., problems might be significant despite arising rarely) (Willis, 1999, p. 28) and differences and similarities by gender and year-group. The detailed notes and written summaries were then subjected to thematic analysis (Green & Thorogood, 2018). Informed by the notion of constant comparison (Green & Thorogood, 2018), data and codes were compared throughout the analysis process, and newly emerging codes were applied to the full dataset.

When responding to a survey item, a survey participant must comprehend the question, retrieve information from long-term memory, make a judgment about how to answer, and then select from among the response options provided (Collins, 2003). Drawing on Young et al.'s work developing survey measures for young people (Young et al., 2016), we conceptualized these processes as falling within two distinct analytic categories: understandability (encompassing comprehension) and answerability (encompassing retrieval, judgment, and response). Individual codes were developed inductively under the headings of "understandability" and "answerability," with two sets of such codes: one applied to data on social norms items and the other (for comparison) applied to data on attitude items. Interview data were coded for evidence of good or poor understandability and answerability and for aspects of the tested items that enhanced or detracted from understandability and answerability. Axial coding drew together initial codes relating to the same themes—for example, the role of the framing of the reference group or observed manifestations of social norms—facilitating analysis within these themes.

Analysis of written summaries provided an overview of our findings and facilitated comparison by gender and year-group. Further analysis of notes on individual interviews identified the evidence supporting overall findings.

Results

In all, 11 students took part in cognitive testing of social norms and attitudes items (Table 2). All were cisgender comprising seven girls and four boys. Participants were spread across year-groups with three in year 8 (age 13), five in year 9 (ages 13–14), and three in year 10 (ages 14–15). All but one identified as White British and all but two selected "none" for a religious group. Injunctive gender norms items were skipped with one participant, who did not reach these items before having to return to class; all other items were tested with all 11 participants.

Summary results relating to understandability and answerability of each tested measure, and refinements made based on these findings, are available in Supplemental Appendix A.

Understandability

Item Clarity. There was some initial difficulty with understanding the intended meaning of all three injunctive norms measures for some participants. Rather than difficulty with specific terms or phrases, some confusion appeared to stem from the framing of the items which, when the instructions and item were read aloud together, were somewhat lengthy (see Table 1). When presented with injunctive norms measures, participants from all year-groups often asked whether the item was asking for their own or others' views, or answered initially in terms of their own views. In an example of the former, one boy asked the interviewer to clarify whether the injunctive DRV norms measure was asking for his views, those of other boys in

Characteristics	Year 8 N	Year 9 N	Year 10 N	Total N (%)
Gender				
Girls	2	3	2	7 (63.6)
Boys	1	2	1	4 (33.4)
Age In years				
13	3	1	0	4 (33.4)
14	0	4	2	6 (54.5)
15	0	0	1	I (9.1)
Ethnicity				
Black African, Caribbean,	1	0	0	l (9.1)
or Black British				
White British	2	5	3	10 (90.9)
Religion				
Christian	1	0	1	2 (18.2)
None	2	5	2	9 (81.8)
Total N (%)	3 (27.3)	5 (45.5)	3 (27.3)	11 (100.0)

Table 2. Background Characteristics of Cognitive Interview Participants.

the school or those of girls. In an example of the latter, in response to the item assessing injunctive gender norms relating to a girl with many sex partners, a girl said, "If it was my friend I'd disapprove, but if it was someone I didn't know, I wouldn't care." Similarly, another girl described the measure of injunctive DRV norms as easy to answer because "I just think boys shouldn't hit girls," suggesting that she had interpreted the item as assessing her own views on DRV.

These findings suggest that injunctive norms items tended not to work well in their tested form, as their meanings were often not initially clear to participants. Where this was the case, the interviewer explained the intended meaning of the item, including (where needed) explaining that the question was asking about the participant's perception of others' views. These explanations were effective in clarifying item meaning, suggesting that for participants who had difficulty with understandability, this reflected a lack of clarity of wording rather than a more fundamental inability to distinguish injunctive norms from personal views. Students' ability to make this distinction was especially apparent in responses to the parallel items (see Table 1) that explored both personal attitudes and injunctive social norms regarding the gendered behavior of a boy paying the expenses on a date with a girl. In their responses, two girls highlighted where they personally disagreed with others' views (as they perceived them).

Answerability

Level of Certainty About Others' Behavior and Views. Participants tended to have difficulty responding to the measure of descriptive DRV norms because they were uncertain about the prevalence of psychological DRV perpetration among their peers. Some qualified their answers; for example, participants added "that I know of" and one of these participants also specified that they were responding with estimates among people whom they knew. Furthermore, asking about multiple behaviors within the same item detracted from answerability. For example, one girl commented that some boys might swear at their partner but would not necessarily insult or try to control her. Contributing to this uncertainty was that psychological DRV perpetration might be unobservable. As one boy explained, some might try to control their partner due to jealousy but he did not think they would "broadcast" this behavior because people would disapprove and the person would feel embarrassed by others' disapproval.

Overall, participants tended to report that they could respond more easily to measures of attitudes than to measures of injunctive norms. They explained that they knew their own mind better while imagining what others thought was more difficult. The level of difficulty in answering injunctive norms items varied based on the specified reference group and on the observability of social sanctions for, and on the strength of, the assessed norm, as described below.

Reference group. The reference group for injunctive norms measures, "most other students in your school," brought to mind a range of different groups for participants. A few said they thought of their friends when responding to these items; others reported thinking of older students or their own year-group. Some reported thinking of other students of the same gender, including older or popular boys. Our data suggest that responses to injunctive norms items would differ depending on the gender of the reference group students had in mind. For example, regarding an injunctive DRV norms item, one girl responded, "I know a lot of the girls would disapprove. I think it depends on who the boy's friends are." A gendered distinction arose also for the descriptive norms items, where levels of perceived DRV differed for items asking about perpetration by girls and by boys.

When asked about changing the reference group to "your friends," some participants said this could make some norms items easier to answer. This change tended to improve the answerability of the injunctive DRV norms measure, with one boy suggesting that this was because he would be more likely to know the views of his friends than views among the broader reference group. However, it made less of a difference to the answerability of the measure of injunctive gender norms relating to paying on a date because the absence of a strong norm governing this behavior also detracted from that item's answerability.

Observability of Behaviors and Social Sanctions. Answerability was improved where participants could draw concretely on past observations and conversations to respond to social norms items. For example, in discussing how she arrived at her response to the descriptive DRV norms item, a girl recalled seeing a boy screaming at another girl because she had thrown away a ring he had given to her. Discussing how they became aware of injunctive gender norms governing sexual behavior, participants described the public visibility of social sanctions. For example, one girl reported that a boy with many sexual partners would be high-fived while a girl would be called a "slag." Similarly, a boy explained that "all the students in the school" would talk judgmentally with their friends about a student with many sexual partners or when a nude image of a student was circulated.

Presence of a Strong Norm. Once the intention to assess others' views had been clarified where needed, participants answered the measure assessing injunctive gender norms relating to a girl with many sexual partners more easily and confidently than they did other social norms measures. Participants were able to describe social repercussions for violating this norm, suggesting that the item taps a norm that is strong in the reference group and they could thus easily draw on examples of observed behavior related to this norm. By contrast, participants tended to have more difficulty responding to the injunctive gender norms item assessing expectations of who should pay on a date between a girl and a boy. While some answered this item with little apparent difficulty, participants often expressed some uncertainty about their response, for example using words like "maybe" or "probably" or describing variable views among the reference group.

Responses to the parallel measure of attitudes provided further insight. For several participants, the response depended on context, for example, who had paid last time, the cost of the bill, or whether this was a first date. Responses to these paired attitude and injunctive norms items suggest that the injunctive norms measure did not tap a strong social norm among this population.

Measure Refinements

Based on our cognitive interview findings, we made a number of refinements to social norms and attitude measures in preparation for piloting. To address variability in who the reference group "most other students in your school" brought to mind, and difficulty reporting perceived views of this reference group, all social norms measures were adapted to ask about a reference group of "your friends." To improve clarity and readability, we simplified the instructions for all measures and made minor changes to wording to reflect common parlance in England. To clarify that injunctive norms items ask about the views of others, we simplified the measures' instructions and adapted items and response options to mirror corresponding attitudes measures (i.e., "Please tick a box to show whether your friends would agree or disagree with each statement"). Refined survey measures thus asked about one's own and others' views on the same behaviors and presented similar Likert scale response options that reinforced the perspective in question (e.g., "I agree" for attitudes items and "My friends would agree" for injunctive norms items). While attitudes items had four response options (two levels of agreement and two levels of disagreement), refined injunctive norms measures had three (agreement, disagreement, or neither) to improve answerability and to accommodate the possibility of items representing weak or absent norms. We also removed three items from the injunctive DRV norms measure (and corresponding attitudes measure) that specified a rationale for DRV (e.g., "A girl who makes her boyfriend jealous on purpose deserves to be hit"), which we judged to be less readable than other items and more difficult to respond to from the perspective of a reference group; and reframed two items from pro- to anti-DRV statements to ensure a mixture of statements supporting and opposing DRV.

For the descriptive DRV norms measure, we added instructions to "show your best guess" and we added a filter question so that only participants reporting more than one friend with a partner would be routed to these items. Based on findings that items assessing behaviors that were more likely to be observed were easier to answer and that assessing multiple behaviors within one item reduced answerability, we removed the item about sexual DRV and separated items on controlling behavior and insulting/swearing at a partner. We also adapted descriptive DRV norms items to be gender-neutral, more closely reflecting the original measure (see Table 1) and enabling us to ask about a more meaningful reference group (friends with partners, as opposed to smaller groups specifying female friends with boy friends and male friends with girlfriends) while reducing the number of items in the measure. Attitude and injunctive gender norms items relating to who should pay on a date were dropped. To reduce the length of the injunctive gender norms measure and the corresponding attitudes measure, we removed items about gender roles among adults (father/mother, husband/wife) and items we judged to be less likely to represent strong norms among young people in England.

Discussion

Our findings suggest that participants were able to understand both descriptive and injunctive norms items and distinguish between the latter and their own views. Some participants showed initial confusion about whether injunctive norms items were asking for their own views or the views of others, suggesting that the wording of tested measures should be refined to improve clarity. To this end, injunctive norms measures were adapted to mirror attitude measures, that is, to ask about the same behaviors using similar Likert scale response options, a format used in other research with young people (Shamu et al., 2016).

We found that inconsistency in who the reference group "most other students in your school" brought to mind, lack of certainty about DRV behaviors, and perceived heterogeneity in views among this population, detracted from the answerability of norms items. Our data suggest that narrowing the reference group to "your friends" might improve answerability where norms are salient among this reference group and where the behaviors in question are likely to have been discussed or observed. The use of this narrower reference group is supported by evidence from a recent systematic review of social norms measures in DRV research showing that for nearly all included measures, DRV supportive norms among friends were associated with young people's own experience of DRV (Meiksin et al., 2023).

Research with young people finds that levels of support for DRV (Pöllänen et al., 2018) and DRV prevalence (Barter et al., 2014; Leen et al., 2013) can both vary by gender and that girls tend to report less support for inequitable gender norms (Kågesten et al., 2016) than do boys. Findings from our study suggest that young people are sensitive to these differences among their peers: participants consider the gender of reference group members in their responses to norms items. Where they thought that the views of girls and boys differed, this detracted from the answerability of tested injunctive norms items. However, repeating all norms measures for reference groups of girls and boys separately could result in lengthy scales that would be unfeasible to include in surveys. Narrowing the reference group to "your friends" allows participants to bring to mind a smaller social group. Peers with whom young people identify or feel connected (such as friends) can be particularly influential (Wolfe & Temple, 2018). While the majority of adolescents' friends are those of the same sex (Deutsch et al., 2014), this approach also allows for individual variation in the gender composition of the reference group while minimizing the number of measure items. Piloting these measures among a representative sample of young people in England will provide important information about their acceptability, reliability, and validity.
During the development of gender norms measures for the Global Early Adolescent Study, participants aged 10 to 14 years were surveyed about their own attitudes and those of their friends toward the same gender norms (Moreau, 2018). The study found that many reported not knowing what their friends thought and that, overall, participants tended to report their friends' views as very similar to their own, raising concerns about whether data collected via surveys can distinguish between these two concepts in this age group (Moreau, 2018; Moreau et al., 2021). The present study builds on these findings by identifying features of norms measures that improve answerability. We found that participants were most easily and confidently able to respond where norms appeared relatively strong and where they could draw on concrete experiences of seeing norms on public display; that is, where they had discussed or seen the specified behaviors or where they had observed social rewards/repercussions for complying with/ violating injunctive norms.

As with any research on social norms important to a particular health outcome, norms measures in DRV research should focus on social norms that are linked theoretically or empirically to DRV outcomes. Based on our findings, we recommend that decisions about which social norms items to include in DRV research should be based on local formative research identifying norms (a) held among a cohesive and influential reference group; (b) strong enough among the reference group for respondents to discern; and (c) for which the relevant behavior (for descriptive norms) or social sanctioning (injunctive norms) is discussed or directly observable. Measures should be worded as clearly and concisely as possible and, where surveys include corresponding attitudinal measures, researchers should consider using parallel formatting, items, and response options for both types of measures to improve the understandability of injunctive norms measures.

Limitations

As the interviewer took notes during cognitive interviews, it is possible that some nonverbal signs of participants' confusion or tentativeness could have been missed.

Our data come from a sample of 11 participants aged 13 to 15 years recruited from one school in England. While our sample included cisgender girls and boys, it was not diverse by ethnicity, religion, or other gender identities and no data were available on sexual orientation. In addition, given time constraints on the length of the interviews as well as the early stage of our work to refine social norms measures, only two descriptive norms items and three injunctive norms items were tested and this did not include testing of refined items. We were, therefore, unable to test directly whether refinements improved understandability and answerability, including assessing whether the gender-neutral framing of the reference group "your friends" detracts from answerability due to the gendered nature of the tested items.

Conclusions

Our findings suggest that it is possible to develop social norms measures about gender and DRV that are understandable and answerable for young people aged 13 to 15 years in England. Future research should cognitively test a broader range of items, including those assessing norms suspected to be more and less publicly manifest, and should do so among a sample of young people that is diverse in terms of backgrounds and sexual and gender identities. The acceptability, reliability, and validity of the social norms measures refined through cognitive testing in the present study should be assessed among a representative sample of young people in England. Where new social norms measures are developed or existing measures adapted for DRV research, these should assess norms which are salient and publicly manifest among a cohesive, influential reference group. Careful consideration is needed to establish the value of including measures of social norms where this is not known to be the case.

Acknowledgments

The authors gratefully acknowledge the young people who generously gave their time and shared their views for this study, the staff who warmly welcomed us into their school and facilitated data collection, and Ben Pelhan for his invaluable support.

Declaration of Conflicting Interests

The author(s) declared no potential conflicts of interests with respect to the authorship and/or publication of this article.

Funding

The author(s) disclosed receipt of the following financial support for the research and/ or authorship of this article: This study was funded by the NIHR Public Health Research Board (project 15/03/09). The views expressed are those of the authors and not necessarily those of the NIHR or the Department of Health and Social Care. The funder was not involved in study design; collection, analysis, or interpretation of data; the writing of this article; or the decision to submit this article for publication.

ORCID iD

Rebecca Meiksin (D) https://orcid.org/0000-0002-5096-8576

Supplemental Material

Supplemental material for this article is available online.

References

- Alexander-Scott, M., Bell, E., & Holden, J. (2016). Shifting social norms to tackle violence against women and girls (VAWG). VAWG Helpdesk.
- Barter, C., McCarry, M., Berridge, D., & Evans, K. (2009). Partner exploitation and violence in teenage intimate relationships. NSPCC.
- Barter, C., & Stanley, N. (2016). Inter-personal violence and abuse in adolescent intimate relationships: Mental health impact and implications for practice. *International Review of Psychiatry*, 28(5), 485–503. https://doi.org/10.1080/09 540261.2016.1215295
- Bonell, C., Blakemore, S.-J., Fletcher, A., & Patton, G. (2019). Role theory of schools and adolescent health. *The Lancet Child & Adolescent Health*, 3(10), 742–748. https://doi.org/10.1016/S2352-4642(19)30183-X
- Barter C., Aghtaie N., Wood M., Stanley N., Pavlou S., Apostolov G., et al. (2014). Safeguarding teenage intimate relationships (STIR): Connecting online and offline contexts and risks Briefing Paper 2: Incidence Rates and Impact of Experiencing Interpersonal Violence and Abuse in Young People's Relationships. (2015). University of Bristol.
- Castellví, P., Miranda-Mendizábal, A., Parés-Badell, O., Almenara, J., Alonso, I., Blasco, M. J., Cebrià, A., Gabilondo, A., Gili, M., Lagares, C., Piqueras, J. A., Roca, M., Rodríguez-Marín, J., Rodríguez-Jimenez, T., Soto-Sanz, V., & Alonso, J. (2017). Exposure to violence, a risk for suicide in youths and young adults. A meta-analysis of longitudinal studies. *Acta Psychiatrica Scandinavica*, 135(3), 195–211. https://doi.org/10.1111/acps.12679
- Cislaghi, B. (2016). LSHTM group on social norms and GBV: Strategies to diagnose and measure social norms related to gender-based violence: Key lessons from the Baltimore working meeting. STRIVE.
- Cislaghi, B., & Heise, L. (2018). Four avenues of normative influence: A research agenda for health promotion in low and mid-income countries. *Health Psychology*, 37(6), 562–573. https://doi.org/10.1037/hea0000618
- Collins, D. (2003). Pretesting survey instruments: An overview of cognitive methods. Quality of Life Research: An International Journal of Quality of Life Aspects of Treatment, Care and Rehabilitation, 12(3), 229–238. https://doi. org/10.1023/a:1023254226592
- Cook-Craig, P. G., Coker, A. L., Clear, E. R., Garcia, L. S., Bush, H. M., Brancato, C. J., Williams, C. M., & Fisher, B. S. (2014). Challenge and opportunity in evaluating a diffusion-based active bystanding prevention program: green dot in high schools. *Violence Against Women*, 20(10), 1179–1202. https://doi. org/10.1177/1077801214551288
- Costenbader, E., Cislaghi, B., Clark, C. J., Hinson, L., Lenzi, R., McCarraher, D. R., McLarnon-Silk, C., Pulerwitz, J., Shaw, B., & Stefanik, L. (2019). Social norms

measurement: Catching up with programs and moving the field forward. The Journal of Adolescent Health: Official Publication of the Society for Adolescent Medicine, 64(4S), S4–S6. https://doi.org/10.1016/j.jadohealth.2019.01.001

- de Leeuw, E., Borgers, N., & Strijbos-Smits, A. (2002, November). Children as respondents: Developing, evaluating, and testing questionnaires for children. In: *International conference in questionnaire development evaluation and testing methods*. Charleton, SC.
- Deutsch, A. R., Steinley, D., & Slutske, W. S. (2014). The role of gender and friends' gender on peer socialization of adolescent drinking: A prospective multilevel social network analysis. *Journal of Youth and Adolescence*, 43(9), 1421–1435. https://doi.org/10.1007/s10964-013-0048-9
- Dietz, J. (2019). Dating and relationship violence in sexual and gender minority youth: A systematic review of the literature and cost-effectiveness analysis of a targeted school-based preventative intervention. London School of Hygiene & Tropical Medicine.
- Exner-Cortens, D., Eckenrode, J., & Rothman, E. (2013). Longitudinal associations between teen dating violence victimization and adverse health outcomes. *Pediatrics*, 131(1), 71–78. https://doi.org/10.1542/peds.2012-1029
- Foshee, V. A., Linder, F., MacDougall, J. E., & Bangdiwala, S. (2001). Gender differences in the longitudinal predictors of adolescent dating violence. *Preventive Medicine*, 32(2), 128–141. https://doi.org/10.1006/pmed.2000.0793
- Foshee, V. A., Linder, G. F., Bauman, K. E., Langwick, S. A., Arriaga, X. B., Heath, J. L., McMahon, P. M., & Bangdiwala, S. (1996). The safe dates project: Theoretical basis, evaluation design, and selected baseline findings. *American Journal of Preventive Medicine*, 12(5 Suppl), 39–47.
- Gage, A. J. (2016). Exposure to spousal violence in the family, attitudes and dating violence perpetration among high school students in Port-au-Prince. *Journal of Interpersonal Violence*, 31(14), 2445–2474. https://doi. org/10.1177/0886260515576971
- Green, J., & Thorogood, N. (2018). Qualitative methods for health research. SAGE.
- Kågesten, A., Gibbs, S., Blum, R. W., Moreau, C., Chandra-Mouli, V., Herbert, A., & Amin, A. (2016). Understanding factors that shape gender attitudes in early adolescence globally: A mixed-methods systematic review. *PLoS One*, 11(6), e0157805. https://doi.org/10.1371/journal.pone.0157805
- Leen, E., Sorbring, E., Mawer, M., Holdsworth, E., Helsing, B., & Bowen, E. (2013). Prevalence, dynamic risk factors and the efficacy of primary interventions for adolescent dating violence: An international review. *Aggression and Violent Behavior*, 18(1), 159–174. https://doi.org/10.1016/j.avb.2012.11.015
- Lewis, C. C., Stanick, C. F., Martinez, R. G., Weiner, B. J., Kim, M., Barwick, M., & Comtois, K. A. (2015). The society for implementation research collaboration instrument review project: A methodology to promote rigorous evaluation. *Implementation Science*, 10(1), 1–18. https://doi.org/10.1186/s13012-014-0193-x
- Meiksin, R., Bonell, C., Bhatia, A., Melendez-Torres, G. J., Kyegombe, N., & Kohli, A. (2023). Social norms about dating and relationship violence and

gender among adolescents: Systematic review of measures used in dating and relationship violence research. *Trauma, Violence, & Abuse*, 152483802311555. https://doi.org/10.1177/15248380231155526

- Meiksin, R., Crichton, J., Dodd, M., Morgan, G. S., Williams, P., Willmott, M., Allen, E., Tilouche, N., Sturgess, J., Morris, S., Barter, C., Young, H., Melendez-Torres, G., Taylor, B., Reyes, H. L. M., Elbourne, D., Sweeting, H., Hunt, K., Ponsford, R., & Bonell, C. (2020). A school intervention for 13- to 15-year-olds to prevent dating and relationship violence: The Project Respect pilot cluster RCT. *Public Health Research*, 8(5), 1–338. https://doi.org/10.3310/phr08050
- Moreau, C. (2018, February 1). Development and validation of cross-cultural gender norms scales for early adolescents [Webinar]. http://www.geastudy.org/webinars/
- Moreau, C., Li, M., Ahmed, S., Zuo, X., & Cislaghi, B. (2021). Assessing the spectrum of gender norms perceptions in early adolescence: A cross-cultural analysis of the global early adolescent study. *Journal of Adolescent Health*, 69(1), S16–S22. https://doi.org/10.1016/j.jadohealth.2021.03.010
- Pöllänen, K., de Vries, H., Mathews, C., Schneider, F., & de Vries, P. J. (2018). Beliefs about sexual intimate partner violence perpetration among adolescents in South Africa. *Journal of Interpersonal Violence*, 36(3–4), NP2056-2078NP. https://doi.org/10.1177/0886260518756114
- Roberts, T. A., Klein, J. D., & Fisher, S. (2003). Longitudinal effect of intimate partner abuse on high-risk behavior among adolescents. *Archives of Pediatrics & Adolescent Medicine*, 157(9), 875. https://doi.org/10.1001/archpedi.157.9.875
- Salazar, L. F., Swartout, K. M., Swahn, M. H., Bellis, A. L., Carney, J., Vagi, K. J., & Lokey, C. (2018). Precollege sexual violence perpetration and associated risk and protective factors among male college freshmen in Georgia. *Journal of Adolescent Health*, 62(3), S51–S57. https://doi.org/10.1016/j.jadohealth.2017.09.028
- Shakya, H. B., Cislaghi, B., Fleming, P., Levtov, R. G., Boyce, S. C., Raj, A., & Silverman, J. G. (2022). Associations of attitudes and social norms with experiences of intimate partner violence among married adolescents and their husbands in rural Niger: A dyadic cross-sectional study. *BMC Women's Health*, 22(1), 180. https://doi.org/10.1186/s12905-022-01724-y
- Shamu, S., Gevers, A., Mahlangu, B. P., Jama Shai, P. N., Chirwa, E. D., & Jewkes, R. K. (2016). Prevalence and risk factors for intimate partner violence among Grade 8 learners in urban South Africa: Baseline analysis from the Skhokho Supporting Success cluster randomised controlled trial. *International Health*, 8(1), 18–26. https://doi.org/10.1093/inthealth/ihv068
- Sotiriou, P., Ntinapogias, S., & Petroulaki, K. (2011). Attitudes on gender stereotypes and gender-based violence among youth. Country report: Greece (Daphne III Programme: Youth4Youth: Empowering Young People in Preventing Gender-Based Violence through Peer Education). European Anti-Violence Network.
- Spear, L. P. (2000). The adolescent brain and age-related behavioral manifestations. Neuroscience & Biobehavioral Reviews, 24(4), 417–463. https://doi.org/10.1016/ S0149-7634(00)00014-2

- Stanley, N., Ellis, J., Farrelly, N., Hollinghurst, S., & Downe, S. (2015). Preventing domestic abuse for children and young people: A review of school-based interventions. *Children and Youth Services Review*, 59, 120–131. https://doi. org/10.1016/j.childyouth.2015.10.018
- Streiner, D. L., & Norman, G. R. (2008). Health measurement scales: A practical guide to their development and use (4th ed.). Ox ford University Press.
- Vagi, K. J., Rothman, E. F., Latzman, N. E., Tharp, A. T., Hall, D. M., & Breiding, M. J. (2013). Beyond correlates: A review of risk and protective factors for adolescent dating violence perpetration. *Journal of Youth and Adolescence*, 42(4), 633–649. https://doi.org/10.1007/s10964-013-9907-7
- Wesche, R., & Dickson-Gomez, J. (2019). Gender attitudes, sexual risk, intimate partner violence, and coercive sex among adolescent gang members. *Journal of Adolescent Health*, 64(5), 648–656. https://doi.org/10.1016/j.jadohealth.2018.10.292
- Willis, G. B. (1999). Cognitive interviewing, A "How To" guide. Research Triangle Institute.
- Willis, G. B., & Artino, A. R. (2013). What do our respondents think we're asking? Using cognitive interviewing to improve medical education surveys. *Journal of Graduate Medical Education*, 5(3), 353–356. https://doi.org/10.4300/ JGME-D-13-00154.1
- Wolfe, D. A., & Temple, J. R. (Eds.). (2018). Adolescent dating violence: Theory, research, and prevention. Academic Press.
- Young, H., Költő, A., Reis, M., Saewyc, E. M., Moreau, N., Burke, L., Cosma, A., Windlin, B., Gabhainn, S. N., & Godeau, E. (2016). Sexual Health questions included in the Health Behaviour in School-aged Children (HBSC) Study: An international methodological pilot investigation. *BMC Medical Research Methodology*, 16(1), 169. https://doi.org/10.1186/s12874-016-0270-8
- Young, H., Turney, C., White, J., Bonell, C., Lewis, R., & Fletcher, A. (2017). Dating and relationship violence among 16–19 year olds in England and Wales: A crosssectional study of victimization. *Journal of Public Health (Oxford, England)*, 40(4), 738–746. https://doi.org/10.1093/pubmed/fdx139

Author Biographies

Rebecca Meiksin, MPH, is a Research Fellow in Social Science at the London School of Hygiene & Tropical Medicine. Her research focuses on sexual and reproductive health and on the role of social and gender norms in the prevention of violence in young people's dating and relationships.

Ruth Ponsford, PhD, is an Assistant Professor in School Health Intervention Research at the London School of Hygiene & Tropical Medicine. Her research focuses on the health and well-being of young people and the role of school in promoting health, including evaluations of sexual health interventions for young people in England. Nambusi Kyegombe, PhD, is the Head of Social Sciences at the MRC/UVRI & LSHTM Uganda Research Unit. Her work on adolescent health focuses on low- and middle-income country settings. This includes formative research on the context in which health is located, methodological research to improve measurement, and research on the conduct of high quality, ethical research, particularly among marginalized populations.

Anjalee Kohli, PhD, is an Assistant Professor-adjunct at Georgetown University and an independent consultant. Through participatory and mixed methods research and learning partnerships, Dr. Kohli advances the evidence and practice to prevent violence against women and children, to improve gender equity and sexual and reproductive health.

Professor Chris Bonell, PhD, is a Professor of Public Health Sociology at the London School of Hygiene & Tropical Medicine. His research on adolescent health focuses on how schools and school-based interventions can benefit or harm young people's health. His research on sexual health examines the broader social determinants of adolescent sexual health, and interventions to address these.

5.3. Follow-up to Paper 2: Development and refinement of social norms measures for piloting

This section summarises the development of the three social norms measures discussed in Paper 2 and their refinement in preparation for being piloted in Project Respect.

5.3.1. Descriptive DRV norms measure

The measure of descriptive DRV norms was adapted from an existing three-item measure used with respectable reliability¹⁸⁷ (α =0.70) among secondary school students in a trial of Green Dot, a sexual violence and DRV intervention in the US.²¹⁶ Table 3 presents the original measure, items tested in cognitive interviews and the refined measure subsequently piloted in Project Respect.

We removed the item assessing sexual DRV due to the young age of survey respondents (aged 12-14 years). Drawing on examples of simple social norms items highlighted in the work of an expert group on social norms measurement in GBV research,¹⁰⁴ I simplified response options for cognitive interviews from asking about the number of friends perpetrating DRV to "none", "some", "many" and "most". I further simplified the measure by developing instructions to apply to all items in the measure. I reversed the order in which boyfriends and girlfriends are referenced second item for consistency with the beginning of the third item.

Time constraints prevented cognitively testing of all measure items, and so testing focused on the item assessing psychological DRV in order to complement cognitive testing of another norms measure that addressed physical DRV. I split the item on psychological DRV into two separate items in order to ask about girls and boys separately because reported DRV rates differ by gender.^{22,238} I specified the reference group as girls/boys "in your school" because Project Respect targeted students within intervention schools.

On the basis of cognitive interviews, I added a routing question to direct only students with friends who have partners to descriptive DRV norms items, added instructions to "show your

best guess", reverted to the original reference group ("your friends"), and split the single item assessing psychological DRV into two separate items to ask about verbal abuse and control separately. Like the original measure, response options for the refined measure were scored such that a higher score indicated more pro-DRV norms and these were summed to compute a total score for the measure.

	Title, items, response options and scoring		
Original	Peer support for violence ^a		
measure	 How many of your friends have forced someone to have sexual activity with them that caused their partner to cry, scream, plead, hit or fight back? 		
	2. How many of your friends have used physical force, such as hitting to solve fights with their boyfriends or girlfriends?		
	3. How many of your friends insult their girlfriend or boyfriend, swear at them, or try to control everything their boyfriend or girlfriend does?		
	Response options: 0 friends (=0), 1-2 (=1), 3-5 (=2), 6+ (=3)		
Tested items	Please tick one box on each line to show how many students in your		
used in	school you think has done each of the following:		
cognitive testing	a. How many boys in your school insult their girlfriend, swear at her, or try to control everything she does?		
	b. How many girls in your school insult their boyfriend, swear at him,		
	or try to control everything he does?		
	Response options: None, Some, Many, Most		
Piloted	Please tick a box to show your best guess of how many of your friends		
measure ^b	have done the following:		
refined based on	a. How many of your friends have used physical force, such as hitting,		
cognitive testing	 b. How many of your friends insult or swear at their girlfriend or boyfriend? 		
	c. How many of your friends try to control everything their girlfriend or boyfriend does?		
	Response options: None (=1), Some (=2), Many (=3), Most (=4)		

Table 3.	Descriptive	DRV norms:	Original.	cognitively	tested	and i	oiloted	measures
Tubic J.	Descriptive		onginai,	cognitively	<i>icsicu</i>	ana	photeu	measures

^a Measure from Cook-Craig et al.²¹⁶

^b Items asked of participants answering "yes" to the yes/no routing question, "Do you have friends who have girlfriends or boyfriends?"

5.3.2. Injunctive DRV norms measure

The measure of injunctive DRV norms was developed based on an eight-item measure of DRV attitudes used with minimally acceptable reliability¹⁸⁷ (α =0.69) among 8th and 9th grade students in a trial of the Safe Dates DRV prevention intervention.⁷⁷ I added introductory text and modified wording to ask participants about the views of "most other students in your school" rather than their own views. This was to assess injunctive norms at the site of intervention. I added an additional item by adapting an existing item from the three-item "perceived negative sanctions" measure used in the Safe Dates trial.⁷⁷ I replaced "dating partner" with "boyfriend or girlfriend" to align with other items in the new measure and I modified the wording to assess injunctive norms about breaking up with an abusive partner rather than the participant's expectation of whether perpetrating DRV would result in losing their partner. I did not include the remaining two items from the original measure because they did not assess specifically social sanctions, the type of sanctions with which injunctive norms are concerned, and because the measure had unacceptable reliability¹⁸⁷ (Cronbach's α =0.55) in the trial. Informed by simplified approaches to measuring injunctive norms highlighted in the work of an expert group on social norms measurement in GBV research, ¹⁰⁴ for cognitive interviews I added a neutral response option and simplified response options from a four-point Likert scale to "approve", "disapprove" and "neither". Table 4 presents the original measures, items tested in cognitive interviews and the refined measure subsequently piloted in Project Respect.

Time constraints prevented cognitively testing all measure items, and so testing focused on one item addressing physical DRV to complement cognitive testing of the descriptive DRV norms items addressing psychological DRV. On the basis of cognitive interviews I then simplified the introductory text, structured the measure's format and response options to mirror the DRV attitudes measure also included in the Project Respect survey, simplified the reference group to "your friends", reiterated the reference group within the response options and removed three items specifying a justification for DRV. In consultation with the study team we reverse-worded three items to avoid confronting participants with a barrage of pro-DRV statements. Like the

original measure, response options for the refined measure were scored such that a higher score indicated more pro-DRV norms (reverse-scoring the reverse-worded items) and these were summed to compute a total score for the measure.

Table 4. Inju	nctive DRV norms	: Original, cogr	nitively tested ar	nd piloted measures
		- 0 - 7 - 0		

	Title, items, response options and scoring			
Original	Prescribed norms ^a			
measures	1. It is OK for a boy to hit his girlfriend if she did something to make him mad.			
	2. It is OK for a boy to hit his girlfriend if she insulted him in front of friends.			
	3. Girls sometimes deserve to be hit by the boys they date.			
	4. A girl who makes her boyfriend jealous on purpose, deserves to be hit			
	5. Boys sometimes deserve to be hit by the girls they date.			
	6. Sometimes boys have to hit their girlfriends to get them back under control.			
	7. It is OK for a boy to hit a girl if she hit him first.			
	8. It is OK for a girl to hit a boy if he hit her first.			
	Perceived negative sanctions ^a			
	1. If I hit a dating partner he/she would break up with me.			
	2. Bad things happen to people who are violent to their dating partners			
	3. If I hit a dating partner I would be arrested			
	Response options (both measures): strongly disagree (=0) disagree (=1) agree			
	(=2), strongly agree (=3)			
Tested items	Please tick one box to show how most other students in your school would feel			
used in cognitive	if a student in your school did each of the following:			
testing	a. A boy hit his girlfriend to get her back under control.			
	Response options: Approve, Disapprove, Neither			
Piloted measure	Please tick a box to show whether your friends would agree or disagree with			
refined based on	each statement:			
cognitive testing	a. It is NOT okay for a boy to hit his girlfriend if she did something to make			
	him mad. (rev)			
	b. Girls sometimes deserve to be hit by their boyfriends.			
	c. Boys sometimes deserve to be hit by their girlfriends.			
	d. It is okay for a boy to hit a girl if she hit him first.			
	e. It is NOT okay for a girl to hit a boy if he hit her first. (rev)			
	f. If someone hits their boyfriend or girlfriend, the boyfriend or girlfriend			
	should break up with them. (rev)			
	Response options: My friends would agree (=3), My friends would neither agree			
	nor disagree (=2), My friends would disagree (=1)			

^a Measures from Foshee et al. ⁷⁷

rev=reverse-worded, meaning response options scored in the reverse so that a higher score indicates more pro-DRV norms

5.3.3. Injunctive gender norms measure

The measure of injunctive gender norms was developed based on a 16-item measure of gender attitudes used with very good reliability¹⁸⁷ (α =0.82) among participants aged 15-20 years in Greece.²¹⁷ As with the injunctive DRV norms measure, I added introductory text, modified wording to ask participants about the views of students in their school, added a neutral response option and simplified response options from a four-point Likert scale to "approve", "disapprove" and "neither". Table 5 presents the original measures, items tested in cognitive interviews and the refined measure subsequently piloted in Project Respect.

Time constraints prevented cognitively testing all measure items and survey length precluded including all items. In order to focus on norms proximal to DRV, I selected two items to cognitively test which focused on the context of intimate relationships. Refinements based on cognitive testing mirrored those for the injunctive DRV norms measure: I simplified the introductory text, structured the measure's format and response options to mirror the gender attitudes measure also included in the Project Respect survey, simplified the reference group to "your friends" and reiterated the reference group within the response options. Cognitive interviews informed the selection of the final five items for inclusion, alongside the research team's assessment of items' salience for young people in the UK. Items were also selected to retain a mixture of those assessing different aspects of the gender norms domain.¹⁸⁷

We reverse-worded two items so as not to present participants with a barrage of sexist statements. Response options for the refined measure were scored such that a higher score indicated more anti-equality norms (reverse-scoring the reverse-worded items). These were summed to compute a total score for the measure.

	Title, items, response options and scoring				
Original	Attitudes towards women scale ^a				
measure	1. Swearing is worse for a girl than for a boy.				
	2. On a date, the boy should be expected to pay all expenses.				
	3. On the average, girls are as smart as boys. ^b				
	4. More encouragement in a family should be given to sons than				
	daughters to go to college.				
	5. It is all right for a girl to want to play rough sports like football. ^b				
	6. In general, the father should have greater authority than the mother				
	in making family decisions.				
	7. It is all right for a girl to ask a boy out on a date. ^b				
	8. It is more important for boys than girls to do well in school.				
	9. If both husband and wife have jobs, the husband should do a share of				
	the housework such as washing dishes and doing the laundry. ^b				
	10. Boys are better leaders than girls.				
	11. Girls should be more concerned with becoming good wives and				
	mothers rather than desiring a professional or business career.				
	12. Girls should have the same freedom as boys. ¹				
	13. Most girls like to show off their bodies.				
	14. Most boys like to go out with girls just for sex.				
	15. Wost girls can t be trusted.				
	16. It is more accepted for a boy to have many sexual partners than for a				
	giri.				
	Response options: strongly disagree (=1), disagree (=2), agree (=3),				
	strongly agree (=4)				
Tested items	Please tick one box on each line to show how most other students in your				
used in	school would feel about each of the following scenarios: ^c				
cognitive	a. A girl and a boy go on a date, and the boy pays all the expenses				
testing					
	Please tick one box on each line to show how most other students in your				
	school would feel about a girl or boy in your school who does each of the				
	a A girl in your school who has a lot of sex partners				
	Response options: Approve, Disapprove, Neither				
	(table continued)				

Table 5. Injunctive gender norms: Original, cognitively tested and piloted measures

	Title, items, response options and scoring		
Piloted	Please tick a box to show whether your friends would agree or disagree		
measure	with each statement.		
refined based	a. Swearing is worse for a girl than for a boy.		
on cognitive	b. It is more acceptable for a boy to have a lot of sexual partners than		
testing	for a girl.		
	c. Most girls can't be trusted.		
	d. On average, girls are as smart as boys. (rev)		
	e. Girls should have the same freedom as boys. (rev)		
	Response options: My friends would agree (=3), My friends would neither		
	agree nor disagree (=2), My friends would disagree (=1)		

^a Items from Sotiriou et al. ²¹⁷

^b rev=reverse-worded, meaning response options scored in the reverse so that a higher score indicates more anti-equality attitudes

rev=reverse-worded; response options scored in the reverse so that a higher score indicates more antiequality norms

^c Instructions for items refer to ticking "one box on each line" because the tested items form part of multi-item measures in which each item appears on a separate line. The wording was retained for cognitive testing including where only one item was tested.

5.3.4. Piloted social norms measures

Table 6 shows the social norms measures, refined on the basis of cognitive testing, which were

piloted in Project Respect. Paper 3 (see Chapter 6) reports on the results of this piloting.

Measure, response options,	Items
Descriptive DRV norms	Please tick a box to show your best guess of how many of
None (=1). Some (=2). Many	your friends have done the following:
(=3). Most (=4)	1. How many of your friends have used physical force.
	such as hitting, to solve fights with their girlfriend or boyfriend?
	2. How many of your friends insult or swear at their
	girlfriend or boyfriend?
	3. How many of your friends try to control everything
	their girlfriend or boyfriend does?
Injunctive DRV norms	Please tick a box to show whether your friends would agree
My friends would agree (=3),	or disagree with each statement:
My friends would neither	a. It is NOT okay for a boy to hit his girlfriend if she did
agree nor disagree (=2), My	something to make him mad. (rev)
friends would disagree (=1)	b. Girls sometimes deserve to be hit by their boyfriends.
	c. Boys sometimes deserve to be hit by their girlfriends.
	d. It is okay for a boy to hit a girl if she hit him first.
	e. It is NOT okay for a girl to hit a boy if he hit her first.
	(rev)
	f. If someone hits their boyfriend or girlfriend, the
	boyfriend or girlfriend should break up with them.
	(rev)
Injunctive gender norms	Please tick a box to show whether your friends would agree
My friends would agree (=3),	or disagree with each statement.
My friends would neither	a. Swearing is worse for a girl than for a boy.
agree nor disagree (=2), My	b. It is more acceptable for a boy to have a lot of sexual
friends would disagree (=1)	partners than for a girl.
	c. Most girls can't be trusted.
	d. On average, girls are as smart as boys. (rev)
	e. Girls should have the same freedom as boys. (rev)

Table 6. Piloted social norms measures, refined on the basis of cognitive testing

rev=reverse-worded, meaning response options scored in the reverse so that a higher score indicates more pro-DRV/anti-equality norms

Chapter 6. Paper 3: Reliability and validity testing of new social norms measures

6.1. Introduction to Paper 3

After refining three social norms measures through cognitive testing as reported in Paper 2 (see Chapter 5), we piloted the measures in baseline surveys with year 8 students in five schools taking part in Project Respect. Paper 3 reports on the reliability and validity of these measures and makes recommendations for future refinements, then extends this work to make recommendations for future development and use of social norms measures in DRV research. In doing so, this paper addresses the third research question of my thesis, "Are new and adapted measures of social norms relating to DRV and gender reliable and valid when used in research with adolescents in England, and how can they be refined?"

As noted in Section 3.2.5.3, the weight of available evidence suggests that some norms measures should show an independent relationship to some DRV outcomes when adjusting for personal attitudes but does not pinpoint which norms measures should do so and for which DRV outcomes. As part of my analysis, I explored these relationships in multivariable models, and the results contribute to considerations for potential refinements of the tested norms measures. I considered that if tested norms measures were to show an association with DRV outcomes in univariable but not multivariable regressions, this could indicate that the measures should be refined to increase the opportunity for covariation with DRV outcomes.

These analyses also serve a second purpose in broader DRV research. In the context of Paper 3 as a standalone article, I frame the multivariable regressions as exploratory analyses under the distinct research question, "Are social norms concerning DRV and gender associated with DRV outcomes after adjusting for individual attitudes and sociodemographic factors?" This approach was informed by Clark et al.'s work psychometrically testing a new social norms measure and then modelling its covariate-adjusted relationship to IPV outcomes.¹⁵¹ After confirmation of the

reliability and validity of three norms measures presented in Paper 3, I drew on Project Respect data to add to the limited empirical research on relationships between social norms and DRV.

The supplemental appendices to which Paper 3 refers are provided in Appendix 18 of this thesis.

6.2. Paper 3: Social norms in dating and relationship violence research: testing new and adapted measures and examining their relationships to violence outcomes



London School of Hygiene & Tropical Medicine Keppel Street, London WC1E 7HT

T: +44 (0)20 7299 4646 F: +44 (0)20 7299 4656 www.lshtm.ac.uk

RESEARCH PAPER COVER SHEET

Please note that a cover sheet must be completed for each research paper included within a thesis.

SECTION A – Student Details

Student ID Number	452277	Title	Ms.
First Name(s)	Rebecca		
Surname/Family Name	Meiksin		
Thesis Title	Developing and refining social norms measures for the evaluation of a school-based intervention to reduce dating and relationship violence among adolescents in England		
Primary Supervisor	Professor Chris Bonell		

If the Research Paper has previously been published please complete Section B, if not please move to Section C.

SECTION B – Paper already published

Where was the work published?			
When was the work published?			
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion			
Have you retained the copyright for the work?*	Choose an item.	Was the work subject to academic peer review?	Choose an item.

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C - Prepared for publication, but not yet published

Where is the work intended to be published?	Social Science & Medicine
Please list the paper's authors in the intended authorship order:	Rebecca Meiksin, GJ Melendez-Torres, Charles Opondo, Anjalee Kohli, Nambusi Kyegombe, Chris Bonell

Improving health worldwide

www.lshtm.ac.uk

SECTION D - Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	I conceptualised the research questions, designed the tested measures, conducted the analysis and led interpretation and writing. GJ Melendez-Torres led advising on the analysis. Charles Opondo provided specialist statistical advice on the analysis of clustered data. Chris Bonell designed the overall study and advised on analysis. All authors helped interpret the findings and provided feedback on the mansucript.
---	--

SECTION E

Student Signature	Rebecca Meiksin
Date	15/03/2024

Supervisor Signature	Chris Bonell
Date	15/03/2024

Social norms in dating and relationship violence research: testing new and adapted measures and exploring their relationships to violence outcomes Authors: Rebecca Meiksin, GJ Melendez-Torres, Charles Opondo Anjalee Kohli, Nambusi

Kyegombe, Chris Bonell

Abstract

Dating and relationship violence (DRV) is widespread globally and associated with increased substance use, sexual risk behaviours, depression and suicide attempts. While DRV interventions often aim to change social norms underpinning DRV, a lack of established measures inhibits researchers' ability to assess whether norms are changing, and if so, whether this mediates programme impact. Drawing on baseline survey data from a DRV prevention pilot trial in secondary schools in England, we analysed the factor structure of 14 pilot items assessing descriptive and injunctive DRV norms and injunctive gender norms. We assessed the reliability and validity of the resulting measures and examined whether associations between norms and DRV were moderated by sex for eight DRV outcomes. Finally, we explored whether associations persisted in multivariable regressions that included other norms measures and sociodemographic and attitudinal variables where these were found to be covariates. Surveys took place from June-July 2017 and included 1,426 students aged 12-14 years (82.5% response rate) in five schools. Reverse-worded norms items loaded onto a single factor with low reliability and unclear interpretability. Loadings of remaining items delineated three distinct measures aligned with a priori social-norms constructs. Measures had acceptable reliability and were associated with all DRV outcomes in univariable analyses. However, floor effects suggested they were not sensitive to low-to-moderate levels of the constructs assessed. In multivariable analyses, descriptive DRV norms remained independently associated with most DRV outcomes; injunctive gender norms retained an association with most victimisation outcomes. This research resulted in three brief, reliable, valid measures of distinct social norms associated with a broad range of DRV outcomes which can be used in evaluations. Measures could be further refined with the addition of items indicating more moderate levels of the constructs assessed.

Longitudinal research is needed to explore temporal relationships between social norms and DRV and whether these are moderated by sex.

Introduction

Comprising psychological, physical or sexual abuse in adolescence by a current or former intimate partner,^{16,19} dating and relationship violence (DRV) poses risks to young people's health and wellbeing.^{26,52} While interventions often target harmful social norms sustaining this type of violence,⁸ methods for social norms measurement are underdeveloped, impeding observational and intervention studies.¹¹ Drawing on data from a pilot trial of a DRV prevention programme in England, the present study examines the reliability and validity of new and adapted social norms measures and explores their associations with DRV outcomes.

While reported prevalence varies by study design and sample, reviews suggest that DRV is widespread. In a 2016 systematic review, among the 26 studies using the Conflict Tactics Scale (the most commonly used DRV behaviour measure), median psychological victimisation rates were 88% among girls and 72% among boys.²¹ A meta-analytic 2017 systematic review reported physical DRV victimisation rates of 21% among both girls and boys, and sexual DRV victimisation rates of 14% among girls and 8% among boys.²⁵ In addition to injuries,⁵⁰ DRV victimisation is associated with subsequent antisocial^{26,52} and sexual risk⁵⁴ behaviours, substance use,^{26,52} depression,⁵² suicidal ideation²⁶ and suicide attempts,⁵⁵ and intimate partner violence (IPV) victimisation⁴⁹ and perpetration in adulthood.⁵⁶ Compared to boys, girls report additional harms including fear³⁴ and greater injury.⁵⁰

Social norms and DRV

Theory and empirical research suggest that social norms might play an important role in sustaining DRV. A common construct across several prominent behavioural theories,^{100,108,239,240} social norms are informal rules that determine acceptable behaviour within a social group.¹⁰³ Theorists distinguish between two types of social norms: beliefs about what people within a reference group of influential others¹⁰¹ (1) typically do (descriptive norms), and (2) consider to

be appropriate (injunctive norms).^{15,102} Anticipation of social rewards and punishments by the refence group are said to maintain these social expectations.^{15,106}

In a 2023 systematic review of DRV research, we explored measures of social norms concerning DRV and gender and their associations with DRV outcomes.¹⁹³ Pro-DRV descriptive and injunctive norms (i.e., those supportive of DRV) were associated with increased DRV risk in the vast majority of included studies, including longitudinally.¹⁹³ Evidence suggests these relationships can, but don't always, vary by sex.^{76–78,128} Gender norms, which "distinguish expected behaviour on the basis of gender", ^{97(p4)} are a particularly powerful type of social norm.⁹⁴ The few quantitative studies examining associations between gender norms and DRV have mostly focused on norms concerning gender-based violence (GBV), missing broader gendered expectations underpinning DRV.¹⁹³ Nonetheless, the limited research available suggests that social norms supporting male-perpetrated GBV are associated with girls' DRV victimisation^{136,241} and boys' perpetration.¹³⁶ The two studies examining broader gender norms reported that gender-inequitable sexual and household norms are associated with increased DRV.^{133,135}

Qualitative research offers further insights into relationships between gender norms and DRV. For example, Marston and King's global systematic review of qualitative research on factors shaping young people's sexual behaviour found that social expectations of female chastity can hinder young women's expression of sexual desire, undermining clear communication about sexual consent.¹¹⁷ Considering UK research, in a context where boys are valourised but girls are stigmatised for sexual activity¹³⁸ girls report that resisting sex can precipitate physical DRV and boys report that girls who are perceived to be sexually experienced are considered more acceptable targets of sexual pressure.³⁴ Their accounts further suggest that norms linking boys' social status to their partners' faithfulness can drive jealousy and controlling behaviours;¹³⁸ while for girls, leaving an abusive relationship could result in loss of social status.³⁴

In contrast to social norms, which hinge on social expectations, personal attitudes are preferences or judgements that are motivated internally.^{101,112} While attitudes and norms are thought to influence each other reciprocally,¹⁰⁰ theorists posit that social norms can also

influence behaviour directly and in some cases override the influence of personal attitudes.^{101,104} Offering empirical support, several studies find that DRV^{76–78,82,83,122,123,127} and gender^{133,135} norms remain associated with DRV in models adjusted for personal attitudes, including (for descriptive DRV norms) longitudinally.^{77,82} Considering the relative strength of attitudinal and normative influences, this appears to vary, as behavioural theory would suggest:¹⁰⁰ some studies find that social norms are more strongly associated with DRV than are attitudes^{78,83,127,130} while in other studies it is attitudes that show a stronger relationship to DRV.^{76,127,131,133,135} There is little evidence, however, to suggest which norms are most important for which DRV outcomes or among whom. Few studies have modelled attitudes and norms together across DRV types^{76,132} or for both perpetration and victimisation.^{76,78} According to our review, no studies share norms measures, limiting comparability; and none have included distinct measures of descriptive and injunctive norms, or DRV and gender norms, in the same DRV models.¹⁹³

While DRV interventions often seek to harness the "power of the peer group"^{8(p126)} to reduce DRV, a limited evidence base offers little guidance in terms of which norms are most important and how they might work together, alongside attitudes, to shape DRV outcomes.⁸² Furthermore, though systematic reviews find that DRV interventions can change personal attitudes,^{22,30} evaluations of DRV interventions rarely measure social norms directly.^{11,193} Those that have offer emerging evidence that interventions can shift norms,^{11,149} though none to date have assessed the role of social norms in mediating impact on DRV outcomes.^{11,193} Both observational and intervention studies are hampered by a lack of established, validated social norms measures.^{11,193}

Social norms measurement in DRV research

Our 2023 review of DRV research identified 40 measures of norms concerning DRV and gender. It assessed their quality, including the extent to which their assessments were consistent (reliability) and measured the intended constructs (validity).^{187,193} Tests of a scale's reliability and validity assume that the scale is unidimensional, i.e., that its items tap a single domain. Where they are multidimensional – i.e., responses to items cluster in a way that

indicates the presence of subscales each tapping a different domain – reliability and validity tests should be carried out on emergent subscales.¹⁸⁷ However, our review found evidence of an analysis to confirm unidimensionality for only one included measure. Setting this limitation aside, most included measures did show evidence of an association with one or more DRV outcome, offering some support for their validity. Indicators of reliability, however, were mixed.¹⁹³ And though research suggests that adolescents might have difficulty responding to questions about the views of others,^{158,159} fewer than a quarter of included measures were reported to have been informed by youth engagement. Finally, no measure had been used in more than one study, reflecting a disjointed and unconsolidated body of literature.¹⁹³

The present study

Project Respect was a DRV prevention programme piloted in England.³⁷ Informed by the programme's theory of change, the pilot trial tested three measures of social norms: a measure of descriptive DRV norms was adapted from an existing scale²¹⁶ and new measures of injunctive DRV and gender norms were developed based on existing measures of personal attitudes.^{77,217} Original measures had previously demonstrated reliability that was minimally acceptable¹⁸⁷ or better among adolescents. Adapted and new measures were cognitively tested with young people in England and refined before piloting.²¹² In the present study, we first assessed the performance of piloted measures in terms of their reliability and validity. We then conducted exploratory analyses of associations between the validated social norms measures and DRV, examining whether associations persisted after adjusting for the other norms measures, attitudes and sociodemographic characteristics in multivariable analyses. This study addresses two research questions (RQs):

- 1) Are new and adapted measures of social norms concerning DRV and gender reliable and valid when used in cross-sectional research with adolescents in England?
- 2) Are social norms concerning DRV and gender associated with DRV after adjusting for each other, personal attitudes and sociodemographic factors?

In the discussion we reflect on these findings to consider what refinements of adapted measures, if any, are indicated.

Methods

Overview

Project Respect was a pilot cluster randomised controlled trial to assess the feasibility and acceptability of a DRV prevention programme and trial methods for its evaluation in secondary schools in England, including the performance of survey measures.³⁷ Recruited via email and telephone calls, eligible schools comprised state secondary schools in southern England excluding pupil referral units and schools exclusively for students with learning disabilities. Of schools expressing interest, we selected three in south-east England and three in south-west England, varying by school value-added attainment and local deprivation.³⁷ Before random allocation, students in years 8 and 9 (aged 12-14 years) from five of the selected schools took part in baseline surveys. The present study draws on the resulting attitudes, norms, DRV outcomes and sociodemographic data to assess the performance of the piloted social norms measures. Further information on trial methods¹ and findings^{2,37} are published elsewhere.

Fieldwork and sample

Baseline surveys were conducted in participating schools from June to July 2017. All year-8 and year-9 students were eligible except students whom school staff judged unable to assent to take part. Students and their parents/carers received information sheets one week before data collection, including information on how to opt out. On the day of data collection, a trained fieldworker orally described the study and students had the opportunity to ask questions before completing assent forms. Surveys were completed anonymously on electronic tablets. Sessions took place during a lesson period and typically lasted approximately 45-60 minutes. Eligible students absent during data collection received assent forms and paper surveys to complete and return by post. The London School of Hygiene & Tropical Medicine (11986) and NSPCC (R/17/106) Ethics Committees approved this research.

Measures

One social norms measure assessing descriptive DRV norms was adapted from an existing measure.²¹⁶ One new measure assessing injunctive DRV norms and one assessing injunctive gender norms were developed based on existing measures of DRV and gender attitudes, respectively.^{77,217} Tables 1 and 2 present survey measures of descriptive and injunctive norms, respectively, while the original and adapted norms measures are shown in Appendix A. No measure of descriptive gender norms was included as we did not identify a suitable existing measure nor did we find theory or evidence suggesting which descriptive gender norms are likely to be associated with DRV.

Our analyses also drew on measures of personal attitudes, sociodemographic characteristics and DRV outcomes. Appendix A presents attitudes measures and documents adaptations to the original measures on which they are based.^{77,217} Social norms and attitude measures were refined through cognitive testing and pre-piloted with one year-9 class (N=25), as reported elsewhere,^{37,212} before piloting in baseline surveys. Sociodemographic measures and DRV outcome measures are presented in Appendix B. The tablets used for survey administration required a response to each item before moving on to the next item. Given their sensitivity, items assessing sexual and gender identity included a "prefer not to answer" option which enabled participants to skip these items.

Descriptive DRV norms

Descriptive DRV norms refer to perceived DRV behaviours among a reference group. Evaluators of a US-based DRV and sexual violence prevention programme report a measure of descriptive DRV norms used with US secondary school students (α =0.70).²¹⁶ Our adaptations for this three-item measure (see Appendix A), reported in detail elsewhere,²¹² included removing the item on sexual DRV and simplifying response options from the number of friends to "none", "some", "many", or "most", reflecting a simplified set of response options used by social norms experts.¹⁰⁴ Cognitive testing informed the following additional adaptations for use with young people in England:²¹² adding a routing question to exclude respondents who do not have friends with partners; adding instructions to "show your best guess"; and splitting one item assessing two types of psychological DRV into two separate items.²¹² Responses were scored 1-4 and summed to compute a total score; a higher scores represented higher levels of perceived DRV among friends.

Injunctive DRV norms

As no suitable scales assessing injunctive DRV norms (perceptions of social expectations relating to DRV) were identified, we developed a new measure by drawing on an eight-item existing measure of DRV attitudes. The original measure was used with 8th- and 9th-grade students (typically aged 13-15 years) in a trial of the Safe Dates DRV intervention (α =0.69) (see Appendix A), which had informed the development of Project Respect.^{1,77} Informed by cognitive interviews,²¹² adaptations: asked about respondents' friends' rather than their own views; reduced the number of response options from four to three; added a neutral response option; reiterated the reference group within the response options (e.g., changing "Agree" to "My friends would agree"); and removed three items that specified a rationale for DRV. Our measure's sixth item was adapted from the Safe Dates trial's "perceived negative sanctions" scale to assess friends' rather than respondents' attitudes.⁷⁷ We reverse-worded three items to avoid presenting participants with a barrage of pro-DRV statements. Responses to each item were scored 1-3. To compute a total score, responses to reverse-worded items were reverse-scored and all responses were summed; a higher value represented more perceived support for DRV among friends.

Injunctive gender norms

No suitable existing measure of injunctive gender norms (perceptions of social expectations relating to gender) used in DRV research among adolescents was identified. The piloted five-item measure was developed by adapting an existing measure of attitudes towards gender roles and stereotypes used in DRV and gender-based violence research with young people aged 15-20 years in Greece (α =0.82) (see Appendix A).²¹⁷ Informed by cognitive interviews,²¹² adaptations: asked respondents about their friends' rather than their own views; simplified language; and aligned response options to the format of injunctive DRV norms items.

We shortened the measure by selecting five items judged as most salient among young people in the UK, drawing on cognitive interviews and retaining items assessing different aspects of the domain.²¹² We reverse-worded two items to avoid presenting participants with a barrage of sexist statements. Responses were scored 1-3. To compute a total score, responses to reverseworded items were reverse-scored and all responses were summed; a higher value represented more perceived gender-inequitable views among friends.

DRV outcomes

Measures of DRV victimisation and perpetration were used in analyses testing the validity of piloted norms measures and exploring associations between norms and DRV after adjusting for covariates. DRV was assessed via slightly adapted versions of the Safe Dates (58 items)¹⁶⁷ and short Conflict in Adolescent Dating Relationships Inventory ("CADRI-s", 18 items) scales,²²² the two most commonly used DRV measures developed for adolescents.²³⁰ While the Safe Dates measure is more sensitive, the CADRI-s is more established and potentially more suitable in evaluation studies because it is shorter.³⁷ Informed by cognitive testing, minor adaptations to the original measures were made to clarify wording and to improve flow and appropriateness for adolescents in England. To improve answerability, the Safe Dates measure was also adapted to offer uniform response options. The CADRI-s was also adapted to add controlling behaviours; and to capture both online and offline abuse within or outside the context of a conflict from all partners within the reporting period. Sexual abuse items were excluded from both measures. Adaptations, cognitive testing and pre-piloting of DRV measures among adolescents in England are reported in full elsewhere.³⁷ Final measures are presented in Appendix B.

Safe Dates items were asked of participants who reported ever having had a girlfriend/boyfriend or gone out with or dated someone. The measure comprises six subscales assessing overall victimisation, overall perpetration, and physical and psychological victimisation and perpetration. CADRI-s items were asked of the smaller subsample of participants reporting a serious or casual girlfriend and/or boyfriend in the last 12 months. Items from its two subscales, victimisation and perpetration, tap physical and psychological

DRV. DRV outcome measures' interitem reliability ranged from ordinal alpha=0.77-0.94 in Project Respect baseline surveys.^{bb,37,187} For each measure, four response options ranged from "Never" to "Often", scored from 1 to 4 such that a higher score represented more DRV. Items were summed for each scale and subscale to compute total scores.

DRV and gender attitudes

We used measures of attitudes towards DRV (five items) and gender (five items) in analyses exploring associations between norms and DRV behaviour adjusting for individual attitudes and socio-demographics. Measures were adapted from the same original attitude measures as were the parallel injunctive norms measures (see Appendix A).^{77,217} Like the original measures, our attitude measures asked for respondents' own views. Response options were adapted to reinforce this perspective (e.g., "I agree") based on findings of cognitive interviews.²¹² Changes to wording and item selection mirrored the changes made for parallel social norms measures. As reported elsewhere, DRV and gender attitudes measures achieved respective ordinal alphas of 0.65 and 0.68 in Project Respect baseline surveys.³⁷ Response options were scored 1-4 such that a higher score represented more pro-DRV or genderinequitable views. Mean response scores were computed for each measure, representing item response scores averaged across all items in the measure for all participants who responded to the full measure.

Sociodemographic variables

Mean response scores of norms measures were compared by sex, and validation analyses explored moderation by sex. Sociodemographic variables, including sex where models were not stratified by sex, were included as covariates in analyses examining associations between norms and DRV. Survey questions for these variables are reported in Appendix B. Age in years was a continuous variable. Socioeconomic status was a continuous variable assessed via a slightly modified version of the Family Affluence Scale II; responses were summed for a

^{bb} Ordinal alpha is a less biased estimate of internal consistency for Likert data than the more typically reported Cronbach's alpha.¹⁸⁷

score of 0 to 9 with a higher score indicating higher affluence.²²⁶ All other sociodemographic variables were categorical: sex assigned at birth, gender identity, year group, ethnicity, sexual identity and religion. Ethnicity was categorised as White or ethnically minoritised. Responses to items assessing sex assigned at birth, gender identity and sexual identity were used to construct a "sexual and/or gender minority" (SGM) variable. Participants were classified as a SGM if they: reported a sexuality of gay or lesbian, bisexual, other, or unsure/questioning; reported a gender identity of non-binary, unsure/questioning, or other; and/or reported a gender identity of female if they were assigned male at birth or of male if they were assigned female at birth.

Response options for participants identifying as White but not White British did not distinguish between ethnically minoritised White groups (e.g., White Roma, Gypsy or Irish Traveller) and other White non-British groups. A binary ethnicity variable was constructed, categorising participants as White (those selecting White British or Any other White background) or ethnically minoritised (those selecting Asian or Asian British; Black, African, Caribbean or Black British; Mixed/multiple ethnic background; or any other ethnic group). For use in a sensitivity analysis, an alternative binary ethnicity variable was constructed which categorised participants as White British or (for those selecting any other response option) Other ethnic group.

Statistical analysis

Handling of missing data

Because tablets on which the survey was administered moved participants systematically through survey items, data could be missing if a participant stopped the survey before reaching the item or if they selected "prefer not to answer" for gender or sexual identity items. We handled missing data by listwise deletion where missing data were not expected to affect results, for example, where less than 10% of a variable's data were missing.²⁴²

Sample and item characteristics

Analyses of descriptive DRV norms drew on data from participants who reported having friends with girlfriends/boyfriends ("descriptive norms sample"); other analyses drew on the

full sample. Following examination of each variable's distribution, we conducted descriptive analyses to characterise our two analytic samples and explore DRV prevalence by group for categorical sociodemographic variables. Relationships between continuous sociodemographic and attitude variables and DRV outcomes were assessed using univariable linear regressions. Descriptive analyses assessed completion rates and distributions of individual norms items, including mean response scores for each item across all participants who responded to the item.

Addressing RQ1: Are new and adapted measures of social norms concerning DRV and gender reliable and valid when used in cross-sectional research with adolescents in England?

Assessing factor structure and reliability. Our first step was to assess the latent factor structure among the 14 tested norms items. This refers to the pattern of clustering that indicates which items tap a shared, single underlying construct. In psychometric testing, items loading onto each latent factor are grouped together and treated as a unidimensional measure which can then be subject to reliability and validity testing. Assessing factor structure allowed us to explore whether the tested items loaded onto the three distinct factors as theorised (descriptive DRV norms, injunctive DRV norms and injunctive gender norms) or whether a different factor structure emerged from our data.

Parallel analysis on the polychoric correlation matrix was used to identify the number of latent factors among the 14 tested items. Findings were used to determine the number of factors to extract in exploratory factor analysis (EFA) with oblique rotation, which was conducted to obtain factor loadings for each item. As indicators of how similar each item is to each latent factor, factor loadings were used to determined which norms items should be grouped into distinct measures for subsequent psychometric testing. We conducted confirmatory factor analysis (CFA) to statistically assess how well the resulting factor structure fit our data using three estimates: root mean square error of approximation (RMSEA), which assesses how close the factor structure is to a perfect model of the data, and comparative fit index (CFI) and Tucker-Lewis Index (TLI) estimates, which compare the factor structure to a "baseline model" representing the worst possible fit.^{243(p409)}

We examined the reliability of each resulting measure by calculating internal consistency, a test of the measure's items' covariance, which indicates the extent to which these items tap the same underlying construct. We assessed ordinal alphas according to Devellis's criteria for unacceptable, undesirable, minimally acceptable, respectable or very good reliability.¹⁸⁷ We then examined whether removing any single item would improve reliability. Measures resulting from factor analysis and with minimally acceptable or better internal consistency (ordinal alpha>=0.65)¹⁸⁷ were retained for subsequent analyses.

Exploring statistical properties. We first examined the distribution of each retained measure's total and mean response scores to assess sensitivity to different levels of social norms. When a measure can discriminate between different levels of its underlying construct its scores will vary, which allows the opportunity to covary and correlate with other measures.¹⁸⁷ We compared mean social norms response scores by sex using t-tests that accounted for clustering within schools. We did not formulate a hypothesis about this relationship because findings from existing research are sparse and mixed: studies variably show girls reporting higher levels than boys of pro-DRV descriptive¹²⁰ and gender-equitable injunctive norms,¹³³ boys reporting higher levels of pro-DRV injunctive norms^{127,132} and no gender difference in injunctive DRV norms.^{78,131}

Assessing validity. Tests of construct validity examined each measure's relationship to theoretically associated measures¹⁸⁷ by testing three *a priori* hypothesis: (1) pro-DRV and gender-inequitable norms will be associated with Safe Dates measures of DRV victimisation and perpetration in univariable analyses; (2) pro-DRV and gender-inequitable norms will be associated with CADRI-s measures of DRV victimisation and perpetration in univariable analyses; and (3) Sex will moderate relationships between DRV/gender norms and DRV outcomes. Given mixed findings in a limited existing evidence base,^{76–78,128} this third hypothesis did not specify how many or which norm-DRV relationships would be moderated by sex. Tests were conducted using linear regressions that included mean social norms measure response scores as their independent variables and DRV victimisation and perpetration scores as their outcome variables.

We assessed the normality of the distribution of the outcome variables to inform our regressions. Finding skewed distributions of DRV scores, we examined residual plots for a sample of our univariable regressions. Finding non-normal distributions of residuals in these models, we bootstrapped these and subsequent regressions to improve the resulting estimates. We then ran bootstrapped linear regressions, accounting for clustering within schools by using robust cluster standard errors, to test our three *a priori* hypotheses. Appendix C provides histograms of DRV outcome scores, residual plots for a sample of univariable regressions and examples of the code used to account for clustering in non-bootstrapped and bootstrapped regressions.

Following unadjusted univariable regression analyses, sex and then sex*social norms interaction terms were added step-wise to each DRV outcome model. Social norms measures were considered significantly correlated with DRV outcomes where p<0.10 for comparability with subsequent multivariable regressions (discussed in the next section).

Addressing RQ2: Are social norms concerning DRV and gender associated with DRV after adjusting for each other, personal attitudes and sociodemographic factors?

Linear regression with addition of covariates. We built regression models for each DRV outcome to address our second research question, exploring the association between mean social norms (mean measure response scores) and DRV (victimisation and perpetration scores), adjusting for covariates. Given this early stage of this research, these analyses were considered exploratory and the threshold for significance set at p<0.10. Where sex moderated the relationship between the norms measure and DRV outcome assessed, models were sex-stratified. To determine which covariates to include in each model, we first ran bootstrapped univariable linear regressions to examine unadjusted relationships between potential covariates (sex, age, SGM, ethnicity and attitude variables) and each DRV outcome. We then ran similar regressions to examine unadjusted relationships between potential covariates and each social norm measure. The latter set of regressions also assessed relationships between

social norms measures.^{cc} In each model variables associated with both the assessed social norms measure and the DRV outcome were retained. The threshold for statistical significance was set at p<0.10 in order to avoid discarding potential covariates prematurely.⁷⁷ We then ran bootstrapped multivariable linear regressions to examine relationships between each social norms measure and each DRV outcome, adjusting each model for its retained covariates. All regressions accounted for clustering within schools by using robust cluster standard errors. Analyses were conducted using Stata/SE 18.0.²⁴⁴

Sensitivity analysis. We conducted a sensitivity analysis using the alternative construction of the ethnicity variable to assess whether categorising participants who selected "Any other White background" with other participants who did not select "White British" changed the results of the multivariable regressions.

Results

Sample and item characteristics

Of the 1,728 students invited, 82.5% (N=1,426) took part in the Project Respect baseline survey. The proportion of missing data was greater than 10% for religion and family affluence variables, both used only to describe the sample. The proportion of missing data was minimal (7% or lower) for the remaining variables, including those used for psychometric testing (Table 3) and therefore not expected to bias results.²⁴² Missing data were therefore handled using listwise deletion.

Characteristics of the participating full (N=1,426) and descriptive DRV norms (N=917) samples were similar (see Table 3): in both, just under half of participants were female and nearly 30% were categorised as SGM with a mean age of 13.4 years. Respectively by sample, 59% and 66% identified as White and 40% and 46% reported having no religious affiliation. Appendix D shows DRV prevalence among eligible samples overall and by participant characteristics for categorical independent variables. Overall DRV victimisation rates were 78%

^{cc} To assess the relationship between social norms measures and DRV outcomes for the purpose of building the multivariable models, we referred to the results of the previous univariable regressions conducted as part of testing the validity of the social norms measures.

and 73% according to the Safe Dates and CADRI-s measures, respectively. Safe Dates and CADRI-s DRV perpetration rates were 66% and 62%, respectively.

Norms item completion rates were high (92%-100%) and participants used all response options for all descriptive (see Table 1) and injunctive (see Table 2) norms items, but mean response scores for most items fell closer the anti-DRV and gender-equitable ends of scale ranges.

[Insert Tables 1, 2 and 3 around here]

Results for RQ1: Are new and adapted measures of social norms concerning DRV and gender reliable and valid when used in cross-sectional research with adolescents in England? *Assessing factor structure and reliability*

Parallel analysis indicated the need to extract four latent factors in EFA. As shown in Table 4, all items had a clear primary loading, with higher loadings (0.50 or greater) for factors 1, 2 and 3. Factor 1, "pro-DRV injunctive norms", comprised the three items developed to assess injunctive DRV norms and worded in support of DRV. Factor 2, "gender-inequitable injunctive norms", comprised the three items developed to assess injunctive gender norms and worded in opposition to gender equality. Factor 3, "pro-DRV descriptive norms", comprised all three items developed to assess descriptive DRV norms. Factor 4, "gender-equitable/anti-DRV injunctive norms", was less conceptually consistent: it comprised three items developed as injunctive DRV norms and two as injunctive gender norms, all reverse-worded (i.e. supporting gender equality/opposing DRV). Estimates from CFA suggested good fit (RMSEA=0.030, CFI=0.950, TLI=0.936).²⁴⁵ Ordinal alphas were minimally acceptable for factors 1 (alpha=0.68), 2 (alpha=0.69) and 3 (alpha=0.65) and unacceptable for factor 4 (alpha=0.56).¹⁸⁷ Internal consistency did not improve by removing any items (see Table 4). We dropped the measure defined by factor 4 and conducted further analyses on the three retained measures.

[Insert Table 4 around here]
Exploring statistical properties

More than 15% of participants scored the lowest possible scores for pro-DRV descriptive norms, pro-DRV injunctive norms and gender-equitable injunctive norms, suggesting that these measures are not sensitive to moderate levels of the constructs they assess (Table 5). Norms differed by sex. In Table 6 we report the mean response scores for each tested measure by sex, showing that female participants reported higher levels of pro-DRV descriptive norms while males reported higher levels of pro-DRV and gender-inequitable injunctive norms. The p-values for the t-tests comparing mean response scores for each tested measure by sex are <0.10, shown in the last column, indicating that mean response score differences by sex are statistically significant and unlikely to have occurred by chance.

[Insert Tables 5 and 6 around here]

Assessing validity

Table 7 reports regression coefficient estimates for regressions assessing the validity of the tested social norms measures. These estimates represent the mean change in DRV score for each one-point increase in the mean response score of the social norms measure. Symbols indicate whether the relationship between the social norms measure and DRV outcome is significant at the level of the p-value specified in the table endnotes. Coefficients indicating relationships significant at the level of p<0.10 are presented in bold.

Tests of construct validity aligned with our hypotheses. For all three norms measures, pro-DRV and gender-inequitable norms were associated with higher levels of all DRV outcomes in univariable regressions (Table 7, first column of results for each social norms measure). These relationships persisted when controlling for sex (Table 7, second column of results for each social norms measure). With the addition of interaction terms for norms*sex in each model, all three norms measures lost their association with overall DRV perpetration (CADRI-s) but other norms-DRV associations persisted (Table 7, third column of results for each social norms measure). All interaction term coefficients for DRV perpetration (CADRI-s) were significant, indicating moderation by sex. Sex also moderated the relationship between descriptive DRV norms and four Safe Dates DRV outcomes: overall, psychological and physical perpetration, and psychological victimisation.

[Insert Table 7 around here]

Results for RQ2: Are social norms concerning DRV and gender associated with DRV after adjusting for each other, personal attitudes and sociodemographic factors? *Linear regression with addition of covariates*

The results of exploratory multivariable linear regressions of DRV outcomes are presented in Table 8. Where the norms-outcome relationship had been found to be moderated by sex (see Table 7) we report sex-stratified estimates. The results of univariable linear regressions used to determine which covariates to retain are reported in Appendix E and Appendix F.

Like Table 7, Table 8 reports regression coefficient estimates representing the mean change in DRV score for each one-point increase in the mean response score of the social norms measure. Grouped by norms measure, results are presented according to whether analyses were conducted within full (first column for each norms measure) or sex-stratified (second and third columns for each norms measure) samples. Symbols and the use of bold text indicate statistical significance.

When adjusting for other norms measures, personal attitudes and sociodemographic factors assessed as covariates, pro-DRV descriptive norms retained an association with higher levels of overall (Safe Dates and CADRI-s measures) and physical DRV victimisation in the full descriptive norms sample. In sex-stratified models, descriptive DRV norms were associated with psychological victimisation and perpetration among both females and males (with higher regression coefficients among females); and with overall (Safe Dates and CADRI-s measures) and physical perpetration among females. Pro-DRV injunctive norms did not retain an association with any DRV outcomes in adjusted models. Gender-inequitable injunctive norms retained an association with higher levels of overall (Safe Dates and CADRI-s measures) and psychological DRV victimisation but were no longer associated with DRV perpetration or physical victimisation.

Sensitivity analysis

The sensitivity analysis using the alternative construction of the ethnicity variable did not result in any changes to the pattern of significance of the relationships between the social norms measures and DRV outcomes. We report the full results of the sensitivity analysis in Appendix G.

[Insert Table 8 around here]

Discussion

Summary and interpretation

Research question 1: Are new and adapted measures of social norms concerning DRV and gender reliable and valid when used in cross-sectional research with adolescents in England?

Descriptive DRV norms and pro-DRV and gender-inequitable injunctive norms items loaded well onto three conceptually distinct factors. Reverse-worded injunctive norms items, however, loaded poorly and onto a single factor with unacceptable reliability¹⁸⁷ and lacking clear interpretability. Item loading onto this fourth factor meant that the injunctive DRV and injunctive gender norms scales were shorter than designed, which tends to compromise reliability. Nonetheless, measures defined by the first three factors demonstrated minimally acceptable reliability and were retained.¹⁸⁷ All three retained measures were associated with all tested DRV outcomes with some relationships moderated by sex, providing evidence in support of construct validity.

Our findings suggest that reverse-worded items do not contribute to reliable and valid measurement of social norms relating to DRV and gender, supporting findings in other fields that reverse-worded items tend to perform poorly¹⁸⁷ and load onto a separate, methods- rather than conceptually-defined factors.²⁴⁶

Research question 2: Are social norms concerning DRV and gender associated with DRV after adjusting for each other, personal attitudes and sociodemographic factors?

When adjusting for covariates, pro-DRV descriptive norms retained independent associations with higher levels of all eight DRV outcomes and tended to be more strongly associated with DRV among females than males, particularly for perpetration. Among males, pro-DRV descriptive norms were associated with psychological but no other type of DRV perpetration, reflecting previous attitude-adjusted analyses reporting correlation with psychological perpetration¹²³ and inconsistent relationships with other types of male DRV perpetration.^{77,82} Conversely, consistent and robust associations in the present study provide new evidence that descriptive DRV norms might play an important role in female perpetration, in contrast to previous findings.⁷⁷ This is the first study to our knowledge to explore descriptive DRV norms in DRV victimisation models accounting for personal attitudes and it provides new evidence of consistent and robust relationships between descriptive DRV norms and DRV victimisation. Building on a limited evidence base, our findings suggest that pro-DRV descriptive norms are independently associated with increased DRV and that this relationship might be particularly strong among females, particularly for perpetration.

Injunctive DRV norms showed no association with DRV outcomes in multivariable models. Previous research adjusting for personal attitudes reports an association between girls' victimisation^{76,78} and perpetration^{76,127} but mixed evidence among boys^{76,78,127} and overall.^{78,83,131,132} Considering injunctive gender norms, a large and global body of literature documents mechanisms through which patriarchal gender norms contribute to male-perpetrated violence against women and girls,^{88,89} including DRV.^{34,82,138,247} Within this framework we might expect gender-inequitable injunctive norms to correlate most strongly with male perpetration and female victimisation. The limited evidence from studies that account for the role of personal attitudes reports an association between gender-inequitable injunctive norms and female victimisation;¹³⁵ and with overall perpetration but not victimisation.¹³³ However, in our adjusted analyses these patterns differed: norms were associated with overall and psychological victimisation but no other DRV outcomes, with no sex differences in these patterns. Varied findings in this sparse body of literature highlight gaps in

understanding of the complex, gender-specific pathways between gender norms and violence.²⁴⁸

Though we found that norms themselves differed significantly by sex, patterns of associations between injunctive norms and DRV did not, suggesting that the social pressures we measured might function similarly in relationship to DRV across both groups. We did not find evidence of independent relationships between DRV injunctive norms and DRV outcomes or between injunctive gender norms and perpetration, suggesting a few possibilities. First, these relationships might be fully attenuated by the covariates retained in the model. Second, adjusted models might underestimate the relationship between social norms and DRV outcomes if, as theory suggests, ^{100,134} attitudes could mediate this relationship. A third possibility is that relationships between some norms and DRV outcomes are indirect, interacting with other norms¹⁰² or attitudes.⁸² Longitudinal research with adolescent boys provides some support for this possibility. Reyes et al. found that gender-inequitable attitudes predicted DRV perpetration only for those with supportive attitudes towards DRV, while descriptive DRV norms retained an independent association with DRV perpetration. The study found no interaction between gender attitudes and descriptive norms but did not measure injunctive norms. A fourth possibility is that norms measures did not detect underlying associations in adjusted analyses due to sample size or limitations or to limited sensitivity to different levels of their respective constructs.

Implications for future research

Evaluation of DRV interventions that aim to shift social norms relies on having valid, reliable measures that are sensitive to changes to modifiable social norms that contribute to or protect against DRV. Adding items and additional Likert scale response options to our measures would "increase opportunities"^{187(p123)} for score variability and therefore covariance among scale items, which provides the opportunity for increased reliability.¹⁸⁷ Floor effects²⁰² and skewed mean response scores observed for each measure suggest that survey items represent more pro-DRV/gender-inequitable norms than are held among friends of the average participant. Many of the original items included in the piloted measures were developed in the

1990s and earlier,^{77,216,217} and changes in gender role expectations over time might mean that these reflect ideas that have become less mainstream.⁸² Our findings suggest that piloted measures might perform better with the addition of items representing more moderate indicators of these constructs.¹⁸⁷ The addition of items about psychological DRV could be particularly appropriate in light of research finding that young people tend to oppose physical DRV but are less likely to identify psychological DRV as abuse.²⁴⁹ Measures that allow for more variance would also address the possibility that sensitivity to relationships between injunctive norms and DRV was limited due to measurement or sample size limitations.

Social norm theory¹⁰³ and empirical research^{77,124,127,241,250,251} suggest that social norms relating most closely to DRV outcomes tend to yield the strongest influence. Due to limited space in a long evaluation survey, our tested measures of DRV norms were gender-neutral and combined physical and psychological DRV (descriptive DRV norms), or combined items concerning female-and male-perpetrated physical DRV (injunctive DRV norms). Distinct, gender-specific measures focusing on psychological, physical or sexual DRV could be expected to more strongly and robustly correlate with outcomes matched on gender and DRV type,^{48,124} providing more granular evidence on norms underpinning specific DRV outcomes.

Considering the gender norms most closely linked to DRV outcomes, our measure of injunctive gender norms might also be usefully refined by focusing new items on sexual and romantic gendered expectations⁸² and on gender norms implicated in qualitative DRV research. A review by McCarthy et al. found that attitude measures that include items on male sexual entitlement were more consistently associated with IPV perpetration than were those addressing male control over finances or other generic measures of gender role attitudes.²⁵² Drawing on global and UK research, gender norms restricting girls' sexuality,¹¹⁷ prescribing their romantic and sexual passivity¹¹⁷ and pinning their social value to stable heterosexual partnership^{34,92,117,139} might be particularly salient. Among boys, norms prescribing toughness,^{92,138} sexual and romantic dominance⁹² and sexual promiscuity⁹² while threatening loss of social status if their girlfriend is unfaithful¹³⁸ can be considered proximal to DRV perpetration. Qualitative research with young people designed to explore the range of social norms influencing DRV would provide valuable insights for further refinement.

While lengthier measures might not be feasibly used for evaluation research, development of longer-form measures could offer a higher level of sensitivity valuable in epidemiological research. It may be that more detailed measures can be used where feasible and indicated while less sensitive yet valid and reliable measures will be useful where survey objectives require assessment of a broad battery of measures.²²²

Our findings offer three key insights to strengthen the measurement of social norms in DRV research:

- Descriptive DRV norms, pro-DRV injunctive norms and gender-inequitable injunctive norms comprise empirically distinct latent constructs.
- 2. As in other areas of health research,^{187,246} reverse-wording items in DRV and gender norms measures detracts from construct validity.
- Scale reliability and validity might be strengthened by (a) providing more than four response options and (b) where survey length allows, including more than three items tapping multiple domains of DRV and gender norms constructs.

Results of psychometric testing support the reliability and validity of three brief social norms measures and indicate potential refinements to further improve their salience and reliability. These measures should next be incorporated into evaluations of norms-based DRV interventions. These evaluations should assess changes to norms and whether these mediate impact on DRV.

Our analysis suggests that descriptive norms might be an important mediator of DRV outcomes, highlighting the urgent need for a measure of descriptive gender norms.¹⁹³ This gap limits researchers' ability to measure the extent to which young people see others in their reference group modelling non-traditional gendered behaviours that could theoretically offer protection against DRV. Qualitative research should explicitly explore gender-specific mechanisms through which gender norms shape DRV outcomes. Findings could also fill critical gaps in the evidence on the relationship between gender norms and DRV among sexual-minority young people, who despite disproportionate DRV risk¹⁶ have been neglected in research on social norms and DRV.¹⁹³

Longitudinal research is needed to further probe relationships between social norms and DRV, including temporal patterns and whether personal attitudes confound, mediate or moderate this relationship. This research should explore psychological, physical and sexual DRV victimisation and perpetration outcomes with well-powered samples, assess for moderation by sex and explore the relative influence of social norms by type and population. Assessing the relative contribution of distinct norms would help intervention developers pinpoint modifiable behavioural antecedents on which to focus.

Limitations

This study uses cross-sectional data which cannot be used to infer causality. Our sample size is more than adequate for examining the factor structure of piloted social norms items¹⁸⁷ but might be underpowered to detect significance in multivariable analyses, especially for CADRI-s outcomes and in sex-stratified models. Our multivariable analyses are exploratory, providing new evidence to inform the development and testing of hypotheses in future research. However, their results cannot be used to determine which factors are responsible for the observed attenuation of norms-outcome associations and regression coefficients for descriptive norms cannot be directly compared to those of injunctive norms due to differing numbers of response options for these two types of measures.

Conclusions

This study presents three brief, reliable and valid measures of distinct social norms associated with a broad range of DRV outcomes. These measures should be incorporated into evaluations of norms-based interventions to assess their sensitivity to change. Quality of these and future measures might be further improved by the addition of more nuanced response options, gender-specific items tapping more moderate DRV norms and items assessing relationship-specific gender norms. Qualitative research is needed to shed light on genderspecific mechanisms through which gender norms shape DRV outcomes.

This study also offers new evidence that descriptive and injunctive norms, and norms relating to DRV and to gender, each retain independent relationships with some DRV outcomes

when adjusting for personal attitudes and other normative factors. It highlights the importance of descriptive DRV norms across DRV outcomes, particularly for girls, and of gender-inequitable injunctive norms in DRV victimisation. Well-powered, quantitative and theory-driven longitudinal research is needed to examine social norms as potential predictors of subsequent DRV outcomes and possible interactions amongst normative and attitudinal constructs.

Acknowledgements

We gratefully acknowledge the students who took part in this research and the school staff who welcomed us into their schools and facilitated data collection. We would also like to thank Liz Allen, Matt Dodd and Joanna Sturgess and for their expert data management and analytic support.

Funding

This study was funded by the NIHR Public Health Research Board (project 15/3/09). The views expressed are those of the authors and not necessarily those of the NIHR or the Department of Health and Social Care. The funder was not involved in study design; collection, analysis, or interpretation of data; the writing of this article; or the decision to submit this article for publication.

	Table 1. Completion and	distribution of	piloted DRV	descriptive nor	ms items (N=947)
--	-------------------------	-----------------	-------------	-----------------	------------------

Item	15	Completion rate (%)	Range	Mean response score (SD) ^c	None (%)	Some (%)	Many (%)	Most (%)
Desc	criptive DRV norms ^{a, b}							
Plea frien	se tick a box to show your best guess of how many of your Ids have done the following:							
A1.	How many of your friends have used physical force, such as hitting, to solve fights with their girlfriend or boyfriend?	99.7	1-4	1.25 (0.59)	81.3	14.8	1.9	2.0
A2.	How many of your friends insult or swear at their girlfriend or boyfriend?	99.5	1-4	1.49 (0.68)	60.1	33.0	5.1	1.8
A3.	How many of your friends try to control everything their girlfriend or boyfriend does?	99.7	1-4	1.34 (0.64)	73.5	21.2	3.4	1.9

^a Items asked of participants answering "yes" to the yes/no routing question, "Do you have friends who have girlfriends or boyfriends?"

^b Scoring: 1=None, 2=Some, 3=Many, 4=Most

^c Mean response scores across all participants who responded to the item

DRV=dating and relationship violence

SD=standard deviation

Table 2. Completion and	distribution of piloted D	RV and gender injunctive no	rms items (N=1,426)
			· · · · ·

Item	S	Completion rate (%)	Range	Mean response score (SD) ^b	My friends would agree (%)	My friends would neither agree nor disagree (%)	My friends would disagree (%)
Injur	nctive DRV norms ^a						
Plea	se tick a box to show whether your friends would agree or						
disa	gree with each statement.						
B1.	It is NOT okay for a boy to hit his girlfriend if she did	94.9	1-3	2.38 (0.87)	64.4	9.3	26.3
	something to make him mad. (rev)						
B2.	Girls sometimes deserve to be hit by their boyfriends.	94.9	1-3	1.29 (0.58)	6.8	15.2	78.1
B3.	Boys sometimes deserve to be hit by their girlfriends.	94.9	1-3	1.50 (0.75)	15.3	19.2	65.5
B4.	It is okay for a boy to hit a girl if she hit him first.	94.9	1-3	1.52 (0.74)	14.6	23.0	62.5
B5.	It is NOT okay for a girl to hit a boy if he hit her first. (rev)	94.4	1-3	1.94 (0.87)	35.1	23.5	41.4
B6.	If someone hits their boyfriend or girlfriend, the boyfriend or	94.4	1-3	2.36 (0.79)	55.8	24.6	19.7
	girlfriend should break up with them. (rev)						
Injur	nctive gender norms ^b						
Plea	se tick a box to show whether your friends would agree or						
disa	gree with each statement.						
C1.	Swearing is worse for a girl than for a boy.	91.9	1-3	1.53 (0.073)	13.9	25.7	60.4
C2.	It is more acceptable for a boy to have a lot of sexual	91.9	1-3	1.45 (0.69)	11.8	21.5	66.8
	partners than for a girl.						
C3.	Most girls can't be trusted.	91.9	1-3	1.58 (0.78)	18.4	21.1	60.5
C4.	On average, girls are as smart as boys. (rev)	91.8	1-3	2.44 (0.81)	64.0	15.5	20.5
C5.	Girls should have the same freedom as boys. (rev)	91.9	1-3	2.77 (0.57)	84.1	8.5	7.4

^a Scoring: 3=My friends would agree, 2=My friends would neither agree nor disagree, 1=My friends would disagree

DRV=dating and relationship violence

^b Mean response scores across all participants who responded to the item

SD=standard deviation

rev=reverse-worded item (against DRV/pro-equality)

Table 3.	Participant	characteristics	and DRV	prevalence
----------	-------------	-----------------	---------	------------

Characteristics	Descriptive norms	Full sample
	sample	N=1,426
	N=917	
Sex, %		
Female	49.3	48.5
Male	50.7	51.5
Missing	0.0	0.1
Gender, %		
Female (including trans girl)	41.3	41.0
Male (including trans boy)	39.9	40.1
Non-binary	4.1	3.7
Unsure/questioning	5.2	4.8
Other	4.3	4.4
Prefer not to say	4.8	5.6
Missing	0.3	0.4
Age (years), mean (SD)	13.4 (0.63)	13.4 (0.63)
Missing, %	0	0.1
Year group, %		
Year 8	44.0	46.2
Year 9	56.1	53.7
Missing	0	0.1
Ethnicity, %		
White British	52.8	46.8
White other	12.8	12.1
Asian/Asian British	5.2	5.6
Black/Black British	11.0	10.5
Mixed ethnicity	8.4	5.5
Any other ethnic group	5.6	5.5
Missing	4.3	1.8
Sexual identity, %		
Heterosexual/straight	84.6	84.2
Gay or lesbian	2.5	2.7
Bisexual	6.1	5.1
Unsure/questioning	2.9	3.4
Other	2.3	2.1
Prefer not to say	1.3	2.2
Missing	0.2	0.4
Sexual or gender minority, %		
Yes	27.8	26.9
No	67.3	66.6
Missing	4.9	6.5
Religion, %		
None	45.6	39.9
Christian	24.9	22.8
Jewish	3.2	2.7
Muslim/Islam	9.9	10.7
Hindu	1.4	1.4
Buddhist	0.9	0.8
L		1

Characteristics	Descriptive norms sample N=917	Full sample N=1,426
Sikh	0.3	0.4
I don't know/not sure	6.5	6.4
Other religious group	2.6	3.0
Missing	4.7	12.1
Family affluence scale, mean (SD)	6.2 (1.6)	6.1 (1.6)
Missing, %	29.6	35.3
Ever dated – Safe Dates sample, %		
Yes	76.0	71.7
No	23.9	28.1
Missing	0.1	0.2
Girlfriend/boyfriend last 12 months – CADRI-s sample, %		
Yes	49.0	44.0
No	50.7	55.5
Missing	0.3	0.4

SD=Standard deviation

Table 4. Rotated factor loadings and ordinal alphas (last row of table) of tested DRV and gender items (N=917)

Bold indicates primary factor loadings

Item				Ordinal alpha		
		Factor 1	Factor 2	Factor 3	Factor 4	of primary
		Pro-DRV	Gender-	Pro-DRV	Gender-	factor if item
		injunctive	inequitable	descriptive	equitable/anti-	removed
		DRV norms	injunctive	norms	DRV injunctive	
			norms		norms	
Pro-D	RV injunctive norms	•	·	·	·	
B2.	Girls sometimes deserve to be hit	0.668	0.207	-0.019	0.065	0.49
	by their boyfriends.					
B3.	Boys sometimes deserve to be hit	0.675	0.018	0.079	0.006	0.62
	by their girlfriends.					
B4.	It is okay for a boy to hit a girl if	0.544	0.037	-0.019	0.013	0.53
	she hit him first.					
Gend	er-inequitable injunctive norms	•			·	
C1.	Swearing is worse for a girl than for	0.086	0.618	-0.043	-0.009	0.47
	a boy.					
C2.	It is more acceptable for a boy to	0.077	0.671	0.056	-0.056	0.53
	have a lot of sexual partners than					
	for a girl.					
C3.	Most girls can't be trusted.	0.109	0.506	-0.031	-0.115	0.49
Pro-D	RV descriptive norms					
A1.	How many of your friends have	-0.044	0.004	0.643	-0.058	0.49
	used physical force, such as hitting,					
	to solve fights with their girlfriend					
	or boyfriend?					
A2.	How many of your friends insult or	0.113	-0.047	0.566	0.008	0.50
	swear at their girlfriend or					
	boyfriend?					
A3.	How many of your friends try to	0.023	0.048	0.630	0.007	0.44
	control everything their girlfriend					
	or boyfriend does?					
Gend	er-equitable/anti-DRV injunctive norm	ıs				
B1.	It is NOT okay for a boy to hit his	-0.112	0.124	-0.023	0.476	0.43
	girlfriend if she did something to					
	make him mad. (rev)					
B5.	It is NOT okay for a girl to hit a boy	-0.129	0.223	0.072	0.385	0.42
	if he hit her first. (rev)					
B6.	If someone hits their boyfriend or	-0.048	0.030	-0.014	0.375	0.60
	girlfriend, the boyfriend or					
	girlfriend should break up with					
	them. (rev)					
C4.	On average, girls are as smart as	0.091	-0.167	0.097	0.535	0.49
	boys. (rev)					
C5.	Girls should have the same	0.125	-0.231	-0.134	0.576	0.44
	freedom as boys. (rev)					
Ordin	al alpha	0.68	0.69	0.65	0.56	

DRV=dating and relationship violence

rev=reverse-worded item (against DRV/pro-equality)

Table 5. Distribution of EFA-derived social norms measures

Scale	N	Possible mean response score range	Mean response score (SD) ^b	Possible score range	Score range	% lowest possible score	% highest possible score
Pro-DRV descriptive norms ^a	942	1-4	1.36 (0.46)	3-12	3-12	30.1	0.2
Pro-DRV injunctive norms	1353	1-3	1.44 (0.51)	3-9	3-9	41.6	2.5
Gender-inequitable injunctive norms	1310	1-3	1.52 (0.55)	3-9	3-9	34.7	3.4

^a Asked of participants answering "yes" to the yes/no routing question, "Do you have friends who have girlfriends or boyfriends?"

^b Item response scores averaged across all items in the measure for all participants who responded to the full measure

DRV=dating and relationship violence

SD=standard deviation

Table 6. Characteristics of EFA-derived social norms measures by sex

Scale	F	emale		Male	<i>p</i> -value
	N	Mean response	N	Mean response	
		score (SE) [®]		score (SE) [®]	
Pro-DRV descriptive norms ^a	459	1.41 (0.03)	482	1.31 (0.03)	p=0.035
Pro-DRV injunctive norms	655	1.37 (0.02)	698	1.50 (0.02)	p=0.002
Gender-inequitable injunctive norms	643	1.45 (0.05)	666	1.59 (0.05)	p=0.093

^a Asked of participants answering "yes" to the yes/no routing question, "Do you have friends who have girlfriends or boyfriends?"

^b Item response scores averaged across all items in the measure for all participants who responded to the full measure

DRV=dating and relationship violence

SE=standard error

Table 7. Unadjusted and adjusted regression coefficients showing relationships between social norms measures and DRV outcomes

	-					Ind	lependent varia	ble					
				Pro-D	RV descriptive r	norms ^a	Pro-D	Pro-DRV injunctive norms			Gender-inequitable injunctive norms		
				Unadjusted [95%Cl]	Adjusted for sex [95%Cl]	Adjusted for norms*sex interaction [95%CI]	Unadjusted [95%Cl]	Adjusted for sex [95%Cl]	Adjusted for norms*sex interaction [95%CI]	Unadjusted [95%Cl]	Adjusted for sex [95%Cl]	Adjusted for norms*sex interaction [95%CI]	
			Norm	7.69***	7.59***	6.44**	3.42***	3.67***	3.44***	3.71***	4.01***	3.00**	
				[4.86, 10.51]	[4.73, 10.44]	[2.22, 10.66]	[2.41, 4.44]	[2.63, 4.71]	[2.32, 4.56]	[1.68, 5.73]	[2.09, 5.92]	[0.91, 5.08]	
		÷	F sex		1.00**	-2.11		2.15**	1.34		2.32**	-1.42	
		٧i			[0.38, 1.62]	[-8.10, 3.89]		[0.76, 3.54]	[-1.39 <i>,</i> 4.08]		[0.86, 3.79]	[-5.49, 2.65]	
	_		Norm*sex			2.27			0.56			2.44	
	eral					[-2.12, 6.65]			[-1.75, 2.88]			[-1.08, 5.97]	
	Š		Norm	3.83**	3.80**	1.87*	1.58***	1.70***	1.47***	1.57**	1.73**	1.46**	
sure	•			[1.41, 6.25]	[1.41, 6.17]	[0.41, 3.34]	[0.82, 2.34]	[0.90, 2.50]	[0.84, 2.09]	[0.41, 2.73]	[0.55, 2.90]	[0.46, 2.46]	
		rp.	F sex		0.37^	-4.79*		1.15*	0.31		1.21*	0.22	
eas		Ре			[-0.01, 0.75]	[-8.78, -0.81]		[0.13, 2.17]	[-1.85, 2.46]		[0.14, 2.27]	[-1.32, 1.75]	
۳ ۲			Norm*sex			3.77*			0.59			0.64	
DRV						[0.68, 6.85]			[-0.89, 2.07]			[-0.97, 2.26]	
] Se			Norm	4.83***	4.73***	3.53**	2.06***	2.26***	2.20***	2.13***	2.36***	1.81**	
ate				[3.31, 6.35]	[3.19, 6.27]	[1.19, 5.88]	[1.38, 2.75]	[1.59, 2.93]	[1.35, 3.04]	[1.03, 3.23]	[1.38, 3.33]	[0.65, 2.98]	
e E		Ċ.	F sex		1.03***	-2.20		1.66**	1.46*		1.73**	-0.28	
Saf		ž			[0.61, 1.46]	[-5.86, 1.47]		[0.70, 2.62]	[0.06, 2.86]		[0.75, 2.71]	[-2.76, 2.20]	
			Norm*sex			2.36^			0.14			1.31	
	, Zch					[-0.29, 5.00]			[-1.31, 1.60]			[-0.68, 0.30]	
	Ps		Norm	2.36**	2.33**	1.12*	0.76**	0.83**	0.63***	0.87***	0.97***	0.81***	
				[0.86, 3.86]	[0.86, 3.80]	[0.25, 1.99]	[0.20, 1.31]	[0.25, 1.41]	[0.33, 0.94]	[0.44, 1.30]	[0.56, 1.37]	[0.38, 1.23]	
		erp	F sex		0.33**	-2.92**		0.67*	-0.03		0.71*	0.12	
		Pe			[0.10, 0.57]	[-5.06, -0.79]		[0.06, 1.28]	[-0.84, 0.78]		[0.11, 1.31]	[-0.70, 0.94]	
			Norm*sex			2.37**			0.49			0.39	
						[0.78, 3.97]			[-0.21, 1.19]			[-0.43, 1.21]	

					Ind	lependent varia	ble						
				Pro-D	RV descriptive r	normsª	Pro-E	Pro-DRV injunctive norms			Gender-inequitable injunctive norms		
				Unadjusted [95%Cl]	Adjusted for sex [95%Cl]	Adjusted for norms*sex interaction [95%CI]	Unadjusted [95%CI]	Adjusted for sex [95%Cl]	Adjusted for norms*sex interaction [95%CI]	Unadjusted [95%Cl]	Adjusted for sex [95%Cl]	Adjusted for norms*sex interaction [95%CI]	
			Norm	2.84***	2.85***	2.90**	1.35***	1.41***	1.25***	1.56**	1.63**	1.17*	
				[1.47, 4.22]	[1.46, 4.23]	[0.94, 4.87]	[0.98, 1.73]	[1.00, 1.82]	[0.73, 1.78]	[0.55, 2.57]	[0.61, 2.66]	[0.09, 2.24]	
		÷	F sex		-0.04	0.11		0.47*	-0.08		0.59*	-1.16	
		Vic			[-0.28, 0.20]	[-2.22, 2.44]		[0.02, 0.93]	[-1.85 <i>,</i> 1.70]		[0.04, 1.14]	[-3.26, 0.95]	
			Norm*sex			-0.11			0.38			1.13	
	ys.					[-1.84, 1.62]			[-0.86, 1.62]			[-0.57, 2.83]	
	Ph		Norm	1.46**	1.46**	0.75*	0.81***	0.86***	0.82**	0.69^	0.75^	0.66*	
				[0.54, 2.38]	[-0.55, 2.37]	[0.08, 1.43]	[0.57, 1.06]	[0.60, 1.12]	[0.31, 1.33]	[-0.08, 1.47]	[-0.04, 1.54]	[0.04, 1.29]	
		rp.	F sex		0.05	-1.85^		0.47*	0.34		0.47*	0.15	
		Ре			[-0.15 <i>,</i> 0.25]	[-3.84, 0.14]		[0.08, 0.86]	[-1.16, 1.83]		[0.03, 0.92]	[-0.77, 1.06]	
			Norm*sex			1.38^			0.09			0.21	
						[-0.21, 2.97]			[-0.88, 1.06]			[-0.64, 1.06]	
			Norm	2.60***	2.52***	2.75**	1.53***	1.59***	1.62**	1.88***	2.02***	1.82***	
				[1.71, 3.48]	[1.57, 3.47]	[0.91, 4.60]	[1.10, 1.97]	[1.15, 2.04]	[0.55, 2.69]	[1.32, 2.44]	[1.53, 2.51]	[0.99, 2.65]	
Ire		ct.	F sex		0.69*	1.31		0.78*	0.87		1.01**	0.25	
asu		٧i			[0.15, 1.23]	[-1.04, 3.65]		[0.10, 1.47]	[-2.13, 3.87]		[0.35, 1.67]	[-1.10, 1.61]	
me	=		Norm*sex			-0.45			-0.06			0.48	
2	eral					[-2.32, 1.42]			[-2.01, 1.89]			[-0.75, 1.71]	
D	ŇŎ		Norm	1.56***	1.48***	0.73	0.80*	0.88*	0.30	0.77*	0.92*	0.19	
R-5	-			[0.79, 2.33]	[0.80, 2.16]	[-0.38, 1.84]	[0.10, 1.50]	[0.14, 1.61]	[-0.36, 0.96]	[0.05, 1.50]	[0.22, 1.62]	[-0.30, 0.68]	
ADI		rp.	F sex		0.74**	-1.20		0.96*	-1.01		1.05**	-1.72**	
J		Pe			[0.25, 1.23]	[-3.00, 0.60]		[0.20, 1.72]	[-2.46, 0.43]		[0.27, 1.82]	[-3.01, -0.43]	
			Norm*sex			1.40*			1.36*			1.77**	
						[0.24, 2.56]			[0.33, 2.39]			[-0.63, 2.90]	

^a Asked of participants answering "yes" to the yes/no routing question, "Do you have friends who have girlfriends or boyfriends?"

^0.10, *p<0.05, **p<0.01, ***p<0.001

Cl=confidence interval; DRV=dating and relationship violence; F=female; Perp.=perpetration; Phys=physical; Psych=psychological; Vict.=victimisation

Table 8. Regression coefficients showing relationships between social norms measures and DRV outcomes, adjusted for covariates

			Independent variable								
			Pro-DRV descriptive norms ^a			Pro-DRV injunctive norms		Gender-inequitable injunctive norms			
			All ^{b,c} [95%Cl]	Females ^{d,e} [95%Cl]	Males ^{d,e} [95%CI]	All ^{b,c} [95%Cl]	Females ^{d,e} [95%CI]	Males ^{d,e} [95%Cl]	All ^{b,c} [95%Cl]	Females ^{d,e} [95%Cl]	Males ^{d,e} [95%Cl]
CADRI-s Safe Dates DRV measure DRV	Overall	٧	6.62 *** [3.65, 9.58]	N/A	N/A	-0.53 [-2.41, 1.35]	N/A	N/A	1.97* [0.08, 3.87]	N/A	N/A
		Р	N/A	5.64** [2.04, 9.24]	1.17 [-0.71, 3.05]	-0.82 [-2.30, 0.67]	N/A	N/A	0.34 [-1.23, 1.92]	N/A	N/A
	Psych.	٨	N/A	5.40 *** [3.54, 7.25]	3.04* [0.56, 5.53]	0.08 [-0.95, 1.11]	N/A	N/A	1.30 ** [0.41, 2.20]	N/A	N/A
		Р	N/A	3.57** [1.43, 5.71]	0.73^ [-0.11, 1.57]	-0.54 [-1.28, 0.20]	N/A	N/A	0.20 [-0.60, 1.00]	N/A	N/A
	Phys.	٨	2.49 ** [1.05, 3.93]	N/A	N/A	-0.60 [-1.52, 0.32]	N/A	N/A	0.62 [-0.44, 1.68]	N/A	N/A
		Р	N/A	2.25 ** [0.59, 3.90	0.45 [-0.48, 1.38]	-0.28 [-1.16, 0.61]	N/A	N/A	0.03 [-0.72, 0.78]	N/A	N/A
	Overall	>	1.94 *** [0.91, 2.97]	N/A	N/A	-0.02 [-0.80, 0.75]	N/A	N/A	0.89 *** [0.43, 1.36]	N/A	N/A
		Р	N/A	1.44 *** [0.94, 1.94]	0.62 [-0.45, 1.68]	N/A	-0.26 [-0.88, 0.35]	-0.22 [-0.80, 0.37]	N/A	0.58 [-0.59, 1.75]	-0.19 [-0.74, 0.36]

^a Asked of participants answering "yes" to the yes/no routing question, "Do you have friends who have girlfriends or boyfriends?"

^b Models control for other social norms measures, DRV attitudes, gender attitudes, sex, age, sexual/gender minority status and ethnicity where each is associated with the specified social norms measure and DRV outcome at p<0.10.

^c Number of observations with complete data, based on non-bootstrapped regressions, ranges from 623 to 695 for Safe Dates measure outcomes and 422 to 444 for CADRI-s measure outcomes.

^d Models control for other social norms measures, DRV attitudes, gender attitudes, age, sexual/gender minority status and ethnicity where each is associated with the specified social norms measure and DRV outcome at p<0.10.

^e Number of observations with complete data, based on non-bootstrapped regressions, ranges from 306 to 355 for Safe Dates measure outcomes and 205 to 229 for CADRI-s measure outcomes.

^p<0.10; *p<0.05; **p<0.01; ***p<0.001

CI=confidence interval; DRV=dating and relationship violence; P=perpetration; Phys.=physical; Psych.=psychological; V=victimisation

6.3. Follow-up to Paper 3: social norms measures refined on the basis of psychometric

testing

As reported in Paper 3, on the basis of psychometric testing three items were removed from the piloted injunctive DRV norms measure and two were removed from the piloted injunctive gender norms measure. The descriptive DRV norms measure remained unchanged. Table 9 shows the refined versions all three measures following psychometric testing.

Measure, response options,	Items			
scoring				
Descriptive DRV norms ^a	Please tick a box to show your best guess of how many of			
None (=1), Some (=2), Many	your friends have done the following:			
(=3), Most (=4)	1. How many of your friends have used physical force,			
	such as hitting, to solve fights with their girlfriend or bovfriend?			
	2. How many of your friends insult or swear at their			
	girlfriend or boyfriend?			
	3. How many of your friends try to control everything			
	their girlfriend or boyfriend does?			
Injunctive DRV norms	Please tick a box to show whether your friends would agree			
My friends would agree (=3),	or disagree with each statement:			
My friends would neither	a. Girls sometimes deserve to be hit by their boyfriends.			
agree nor disagree (=2), My	b. Boys sometimes deserve to be hit by their girlfriends.			
friends would disagree (=1)	c. It is okay for a boy to hit a girl if she hit him first.			
Injunctive gender norms	Please tick a box to show whether your friends would agree			
My friends would agree (=3),	or disagree with each statement.			
My friends would neither	a. Swearing is worse for a girl than for a boy.			
agree nor disagree (=2), My	b. It is more acceptable for a boy to have a lot of sexual			
friends would disagree (=1)	partners than for a girl.			
	c. Most girls can't be trusted.			

Table 9. Final social norms measures following psychometric testing

^a Items asked of participants answering "yes" to the yes/no routing question, "Do you have friends who have girlfriends or boyfriends?"

Chapter 7. Paper 4: Relationships between social norms and DRV in student, staff and parent/carer accounts

7.1. Introduction to Paper 4

Project Respect yielded rich data on social norms in schools, which (1) participants linked to DRV in their accounts, or (2) existing theoretical or empirical literature suggests could play a role in DRV. I conceptualised the analysis for Paper 4 as an opportunity to assess and improve the content validity of the final social norms measures presented in Paper 3. To do so, I aimed to explore relevant norms from participant accounts and assess whether these domains were covered in the measures, informing new items for further refinement.

As my analysis progressed I saw that the data we had collected offered insights into ways in which gendered practices in schools both were facilitated by, and also reinforced, prevailing gender-inequitable norms. While retaining findings on social norms and DRV as intended, I oriented the reporting of my findings to an integrative analysis of interactions between these norms and practices. In the Discussion section of my thesis (Chapter 8) I draw on the findings of Paper 4 to consider the content validity of the norms measures presented in Paper 3 and I reflect on implications for refining these measures.

Paper 4 refers to Appendices 1 and 2 showing semi-structured guides used for optimisation and process evaluation data collection, respectively. These tools are provided in Appendix 13 of this thesis.

7.2. Paper 4: Social norms relating to gender and dating and relationship violence in English secondary schools: exploring student, staff and parent/carer accounts



London School of Hygiene & Tropical Medicine Keppel Street, London WC1E 7HT

T: +44 (0)20 7299 4646 F: +44 (0)20 7299 4656 www.lshtm.ac.uk

RESEARCH PAPER COVER SHEET

Please note that a cover sheet must be completed for each research paper included within a thesis.

SECTION A - Student Details

Student ID Number	452277	Title	Ms.	
First Name(s)	Rebecca			
Surname/Family Name	Meiksin			
Thesis Title	Developing and refining social norms measures for the evaluation of a school-based intervention to reduce dating and relationship violence among adolescents in England			
Primary Supervisor	Professor Chris Bonell			

If the Research Paper has previously been published please complete Section B, if not please move to Section C.

SECTION B - Paper already published

Where was the work published?			
When was the work published?			
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion			
Have you retained the copyright for the work?*	Choose an item.	Was the work subject to academic peer review?	Choose an item.

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C - Prepared for publication, but not yet published

Where is the work intended to be published?	Social Science & Medicine
Please list the paper's authors in the	Rebecca Meiksin, Ruth Ponsford, Nambusi Kyegombe,
intended authorship order:	Chris Bonell

Improving health worldwide

www.lshtm.ac.uk

Stage of publication	Not yet submitted

SECTION D - Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	I conceptalised the research questions, conducted the analysis and led interpretation and writing. Chris Bonell designed the overall study and advised on analysis and interpretation. All authors helped interpret the findings and provided feedback on the manuscript.
---	---

SECTION E

Student Signature	Rebecca Meiksin
Date	15/03/2024

Supervisor Signature	Chris Bonell
Date	15/03/2024

Social norms relating to gender and dating and relationship violence in English secondary schools: exploring student, staff and parent/carer accounts

Authors: Rebecca Meiksin, Ruth Ponsford, Nambusi Kyegombe, Chris Bonell

Abstract

Dating and relationship violence (DRV), which refers to intimate partner violence among young people, is widespread. DRV is associated with subsequent mental ill health, substance use and sexual risk among girls and boys and is a leading cause of morbidity and mortality among girls globally. Harmful social norms are widely recognised for their role in sustaining DRV, and interventions often seek to change these. However, little evidence is available to suggest which specific norms are most salient and where protective norms might be strengthened. We conducted, audio-recorded and transcribed focus groups and semi-structured interviews with students (years 9 and 10), staff and parents/carers from ten secondary schools in England. We also audio-recorded discussions in staff DRV trainings in four of these schools. Data collection took place from April 2017 and July 2018. This research explored participant accounts of social norms relating to gender and to DRV in schools and their influence on DRV behaviours. Drawing on Giddens' structuration theory, our thematic analysis found that sexist social norms subjugating girls to boys facilitated gendered practices of harassment and abuse, including DRV; and that these practices, in turn, reproduced this gendered power structure. Our data suggest that while physical DRV is socially proscribed, norms supporting controlling behaviours and inhibiting disclosure of victimisation directly underpin DRV. They further suggest that indirectly, gender norms concerning cross-gender friendships; sexual harassment; the policing of girls' sexuality; homophobic abuse; and dominance, control and sexual activity as masculine ideals indirectly sustain DRV. Accounts demonstrated that students and staff challenge harmful norms, but that these efforts can be ineffective and socially punished. Our findings can inform DRV interventions, which should draw on evidence to foster protective norms and shift those that sustain DRV.

Introduction

Dating and relationship violence (DRV) refers to intimate partner violence (IPV) among young people¹⁶. Comprising psychological, physical and sexual violence by a current or former partner, this type of abuse is widespread^{21,25} and poses serious threats to young people's health and well-being.^{26,27,52} Globally, 24% of girls aged 15-19 years have experienced physical and/or sexual DRV.²⁵³ Though global prevalence data on boys are unavailable,²⁵⁴ a 2017 meta-analytic review estimates victimisations rates of 21% for physical DRV and 8% for sexual DRV among boys aged 13-18 years.²⁵ Estimates for psychological DRV victimisation vary widely but also tend to be high among both girls and boys across studies.^{21,22} In terms of its impacts, DRV is the fourth leading risk factor for mortality and the third for morbidity among girls aged 15-19 years globally.²⁷ In addition to causing injuries,¹⁶⁴ DRV is associated with subsequent mental ill health,^{26,52,55} substance use^{26,52} and sexual risk behaviour⁵⁴ among girls and boys as well as with IPV victimisation and perpetration in adulthood.⁵⁶

Public health researchers and practitioners are increasingly using social norms theory as a framework for considering modifiable social factors contributing to gender-based violence and adolescent health.^{6,15,150,255} Social norms theorists distinguish between descriptive norms (beliefs about what behaviour is typical in a reference group of important others) and injunctive norms (beliefs about what others in the reference group think constitutes appropriate behaviour).^{15,106} Though theorists disagree about the relationship between these two constructs,⁶ descriptive and injunctive norms are thought to play discrete roles in influencing behaviour^{6,102} and each type has been found to be associated with both experience^{76,78,121,122,125,126} and perpetration^{77,82,83,120–124,127} of DRV.

Theorists posit that social norms are primarily sustained by anticipation of social sanctioning by the reference group, including social rewards for complying with prevailing norms and social punishments for deviating from them.^{15,103} There is evidence to suggest that social norms relating directly to DRV and also more generally to gender expectations are associated with DRV outcomes. In a systematic review we conducted, the vast majority of studies found that young people who report descriptive or injunctive norms supportive of DRV are at increased risk for DRV involvement themselves.¹⁹³ Quantitative research on gender

norms tends to focus on injunctive norms and male-perpetrated DRV against girls, reporting significant associations between gender-inequitable norms and increased risk.¹⁹³ UK qualitative research provides insights into potential mechanisms of influence, suggesting that gendered expectations relating to sexual behaviour, heterosexual partnerships and masculinity contribute to perpetuating this type of violence.^{34,138,139}

DRV prevention interventions have long sought to address gender-inequitable norms and the social acceptability of DRV.^{8,32,256} Interventions often take place in schools, which offer a good foundation to foster norms change because schools are key sites of gender socialisation⁹¹ and because there is evidence that group discussions involving critical reflection can be effective in promoting more gender-equitable norms.^{96,97} However, evaluations of DRV interventions rarely measure impact on descriptive or injunctive norms and none, according to a recent systematic review, has assessed them as a mediator of DRV reductions.¹¹ Efforts to do so are limited by gaps in existing research. In addition to a lack of appropriate, established measures of social norms,^{11,193} there is little evidence to suggest which specific norms are the most important for DRV and which existing, protective norms might be strengthened. While review evidence suggests that many norms are shared across settings,¹¹⁷ answers to these questions are also likely to vary by context, at least in part.⁶. In the UK specifically, exploratory research suggests that social norms play an important role in enabling initiation and perpetuation of DRV.^{34,138,139} However, this research has not sought to explore the range of social norms underpinning DRV, distinguished between descriptive and injunctive norms or explored how each relates to each other, the school environment or DRV. In order to determine which DRV and gender norms operating in UK secondary schools should be prioritised for DRV prevention, there is a need for research informed by social norms theory and distinguishing between descriptive and injunctive norms. We aim to address this gap by exploring student, school staff and parent/carer accounts of (1) social norms relating to gender and to DRV in schools; and (2) how these social norms appear to influence practices of abuse and harassment, including DRV.

Informed by Jamal et al.'s research on girls' bullying in London secondary schools, our analysis draws on Giddens' structuration theory.¹⁰⁷ According to Giddens, structure consists of

"rules and resources" that, in addition to enabling and constraining social practices, are themselves reproduced or modified by those practices.^{107(p25)} Giddens suggests that people, while not always completely aware of all the conditions for or consequences of their actions, can deliberately seek to "keep things as they are" or transform the social structures in which they operate.^{107(p28)} We draw on this theory to examine social norms concerning gender and DRV as a structural feature of secondary schools, and the practices relating to gender and DRV which they enable and constrain.

Methods

The Project Respect study

Our data come from Project Respect, a study to optimise and pilot the Project Respect DRV intervention in secondary schools in England. This pilot cluster RCT was guided by a study protocol registered on the ISRCTN registry (reference ISRCTN65324176) and included an integral process evaluation.¹ Underpinned by the Theory of Planned Behavior¹⁰⁰ and the Social Development Model,¹⁶⁹ Project Respect was a whole-school intervention aiming to promote changes to the school environment and among the school community to reduce DRV. Informed by the Safe Dates¹⁶⁴ and Shifting Boundaries¹⁴⁶ interventions, both effective in reducing DRV in the United States,^{147,165} Project Respect comprised expert-led training for key school staff, led by our delivery partner The National Society for the Prevention of Cruelty to Children (NSPCC); cascaded training to all school staff; school policy review; mapping of 'hotspots' for DRV and harassment and shifting of patrol patterns to address these; written information for parents/carers; and a classroom curriculum for students in years 9 and 10 (typically aged 13-15 years). The curriculum included the development of student-led campaigns and the opportunity to download an existing smartphone app²⁵⁷ to support help-seeking. Detailed information on the intervention and findings on its optimisation and piloting have been published elsewhere.^{2,37} The present study draws on qualitative data conducted in two waves of intervention optimisation sessions and for the process evaluation; and on audio-recordings of the expert-led trainings.

Sampling and recruitment

We recruited schools via emails and telephone calls to those eligible, which were mainstream state secondary schools in southern England. Of those expressing interest in participating in the pilot RCT we purposively selected six schools stratified by region (south-east/south-west England) and varying by deprivation and by value-added attainment, which is a school-level indicator of students' progress between school entry and their final secondary school exams.¹⁷³ One school withdrew from the study before baseline assessments and was replaced. After baseline assessments, pilot RCT schools were stratified by region and randomised 2:1 to the intervention or control arm. We selected four optimisation schools from among those expressing interest but not selected for the pilot RCT, stratified by region and varying by deprivation. Head teachers in both cases signed a consent form.

For student/staff intervention optimisation sessions, lead school contacts were each asked to recruit at least three girls and three boys from each of years 9 and 10, and three or more members of staff, prioritising: a school safeguarding lead; personal, social and health education staff; and senior leadership. Wave 1 participants also took part in wave 2 sessions where feasible. The expert-led training sessions were for school staff delivering Project Respect and for school senior leadership. For individual interviews in intervention schools, we aimed to recruit four staff, purposively sampled by seniority and programme involvement, and two parents/carers from intervention schools, purposively sampled by their child's year-group (years 9 and 10) and their child's gender. We aimed to recruit eight students from intervention schools and four from control schools, purposively sampled by year-group (years 9 and 10) and gender. Finally, for individual interviews in control schools we aimed to recruit two staff, purposively sampled by seniority.

The trainer and participants involved in the expert-led training, and those invited to take part in optimisation sessions and interviews, received a study information sheet prior to data collection which included information on local support resources and on how to opt out of the study if they wished. They had the opportunity to ask questions to a trained member of the research team before signing a study assent (for students) or consent (for adults) form.

Parents/carers of students invited to take part received an information sheet before data collection and had the opportunity to opt out their child. Interview participants also received a copy of the information sheet at the start of their interview. Information and assent/consent materials made clear that participation was voluntary and confidential, and that participants could stop taking part at any time.

Data collection

Optimisation sessions aimed to inform intervention content and format (wave 1) and to gather feedback on intervention materials and inform delivery (wave 2). Led by members of the intervention and evaluation teams, each session included an introductory slide presentation followed by discussions using a semi-structured guide (Appendix 1). The first session included a plenary discussion followed by separate staff and student group discussions. In subsequent sessions, we led discussions in three groups: year-9 students, year-10 students and staff. Facilitators took notes and wave-2 sessions were also audio-recorded and transcribed. Trainings were audio-recorded and included discussion questions for staff about signs of DRV, and about concerning behaviours in their schools that they would like to address through Project Respect. Researchers took notes on the audio-recordings of trainings.

Interviews were conducted by experienced researchers using semi-structured guides on which they were trained (Appendix 2). We primarily conducted individual interviews but also conducted student interviews in pairs or small groups when schools requested this. All interviews explored school context. Intervention school interviews also explored programme implementation, costs, receipt, mechanisms of change and impacts. Control school interviews also explored provision of violence prevention, relationship and sex education and social/emotional learning. The present analysis draws on interview data about school context, social norms and gender-based harassment and abuse. Data were recorded in interview notes and, where participants assented/consented, audio-recordings which were transcribed verbatim.

Data analysis

Notes and transcripts were loaded into the qualitative analysis software Nvivo 12 to aid analysis.²⁵⁸ Data were analysed using thematic analysis, complemented with techniques from grounded theory.¹⁷⁸ After reviewing and re-reviewing transcripts to gain familiarity with the data, RM led initial coding, and CB and RM reviewed and interpreted the coded data. Starting codes were informed by social norms theory and existing research on relationships between social norms and DRV, and these included: descriptive and injunctive gender norms, other descriptive and injunctive norms, reference groups and social sanctions. New codes were developed inductively as new themes emerged, with special attention to "deviant cases" that challenged or provided deeper insight into emerging themes.^{178(p294)} Axial coding built on initial coding and was used to explore relationships between codes, combining those with significant overlap and separating codes that represented distinct constructs, to arrive at a final coding scheme. Informed by Jamal et al.'s study of bullying in schools²⁵⁹ and by Giddens,¹⁰⁷ our analysis was sensitised to where norms constrained or enabled actions that made DRV more likely, to ways in which norms were reproduced by abusive practices and to where people appeared to exercise agency in reshaping these norms to transform existing structures. Informed by the notion of "constant comparison", ^{178(p290)} codes and transcripts were compared throughout the analysis process.

Where participant accounts directly or indirectly linked social norms to DRV, we drew this out in our analysis. In our interpretation, we distinguished between norms linked to DRV in participant accounts and norms emerging from participant accounts and for which other existing evidence supports a relationship with DRV. We explored norms in both intervention and control schools and we treated data from these in the same way.

Confidentiality and ethical review

To maximise retention in the study, control schools received a £500 payment. No individual participants received compensation for taking part. Information and assent/consent materials informed participants that if safeguarding concerns arose during data collection, then confidentiality would be removed to address safeguarding concerns. Data were stored in password-protected folders on the London School of Hygiene & Tropical Medicine's (LSHTM) secure servers and anonymised using study ID numbers. Further details on safeguarding and data management procedures have been reported previously.³⁷ This research was approved by the LSHTM (reference: 11986) and NSPCC (reference: R/17/106) Ethics Committees.

Results

Participation

We have previously reported on optimisation, fieldwork and participation in Project Respect.^{37,260} For the present study we analysed data from ten schools (four optimisation, four intervention and two control).

Optimisation sessions

Optimisation-session participants are described in Table 1. Four schools took part in the first wave of optimisation sessions (April 2017). Of these, one was unable to arrange an optimisation session and instead a member of staff took part by telephone. Participants were 31 students, mixed by gender and year-group, and nine staff. Three schools took part in the second wave (July 2017), which included 35 students and six staff, some of whom had also taken part in the first wave. Across waves, staff participants were a mix of teachers, support staff and senior leadership, and included a member of the safeguarding team in most schools.

Participants	Wave 1	Wave 2 [*]
Year-9 students		
Girls	8	11
Boys	7	12
Year-10 students		
Girls	9	6
Boys	7	6
Total students	31	35
Staff		
Female	7	4
Male	0	1
Not reported	2	1
Total staff	9	6

Table 1. Optimisation session participants

*Some Wave 2 participants had also taken part in Wave 1

Process evaluation – trainings and interviews

Expert-led training sessions were audio-recorded in all four intervention schools. Process evaluation interview participant characteristics are shown in Table 2. Forty students from six schools participated in interviews, a mixed sample of girls and boys in years 9 and 10. Student interviews were conducted primarily individually (N=9) or in pairs (11 interviews). Two were conducted in small groups of three and six. Individual interviews were conducted with 21 members of staff (16 female, five male) from six schools and with five parents/carers (all female) of a mix of year-9 and -10 students, predominantly girls, from three intervention schools. Staff participants were a mix of those responsible for student well-being initiatives, teachers and senior leadership and the sample included a member of the safeguarding team from most schools.

Participants	Intervention arm	Control arm	Total
Year-9 students			
Girls	11	2	13
Boys	10	2	12
Year-10 students			
Girls	5	2	7
Boys	6	2	8
Total students	32	8	40
Staff	17	4	21
Female	13	3	16
Male	4	1	5
Parents/carers	5	Not applicable	5

Table 2. Interview participants

Thematic analysis

Norms about how girls and boys should behave

Participant accounts identified a range of gender norms operating in secondary schools to shape girls' and boys' behaviours and their physical and personality attributes. These appeared to support a sexist, gendered hierarchy placing girls in an inferior position to boys. References to normative expectations and sanctions mainly centred on peer influences, while some participants also made references to parent and broader cultural influences on gender socialisation.

Sexist norms about gendered attributes. Participant accounts revealed a range of social expectations in schools governing girls' and boys' physical and personality attributes, which were more pronounced in some settings and populations than in others. These injunctive norms generally functioned to maintain girls' positions as inferior and subordinate to boys, while maintaining boys' position of superiority and dominance over girls, through systems of social rewards and punishments. As summed up by one girl, "Boys are in control and girls do what they say" (optimisation school A, wave 1).

Scrutiny and judgement of girls' looks was a common theme. Participants described social expectations of girls to wear make-up and attend to their physical appearance. At the same time, girls perceived to be too made-up could face social repercussions. As one year-9 girl described girls' confinement, "...if you're not styled up properly you're a tramp, if you're wearing too much make-up you're still a tramp, like there's no in between" (year-9 girls, intervention school D, IDs S1-S2). Injunctive norms governing boys' attributes, on the other hand, generally functioned to promote displays of emotional and physical strength and toughness. Accounts suggested that, while some gender norms were beginning to shift and not all young people supported them, boys nonetheless faced pressure to "prove their masculinity" (staff training, school F) act "tough" (staff, intervention school D, ID T3) and "act like a man" (staff training, school F). Boys were expected to hide emotional vulnerability and not to cry, and they could be teased for showing weakness, wearing make-up or displaying other ostensibly feminine characteristics.

Students also described gendered stereotypes endorsed by their peers, a form of descriptive norm. These included beliefs that girls dressed in feminine clothes, were physically weaker and less emotionally stable than boys and that they were not athletic or as smart as boys in specific subjects. Students reported that boys, on the other hand, were seen as "the stronger sex" (year-10 girl, control school G, ID S1), smarter than girls in some subjects, more athletic and less prone to sadness or agitation.

Sexist norms governing sexual behaviours. Scrutiny and judgement of girls' sexual behaviours were prevalent in participant accounts. Students and staff across several schools described injunctive norms supporting a sexual double-standard among students²²¹ involving social rewards for boys seen as sexually active but stigmatisation of girls seen this way. This was described in relation to both sexual/romantic activity and to the circulation of sexual images. In the case of sexual activity, boys were congratulated for what was seen as sexual or romantic success while girls were reprimanded or derided. Year-9 girls described the phenomenon this way (year-9 girls, intervention school D, IDs S1-S2):

Participant: Like when boys go, 'Oh we're dating five girls,' it's okay, and then if a girl could maybe be talking to two boys and a friend, they're automatically a slag^{dd} or something.

This policing of female behaviour could foster insecurity and limit girls' self-expression, as a year-9 girl described:

I think it lowers their self-esteem kind of, even if it doesn't make it visible or something, it does lower it. Like it is kind of sad because some girls will stop doing, like, stop being confident, maybe stop going to dance, just because they got called this or that, and then they just feel insecure and stuff. (Year-9 girls, intervention school D, IDs S1-S2)

Accounts of students in a few schools suggested that cross-gender friendships were uncommon and that norms governing sexual behaviour could play a role in proscribing them: peers would assume cross-gender friendships were romantic or sexual and if a girl had many male friends, a year-9 student explained that some people might see her as a "slut", "ho" or "sket"^{ee} (optimisation school B, SE, wave 2).

Resistance to sexist norms and gender stereotypes. Although social pressure could make it difficult, resistance to gendered expectations and stereotypes was common in student accounts across several schools. One manifestation was citing examples that challenged restrictive descriptive norms, as reported by this year-9 girl reflecting on a classroom discussion:

Boys have a stereotype of not being sad or not getting annoyed as easy as girls do. But then when we discuss that, the boys were like 'No, I've never cried in front of anyone, I've

^{dd} "Slag" is a derogatory term for a girl or woman who is seen as promiscuous.

^{ee} "Sket" is a derogatory term for a girl or woman who is seen as promiscuous.

never cried'... And then everyone will start saying 'But you probably have, you probably have been upset'. And, like, when it says girls are more agitated, we probably aren't, boys get annoyed when one of their football teams loses and like girls don't care about them things. And then they'll say about how we care about our makeup... but then we all care about something and like we all get the same amount of agitated and we all like cry, we all cry, we all get upset. And it's just common and it's not something that should be stereotyped because boys do get upset and girls do get agitated and it's like common. (year-9 girl, intervention school C, ID S7)

In participant accounts, resistance could also take the form of students disregarding gendered expectations, either in terms of their own behaviour or by not sanctioning others who transgress them.

Maintaining gender norms compliance through homophobic practices. Accounts suggest that enactments of homophobia were common in schools, were perpetuated by both girls and boys, and fell into three main categories. First, accounts indicate that direct bullying of gay and bisexual young people operated as a social sanction supporting an injunctive norm of heterosexuality. Second, homophobic comments appeared to operate as a social sanction sustaining sexist norms. For example, students in a paired interview described the use of homophobic insults to shape masculinity among boys (year-9 girl and boy, control school H, IDs S3-S4):

Participant: It's not as, because there's like the stereotype that girls used to be all fairies and pink and holding hands and that. But if you're a boy with fairies and pink and holding hands, it would be a lot more strange at that moment. Participant: You'd be gay. Yeah, you'd just be named gay.

These insults also appeared to play a role in regulating the boundaries of same-gender friendships, particularly among boys, e.g.:

Or if, let's say, two girls, say, like, really close friends, and they're always hugging, they'll like call them lesbian or 'You're a queer' or something like that. Like if you're holding hands with a girl, like, it's like... But I think boys, if they hold the hands of a boy, it would be more weirder than two girls holding hands. (year-9 girl and boy, control school H, IDs S3-S4)

Third, some students framed the use of homophobic language as an insult with, as one student put it, "no tag to sexuality at all" (year-9, mixed-gender group, intervention school E, IDs S3-S8). Minimising this type of behaviour, these participants framed homophobic comments

as "jokes" (year-9 boy, intervention school D, IDs S3-S4) or "banter" (year-9 girl and boy, intervention school F, IDs S4 and S8). As one student explained, "It's an insult but it's not like a horrible insult in their mind, it's just saying it because it's just what comes to mind" (year-9 girl and boy, intervention school F, IDs S4 and S8). In contrast, other students expressed frustration or disapproval, for example describing homophobic practices as "disgusting" (year-9 girls, intervention school D, IDs S1-S2) or suggesting that more should be done to address them:

Once, in one class, someone said, "That's so gay", and the teacher just went, "What exactly do you mean by that?", or something, and they had their little say about that and it was, but that was it, it was like nothing else. (year-10 girls, intervention school D, IDs S7-S8)

We found little evidence of significant negative sanctioning of homophobic comments. Accounts suggest that, where staff or students did intervene, this tended to be ineffective. Staff reactions, discussed primarily by students in one school but also by a staff-member in a second, were characterised as inadequate in that they failed to engage students in a meaningful or transformative way, e.g.: "That's how it feels. Like it's, like, they've been pulled over, 'You shouldn't say this, it's bad'. But like they don't actually care too much about it so they don't explain why" (year-9 girls, intervention school D, IDs S1-S2).

A girl who had challenged homophobic behaviour with peers in her school recounted being made a target herself, which functioned as an effective sanction: "Actually in year 7, this is a personal experience, I tried to step in and they accused me, and then they tried to push me down the stairs...from now on I just don't step in" (year-9 girls, intervention school D, IDs S1-S2).

Gendered expectations and abuse involving sexual images. Students and staff discussed the often persistent pressure on girls to send sexual images of themselves to boys and the often severe social consequences of doing so. In these accounts, expectations of girls' sexual availability and obedience were coupled with acute stigmatisation of girls whose images were subsequently posted publicly or circulated among classmates without their consent. Pressure on girls could come from partners and from other, sometimes older, boys. This could be overtly forceful or more subtle and protracted involving what one teacher described as "...a kind of gentle kind of tapping on the window type of effect over a period of time, where that
young person has just felt 'I need to do this...'" (staff, control school H, ID T2). A staff-member from one optimisation school demonstrated the significant influence that social norms with regard to sending sexual imagery could have:

Although we've done a lot of work in school on the legal implications of sending and asking for images and sharing images and all that kind of thing, they do it anyway. Just because everybody does it... It is predominantly the boys who are asking the girls for the images and the girls who are sending them because that's what they're supposed to do and that's what the boys like... (staff, optimisation school B, wave 2).

A year-9 student's account illustrated how pressure to send sexual images could operate both as an abusive practice in itself and as a social sanction reinforcing an expectation of girls' sexual availability: "Yeah, the boys just call you frigid and stuff, and they call you boring, 'Blah blah blah'. And then the girls just give in to it." (optimisation school A, wave 2)

Referring to the circulation of girls' images, one student commented that "People in our school think it's okay to expose a sket" (year-9 girls, intervention school D, IDs S1-S2). This frames the circulation of these images itself as a form of social punishment for violating injunctive norms of sexual modesty. Participants reported that sexual images of both girls and boys were posted publicly or circulated among students without their permission, with frequency varying between schools. Where accounts described gendered impacts, these were manifestly much more severe for girls. Reflecting the sexual double-standard described above, participants reported that boys tended to be "more blasé", or bragged about their bodies, while girls were "absolutely affronted" (Staff, control school G, ID T1) and typically ridiculed. As a year-9 girl explained, "If a boy's one gets spreaded, I guess it's just like, 'Oh, well done, mate'. But if a girl's one gets sent, like you're a slag or summat." (year-9 girl and boy, control school H, IDs STK3-STK4).

Other abusive practices were facilitated by and reproduced sexist norms

Gendered environments in participating schools were characterised in part by interrelated and overlapping sexual harassment practices, and by use of language that objectified, degraded or subordinated girls. Our interpretation is that these practices were both enabled by, and reproduced, sexist norms.

Sexual harassment

Perceptions of the prevalence of sexual harassment varied across individuals, settings and groups of students. Some participants expressed uncertainty about how common it was generally or among students outside their social group. Two students commented that sexual harassment was perpetrated by a minority of students, though descriptive norms within and outside of school were seen as contributing to both its perpetration and social acceptance. In the words of one staff-member, "They do it because everybody does..." (optimisation school B, wave 2).

While participants acknowledged that sexual harassment could be perpetrated by, and target, both girls and boys, they focused primarily on boys' sexual harassment of girls. Accounts of physical sexual harassment included uninvited or unwelcome sexual touching: for example, boys slapping girls' buttocks was frequently cited. Verbal sexual harassment reportedly often took the form of comments about girls' appearance and our data suggest that girls could face social repercussions for not accepting these comments as compliments. Through a social norms lens, this comprises a form of social sanctioning that reinforces expectations of girls to submit to male judgement.

Accounts portrayed sexual harassment as often minimised (e.g. framed as a joke or compliment) and as tolerated or accepted among students. Accounts also suggested that perpetration could be socially approved, reflecting supportive injunctive norms. Two year-10 girls explained that boys who sexually harassed girls in their school drew confidence from their "friends as back up" and "[did] it to get laughs from their friends," who encouraged them rather than intervening (year-10 girls, intervention school D, IDs S7-S8). According to participants, striving for acceptance among a dominant social group and generally preferring to avoid conflict could both prevent students from challenging or reporting sexual harassment.

However, there were also instances of protective social norms in some groups and settings. For example, a year-10 girl's account of her year-group portrayed an alignment of protective descriptive and injunctive norms:

We get along so well with the boys, but the boys are... not the sort of boys that would just go over to you and touch your bum. Like our girls, if like my group, would be like, 'What are you doing? Don't do that'. (year-10 girl, control school G, ID S1) Additionally, participants gave accounts of both victims and bystanders confronting perpetrators and reporting sexual harassment to school staff. Teacher and parent interviews suggested that schools generally respond to sexual harassment. As a parent described,

...the boys, you know, can be quite sexually aggressive, so I think it, I mean the attitude, there's a kind of general attitude of language which is used, which is just totally inappropriate...and I think the school, I have to say [School D] is brilliant...they would deal with it immediately. (parent/carer, intervention school D, ID P2)

Data from students, however, suggest that schools' response efforts could be hampered by lack of visibility, with sexual harassment often occurring out of their sight or outside of school. Comments by some staff indicated that they viewed girls as partly responsible for their own victimisation. This raises the possibility that sexist norms among staff might sometimes weaken institutional responses to sexual harassment. These staff expressed concern that girls "normalise" (staff, control school G, ID T2) sexual harassment or "think that's a compliment" (staff, intervention school C, ID T1), and expressed the need to work with both victims and perpetrators to address the issue. For example,

Around sexual harassment, I would say there have been cases where girls have sometimes complained that boys have been looking at them inappropriately, making inappropriate comments to them. And within a mainstream school, clearly where there are some young people for whom don't wear the right skirt and so on, or will roll the skirt up and comments and so on are made. Clearly, there is work to do for both the victim but also the perpetrator in those areas in terms of highlighting where the concerns are. (staff, control school H, ID T2)

On the other hand, staff in an intervention school training discussed their concern that the presence of a high level of sexual harassment could contribute to an erosion of physical boundaries among students, normalising "inappropriate" and non-consensual touching, which they suggested might indirectly contribute to DRV (staff training, intervention school D).

Objectification of girls and degrading language

Participants expressed concern about the prevalence in their schools of objectification of girls and of sexist and degrading language used in reference to girls. Staff concerns in some settings included girls adopting and casually using degrading, gendered terms like "bitches" to

refer to each other. However, accounts primarily centred on boys' behaviour towards girls, for example in using "very sexualised language" (staff, intervention school D, ID T3) and executing public judgements of girls' bodies. As one boy described,

Yeah, I feel like, I know this kind of may sound petty in some ways, but I feel like quite a few boys take it into their own hands to make comments about girls body composition. And, like, like take it into their own hands to like start staring and like at girls as such. And like looking them up and down as if they're sort of like objects or trophies. And I feel like sometimes boys will hang around in groups and stereotypically talk about how good-looking girls are... (year-10 girl and boy, control school H, IDs S1-S2)

In our analysis, these behaviours were enabled, in part, by the gender hierarchy formed by prevailing sexist norms. Our data suggest that practices objectifying and degrading girls can also play a role in reproducing that hierarchy, as described by this student: "Some girls would just be called a 'sket' for like doing nothing, because they got on boy's nerves... Like it's just girls are always classified as something so below..." (year-9 girls, intervention school D, IDs S1-S2).

In an intervention school that had also taken other initiatives to improve relationships between students, participants reported recent shifts towards fewer incidents, less social acceptability of sexist behaviours and increasing confidence among students to challenge them (although the last of these could still trigger social repercussions). One teacher described emerging protective injunctive norms among boys in a year-group with which she worked (staff, intervention school D, ID T3):

Participant: ...like boys, particularly in year 11, do not want to be identified as sexist, that's for sure.

Interviewer: Now?

Participant: Yeah, which is amazing. Which is, like, huge. I'm not saying that maybe they aren't sexist sometimes but if somebody [peers or school staff] calls them up on it then they really do not want to be considered sexist.

In line with social norms theory, this account suggested that young people were influenced by social norms among student and staff reference groups even where their personal attitudes had not yet changed.

Dating and relationship violence norms

Participants tended to see DRV as largely unobservable, acknowledging that it could take place outside of school, online or otherwise in private. They were often unsure of how common it was in their school. Our data suggest that embarrassment or fear of losing friends' respect could serve as barriers to disclosing victimisation, and some suggested that disclosure could be particularly difficult for boys. In the words of one student, "I don't think that if anyone in this school is in an abusive relationship or has trouble with sexual harassment then they wouldn't be spreading [it] across the whole school if you know what I mean..." (year-10 girl, control school G, ID S1). Concerns about privacy and about how friends would react were seen as potential deterrents to breaking up with an abusive partner.

Teachers, parents/carers, family, friends and other peers emerged as influential reference groups. Staff saw themselves as playing a protective role and suggested that young people would be embarrassed for a teacher to hear them brag about abusive behaviours. Participants, primarily staff, suggested that young people's expectations about violence in intimate relationships could be influenced by exposure to abusive relationships at home. They emphasised the school's critical role in teaching students about DRV and providing a space to explore their views, as described by this teacher:

Because I think that they quite often don't have any awareness of any, you know, rules, legislation, anything. Quite often they will have experienced domestic violence and things like that, and don't actually know that it's not normal. Because it's just a normal, you know, it happens to lots of people, unfortunately, around here. And it's something that is not really discussed openly with adults at home. And so I think it's important to inform at school so that they know what is right and what is wrong. But I'm not sure that's a conversation that's happening, often. (staff, intervention school F, ID T3)

Participants tended to view physical DRV as rare. They described protective social sanctions such that a young person experiencing DRV would "just break up with" the abusive partner (optimisation school B, wave 2, year-10 discussion) and friends of someone experiencing or perpetrating DRV would intervene. One teacher shared an example of peer intervention with a boy who had been "quite heavy handed" with his girlfriend, which the teacher viewed as particularly impactful:

...about 12 boys came to see the Head of Year and said, this is, you know 'This is totally out of order...he shouldn't have done this. It's really disrespectful. Please can you speak to him?' And in the end the Head of Year actually said, 'Well, why don't you all speak to him?' So they all sat down and told him how they felt about it and that he shouldn't be doing that. And he was pretty embarrassed and he completely understood. And, actually, the fact that that was coming from the students was really nice... (staff, intervention school F, ID T3)

On the other hand, accounts suggested that peer injunctive norms could also contribute to DRV. This was particularly the case for controlling behaviours, which when discussed by gender were described primarily in terms of boys' behaviours towards girls. Controlling behaviours were viewed as more common and visible than physical DRV, especially when they involved control over how a partner presented themselves or interacted with others. In a year-10 focus group, a boy contrasted social intolerance of physical DRV in his school with what he saw as the more typical practice of controlling behaviours, and others agreed:

...There's probably more emotional, like, [a girl agreeing]. They'll, like, kind of like, an example is, like, they might isolate you, so, like, not let you speak to anyone. That'll be, cos that's kind of more subtle, like you don't really realise, but your friends might realise. But if they just, like, full-out, like, slap you or something, then everyone would just like stop you [another girl and boy agreeing]. And then you go on a break or whatever. But yeah, so emotional bullying, that's more common, like yeah. Like, I've seen it, like, you would be isolated, or you would stop talking to people, because of your boyfriend being scared about losing you or something. (optimisation school B, wave 2, year-10 discussion)

Often framed as rooted in jealousy, this type of abuse was reported to manifest as control over a partner's makeup and clothing, whether they went out and who they spent time or talked with. Reflecting gender norms restricting cross-gender friendships, our data also suggest there was an expectation among students that young people in a relationship wouldn't spend time with cross-gender peers other than their partner (see section "Sexist norms governing sexual behaviour").

Staff accounts indicated pro-DRV descriptive norms, reporting that jealousy appeared to be considered "normal" and that, as one participant said, "It seems to be very normal for a lot of the pupils to think that whoever you're dating, should know where you are at all times pretty much..." (staff, intervention school D, ID T3). Contrasting pro-DRV norms in her year-group with protective norms among her friends, one student described injunctive norms condoning controlling behaviours this way: I think it's like small things. Like a lot of small things are acceptable, like, for example...people in my year, like their boyfriend would say, they'd be like, 'Oh, yeah, my boyfriend let me wear shorts for PE today'. And you're like, 'Why would he not let you wear shorts for PE today?' Like that's just normal...it's like that's just what's expected, but then so no one like says anything. (year-10 girls, intervention school D, IDs S7-S8)

Considering sexual DRV, participants described two forms involving sexual images. First, accounts suggest that while young people sometimes shared sexual images of themselves with a partner as a consensual part of their relationship, some did so under pressure from their partner. The second involved threatening to or actually sharing sexual images of a partner or former partner without their consent, abuse which we suggest draws power from sexist norms engendering ridicule of girls whose images are circulated (see section "Sexual abuse involving sexual images").

Discussion

Summary and interpretation

Based on qualitative research with students, staff and parents/carers across ten secondary schools in England, this study describes social norms governing gendered attributes as well as DRV and other forms of abuse among young people. Some DRV norms were discussed in general terms, while others were strongly gendered. Like social practices of sexual harassment, and the objectification and degradation of girls, DRV involving controlling behaviours or sexual images was portrayed as focused largely on dominating girls.

We found Giddens's structuration theory¹⁰⁷ to be a useful framework for examining relationships between the social norms and practices emerging from participant accounts. Drawing on Giddens "duality of structure",^{107(p19)} we theorise from the accounts presented that sexist norms comprise a hierarchical gender structure that enables a host of gendered abusive practices among young people, and that enactment of these practices, in turn, reproduce that structure. Integrating a social norms framework, our data suggest that sexist descriptive and injunctive norms governing how girls and boys behave foster social expectations that subjugate girls to boys. Our data also suggest that homophobic language, pressure to share sexual images and the vilification of girls whose images are then circulated operate not only as abusive

practices but also as social sanctions promoting compliance with sexist norms. We theorise that the resulting recursive gender structure, characterised by the subjugation of girls including policing of their bodies and sexuality, enables other practices that are abusive of girls and were prevalent in participant accounts (sexual harassment, objectification, use of degrading language, and some forms of DRV). Our data indicate that injunctive norms tended to support these abusive practices, manifesting as social tolerance and in some cases encouragement from peers. We also theorise that enactments of these abusive practices subjugate girls to boys' judgement and control, functioning to reproducing the prevailing gender structure.

According to this model, engaging in abusive behaviours and enacting social sanctions to uphold prevailing gender norms represent expressions of agency that are both enabled and constrained by a male-dominated gender structure. But Giddens's work suggests that people can take deliberate action to transform prevailing structures¹⁰⁷ and choose between competing norms, particularly where they can draw on different social networks and sources of knowledge. We found evidence of this in the schools in our study. Participants gave accounts of staff and students adopting practices that challenged established norms, from accepting peer transgressions to actively challenging abusive practices and (among staff) pursuing new, coordinated interventions. While doing so could result in social repercussions, structuration theory suggests that these social practices of rebellion nonetheless recursively impact the "conditions of action"^{107(p5)} in schools. For example, acts of resistance to existing gender structures might increase the social cost of abusive practices against girls and expand opportunities for others to resist them too. These acts of resistance represent protective practices on which normative DRV interventions can build.

In addition to protecting girls from abuse, evidence suggests that transforming sexist norms and abusive practices identified in this study would also benefit boys. A large body of research documents "the emotional and physical costs of patriarchy for boys and men".^{91(p228)} Connell suggests that these include the majority of the burden of school disciplinary practices, underdevelopment of skills for fostering "good human relationships"^{91(p222)} and social sanctions against boys who don't conform to hegemonic forms of masculinity.⁹¹ There is evidence of associations between traditional gender norms and increased sexual risk, limited help-seeking,

substance use and suicide and among men.²⁴⁸ There is also some evidence to suggest that perpetrating DRV predicts subsequent marijuana use and lower academic aspirations, including among boys.⁵⁷

Our findings build on past studies that report on the influential role of schools in the construction and support of gender expectations and hierarchies and in the reproduction (and challenging) of gender inequalities.⁹¹ Others have also reported on the use of sexualised language, including anti-gay and misogynist insults used against girls and boys, as a key feature of the construction of masculinity and of the subjugation of girls in school.^{91,261} Our findings also resonate, as we would expect, with Jamal et al.'s research which drew on structuration theory to explore girls' bullying in secondary schools in England.²⁵⁹ While that research focused mainly on girl-only settings, the "policing of [girls] bodies and sexuality" it documented emerged in our mixed-sex settings. In both studies, students recounted the prevalence of sexual harassment and objectification of girls by boys.²⁵⁹ These qualitative findings echo the results of a 2017 survey in mixed-sex primary and secondary schools in England and Wales, which found high rates among girls of experiencing sexual harassment (37%), experiencing gender discrimination (36%) and being described using sexist language (30%).²⁶²

In line with our findings on recursive interactions between abusive practices and the structural conditions that enable them, Jamal et al. demonstrated how school-based sexual bullying of girls reinforced gendered social categories and traced the role of institutional response in reproducing harmful norms "by sometimes ignoring" these practices.^{259(p736)} Data from staff in our study indicated that some subscribe to beliefs holding girls accountable for their own victimisation, which could undermine a protective response. However, student complaints about institutional tolerance in this study centred on staff responses to homophobic comments, which were characterised as surface-level and ineffective. These findings suggest that transformation of sexist structures in schools might usefully involve work with secondary school teachers, 27% of whom in England and Wales report not feeling confident in responding to sexist incidents in their school.²⁶²

Our analysis is novel in using a social norms framework to examine gendered expectations and abusive practices, including DRV, in UK schools. Delineating between norms

that (1) govern DRV directly, (2) participant accounts link explicitly or indirectly to DRV and (3) broader empirical literature links to DRV, we can suggest some specific areas of focus for normative components of DRV interventions. First, descriptive and injunctive norms in schools in this study were described as mainly protective against physical DRV but supportive of boys' controlling behaviours towards female partners. We also found that fear of others' response could be a barrier to DRV help-seeking, results which are supported by other research that finds that young people fear being blamed^{34,263} and significant proportions of girls and boys in Great Britain who experience DRV tell no one.³⁴ Second, staff explicitly linked sexual harassment to DRV via a breakdown of appropriate physical boundaries between students. Participant accounts also indirectly suggest that norms proscribing cross-gender relationships and policing girls' sexuality could contribute to DRV. For the former, this is by underpinning jealousy, which was identified as a key driver of boys' controlling behaviours in relationships. For the latter, we suggest that this policing underpins sanctions that disparage girls whose sexual images are circulated (but not reportedly of the boys who circulate them), lending power to the form of DRV involving threatening to circulate a female partner's sexual images. Third, drawing on other empirical research, we identify pathways through which norms about how girls and boys are expected to behave identified in our study might underpin DRV. Norms prescribing male dominance and control are implicated directly in young people's accounts of boys' DRV perpetration in Great Britain,^{34,138} while the sexual double-standard prescribing conflicting sexual behaviours for girls and boys appears to play a role in undermining clear communication about sexual consent.¹¹⁷ Further considering the latter, in interviews with girls in the UK "male sexual coercion was perceived as standard" for many^{34(p108)} and refusing sex could precipitate severe physical DRV.³⁴ Reports of peer pressure on boys to be sexually active³⁴ suggest that gendered expectations might play a role in underpinning these types of abuse.

Drawing on a social norms framework also enabled us to conceptualise enactments of homophobia, tactics to pressure girls to share sexual images and the vilification of girls whose images are circulated as manifestations of social sanctions promoting sexist norms. This suggests that interventions to weaken these sanctions, even where personal attitudes have not

yet shifted, could theoretically disrupt cycles whereby they reproduce gender-inequitable structures that facilitate gendered abusive practices, including some forms of DRV.

Limitations

This study collected qualitative data from ten schools in south-east and south-west England. While this type of research can provide valuable theoretical insights and identify areas for further research,²⁵⁹ its findings cannot automatically be generalised to other schools or settings. However, many of our findings resonated with other UK school-based research, contributing to a broader understanding of gendered expectations and abuse in UK schools. This study focused on perceptions of social patterns, expectations and sanctions but did not ask about other factors underpinning personal experiences of DRV victimisation or perpetration. The latter have emerged from previous UK-based DRV research (e.g. Barter et al., 2009; Wood et al., 2011),^{34,138} suggesting that although individuals are not always conscious of the factors driving their behaviour,¹⁰⁷ targeted research in this area could usefully augment our findings.

Conclusions

DRV prevention interventions are often concerned with transforming harmful social norms, but there is limited evidence to inform decisions about which specific norms to target. Drawing on data from ten schools in England, we theorise that sexist norms subjugating girls to boys interact with social practices that are abusive of girls, including DRV against girls, to reproduce gender-inequitable structures in schools. Our findings suggest that school staff and students can, and do, take action to interrupt these cycles, and that physical DRV was seen as uncommon and socially proscribed. Normative DRV interventions might usefully build on these protective factors.

We also found that boys' controlling behaviours towards female partners were considered common and typically socially accepted. Our findings suggest that shifting these norms supporting DRV directly, and weakening the social sanctions used to promote compliance with inequitable gender norms, might both be useful targets of DRV interventions. Finally, when considered alongside other empirical UK evidence, our findings identify several

important candidates for the targeting of normative DRV interventions: norms governing sexual harassment, cross-gender friendships, the sexual double-standard and control and dominance as masculine ideals.

Acknowledgements

We gratefully acknowledge the students, school staff and parents/carers who generously gave their time and shared their views for this research.

Funding

This study was funded by the NIHR Public Health Research Board (project 15/3/09). The views expressed are those of the authors and not necessarily those of the NIHR or the Department of Health and Social Care. The funder was not involved in study design; collection, analysis, or interpretation of data; the writing of this article; or the decision to submit this article for publication.

Chapter 8. Discussion

This chapter discusses the relationship between social norms and DRV and then summarises key findings for each research question that this thesis sought to address. It then provides a synthesis of my findings and interprets these in the context of existing literature. Next it discusses limitations to this thesis research. It concludes with a discussion of implications of this research for future public health research and practice.

8.1. Summary of key findings

8.1.1. Contextualisation

Systematic reviews^{22,40,47,72–74} and meta-analyses^{67–71} have identified a range of risk factors at the individual, relationship, community and societal levels that are associated with DRV outcomes. Situated at the community level, social norms have emerged as an enduring focus for DRV prevention programmes.⁷ This focus is supported both by qualitative research on normative mechanisms underpinning DRV^{34,138,139} and by a large body of quantitative research finding associations between social norms and DRV outcomes (see Paper 1, Chapter 4). Theory and empirical research suggest that norms play a role in influencing individual-level risk factors and behaviours, and that societal factors interact with prevailing norms to influence behaviour.

Supporting the Theory of Planned Behaviour,¹⁰⁰ research in fields closely related to DRV provide insights into norms' impact on individual-level risk factors. Mulla et al.'s research on adolescents' bystander actions to address sexual violence and relationship abuse suggests that social norms influence personal attitudes, which in turn affect subsequent bystander action.¹⁵⁶ Similarly in the GBV field, Jewkes et al. highlight the direct influence of gender norms on individual-level risk factors for perpetration, such as having multiple partners and engaging in transactional sex.⁸⁹ They also posit that expectations created by patriarchal norms help explain why men who are not living up to hegemonic masculine ideals (e.g., men who report difficulties such as depression or economic marginalisation) are also more likely to perpetrate GBV.⁸⁹

Considering the broader levels of the social ecological model, theory and research suggest that features of social structures or institutions, including schools, influence normative expectations within a social system.^{91,107} For example, Connell's scholarship traces ways in which a school's

"gender regime" reinforces gendered expectations,^{91(p213)} while UK research documents pervasive sexism in secondary schools and difficulties that teachers face in challenging this.²⁶² Despite the wealth of evidence documenting direct relationships between social norms and DRV alongside more complex relationships between societal influences, social norms and individual-level factors, research quantifying the importance of social norms to DRV outcomes remains limited. Injunctive norms do not feature in any existing meta-analyses of DRV risk factors, which would enable comparisons with other DRV risk factors,^{67–71} and social norms are only rarely assessed in evaluations of DRV interventions.^{11,148,149} Their role has not been examined using consistent, good-quality measures (see Paper 1, Chapter 4), limiting researchers' ability to assess the influence of social norms across populations and contexts, to model this relationship alongside other factors and ultimately to assess empirically the role of norms change in DRV prevention.

This thesis aimed to address gaps in existing methods for measuring social norms by answering four research questions. The following sections summarise findings for each of these in turn.

8.1.2. Research question 1: Are existing measures of adolescent social norms relating to DRV and gender reliable and valid?

To address this research question, we undertook a systematic review of DRV literature. The review identified 40 eligible measures of DRV norms (19 descriptive, 14 injunctive) and gender norms (1 descriptive, 6 injunctive). Most measures assessed norms among peers or friends, which are appropriate reference groups whose influence intensifies as affiliation shifts from family to peer groups in adolescence.^{32,92,141} Almost all eligible measures were associated with young people's own experience of DRV, supporting their construct validity. However, evidence of reliability and content validity of included measures was mixed. Additionally, we identified no measures used in the UK and none used across more than one study. These findings suggest that it is possible to develop valid, reliable measures of social norms relating to DRV and gender, but existing literature is disjointed and noncomparable.

The review also assessed other indicators of measure quality, identifying potential limitations to the existing suite of measures in terms of their (1) development, (2) reference groups, (3)

assessed norms and (4) analyses of relationships between norms and DRV outcomes. First, considering measure development, few measures drew on participatory work with young people to inform and assess the appropriateness of their content and construction. Second, while most measures asked about a defined reference group, several combined multiple reference groups in the same measure. This limits their usefulness for identifying reference groups for intervention and for assessing the effects of targeted interventions. Third, only one descriptive gender norms measure was identified. Injunctive gender norms measures tended to assess norms about GBV but miss more distal gender norms concerning relationships, sexuality and gendered attributes that qualitative research (e.g. Marston and King 2006, Barter 2009, and Wood 2011),^{34,117,138} including data analysed for this thesis, suggests underpin DRV. Longitudinal DRV research by Reyes et al. further supports the relevance of broader gender norms and the importance of measuring these separately from norms about violence.⁸² In their research, the relationship between traditional gender role attitudes and male DRV perpetration was modified by attitudes towards DRV. Though their findings report on personal attitudes, the distinct ways in which gender and DRV attitudes operate at the individual level suggest the possibility that social norms governing gender and DRV might also have distinct roles in underpinning DRV.

Further considering the types of norms examined, despite elevated risk of DRV among SGM adolescents no included measures specified norms concerning DRV in non-heterosexual relationships.^{42–44} Included measures were often gender-neutral or asked about heterosexual DRV but combined items assessing perpetration by girls and by boys in one composite measure. While most were nonetheless associated with DRV outcomes, UK evidence,⁴⁸ including the qualitative data analysed for this thesis, suggests that norms concerning DRV vary depending on the gender of the perpetrator. Eliding norms across genders might therefore obscure relationships between gender-specific norms and DRV outcomes.

Fourth, several norms measures were tested for their association with outcomes in aggregated samples of girls and boys, without assessing for effect modification by sex or gender. DRV research, however, suggests that social norms might influence DRV behaviours via gender-specific pathways. For example, studies have found some measures of DRV norms to be associated with DRV perpetration among girls but not boys^{77,127} or vice versa,¹²⁷ or have found

differences in the strength of these relationships by gender.¹²⁴ Like eliding norms across genders, combining samples without assessing the need for stratification by sex or gender risks obscuring the nature and magnitude of relationships between social norms and DRV outcomes.

Psychometric testing of new and adapted social norms measures for this thesis provided the opportunity to assess the importance of two of these potential limitations by testing the performance of norms measures that are not gender-specific and by assessing norms-sex interactions.

8.1.3. Research question 2: Are new and adapted measures of social norms relating to DRV and gender understandable and answerable when used in research with adolescents in England? I conducted cognitive interviews to assess the understandability and answerability of social norms items concerning DRV and gender among young people in England. I found that participants could understand descriptive and injunctive norms items and could distinguish between injunctive norms and their personal attitudes in their responses. Our data suggest that some participants, however, had initial difficulty in understanding whether injunctive norms items were asking about their own or others' views. Our data also suggest that the reference group "most other students in your school" was understood differently by different participants. Additionally, uncertainty about and perceived heterogeneity among this broad group detracted from answerability, as did asking about norms that were not strong among the reference group. Answerability was enhanced where participants could draw on concrete experiences of the norms in question being voiced or enacted. These findings informed refinements to the tested measures.

To improve understandability and reduce the cognitive burden of injunctive norms items, I modified their construction to parallel the construction of more easily understood items assessing personal attitudes. Our data suggested that narrowing the reference group to "your friends" might improve answerability. Supported by my literature review, which found that most measures assessing "your friends" were associated with DRV behavioural outcomes, I modified items to ask about this smaller and more specific reference group. Final measures included only items assessing norms that we judged, based on cognitive interviews and knowledge of our

population and context, to be likely to have been discussed or enacted (i.e. in the form of direct behaviours or social rewards or consequences).

Our data suggest that participants considered gender in their responses to social norms items, which might undermine answerability of the gender-neutral descriptive norms items. We anticipated that narrowing the reference group to "your friends" would help to mitigate this because participants are likely to have better knowledge of the experiences of their friends compared to a broader student population, and because most of adolescents' friends are of the same gender,²⁶⁴ suggesting that a gender-neutral item asking about friends might be an acceptable proxy for a gender-specific measure. Psychometric testing provided an opportunity to quantitatively assess the performance of refined measures.

8.1.4. Research question 3: Are new and adapted measures of social norms relating to DRV and gender reliable and valid when used in research with adolescents in England, and how can they be refined?

I tested the refined measures of social norms concerning DRV and gender among a large and diverse sample of young people in England. Reverse-worded items did not perform well, loading on a distinct, methods-related factor which was not conceptually cohesive. Supporting observations by Reyes et al. on the importance of distinguishing between items assessing violence and gender constructs,⁸² the remaining items comprised three empirically distinct constructs as theorised: descriptive DRV norms; injunctive DRV norms; and injunctive gender norms. Estimates from CFA suggested that this four-factor solution fit the data well (RMSEA=0.030, CFI=0.950, TLI=0.936).²⁴⁵ The three conceptually consistent measures of distinct social norms constructs each had an ordinal alpha between 0.65 and 0.70, demonstrating minimally acceptable reliability, and were retained for further testing.¹⁸⁷ I tested the validity of each of these measures by assessing whether its relationship to other, theoretically associated measures followed expected patterns.¹⁸⁷ Pro-DRV and gender-inequitable norms were consistently associated with increased levels of DRV victimisation and perpetration at the level of p<0.10 in univariable analyses and these relationships showed some evidence of effect modification by sex for each tested norms measure, providing evidence in support of the tested measures' validity.

More than 15% of participants scored the lowest possible scores for each of the three tested measures, suggesting that all three measures were insensitive to low-to-moderate levels of pro-DRV or gender-inequitable norms. Findings from multivariable analyses using a threshold for statistical significance of p<0.10 also suggest that measures might be further improved by increasing their sensitivity. Descriptive DRV norms measures, which comprised one item assessing physical DRV and two assessing different forms of psychological DRV, remained associated with victimisation and perpetration overall, and of all DRV types. Sensitivity of this measure might be further improved by the addition of a fifth response option. Injunctive DRV norms, on the other hand, referred only to physical DRV and did not retain an independent association with any assessed DRV outcomes in multivariable analyses. Sensitivity of the injunctive DRV norms measure might be increased by adding items assessing forms of psychological DRV such as controlling behaviours, which young people in the UK are less likely (as compared to physical DRV) to recognise as abuse.²⁴⁹ Considering differences in the peer acceptability of DRV by perpetrator gender,⁴⁸ further refinement might also usefully examine whether the addition of more gender-specific items results in the emergence of distinct constructs for DRV perpetrated by boys and by girls, which might be more sensitive to genderspecific DRV outcomes. The addition of relevant items would also likely improve reliability, which tends to increase with measure length.¹⁸⁷

Finally, injunctive gender norms retained an independent association with overall and psychological DRV victimisation in multivariable analyses. As others have observed, societal changes over time might render older measures of gender norm outdated, and it is important that DRV research uses measures that assess the specific types of beliefs that influence DRV among young people today.⁸² The sensitivity of the injunctive gender norms measure might therefore be increased by adding items assessing domains that qualitative research has identified as salient. These include, for example, romantic and sexual expectations of girls and boys, and norms prescribing passivity for girls and toughness for boys.^{34,92,117,138,139} Refinements should be informed by Project Respect's qualitative research, analysed to address research question 4, which explored salient norms among young people in England.

8.1.5. Research question 4: What are student, staff and parent/carer accounts of social norms relating to DRV and gender in schools, and how are these implicated in DRV? Interviews with students, staff and parents/carers identified a range of social norms that underpin DRV directly or indirectly. Considering social norms concerning DRV itself, our findings suggest that those supporting *controlling behaviours* and inhibiting *disclosure of DRV victimisation* are particularly salient, supporting previous research with UK young people.^{34,249} Considering broader gender norms, our data suggest that those supporting *sexual harassment* and the *policing of girls' sexuality*, and those proscribing *cross-gender friendships*, each contribute to DRV.

Other UK research has demonstrated that *sexist norms about gendered attributes*, especially those prescribing *control and dominance as masculine ideals*, contribute to boys' perpetration of DRV.^{34,138} Global review evidence suggests that the *sexual double-standard* prescribing sexual activity for boys and proscribing this for girls can undermine communication about sexual consent.¹¹⁷ Our interview data provide evidence of these norms operating among young people in England, highlighting their salience for local DRV prevention.

Drawing on Giddens' structuration theory,¹⁰⁷ we theorise from participant accounts that social sanctions enacted to uphold sexist norms about gendered attributes function to reproduce the prevailing gendered hierarchy. Sanctions featured in participant accounts included *enactments of homophobia, tactics to put pressure on girls to comply with boys' requests for sexual images* (e.g. deriding them as "frigid"), and *social punishments for girls whose personal images were then circulated*. We posit that efforts to weaken these sanctions might effectively alter the structural conditions that facilitate some forms of DRV. In line with social norms theory, our data suggest that efforts to shift social sanctions can be effective in addressing gendered abuse even where personal attitudes have not yet shifted.

Finally, participant accounts demonstrate that students and staff can, and do, take action to negatively sanction DRV and to challenge sexist gender norms in school. However, data from staff and students also suggest that staff efforts to address homophobia can be ineffective, and that some staff view girls as responsible in part for their own sexual harassment, which could theoretically undermine institutional response to sexual harassment in schools. In a 2017

survey, more than a quarter of secondary school teachers in England and Wales reported not feeling confident to respond to sexist incidents in their school,²⁶² providing further evidence of the need to work with school staff to transform sexist school structures that support DRV.

8.2. Synthesis and interpretation

This thesis sought to refine measures of social norms relating to DRV and to gender, and to inform methods of social norms measurement in DRV research. An adapted measure of descriptive DRV norms and new measures of injunctive DRV and gender norms were subjected to cognitive testing with young people in England. Findings led to refinements to the structure, content and reference groups of these measures, which were subsequently piloted among 1,462 year-8 and year-9 students across five secondary schools in England. Results of a literature review and of cognitive testing raised questions about the importance of sex- or gender-specificity in DRV norms measures, and in the DRV outcomes against which they are tested, which piloting also provided the opportunity to examine.

All three refined measures were found to be reliable and valid in psychometric testing, supporting their construction, content and reference group framing. Furthermore, the gender-neutral descriptive norms measure retained an independent relationship with DRV outcomes in multivariable analyses. This suggests that using gender-neutral items is a practical solution, at least for lengthy evaluation surveys, for a brief but sensitive measure of this construct. Results of multivariable analyses suggest that injunctive norms measures could potentially be further refined by increasing the number of response options, and by developing longer-form versions for use where survey length allows and research aims require more sensitive measures.²²² Considering outcomes, relationships between social norms and DRV behavioural outcomes were sometimes, but not always, moderated by sex. This suggests that analyses should test for interactions between social norms measures and sex or gender to guide decisions about whether and which models should be sex- or gender-stratified.

Finally, robust independent associations between descriptive DRV norms and DRV outcomes suggest that descriptive norms might play an important role in DRV. While descriptive DRV norms were the most common type of norms measure identified in our systematic review, descriptive gender norms have been neglected in DRV research. Drawing on Social Cognitive Theory, which frames social modelling as an important source of behavioural learning,¹⁰⁸ we posit that perceptions that protective gendered behaviours are typical among an influential reference group could be protective against DRV. Our qualitative research suggests, for example, that perceptions of lower levels of compliance with gendered expectations concerning male sexual activity, dominating and controlling forms of masculinity, female sexual expression, cross-gender friendships and sexual harassment could be beneficial. Our findings suggest that the development and testing of salient descriptive gender norms measures would be a valuable contribution to DRV prevention and evaluation.

Our qualitative research also provides insights into salient social norms that could be incorporated into longer versions of our three piloted measures. Our findings support existing items addressing physical DRV and controlling behaviours in the descriptive DRV norms measure, and suggest that norms concerning disclosure of DRV victimisation could be added. Considering the measure of injunctive DRV norms, our findings support the inclusion of items addressing physical DRV, and the addition of items assessing controlling behaviours and disclosure of victimisation. Finally, considering the measure of injunctive gender norms, our qualitative findings support the inclusion of existing items assessing gendered behavioural attributes as well as the sexual double-standard. If the measure is to be extended, other salient norms in schools include social expectations concerning dominance and control as masculine ideals, cross-gender friendships, homophobic abuse, sexual harassment and the policing of girls' sexuality.

8.2.1. Implications for research

Results of psychometric testing support the reliability and validity of three brief new and adapted measures of descriptive DRV norms, injunctive DRV norms and injunctive gender norms. These measures should be incorporated into evaluations of DRV interventions that aim to shift social norms, which should assess impact on norms and whether changes mediate reductions in DRV. As Reyes et al. recommend, evaluators should use mediation, moderation and moderated mediation analyses both to inform interventions and to improve our understanding of how DRV develops and is sustained.⁸²

Further refinements to increase the sensitivity of the three tested measures could be useful, especially for epidemiological research to inform interventions. This work should be guided by existing research on the specific DRV and gender norms that underpin DRV in the UK, including findings from analysis conducted for this thesis of interviews with students, school staff and parents/carers. Additional work is needed to develop and test a measure of salient descriptive gender norms, drawing on existing research, while exploratory formative research is needed to inform the development of measures of social norms underpinning DRV among SGM young people.

Findings from this thesis research support several recommendations for the development of social norms measures for DRV research among young people. Recommendations relate to the norms assessed, reference groups and measure construction:

Measures should assess norms:

- 1. that are held among a cohesive, influential reference group;
- 2. that are strong enough among the reference group for respondents to discern;
- for which participants can draw on concrete experiences to respond (i.e. descriptive norms governing behaviours that are observable or discussed; or injunctive norms supported by observable social sanctions);
- that include what evidence suggests might be more moderate indicators of the phenomenon of interest;
- that are proximal, or distal where evidence supports the influence of specific distal norms;
- 6. concerning gender-specific forms of DRV, where feasible; and
- concerning a broad range of gendered expectations beyond violence itself that are important to DRV.

Measures should use reference groups that are:

- 1. separate, bounded and sufficiently narrow;^{ff}
- 2. of known importance for the behaviour of interest; and
- 3. **gender-specific**, where feasible and where evidence suggests that norms vary among girls and boys.

In terms of their construction, norms measures:

- 1. should use simple, clear and concise instructions and wording;
- 2. should not use reverse-worded items;
- 3. can use **parallel formatting, items and response options** where surveys assess both norms and attitudes relating to the same phenomena; and
- 4. can include a **response option indicating uncertainty**, which should especially be considered where it is uncertain whether norms are discernible and publicly manifest.

A challenge in norms measurement with young people is the need to balance a measure's length and answerability with its sensitivity and opportunity for co-variation with other variables of interest. Existing reliable and valid norms measures identified via review of the DRV literature review tended to include more than three items (ranging from two^{gg} to 28 items), and to offer more than four response options (ranging from four to seven, where information was provided). The length of norms measures needs to be considered in the context of the length of the full survey in which they are administered, but these findings suggest that longer measures and those with more than four response options can perform well among young people.

Another challenge is the need to balance the contextual specificity of salient norms⁶ with the time and resources required to develop new measures, and the limitations that the proliferation of new measures poses for comparisons across time, settings and populations. Findings from a

^{ff} If there is more than one important reference group, these should be asked about in separate items. Items should be combined into a single measure only if indicated in an analysis of factor structure. ^{gg} This excludes single-item measures, some of which were valid but which were not assessed for reliability.

global systematic review of 268 qualitative studies addressing factors that shape young people's sexual behaviour suggest that many normative social factors are shared across settings.¹¹⁷ In the review, key themes addressing stigma, gender stereotypes, social expectations, social sanctions and the sexual double-standard were present in all study countries. If key norms underpinning DRV are similarly shared across settings, this suggests that existing measures might be reliable and valid across settings or could potentially be refined to be used across settings. This possibility could be explored via cognitive testing on an *ad hoc* basis to assess whether existing reliable, valid measures such as those found to be reliable and valid in this thesis research can be used with minimal changes among similar populations, as has been done with more established measures such as the Safe Dates and CADRI-s measures of DRV.³⁷ Alternatively, multi-site research exploring whether existing measures of salient norms are sufficiently general to perform well across contexts would also make a valuable contribution to the field. The availability of established measures that are reliable and valid across multiple settings would reduce barriers to and improve comparability of norms measurement in DRV research, aiding evidence synthesis and the consolidation of learning. The routine incorporation of norms measures would improve our understanding of the role that social norms play in the development, persistence and reduction of DRV.

8.2.2. Implications for practice

Norms concerning DRV itself and broader gender norms are both independently associated with DRV outcomes. Social norms theory, DRV experts and empirical research suggest that addressing these norms is an important aspect of DRV prevention.^{8,15,34,138} Intervention developers should draw on existing research, and conduct formative research where needed, to explicitly theorise which specific DRV and gender norms are important for interventions to target in their settings. In the UK, our and others'^{34,48} findings suggest these include norms concerning psychological DRV and disclosure of DRV victimisation. Considering broader gender norms, participant accounts in our research and other qualitative research with UK young people have identified norms concerning several gendered phenomena as particularly salient. These include norms concerning: cross-gender friendships; control and dominance as masculine ideals;^{34,138} the policing of girls' sexuality; loss of status for boys if their girlfriend is

unfaithful;^{34,138} durable heterosexual relationships for girls;^{34,139} the sexual doublestandard;^{34,138} and sexual harassment and homophobic abuse. Interventions should also support protective norms evidenced in UK schools, such as those proscribing physical DRV, and those supporting student and staff efforts to challenge gender stereotypes, sexual harassment and homophobic abuse. Finally, interventions should incorporate work with school staff to address harmful views, and to enhance skills and confidence for effectively engaging with students to transform sexist school structures that support DRV.

8.3. Limitations

This thesis research draws on data collected for the Project Respect study, and the scope of feasible data collection activities was determined by the funding, timeline and resources for that overall study. While Project Respect provided an excellent opportunity to address the research questions set out in this thesis, it also posed some limitations. Full-scale measure development beginning with formative research and the testing of a broad range of candidate items¹⁸⁷ was not feasible. There was insufficient time to conduct a systematic review of relevant norms measures before developing and testing new and adapted measures for Project Respect, which meant that measures were developed based on the most relevant measures identified in a rapid, ad hoc review. Similarly, measures were refined based on cognitive testing but the study timeline did not allow for a second round of cognitive testing to evaluate whether these changes improved understandability and answerability. Despite these limitations, I was able to modify existing measures to develop and refine an adapted descriptive DRV norms measure, and new injunctive DRV and gender norms measures. The baseline survey conducted for Project Respect provided the opportunity to psychometrically test the resulting measures among a large, diverse sample of young people in England, and the refined measures were found to be reliable and valid. Testing also identified floor effects, suggesting that measures could be further improved to enhance their sensitivity to low-to-moderate levels of the assessed norms.

Limitations to available resources precluded full dual-screening of systematic review records and dual data extraction. To minimise errors, initial records were instead dual-screened on title and abstract until reaching a high level of agreement and a second reviewer checked all data extraction. However, it remains possible that eligible records were excluded or relevant data missed. Similarly, available resources precluded dual coding of qualitative data. To enhance the internal validity of my analysis, findings and interpretation were discussed at length with colleagues who have expertise in qualitative and school-based research. Considering external validity, it is important to note that, while the qualitative component of this research provides valuable theoretical insights and indicates potential areas for future research²⁵⁹ and measure refinement, its findings come from ten schools in south-east and south-west England, and cannot be assumed to generalise to other settings or populations.

8.4. Conclusions

Descriptive DRV norms, injunctive DRV norms and injunctive gender norms comprise empirically distinct constructs. It is possible to develop reliable and valid measures of these constructs for use with young people. This thesis research involved developing and testing three such measures for use in DRV research in England. These measures should next be incorporated into DRV intervention evaluation research, which should assess changes in social norms and examine the role of any changes in mediating DRV reduction. Further research should build on the tested measures to develop more sensitive, longer-form versions for use in epidemiological research, drawing on what is known about salient DRV and gender norms underpinning DRV victimisation and perpetration. Researchers should draw on existing knowledge about gender norms underpinning DRV to develop a new measure of salient descriptive gender norms, while exploratory research is needed to inform the development of measures assessing social norms underpinning DRV among SGM young people.

Findings from this research support several recommendations for social norms measurement in DRV research.

- 1) Measures should assess relevant and discernible social norms that are publicly manifest among a specific, bounded, cohesive and influential reference group.
- 2) These should include indicators of less severe forms of the phenomenon of interest.
- Measures should assess norms that are proximal to the form of DRV that is of interest, or more distal norms where evidence supports their role in underpinning DRV.
- 4) Gender norms measures should assess gendered expectations beyond violence itself.
- 5) Norms and reference groups should be gender-specific, where indicated and feasible.

- 6) Measures should use simple, clear and concise instructions and wording.
- 7) Reverse-worded items should not be used.
- 8) Measures can be constructed as parallel to attitudes measures where analogous attitudes are assessed in the same survey.
- 9) Measures can include a response option that indicates uncertainty.

Future research should explore the potential for use of measures of salient social norms underpinning DRV across contexts, which if reliable and valid across settings could be routinely incorporated into DRV research.

References

- 1. Meiksin R, Allen E, Crichton J, et al. Protocol for pilot cluster RCT of Project Respect: a school-based intervention to prevent dating and relationship violence and address health inequalities among young people. *Pilot and Feasibility Studies*. 2019;5(1). doi:10.1186/s40814-019-0391-z
- Meiksin R, Campbell R, Crichton J, et al. Implementing a whole-school relationships and sex education intervention to prevent dating and relationship violence: evidence from a pilot trial in English secondary schools. *Sex Education*. Published online March 10, 2020:1-17. doi:10.1080/14681811.2020.1729718
- 3. Meiksin, Rebecca. Systematic review of social norms measures relating to dating and relationship violence. 2020. Accessed September 23, 2020. https://doi.org/10.17605/OSF.IO/UWX9S
- 4. Meiksin R, Bonell C, Bhatia A, Melendez-Torres G, Kyegombe N, Kohli A. Social norms concerning dating and relationship violence and gender among adolescents: a systematic review of survey measures used in dating and relationship violence research. *The Lancet*. 2022;400:S65. doi:10.1016/S0140-6736(22)02275-9
- Meiksin R, Ponsford R, Bonell C. Assessment of survey items on social norms relating to sexual behaviour and dating and relationship violence among young adolescents in England: cognitive interviews within two pilot studies. *The Lancet*. 2018;392:S63. doi:10.1016/S0140-6736(18)32888-5
- Ashburn K, Costenbader B, Igras S, Pirzadeh M, Homan R. Learning Collaborative Background Reader: Advancing Research and Practice on Normative Change for Adolescent Sexual and Reproductive Health and Well-Being. Developed for the Convening Meeting, December 5-6, 2016. Institute for Reproductive Health at Georgetown University and FHI 360; 2016.
- 7. Wekerle C, Wolfe DA. Dating violence in mid-adolescence: theory, significance, and emerging prevention initiatives. *Clin Psychol Rev.* 1999;19(4):435-456.
- 8. Stanley N, Ellis J, Farrelly N, Hollinghurst S, Downe S. Preventing domestic abuse for children and young people: A review of school-based interventions. *Children and Youth Services Review*. 2015;59:120-131. doi:10.1016/j.childyouth.2015.10.018
- Lowe H, Dobbin J, Kiss L, et al. Mechanisms for the prevention of adolescent intimate partner violence: A realist review of interventions in low- and middle-income countries. Singh R, ed. *PLOS Glob Public Health*. 2022;2(11):e0001230. doi:10.1371/journal.pgph.0001230

- Lundgren R, Amin A. Addressing Intimate Partner Violence and Sexual Violence Among Adolescents: Emerging Evidence of Effectiveness. *Journal of Adolescent Health*. 2015;56(1):S42-S50. doi:10.1016/j.jadohealth.2014.08.012
- 11. Melendez-Torres G, Orr N, Farmer C, et al. School-based interventions TO Prevent Dating and Relationship Violence and Gender-Based Violence: STOP-DRV-GBV systematic review. *Public Health Res (Southampt)*. 2024;12(3):1-192. doi:10.3310/KTWR6997
- Verbeek M, Weeland J, Luijk M, van de Bongardt D. Sexual and Dating Violence Prevention Programs for Male Youth: A Systematic Review of Program Characteristics, Intended Psychosexual Outcomes, and Effectiveness. *Arch Sex Behav*. 2023;52(7):2899-2935. doi:10.1007/s10508-023-02596-5
- 13. Whitaker DJ, Morrison S, Lindquist C, et al. A critical review of interventions for the primary prevention of perpetration of partner violence. *Aggression and Violent Behavior*. 2006;11(2):151-166. doi:10.1016/j.avb.2005.07.007
- 14. McNaughton Reyes HL, Graham LM, Chen MS, et al. Adolescent dating violence prevention programmes: a global systematic review of evaluation studies. *The Lancet Child & Adolescent Health*. 2021;5(3):223-232. doi:10.1016/S2352-4642(20)30276-5
- 15. Alexander-Scott M, Bell E, Holden J. *Shifting Social Norms to Tackle Violence against Women and Girls (VAWG)*. VAWG Helpdesk; 2016.
- Young H, Turney C, White J, Bonell C, Lewis R, Fletcher A. Dating and relationship violence among 16-19 year olds in England and Wales: a cross-sectional study of victimization. J Public Health (Oxf). Published online November 10, 2017:1-9. doi:10.1093/pubmed/fdx139
- Singh JA, Siddiqi M, Parameshwar P, Chandra-Mouli V. World Health Organization Guidance on Ethical Considerations in Planning and Reviewing Research Studies on Sexual and Reproductive Health in Adolescents. *Journal of Adolescent Health*. 2019;64(4):427-429. doi:10.1016/j.jadohealth.2019.01.008
- 18. Breiding M, Basile K, Smith S, Black M, Mahendra P. *Intimate Partner Violence Surveillance: Uniform Definitions and Recommended Data Elements, Version 2.0.* National Center for Injury Prevention and Control, Center for Disease Control and Prevention; 2015.
- 19. Barter C, Stanley N. Inter-personal violence and abuse in adolescent intimate relationships: mental health impact and implications for practice. *International Review of Psychiatry*. 2016;28(5):485-503. doi:10.1080/09540261.2016.1215295
- 20. Wolfe DA, Temple JR, eds. *Adolescent Dating Violence: Theory, Research, and Prevention*. Academic Press, is an imprint of Elsevier; 2018.

- Exner-Cortens D, Gill L, Eckenrode J. Measurement of adolescent dating violence: A comprehensive review (Part 1, behaviors). *Aggression and Violent Behavior*. 2016;27:64-78. doi:10.1016/j.avb.2016.02.007
- 22. Leen E, Sorbring E, Mawer M, Holdsworth E, Helsing B, Bowen E. Prevalence, dynamic risk factors and the efficacy of primary interventions for adolescent dating violence: An international review. *Aggression and Violent Behavior*. 2013;18(1):159-174. doi:10.1016/j.avb.2012.11.015
- 23. Sardinha L, Maheu-Giroux M, Stöckl H, Meyer SR, García-Moreno C. Global, regional, and national prevalence estimates of physical or sexual, or both, intimate partner violence against women in 2018. *The Lancet*. 2022;399(10327):803-813. doi:10.1016/S0140-6736(21)02664-7
- Swedo EA, Sumner SA, Hillis SD, et al. Prevalence of Violence Victimization and Perpetration Among Persons Aged 13–24 Years — Four Sub-Saharan African Countries, 2013–2015. MMWR Morb Mortal Wkly Rep. 2019;68(15):350-355. doi:10.15585/mmwr.mm6815a3
- 25. Wincentak K, Connolly J, Card N. Teen dating violence: A meta-analytic review of prevalence rates. *Psychology of Violence*. 2017;7(2):224-241. doi:10.1037/a0040194
- Exner-Cortens D, Eckenrode J, Rothman E. Longitudinal Associations Between Teen Dating Violence Victimization and Adverse Health Outcomes. *Pediatrics*. 2013;131(1):71-78. doi:10.1542/peds.2012-1029
- Mokdad AH, Forouzanfar MH, Daoud F, et al. Global burden of diseases, injuries, and risk factors for young people's health during 1990–2013: a systematic analysis for the Global Burden of Disease Study 2013. *The Lancet*. 2016;387(10036):2383-2401. doi:10.1016/S0140-6736(16)00648-6
- Cornelius TL, Resseguie N. Primary and secondary prevention programs for dating violence: A review of the literature. *Aggression and Violent Behavior*. 2007;12(3):364-375. doi:10.1016/j.avb.2006.09.006
- 29. De Koker P, Mathews C, Zuch M, Bastien S, Mason-Jones AJ. A Systematic Review of Interventions for Preventing Adolescent Intimate Partner Violence. *Journal of Adolescent Health*. 2014;54(1):3-13. doi:10.1016/j.jadohealth.2013.08.008
- 30. De La Rue L, Polanin JR, Espelage DL, Pigott TD. School-based interventions to reduce dating and sexual violence. *Campbell Systematic Reviews*. 2014;(7). doi:10.4073/csr.2014.7
- 31. Fellmeth GL, Heffernan C, Nurse J, Habibula S, Sethi D. Educational and skills-based interventions for preventing relationship and dating violence in adolescents and young adults. The Cochrane Collaboration, ed. *Cochrane Database Syst Rev.* 2013;(6):Cd004534. doi:10.1002/14651858.CD004534.pub3

- 32. Wolfe DA, Jaffe PG. Emerging strategies in the prevention of domestic violence. *Future Child*. 1999;9(3):133-144.
- 33. Arriaga XB, Foshee VA. Adolescent Dating Violence: Do Adolescents Follow in Their Friends', Or Their Parents', Footsteps? *Journal of Interpersonal Violence*. 2004;19(2):162-184. doi:10.1177/0886260503260247
- 34. Barter C, McCarry M, Berridge D, Evans K. *Partner Exploitation and Violence in Teenage Intimate Relationships*. NSPCC; 2009.
- 35. Briefing Paper 2: Incidence Rates and Impact of Experiencing Interpersonal Violence and Abuse in Young People's Relationships.; 2015.
- 36. Fox CL, Corr ML, Gadd D, Butler I. Young teenagers' experiences of domestic abuse. *Journal of Youth Studies*. 2014;17(4):510-526. doi:10.1080/13676261.2013.780125
- 37. Meiksin R, Crichton J, Dodd M, et al. A school intervention for 13- to 15-year-olds to prevent dating and relationship violence: the Project Respect pilot cluster RCT. *Public Health Res.* 2020;8(5):1-338. doi:10.3310/phr08050
- Young H, Long SJ, Melendez-Torres GJ, et al. Dating and relationship violence victimization and perpetration among 11–16 year olds in Wales: a cross-sectional analysis of the School Health Research Network (SHRN) survey. *Journal of Public Health*. 2021;43(1):111-122. doi:10.1093/pubmed/fdz084
- 39. Meiksin R, Allen E, Dodd M, et al. Refining and testing measures of dating and relationship violence (DRV) among adolescents in England: Cognitive interview and baseline findings from the Project Respect DRV prevention programme [oral presentation]. Presented at: European Society for Prevention Research Conference; October 2018; Lisbon, Portugal.
- 40. Capaldi DM, Knoble NB, Shortt JW, Kim HK. A Systematic Review of Risk Factors for Intimate Partner Violence. *Partner Abuse*. 2012;3(2):231-280. doi:10.1891/1946-6560.3.2.231
- 41. World Health Organization. *Violence against Women Prevalence Estimates, 2018: Global, Regional and National Prevalence Estimates for Intimate Partner Violence against Women and Global and Regional Prevalence Estimates for Non-Partner Sexual Violence against Women.* World Health Organization; 2021. Accessed February 8, 2022. https://apps.who.int/iris/handle/10665/341337
- 42. Dank M, Lachman P, Zweig JM, Yahner J. Dating Violence Experiences of Lesbian, Gay, Bisexual, and Transgender Youth. *Journal of Youth and Adolescence*. 2014;43(5):846-857. doi:10.1007/s10964-013-9975-8

- 43. Dietz J. Dating and Relationship Violence in Sexual and Gender Minority Youth: A Systematic Review of the Literature and Cost-Effectiveness Analysis of a Targeted School-Based Preventative Intervention. London School of Hygiene & Tropical Medicine; 2019.
- 44. Luo F, Stone DM, Tharp AT. Physical dating violence victimization among sexual minority youth. *Am J Public Health*. 2014;104(10):e66-73. doi:10.2105/AJPH.2014.302051
- 45. Taylor BG, Mumford EA, Liu W. *The National Survey of Teen Relationships and Intimate Violence (STRiV)*.; 2016:18.
- Vagi KJ, O'Malley Olsen E, Basile KC, Vivolo-Kantor AM. Teen Dating Violence (Physical and Sexual) Among US High School Students: Findings From the 2013 National Youth Risk Behavior Survey. *JAMA Pediatrics*. 2015;169(5):474. doi:10.1001/jamapediatrics.2014.3577
- Vézina J, Hébert M. Risk Factors for Victimization in Romantic Relationships of Young Women: A Review of Empirical Studies and Implications for Prevention. *Trauma, Violence,* & Abuse. 2007;8(1):33-66. doi:10.1177/1524838006297029
- 48. Hird MJ. An Empirical study of adolescent dating aggression in the U.K. *Journal of Adolescence*. 2000;23(1):69-78. doi:10.1006/jado.1999.0292
- 49. Exner-Cortens D, Eckenrode J, Bunge J, Rothman E. Revictimization After Adolescent Dating Violence in a Matched, National Sample of Youth. *Journal of Adolescent Health*. 2017;60(2):176-183. doi:10.1016/j.jadohealth.2016.09.015
- 50. Foshee VA. Gender differences in adolescent dating abuse prevalence, types and injuries. *Health Education Research*. 1996;11(3):275-286. doi:10.1093/her/11.3.275-a
- 51. Adhia A, Kernic MA, Hemenway D, Vavilala MS, Rivara FP. Intimate Partner Homicide of Adolescents. *JAMA Pediatr*. 2019;173(6):571. doi:10.1001/jamapediatrics.2019.0621
- Roberts TA, Klein JD, Fisher S. Longitudinal Effect of Intimate Partner Abuse on High-Risk Behavior Among Adolescents. *Archives of Pediatrics & Adolescent Medicine*. 2003;157(9):875. doi:10.1001/archpedi.157.9.875
- 53. Piolanti A, Waller F, Schmid IE, Foran HM. Long-term Adverse Outcomes Associated With Teen Dating Violence: A Systematic Review. *Pediatrics*. 2023;151(6):e2022059654. doi:10.1542/peds.2022-059654
- 54. Shorey RC, Cohen JR, Kolp H, Fite PJ, Stuart GL, Temple JR. Predicting sexual behaviors from mid-adolescence to emerging adulthood: The roles of dating violence victimization and substance use. *Preventive Medicine*. 2019;129:105844. doi:10.1016/j.ypmed.2019.105844

- 55. Castellví P, Miranda-Mendizábal A, Parés-Badell O, et al. Exposure to violence, a risk for suicide in youths and young adults. A meta-analysis of longitudinal studies. *Acta Psychiatrica Scandinavica*. 2017;135(3):195-211. doi:10.1111/acps.12679
- 56. Manchikanti Gómez A. Testing the Cycle of Violence Hypothesis: Child Abuse and Adolescent Dating Violence as Predictors of Intimate Partner Violence in Young Adulthood. *Youth & Society*. 2011;43(1):171-192. doi:10.1177/0044118X09358313
- 57. Foshee VA, Gottfredson NC, Reyes HLM, et al. Developmental Outcomes of Using Physical Violence Against Dates and Peers. *Journal of Adolescent Health*. 2016;58(6):665-671. doi:10.1016/j.jadohealth.2016.03.002
- 58. Fry D, Fang X, Elliott S, et al. The relationships between violence in childhood and educational outcomes: A global systematic review and meta-analysis. *Child Abuse & Neglect*. 2018;75:6-28. doi:10.1016/j.chiabu.2017.06.021
- 59. Hagan J, Foster H. Youth Violence and the End of Adolescence. *American Sociological Review*. 2001;66(6):874. doi:10.2307/3088877
- Adams AE, Greeson MR, Kennedy AC, Tolman RM. The Effects of Adolescent Intimate Partner Violence on Women's Educational Attainment and Earnings. *J Interpers Violence*. 2013;28(17):3283-3300. doi:10.1177/0886260513496895
- Oliver R, Alexander B, Roe S, Wlasny M. *The Economic and Social Costs of Domestic Abuse*. Home Office; 2019. Accessed December 8, 2023. https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachmen t_data/file/918897/horr107.pdf
- 62. Wolfe DA, Feiring C. Dating Violence Through the Lens of Adolescent Romantic Relationships. *Child Maltreat*. 2000;5(4):360-363. doi:10.1177/1077559500005004007
- 63. Heise LL. Violence Against Women: An Integrated, Ecological Framework. *Violence Against Women*. 1998;4(3):262-290. doi:10.1177/1077801298004003002
- 64. Claussen C, Matejko E, Exner-Cortens D. Exploring risk and protective factors for adolescent dating violence across the social-ecological model: A systematic scoping review of reviews. *Front Psychiatry*. 2022;13:933433. doi:10.3389/fpsyt.2022.933433
- 65. Heise L. What Works to Prevent Partner Violence? An Evidence Overview. STRIVE; 2011.
- 66. Krug EG, Dahlberg L, Mercy J, Zwi A, Lozano R. *World Report on Violence and Health*. World Health Organization; 2002.
- 67. Garthe RC, Sullivan TN, McDaniel MA. A meta-analytic review of peer risk factors and adolescent dating violence. *Psychology of Violence*. 2017;7(1):45-57. doi:10.1037/vio0000040

- Hébert M, Daspe MÈ, Lapierre A, et al. A Meta-Analysis of Risk and Protective Factors for Dating Violence Victimization: The Role of Family and Peer Interpersonal Context. *Trauma, Violence, & Abuse*. Published online August 22, 2017:152483801772533. doi:10.1177/1524838017725336
- Spencer CM, Anders KM, Toews ML, Emanuels SK. Risk Markers for Physical Teen Dating Violence Victimization in the United States: A Meta-Analysis. J Youth Adolescence. 2020;49(3):575-589. doi:10.1007/s10964-020-01194-1
- 70. Spencer CM, Toews ML, Anders KM, Emanuels SK. Risk Markers for Physical Teen Dating Violence Perpetration: A Meta-Analysis. *Trauma, Violence, & Abuse*. 2021;22(3):619-631. doi:10.1177/1524838019875700
- 71. Park S, Kim SH. The power of family and community factors in predicting dating violence: A meta-analysis. *Aggression and Violent Behavior*. 2018;40:19-28. doi:10.1016/j.avb.2018.03.002
- 72. Johnson RM, Parker EM, Rinehart J, Nail J, Rothman EF. Neighborhood Factors and Dating Violence Among Youth: A Systematic Review. *Am J Prev Med*. 2015;49(3):458-466. doi:10.1016/j.amepre.2015.05.020
- Malhi N, Oliffe JL, Bungay V, Kelly MT. Male Perpetration of Adolescent Dating Violence: A Scoping Review. Am J Mens Health. 2020;14(5):1557988320963600. doi:10.1177/1557988320963600
- 74. Vagi KJ, Rothman EF, Latzman NE, Tharp AT, Hall DM, Breiding MJ. Beyond Correlates: A Review of Risk and Protective Factors for Adolescent Dating Violence Perpetration. *Journal of Youth and Adolescence*. 2013;42(4):633-649. doi:10.1007/s10964-013-9907-7
- 75. Ali B, Swahn M, Hamburger M. Attitudes Affecting Physical Dating Violence Perpetration and Victimization: Findings From Adolescents in a High-Risk Urban Community. *Violence and Victims*. 2011;26(5):669-683. doi:10.1891/0886-6708.26.5.669
- 76. Enosh G. Cognition or Involvement? Explaining Sexual-coercion in High-school Dating. *Sex Abuse*. 2007;19(3):311-329. doi:10.1177/107906320701900308
- Foshee VA, Linder F, MacDougall JE, Bangdiwala S. Gender Differences in the Longitudinal Predictors of Adolescent Dating Violence. *Preventive Medicine*. 2001;32(2):128-141. doi:10.1006/pmed.2000.0793
- Hunt KE, Robinson LE, Valido A, Espelage DL, Hong JS. Teen Dating Violence Victimization: Associations Among Peer Justification, Attitudes Toward Gender Inequality, Sexual Activity, and Peer Victimization. *J Interpers Violence*. 2022;37(9-10):5914-5936. doi:10.1177/08862605221085015

- 79. Jain S, Cohen AK, Paglisotti T, Subramanyam MA, Chopel A, Miller E. School climate and physical adolescent relationship abuse: Differences by sex, socioeconomic status, and bullying. *Journal of Adolescence*. 2018;66(1):71-82. doi:10.1016/j.adolescence.2018.05.001
- McCauley HL, Tancredi DJ, Silverman JG, et al. Gender-Equitable Attitudes, Bystander Behavior, and Recent Abuse Perpetration Against Heterosexual Dating Partners of Male High School Athletes. *American Journal of Public Health*. 2013;103(10):1882-1887. doi:10.2105/AJPH.2013.301443
- Reed E, Silverman JG, Raj A, Decker MR, Miller E. Male Perpetration of Teen Dating Violence: Associations with Neighborhood Violence Involvement, Gender Attitudes, and Perceived Peer and Neighborhood Norms. *J Urban Health*. 2011;88(2):226-239. doi:10.1007/s11524-011-9545-x
- 82. Reyes HLM, Foshee VA, Niolon PH, Reidy DE, Hall JE. Gender Role Attitudes and Male Adolescent Dating Violence Perpetration: Normative Beliefs as Moderators. *Journal of Youth and Adolescence*. 2016;45(2):350-360. doi:10.1007/s10964-015-0278-0
- Van Ouytsel J, Ponnet K, Walrave M. Cyber Dating Abuse: Investigating Digital Monitoring Behaviors Among Adolescents From a Social Learning Perspective. J Interpers Violence. Published online July 20, 2017:088626051771953. doi:10.1177/0886260517719538
- 84. Foshee VA, McNaughton Reyes HL, Ennett ST, et al. Risk and Protective Factors Distinguishing Profiles of Adolescent Peer and Dating Violence Perpetration. *Journal of Adolescent Health*. 2011;48(4):344-350. doi:10.1016/j.jadohealth.2010.07.030
- Schnurr MP, Lohman BJ. How Much Does School Matter? An Examination of Adolescent Dating Violence Perpetration. *Journal of Youth and Adolescence*. 2008;37(3):266-283. doi:10.1007/s10964-007-9246-7
- 86. Deitch-Stackhouse J, Kenneavy K, Thayer R, Berkowitz A, Mascari J. The Influence of Social Norms on Advancement Through Bystander Stages for Preventing Interpersonal Violence. *Violence Against Women*. 2015;21(10):1284-1307. doi:10.1177/1077801215592720
- 87. Gupta GR. Gender, sexuality, and HIV/AIDS: The what, the why, and the how. Presented at: XIIIth International AIDS Conference; July 12, 2000; Durban, South Africa.
- 88. Jewkes R. Intimate partner violence: causes and prevention. *Lancet*. 2002;359(9315):1423-1429. doi:10.1016/S0140-6736(02)08357-5
- Jewkes R, Flood M, Lang J. From work with men and boys to changes of social norms and reduction of inequities in gender relations: a conceptual shift in prevention of violence against women and girls. *The Lancet*. 2015;385(9977):1580-1589. doi:10.1016/S0140-6736(14)61683-4

- 90. Fulu E, Kerr-Wilson L, Lang J. What Works to Prevent Violence against Women and Girls? Evidence Review of Interventions to Prevent Violence against Women and Girls.; 2014.
- 91. Connell R. Teaching the boys: New research on masculinity, and gender strategies for schools. *Teachers College Record*. 1996;98(2).
- 92. Kågesten A, Gibbs S, Blum RW, et al. Understanding Factors that Shape Gender Attitudes in Early Adolescence Globally: A Mixed-Methods Systematic Review. Dalby AR, ed. *PLOS ONE*. 2016;11(6):e0157805. doi:10.1371/journal.pone.0157805
- 93. Hamby S, Nix K, De Puy J, Monnier S. Adapting dating violence prevention to francophone Switzerland: a story of intra-western cultural differences. *Violence Vict*. 2012;27(1):33-42.
- 94. Cislaghi B, Heise L. Gender norms and social norms: differences, similarities and why they matter in prevention science. *Sociol Health Illn*. 2020;42(2):407-422. doi:10.1111/1467-9566.13008
- Foxcroft DR, Moreira MT, Almeida Santimano NM, Smith LA. Social norms information for alcohol misuse in university and college students. In: The Cochrane Collaboration, ed. *Cochrane Database of Systematic Reviews*. John Wiley & Sons, Ltd; 2015:CD006748.pub3. doi:10.1002/14651858.CD006748.pub3
- Pulerwitz J, Michaelis A, Verma R, Weiss E. Addressing gender dynamics and engaging men in HIV programs: lessons learned from Horizons research. *Public Health Rep*. 2010;125(2):282-292. doi:10.1177/003335491012500219
- 97. Marcus R, Harper C, Brodbeck S, Page E. How Do Gender Norms Change? ODI; 2015.
- 98. Dworkin SL, Treves-Kagan S, Lippman SA. Gender-Transformative Interventions to Reduce HIV Risks and Violence with Heterosexually-Active Men: A Review of the Global Evidence. *AIDS and Behavior*. 2013;17(9):2845-2863. doi:10.1007/s10461-013-0565-2
- Bicchieri C, Mercier H. Norm and Beliefs: How Change Occurs. In: Xenitidou M, Edmonds B, eds. *The Complexity of Social Norms*. Computational Social Sciences. Springer Cham; 2014. doi:10.1007/978-3-319-05308-0_3
- 100. Ajzen I. The theory of planned behavior. *Organizational behavior and human decision processes*. 1991;50:179-211.
- 101. Mackie G, Moneti F, Shakya H, Elaine D. *What Are Social Norms? How Are They Measured? Working Paper*. UNICEF/Centre on Global Justice, University of San Diego; 2015.
- 102. Chung A, Rimal RN. Social norms: A review. *Review of Communication Research*. 2016;4(1):1-28. doi:10.12840/issn.2255-4165.2016.04.01.008
- 103. Cislaghi B, Heise L. Four avenues of normative influence: A research agenda for health promotion in low and mid-income countries. *Health Psychology*. 2018;37(6):562-573. doi:10.1037/hea0000618
- 104. Cislaghi B, Heise L. *Measuring Gender-Related Social Norms: Report of a Meeting, Baltimore Maryland, June 14-15, 2016.* Learning Group on Social Norms and Gender-based Violence of the London School of Hygiene and Tropical Medicine; 2017.
- 105. Cislaghi B, Shakya H. Social Norms and Adolescents' Sexual Health: An Introduction for Practitioners Working in Low and Mid-income African countries. *African journal of reproductive health*. 2018;22(1):38-46. doi:10.29063/ajrh2018/v22i1.4
- 106. Cislaghi B, Heise L. Theory and practice of social norms interventions: eight common pitfalls. *Globalization and Health*. 2018;14(1). doi:10.1186/s12992-018-0398-x
- 107. Giddens A. The Constitution of Society: Outline of the Theory of Structuration. Polity; 1984.
- 108. Bandura A. Health Promotion from the Perspective of Social Cognitive Theory. *Psychology and Health*. 1998;13:623-649.
- 109. Luszczynska A, Schwarzer R. 3 Changing behavior using social cognitive theory, from part I – theory and behavior change. In: Hagger M, Cameron L, Hamilton K, Hankonen N, Lintunen T, eds. *The Handbook of Behavior Change*. Cambridge University Press; 2020.
- 110. Cislaghi B, Heise L. Using social norms theory for health promotion in low-income countries. *Health Promotion International*. Published online March 22, 2018. doi:10.1093/heapro/day017
- 111. Cislaghi B, Manji K, Heise L. *Social Norms and Gender-Related Harmful Practices, Learning Report 2: Theory in Support of Better Practice.* Learning Group on Social Norms and Gender-related Harmful Practices, London School of Hygiene & Tropical Medicine; 2018.
- 112. Heise L, Manji K. Social Norms. University of Birmingham; 2016.
- 113. Facilitation Guide: Social Norms Mentorship Programme. London School of Hygiene & Tropical Medicine; 2020. https://www.lshtm.ac.uk/research/centres-projects-groups/social-norms-mentorship-project#snmp-resources-and-templates
- 114. Marcus R, Harper C, Brodbeck S, Page E. *Social Norms, Gender Norms and Adolescent Girls: A Brief Guide*. ODI; 2015.
- 115. Bem SL. Gender schema theory: A cognitive account of sex typing. *Psychological Review*. 1981;88(4):354-364.

- 116. Lokot M, Bhatia A, Kenny L, Cislaghi B. Corporal punishment, discipline and social norms: A systematic review in low- and middle-income countries. *Aggression and Violent Behavior*. 2020;55:101507. doi:10.1016/j.avb.2020.101507
- 117. Marston C, King E. Factors that shape young people's sexual behaviour: a systematic review. *Lancet*. 2006;368(9547):1581-1586. doi:10.1016/S0140-6736(06)69662-1
- 118. De Meyer S, Kågesten A, Mmari K, et al. "Boys Should Have the Courage to Ask a Girl Out": Gender Norms in Early Adolescent Romantic Relationships. *Journal of Adolescent Health*. 2017;61(4):S42-S47. doi:10.1016/j.jadohealth.2017.03.007
- 119. Yu C, Zuo X, Blum RW, et al. Marching to a Different Drummer: A Cross-Cultural Comparison of Young Adolescents Who Challenge Gender Norms. *Journal of Adolescent Health*. 2017;61(4):S48-S54. doi:10.1016/j.jadohealth.2017.07.005
- 120. Aizpitarte A, Alonso-Arbiol I, Van de Vijver FJR. An Explanatory Model of Dating Violence Risk Factors in Spanish Adolescents. J Res Adolesc. 2017;27(4):797-809. doi:10.1111/jora.12315
- 121. Helland TA. The role of the peer group on individual use and acceptance of physical aggression in adolescent dating relationships. *Dissertation Abstracts International: Section B: The Sciences and Engineering*. 1998;58(8-B):4450.
- 122. Kinsfogel KM, Grych JH. Interparental Conflict and Adolescent Dating Relationships: Integrating Cognitive, Emotional, and Peer Influences. *Journal of Family Psychology*. 2004;18(3):505-515. doi:10.1037/0893-3200.18.3.505
- 123. Price EL. Risk factors for boys' psychologically abusive behaviour in dating relationships. *Dissertation Abstracts International: Section B: The Sciences and Engineering*. 2002;62(10-B):4800.
- 124. Shorey RC, Wymbs B, Torres L, Cohen JR, Fite PJ, Temple JR. Does change in perceptions of peer teen dating violence predict change in teen dating violence perpetration over time? *Aggr Behav.* 2018;44(2):156-164. doi:10.1002/ab.21739
- 125. Gonzalez-Mendez R, Aguilera L, Ramírez-Santana G. Weighing Risk Factors for Adolescent Victimization in the Context of Romantic Relationship Initiation. J Interpers Violence. Published online April 15, 2019:088626051984328. doi:10.1177/0886260519843284
- 126. Hébert M, Moreau C, Blais M, Oussaïd E, Lavoie F. A three-step gendered latent class analysis on dating victimization profiles. *Psychology of Violence*. 2019;9(5):504-516. doi:10.1037/vio0000225
- 127. Pöllänen K, de Vries H, Mathews C, Schneider F, de Vries PJ. Beliefs About Sexual Intimate Partner Violence Perpetration Among Adolescents in South Africa. J Interpers Violence. 2018;36(3-4):NP2056-2078NP. doi:10.1177/0886260518756114

- 128. Nardi-Rodríguez A, Pastor-Mira MÁ, López-Roig S, Pamies-Aubalat L, Martínez-Zaragoza F, Ferrer-Pérez VA. Predicting Abusive Behaviours in Spanish Adolescents' Relationships: Insights from the Reasoned Action Approach. *IJERPH*. 2022;19(3):1441. doi:10.3390/ijerph19031441
- 129. Gage AJ. Exposure to Spousal Violence in the Family, Attitudes and Dating Violence Perpetration Among High School Students in Port-au-Prince. *Journal of Interpersonal Violence*. 2016;31(14):2445-2474. doi:10.1177/0886260515576971
- Peskin MF, Markham CM, Shegog R, et al. Prevalence and Correlates of the Perpetration of Cyber Dating Abuse among Early Adolescents. J Youth Adolescence. 2017;46(2):358-375. doi:10.1007/s10964-016-0568-1
- 131. Flisher AJ, Myer L, Mèrais A, Lombard C, Reddy P. Prevalence and correlates of partner violence among South African adolescents. J Child Psychol & Psychiat. 2007;48(6):619-627. doi:10.1111/j.1469-7610.2007.01711.x
- Hopper L. Contributions of individual and friend attitudes to dating violence experience in adolescents. *Dissertation Abstracts International: Section B: The Sciences and Engineering*. 2011;73(4-B):2563.
- 133. Wesche R, Dickson-Gomez J. Gender Attitudes, Sexual Risk, Intimate Partner Violence, and Coercive Sex Among Adolescent Gang Members. *Journal of Adolescent Health*. 2019;64(5):648-656. doi:10.1016/j.jadohealth.2018.10.292
- 134. Heise LL, Kotsadam A. Cross-national and multilevel correlates of partner violence: an analysis of data from population-based surveys. *The Lancet Global Health*. 2015;3(6):e332-e340. doi:10.1016/S2214-109X(15)00013-3
- 135. Shakya HB, Cislaghi B, Fleming P, et al. Associations of attitudes and social norms with experiences of intimate partner violence among married adolescents and their husbands in rural Niger: a dyadic cross-sectional study. *BMC Women's Health*. 2022;22(1):180. doi:10.1186/s12905-022-01724-y
- 136. Shamu S, Gevers A, Mahlangu BP, Jama Shai PN, Chirwa ED, Jewkes RK. Prevalence and risk factors for intimate partner violence among Grade 8 learners in urban South Africa: baseline analysis from the Skhokho Supporting Success cluster randomised controlled trial. *Int Health*. 2016;8(1):18-26. doi:10.1093/inthealth/ihv068
- 137. Foshee VA, Bauman KE, Ennett ST, Suchindran C, Benefield T, Linder GF. Assessing the Effects of the Dating Violence Prevention Program "Safe Dates" Using Random Coefficient Regression Modeling. *Prevention Science*. 2005;6(3):245-258. doi:10.1007/s11121-005-0007-0
- 138. Wood M, Barter C, Berridge D. 'Standing on My Own Two Feet': Disadvantaged Teenagers, Intimate Partner Violence and Coercive Control. NSPCC; 2011.

- Barter C. Discourses of blame: deconstructing (hetero)sexuality, peer sexual violence and residential children's homes. *Child & Family Social Work*. 2006;11(4):347-356. doi:10.1111/j.1365-2206.2006.00425.x
- 140. Choudhury S, Blakemore SJ, Charman T. Social cognitive development during adolescence. *Social Cognitive and Affective Neuroscience*. 2006;1(3):165-174. doi:10.1093/scan/nsl024
- 141. Johnson SB, Blum RW, Giedd JN. Adolescent maturity and the brain: the promise and pitfalls of neuroscience research in adolescent health policy. *J Adolesc Health*. 2009;45(3):216-221. doi:10.1016/j.jadohealth.2009.05.016
- 142. Coker AL, Bush HM, Cook-Craig PG, et al. RCT Testing Bystander Effectiveness to Reduce Violence. *American Journal of Preventive Medicine*. 2017;52(5):566-578. doi:10.1016/j.amepre.2017.01.020
- 143. Edwards KM, Banyard VL, Sessarego SN, Waterman EA, Mitchell KJ, Chang H. Evaluation of a Bystander-Focused Interpersonal Violence Prevention Program with High School Students. *Prev Sci.* 2019;20(4):488-498. doi:10.1007/s11121-019-01000-w
- 144. Miller E, Tancredi DJ, McCauley HL, et al. "Coaching Boys into Men": A Cluster-Randomized Controlled Trial of a Dating Violence Prevention Program. *Journal of Adolescent Health*. 2012;51(5):431-438. doi:10.1016/j.jadohealth.2012.01.018
- 145. Miller E, Tancredi DJ, McCauley HL, et al. One-Year Follow-Up of a Coach-Delivered Dating Violence Prevention Program. American Journal of Preventive Medicine. 2013;45(1):108-112. doi:10.1016/j.amepre.2013.03.007
- 146. Taylor B, Stein ND, Woods D, Mumford E. *Shifting Boundaries: Final Report on an Experimental Evaluation of a Youth Dating Violence Prevention Program in New York City Middle Schools*. U.S. Department of Justice; 2011.
- 147. Taylor BG, Mumford EA, Stein ND. Effectiveness of "Shifting Boundaries" Teen Dating Violence Prevention Program for Subgroups of Middle School Students. *Journal of Adolescent Health*. 2015;56(2):S20-S26. doi:10.1016/j.jadohealth.2014.07.004
- 148. Ehrensaft MK, Westfall HK, Niolon PH, et al. Can a Parenting Intervention to Prevent Early Conduct Problems Interrupt Girls' Risk for Intimate Partner Violence 10 Years Later? *Prev Sci.* 2018;19(4):449-458. doi:10.1007/s11121-017-0831-z
- 149. Plourde C, Shore N, Herrick P, et al. You the man: theater as bystander education in dating violence. *Arts & Health*. 2016;8(3):229-247. doi:10.1080/17533015.2015.1091017
- 150. Cislaghi B. LSHTM Group on Social Norms and GBV: Strategies to Diagnose and Measure Social Norms Related to Gender-Based Violence: Key Lessons from the Baltimore Working Meeting. STRIVE; 2016.

- 151. Clark CJ, Ferguson G, Shrestha B, et al. Social norms and women's risk of intimate partner violence in Nepal. *Social Science & Medicine*. 2018;202:162-169. doi:10.1016/j.socscimed.2018.02.017
- 152. Moreau C, Li M, De Meyer S, et al. Measuring gender norms about relationships in early adolescence: Results from the global early adolescent study. SSM - Population Health. 2019;7:100314. doi:10.1016/j.ssmph.2018.10.014
- 153. Perrin N, Marsh M, Clough A, et al. Social norms and beliefs about gender based violence scale: a measure for use with gender based violence prevention programs in low-resource and humanitarian settings. *Conflict and Health*. 2019;13(1):6. doi:10.1186/s13031-019-0189-x
- 154. Top 20 Resources on Social Norms.
- 155. Passages Project: Transforming social norms for sexual and reproductive health. Presented at: 2017.
- 156. Mulla MM, Haikalis M, Orchowski LM, Berkowitz AD. The Prospective Influence of Perceived Social Norms on Bystander Actions Against Sexual Violence and Relationship Abuse: A Multiple Mediation Model. *J Interpers Violence*. Published online July 3, 2020:088626052093303. doi:10.1177/0886260520933035
- 157. Rothman EF, Edwards KM, Rizzo AJ, Kearns M, Banyard VL. Perceptions of Community Norms and Youths' Reactive and Proactive Dating and Sexual Violence Bystander Action. *American Journal of Community Psychology*. 2019;63(1-2):122-134. doi:10.1002/ajcp.12312
- 158. Moreau C. Development and validation of cross-cultural gender norms scales for early adolescents. Webinar presented at: February 1, 2018; Global Early Adolescent Study. http://www.geastudy.org/webinars/
- 159. Moreau C, Li M, Ahmed S, Zuo X, Cislaghi B. Assessing the Spectrum of Gender Norms Perceptions in Early Adolescence: A Cross-Cultural Analysis of the Global Early Adolescent Study. *Journal of Adolescent Health*. 2021;69(1):S16-S22. doi:10.1016/j.jadohealth.2021.03.010
- 160. Social Norms Mentorship Programme training slides. SNMP resources and templates. 2020. Accessed May 3, 2023. https://www.lshtm.ac.uk/research/centres-projects-groups/social-norms-mentorship-project#snmp-resources-and-templates
- 161. Moneti F, Mackie G. The general considerations in measuring social norms. Presented at: 2013. http://strive.lshtm.ac.uk/sites/strive.lshtm.ac.uk/files/Gerry%20Mackie%20General%20co nsiderations%20in% 20measuring%20social%20norms.pdf

- 162. Stefanik L, Hwang T. *Applying Theory to Practice: CARE's Journey Piloting Social Norms Measures for Gender Programming.* Cooperative for Assistance and Relief Everywhere, Inc. (CARE); 2017.
- Craig P, Dieppe P, Macintyre S, Michie S, Nazareth I, Petticrew M. Developing and evaluating complex interventions: the new Medical Research Council guidance. *BMJ*. 2008;337:a1655.
- 164. Foshee VA, Linder GF, Bauman KE, et al. The Safe Dates Project: theoretical basis, evaluation design, and selected baseline findings. *Am J Prev Med*. 1996;12(5 Suppl):39-47.
- 165. Foshee VA, Bauman KE, Ennett ST, Linder GF, Benefield T, Suchindran C. Assessing the long-term effects of the Safe Dates program and a booster in preventing and reducing adolescent dating violence victimization and perpetration. *Am J Public Health*. 2004;94(4):619-624.
- 166. ALPHA: DECIPHer's research advisory group of young people. DECIPHer. Accessed February 16, 2024. https://decipher.uk.net/public-health-improvement-researchnetworks-phirns/alpha/
- 167. Foshee VA, Bauman KE, Arriaga XB, Helms RW, Koch GG, Linder GF. An evaluation of Safe Dates, an adolescent dating violence prevention program. *Am J Public Health*. 1998;88(1):45-50.
- 168. Taylor BG, Stein ND, Mumford E, Woods D. Shifting Boundaries: An experimental evaluation of a dating violence prevention program in middle schools. *Prevention Science*. 2013;14:64-76.
- 169. Hawkins JD, Weis JG. The social development model: An integrated approach to delinquency prevention. *The Journal of Primary Prevention*. 1985;6(2):73-97. doi:10.1007/BF01325432
- 170. Cambron C, Catalano RF, Hawkins JD. The Social Development Model [submitted version]. In: Farrington DP, Kazemian L, Piquero AR, eds. *The Oxford Handbook of Developmental and Life-Course Criminology*. Oxford University Press; 2019:223-247. doi:10.1093/oxfordhb/9780190201371.013.13
- 171. Moore G, Audrey S, Barker M, et al. *Process Evaluation of Complex Interventions: UK Medical Research Council (MRC) Guidance*. MRC Population Health Science Research Network; 2014.
- 172. DCLG Indices of Deprivation 2015: i. Income Deprivation Affecting Children Index (IDACI). Open data; Department for Levelling Up, Housing and Communities. Accessed March 24, 2024. https://opendatacommunities.org/def/concept/general-concepts/imd/idaci

- 173. Department for Education. *Progress 8: How Progress 8 and Attainment 8 Measures Are Calculated*. Department for Education; 2016. Accessed March 24, 2024. https://dera.ioe.ac.uk/id/eprint/28143/
- 174. Bonell C. Pilot evaluation to assess the feasibility and acceptability of Project Respect: a school-based intervention to prevent dating and relationship violence among young people. doi:10.1186/ISRCTN65324176
- 175. Buller AM, Pichon M, McAlpine A, Cislaghi B, Heise L, Meiksin R. Systematic review of social norms, attitudes, and factual beliefs linked to the sexual exploitation of children and adolescents. *Child Abuse & Neglect*. 2020;104:104471. doi:10.1016/j.chiabu.2020.104471
- 176. Alderson P. Childhoods Real and Imagined: An Introduction to Childhood Studies and Critical Realism, Volume 1. Routledge; 2013.
- 177. Gorski PS. Review: "What is Critical Realism? And Why Should You Care?" *Contemporary Sociology*. 2013;42(5):658-670.
- 178. Green J, Thorogood N. Qualitative Methods for Health Research. Sage Publications; 2018.
- 179. Maxwell J, Mittapalli K. Realism as a stance for mixed methods research. In: Tashakkori A, Teddlie C, eds. *SAGE Handbook of Mixed Methods in Social & Behavioral Research*. 2nd ed. SAGE Publications, Inc; 2010:145-168.
- 180. Brunson L, Lauzier-Jobin F, Olson B, Côté L. Seven key insights from critical realism and their implications for ecological thinking and action in community psychology. *Journal Community Psychology*. Published online June 5, 2023: jcop.23054. doi:10.1002/jcop.23054
- 181. Bhaskar R. On the possibility of social scientific knowledge and the limits of naturalism. Journal for the theory of social behaviour. 1978;8(1):1-28. doi:10.1111/j.1468-5914.1978.tb00389.x
- 182. Manyukhina Y, Alderson P. Healthcare and critical realism, part 2. Presented at: Introductory and basic refresher day course; March 3, 2018; University College London Institute of Education.
- 183. Dalkin SM, Greenhalgh J, Jones D, Cunningham B, Lhussier M. What's in a mechanism? Development of a key concept in realist evaluation. *Implementation Science*. 2015;10(1). doi:10.1186/s13012-015-0237-x
- 184. Sayer RA. Method in Social Science: A Realist Approach. Rev. 2nd ed. Routledge; 2010.
- 185. Bonell C, Melendez-Torres GJ, Warren E. *Realist Trials and Systematic Reviews: Rigorous,* Useful Evidence to Inform Health Policy. Cambridge University Press; 2024.

- 186. Creswell JW, Klassen AC, Clark VLP, Smith KC. *Best Practices for Mixed Methods Research in the Health Sciences*. Office of Behavioral and Social Sciences Research
- 187. DeVellis RF. Scale Development: Theory and Applications. Fourth edition. SAGE; 2017.
- 188. Cappelleri JC, Jason Lundy J, Hays RD. Overview of Classical Test Theory and Item Response Theory for the Quantitative Assessment of Items in Developing Patient-Reported Outcomes Measures. *Clinical Therapeutics*. 2014;36(5):648-662. doi:10.1016/j.clinthera.2014.04.006
- 189. Streiner DL, Norman GR. Health Measurement Scales: A Practical Guide to Their Development and Use. 4th ed. Oxford University Press; 2008. doi:10.1093/acprof:oso/9780199231881.001.0001
- 190. Prinsen CAC, Mokkink LB, Bouter LM, et al. COSMIN guideline for systematic reviews of patient-reported outcome measures. *Quality of Life Research*. 2018;27(5):1147-1157. doi:10.1007/s11136-018-1798-3
- 191. Hill L, Maucione K, Hood B. A Focused Approach to Assessing Program Fidelity. *Prevention Science*. 2007;8:25-34.
- 192. Hughes DJ. Psychometric Validity: Establishing the Accuracy and Appropriateness of Psychometric Measures. In: Irwing P, Booth T, Hughes DJ, eds. *The Wiley Handbook of Psychometric Testing*. 1st ed. Wiley; 2018:751-779. doi:10.1002/9781118489772.ch24
- 193. Meiksin R, Bonell C, Bhatia A, Melendez-Torres GJ, Kyegombe N, Kohli A. Social Norms About Dating and Relationship Violence and Gender Among Adolescents: Systematic Review of Measures Used in Dating and Relationship Violence Research. *Trauma, Violence,* & Abuse. Published online February 24, 2023:152483802311555. doi:10.1177/15248380231155526
- 194. Chalmers I, Hedges LV, Cooper H. A Brief History of Research Synthesis. *Eval Health Prof.* 2002;25(1):12-37. doi:10.1177/0163278702025001003
- 195. Aromataris E, Pearson A. The Systematic Review: An Overview. *AJN, American Journal of Nursing*. 2014;114(3):53-58. doi:10.1097/01.NAJ.0000444496.24228.2c
- 196. Munn Z, Stern C, Aromataris E, Lockwood C, Jordan Z. What kind of systematic review should I conduct? A proposed typology and guidance for systematic reviewers in the medical and health sciences. *BMC Med Res Methodol*. 2018;18(1):5. doi:10.1186/s12874-017-0468-4
- 197. The history of social media: Social networking evolution! History Cooperative. Accessed June 3, 2019. https://historycooperative.org/the-history-of-social-media/

- 198. McGeeney E, Hanson E. Digital Romance: A Research Project Exploring Young People's Use of Technology in Their Romantic Relationships and Love Lives. National Crime Agency and Brook; 2017. https://www.thinkuknow.co.uk/professionals/guidance/digital-romance/
- 199. McGowan J, Sampson M, Salzwedel DM, Cogo E, Foerster V, Lefebvre C. PRESS Peer Review of Electronic Search Strategies: 2015 Guideline Statement. *Journal of Clinical Epidemiology*. 2016;75:40-46. doi:10.1016/j.jclinepi.2016.01.021
- 200. Shamseer L, Moher D, Clarke M, et al. Preferred reporting items for systematic review and meta-analysis protocols (PRISMA-P) 2015: elaboration and explanation. *BMJ*. 2015;350(jan02 1):g7647-g7647. doi:10.1136/bmj.g7647
- 201. Removing duplicates from an EndNote library. Library & Archives Service Blog. December 7, 2018. Accessed February 26, 2020. http://blogs.lshtm.ac.uk/library/2018/12/07/removing-duplicates-from-an-endnotelibrary/
- 202. Terwee CB, Bot SDM, de Boer MR, et al. Quality criteria were proposed for measurement properties of health status questionnaires. *Journal of Clinical Epidemiology*. 2007;60(1):34-42. doi:10.1016/j.jclinepi.2006.03.012
- 203. Moher D, Liberati A, Tetzlaff J, Altman DG, for the PRISMA Group. Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. *BMJ*. 2009;339(jul21 1):b2535-b2535. doi:10.1136/bmj.b2535
- 204. de Leeuw E, Borgers N, Strijbos-Smits A. Children as respondents: Developing, evaluating, and testing questionnaires for children. In: ; 2002.
- 205. DeGue S, Valle LA, Holt MK, Massetti GM, Matjasko JL, Tharp AT. A systematic review of primary prevention strategies for sexual violence perpetration. *Aggression and Violent Behavior*. 2014;19(4):346-362. doi:10.1016/j.avb.2014.05.004
- 206. Agnew-Brune CB. "It's a Bad Thing...but It's a Good Thing Too": A Mixed Methods Examination of Technology Use and Cyber Dating Abuse Perpetration in Adolescent Romantic Relationships. Chapel Hill; 2016.
- 207. Hennegan J, Brooks DJ, Schwab KJ, Melendez-Torres GJ. Measurement in the study of menstrual health and hygiene: A systematic review and audit. Pant Pai N, ed. *PLoS ONE*. 2020;15(6):e0232935. doi:10.1371/journal.pone.0232935
- 208. Costenbader E, Lenzi R, Hershow RB, Ashburn K, McCarraher DR. Measurement of Social Norms Affecting Modern Contraceptive Use: A Literature Review. *Stud Fam Plann*. 2017;48(4):377-389. doi:10.1111/sifp.12040
- 209. Doherty S, Oram S, Siriwardhana C, Abas M. Suitability of measurements used to assess mental health outcomes in men and women trafficked for sexual and labour exploitation:

a systematic review. *The Lancet Psychiatry*. 2016;3(5):464-471. doi:10.1016/S2215-0366(16)30047-5

- 210. Lewis CC, Stanick CF, Martinez RG, et al. The Society for Implementation Research Collaboration Instrument Review Project: A methodology to promote rigorous evaluation. *Implementation Science*. 2015;10(1). doi:10.1186/s13012-014-0193-x
- 211. Pocock NS, Chan CW, Zimmerman C. Suitability of Measurement Tools for Assessing the Prevalence of Child Domestic Work: A Rapid Systematic Review. *Int J Environ Res Public Health*. 2021;18(5):2357. doi:10.3390/ijerph18052357
- 212. Meiksin R, Ponsford R, Kyegombe N, Kohli A, Bonell C. Assessing Survey Items on Social Norms Relating to Dating and Relationship Violence and to Gender: Cognitive Interviews with Young People in England. *J Interpers Violence*. Published online October 21, 2023:08862605231204561. doi:10.1177/08862605231204561
- 213. Willis GB, Artino AR. What Do Our Respondents Think We're Asking? Using Cognitive Interviewing to Improve Medical Education Surveys. *Journal of Graduate Medical Education*. 2013;5(3):353-356. doi:10.4300/JGME-D-13-00154.1
- 214. Collins D. Pretesting survey instruments: an overview of cognitive methods. *Qual Life Res*. 2003;12(3):229-238. doi:10.1023/a:1023254226592
- 215. Willis GB. Cognitive Interviewing, A "How To" Guide. Research Triangle Institute; 1999.
- 216. Cook-Craig PG, Coker AL, Clear ER, et al. Challenge and Opportunity in Evaluating a Diffusion-Based Active Bystanding Prevention Program: Green Dot in High Schools. *Violence Against Women*. 2014;20(10):1179-1202. doi:10.1177/1077801214551288
- 217. Sotiriou P, Ntinapogias S, Petroulaki K. *Attitudes on Gender Stereotypes and Gender-Based Violence among Youth. Country Report: Greece*. European Anti-Violence Network; 2011.
- 218. Young H, Költő A, Reis M, et al. Sexual Health questions included in the Health Behaviour in School-aged Children (HBSC) Study: an international methodological pilot investigation. *BMC Med Res Methodol*. 2016;16(1):169. doi:10.1186/s12874-016-0270-8
- 219. Lever J, Frederick DA, Hertz R. Who Pays for Dates? Following Versus Challenging Gender Norms. *SAGE Open*. 2015;5(4):215824401561310. doi:10.1177/2158244015613107
- 220. Glick P, Fiske ST. An ambivalent alliance: Hostile and benevolent sexism as complementary justifications for gender inequality. *American Psychologist*. 2001;56(2):109-118. doi:10.1037/0003-066X.56.2.109
- 221. Crawford M, Popp D. Sexual double standards: a review and methodological critique of two decades of research. *The Journal of Sex Research*. 2003;40(1):13-26.

- 222. Fernández-González L, Wekerle C, Goldstein AL. Measuring adolescent dating violence: Development of 'conflict in adolescent dating relationships inventory' short form. Advances in Mental Health. 2012;11(1):35-54. doi:10.5172/jamh.2012.11.1.35
- 223. The GenIUSS Group. Best Practices for Asking Questions to Identify Transgender and Other Gender Minority Respondents on Population-Based Surveys. (Herman J, ed.). The Williams Institute; 2014.
- 224. Harmonised Concepts and Questions for Social Data Sources: Primary Principles, Ethnic Group, Version 3.3. Office for National Statistics; 2015. Accessed February 5, 2024. https://analysisfunction.civilservice.gov.uk/wp-content/uploads/2016/03/P3-Ethnic-Group-June-16-1.pdf
- 225. Harmonised Concepts and Questions for Social Data Sources: Secondary Principles, National Identity and Religion, Version 4.2. Office for National Statistics; 2015. Accessed February 5, 2024. https://analysisfunction.civilservice.gov.uk/wpcontent/uploads/2016/03/S11-National-Identity-and-Religion-June-16.pdf
- 226. Currie C, Molcho M, Boyce W, Holstein B, Torsheim T, Richter M. Researching health inequalities in adolescents: The development of the Health Behaviour in School-Aged Children (HBSC) Family Affluence Scale. *Social Science & Medicine*. 2008;66(6):1429-1436. doi:10.1016/j.socscimed.2007.11.024
- 227. Zumbo BD, Gadermann AM, Zeisser C. Ordinal Versions of Coefficients Alpha and Theta for Likert Rating Scales. *Journal of Modern Applied Statistical Methods*. 2007;6(1):21-29. doi:10.22237/jmasm/1177992180
- 228. Boulton AJ, Williford A. Analyzing Skewed Continuous Outcomes With Many Zeros: A Tutorial for Social Work and Youth Prevention Science Researchers. *Journal of the Society for Social Work and Research*. 2018;9(4):721-740. doi:10.1086/701235
- 229. Fox J. Bootstrapping regression models. In: *Applied Regression Analysis and Generalized Linear Models*. 3rd ed. SAGE Publications Inc.; 2015:587-606. Accessed October 13, 2024. https://www.sagepub.com/sites/default/files/upm-binaries/21122_Chapter_21.pdf
- Smith J, Mulford C, Latzman NE, Tharp AT, Niolon PH, Blachman-Demner D. Taking Stock of Behavioral Measures of Adolescent Dating Violence. *Journal of Aggression, Maltreatment* & Trauma. 2015;24(6):674-692. doi:10.1080/10926771.2015.1049767
- 231. Pawson R, Tilley N. Realist Evaluation.; 2004.
- 232. Meiksin R, Melendez-Torres GJ, Falconer J, Witzel TC, Weatherburn P, Bonell C. Theories of change for e-health interventions targeting HIV/STIs and sexual risk, substance use and mental ill health amongst men who have sex with men: systematic review and synthesis. *Syst Rev.* 2021;10(1):21. doi:10.1186/s13643-020-01523-2

- 233. Meiksin R, Bonell C, Bhatia A, Melendez-Torres G, Kyegombe N, Kohli A. Social norms concerning dating and relationship violence and gender among adolescents: a systematic review of survey measures used in dating and relationship violence research [poster]. Presented at: Lancet UK Public Health Science Conference; 2022; Glasgow, UK.
- 234. Meiksin R, Bonell C, Bhatia A, et al. Measuring social norms relating to sexual and reproductive health and dating and relationship violence among young people: Preliminary findings from a systematic review and cognitive testing. Presented at: LINEA webinar series; October 7, 2020; Webinar.
- 235. Ponsford R, Bragg S, Allen E, et al. A school-based social-marketing intervention to promote sexual health in English secondary schools: the Positive Choices pilot cluster RCT. *Public Health Res.* 2021;9(1):1-190. doi:10.3310/phr09010
- 236. Meiksin R, Ponsford R, Bonell C. Cognitive interviews with adolescents in England to develop survey items measuring social norms relating to sexual behaviour, dating and relationship violence and gender. Presented at: Third LINEA Biennial Meeting; October 13, 2020; Virtual.
- 237. Ponsford R, Crichton J, Meiksin R, et al. Stakeholder involvement in the systematic optimisation of the Project Respect and Positive Choices school based relationships and sex education interventions [poster]. Presented at: Lancet UK Public Health Science Conference; November 2018; Blefast, Northern Ireland.
- 238. Barter C, Aghtaie N, Larkins C, et al. Safeguarding Teenage Intimate Relationships (STIR). Connecting Online and Offline Contexts and Risks. Briefing Paper 2: Incidence Rates and Impact of Experiencing Interpersonal Violence and Abuse in Young People's Relationships. University of Bristol; 2014.
- 239. Rogers EM. Diffusion of Innovations. 3rd ed. The Free Press; 1983.
- 240. Hawkins AJ. Realist evaluation and randomised controlled trials for testing program theory in complex social systems. *Evaluation*. 2016;22(3):270-285. doi:10.1177/1356389016652744
- 241. Gagné MH, Lavoie F, Hébert M. Victimization during childhood and revictimization in dating relationships in adolescent girls. *Child Abuse Negl*. 2005;29(10):1155-1172.
- 242. Bennett DA. How can I deal with missing data in my study? *Australian and New Zealand Journal of Public Health*. 2001;25(5):464-469. doi:10.1111/j.1467-842X.2001.tb00294.x
- 243. Xia Y, Yang Y. RMSEA, CFI, and TLI in structural equation modeling with ordered categorical data: The story they tell depends on the estimation methods. *Behav Res Methods*. 2019;51(1):409-428. doi:10.3758/s13428-018-1055-2
- 244. StataCorp. Stata statistical software: release 18. Published online 2023.

- 245. Hu L, Bentler PM. Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives. *Structural Equation Modeling: A Multidisciplinary Journal*. 1999;6(1):1-55. doi:10.1080/10705519909540118
- 246. Sonderen E van, Sanderman R, Coyne JC. Ineffectiveness of Reverse Wording of Questionnaire Items: Let's Learn from Cows in the Rain. Baradaran HR, ed. *PLoS ONE*. 2013;8(7):e68967. doi:10.1371/journal.pone.0068967
- 247. Reidy DE, Smith-Darden JP, Cortina KS, Kernsmith RM, Kernsmith PD. Masculine Discrepancy Stress, Teen Dating Violence, and Sexual Violence Perpetration Among Adolescent Boys. *Journal of Adolescent Health*. 2015;56(6):619-624. doi:10.1016/j.jadohealth.2015.02.009
- 248. Fleming PJ, Agnew-Brune C. Current trends in the study of gender norms and health behaviors. *Current Opinion in Psychology*. 2015;5:72-77. doi:10.1016/j.copsyc.2015.05.001
- 249. Fox CL, Hale R, Gadd D. Domestic abuse prevention education: listening to the views of young people. *Sex Education*. 2014;14(1):28-41. doi:10.1080/14681811.2013.816949
- 250. Exner-Cortens D, Gill L, Eckenrode J. Measurement of adolescent dating violence: A comprehensive review (Part 2, attitudes). *Aggression and Violent Behavior*. 2016;27:93-106. doi:10.1016/j.avb.2016.02.011
- 251. Pusch N. A Meta-Analytic Review of Social Learning Theory and Teen Dating Violence Perpetration. *Journal of Research in Crime and Delinquency*. 2024;61(2):171-223. doi:10.1177/00224278221130004
- 252. McCarthy KJ, Mehta R, Haberland NA. Gender, power, and violence: A systematic review of measures and their association with male perpetration of IPV. Dalby AR, ed. *PLOS ONE*. 2018;13(11):e0207091. doi:10.1371/journal.pone.0207091
- 253. Sardinha L, Nájera Catalán HE. Attitudes towards domestic violence in 49 low- and middleincome countries: A gendered analysis of prevalence and country-level correlates. Uthman O, ed. *PLOS ONE*. 2018;13(10):e0206101. doi:10.1371/journal.pone.0206101
- 254. Johnson SL, Mootz J, Waller B, Fortunato Dos Santos P, Jaguga F, Giusto A. A global call for adolescent intimate partner violence prevention. *The Lancet Psychiatry*. Published online January 2024:S2215036623004352. doi:10.1016/S2215-0366(23)00435-2
- 255. Costenbader E, Cislaghi B, Clark CJ, et al. Social Norms Measurement: Catching up With Programs and Moving the Field Forward. *J Adolesc Health*. 2019;64(4S):S4-S6. doi:10.1016/j.jadohealth.2019.01.001
- 256. Offenhauer P, Buchalter A. Teen dating violence: a literature review and annotated bibliography. Published online 2011. Accessed May 13, 2018. https://www.ncjrs.gov/pdffiles1/nij/grants/235368.pdf

- 257. Circle of 6. Devpost. Accessed February 28, 2024. https://appsagainstabuse.devpost.com/submissions/4900-circle-of-6
- 258. Lumivero. NVivo 12. Published online 2017. www.lumivero.com
- 259. Jamal F, Bonell C, Harden A, Lorenc T. The social ecology of girls' bullying practices: exploratory research in two London schools. *Sociology of Health & Illness*. 2015;37(5):731-744. doi:10.1111/1467-9566.12231
- 260. Ponsford R, Meiksin R, Bragg S, et al. Co-production of two whole-school sexual health interventions for English secondary schools: Positive Choices and Project Respect. *Pilot Feasibility Stud*. 2021;7(1):50. doi:10.1186/s40814-020-00752-5
- 261. Pascoe CJ. 'Dude, You're a Fag': Adolescent Masculinity and the Fag Discourse. *Sexualities*. 2005;8(3):329-346. doi:10.1177/1363460705053337
- 262. *"It's Just Everywhere": A Study on Sexism in Schools and How We Tackle It*. National Education Union and UK Feminista; 2017.
- 263. Barter C. In the Name of Love: Partner Abuse and Violence in Teenage Relationships. *British Journal of Social Work*. 2009;39(2):211-233. doi:10.1093/bjsw/bcm127
- 264. Deutsch AR, Steinley D, Slutske WS. The role of gender and friends' gender on peer socialization of adolescent drinking: a prospective multilevel social network analysis. J Youth Adolesc. 2014;43(9):1421-1435. doi:10.1007/s10964-013-0048-9